# **UNITED STATES** SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# EODM 10 IZ

|                         | FOR  | NI 10-K   |                               |
|-------------------------|--|---|-------------------------------|
| (Mark                   | One)   |   |                               |
| × AN                    | NUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SEC   | CURITIES EXCHANGE ACT OF 1934   |                               |
| For                     | r the fiscal year ended December 31, 2018  |   |                               |
|                         |  | OR  |                               |
| □ TR                    | ANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE   | SECURITIES EXCHANGE ACT OF 1934   |                               |
| For                     | r the transition period from to  |   |                               |
|                         | Commission Fil   | e Number 000-25131  |                               |
|                         | BLUCO  | PRA, INC.   |                               |
|                         |  | nt as specified in its charter)   |                               |
|                         | Delaware   | 91-1718107  |                               |
|                         | (State or other jurisdiction of  | (IRS Employer   |                               |
|                         | incorporation or organization)   | Identification No.)   |                               |
|                         | (Address of principal o<br>Registrant's telephone n  | h Floor, Irving, Texas 75038<br>executive offices) (Zip code)<br>number, including area code:<br>870-6400   |                               |
|                         | Securities registered purs   | uant to Section 12(b) of the Act:   |                               |
|                         | Title of each class  | Name of each exchange on which registere  | ed                            |
|                         | Common Stock, par value \$0.0001 per share   | NASDAQ Global Select Market   |                               |
|                         | U 1  | at to Section 12(g) of the Act: None e of Class)  |                               |
| In                      | ndicate by check mark whether the registrant is a well-known seasoned issuer, as No $\Box$   |   |                               |
| In<br>12 montl<br>In    | ndicate by check mark whether the registrant is not required to file reports pursual dicate by check mark whether the registrant (1) has filed all reports required to I hs (or for such shorter period that the registrant was required to file such reports) dicate by check mark whether the registrant has submitted electronically every I eding 12 months (or for such shorter period that the registrant was required to su   | pe filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 du<br>t, and (2) has been subject to such filing requirements for the past 90 days<br>interactive Data File required to be submitted pursuant to Rule 405 of Reg | s. Yes ■ No □                 |
| In<br>the best of<br>In | ndicate by check mark if disclosure of delinquent filers pursuant to Item 405 of F of the registrant's knowledge, in definitive proxy or information statements incondicate by check mark whether the registrant is a large accelerated filer, an acceled See See definitions of "large accelerated filer," "accelerated filer," "smaller reporting the second s | tegulation S-K (§229.405 of this chapter) is not contained herein, and will prorated by reference in Part III of this Form 10-K or any amendment to rated filer, a non-accelerated filer, a smaller reporting company, or emergence.    | this Form 10-K. 🗷 ging growth |
| Large                   | e accelerated filer  | Accelerated filer   |                               |
| Non-                    | -accelerated filer   | Smaller reporting company   |                               |
|                         |  | Emerging growth company   |                               |
| accountii<br>In         | an emerging growth company, indicate by check mark if the registrant has elect ng standards provided pursuant to Section 13(a) of the Exchange Act.   Indicate by check mark whether the registrant is a shell company (as defined in Rubbe aggregate market value of the Company Stock held by non-offiliates of the re-  | tle 12b-2 of the Act). Yes □ No 🗷   |                               |

June 30, 2018 as reported on the NASDAQ Global Select Market, was \$1,736.6 million. Common Stock held by each officer and director (or his or her affiliate) has been excluded because such persons may be deemed to be affiliates. This determination of affiliate status is not necessarily a conclusive determination for other purposes.

As of February 22, 2019, 48,203,356 shares of the registrant's Common Stock were outstanding.

# DOCUMENTS INCORPORATED BY REFERENCE

Part III incorporates certain information by reference from the Definitive Proxy Statement to be filed by the registrant in connection with the 2019 Annual Meeting of Stockholders.

# TABLE OF CONTENTS

|  | Page  |
|--|---|
| <u>Business</u>  | 4   |
| Risk Factors   | <u>11</u>   |
| <u>Unresolved Staff Comments</u>   | <u>29</u>   |
| <u>Properties</u>  | <u>29</u>   |
| <u>Legal Proceedings</u>   | <u>29</u>   |
| Mine Safety Disclosures  | <u>29</u>   |
|  |   |
|  | <u>30</u>   |
| Selected Financial Data  | <u>31</u>   |
| Management's Discussion and Analysis of Financial Condition and Results of Operations          | <u>32</u>   |
| Quantitative and Qualitative Disclosures About Market Risk                                     | <u>54</u>   |
| Financial Statements and Supplementary Data  | <u>55</u>   |
| Changes in and Disagreements with Accountants on Accounting and Financial Disclosure           | <u>88</u>   |
| Controls and Procedures  | <u>88</u>   |
| Other Information  | <u>89</u>   |
|  |   |
| <u>Directors, Executive Officers and Corporate Governance</u>                                  | <u>90</u>   |
| Executive Compensation   | <u>90</u>   |
| Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters | <u>90</u>   |
| Certain Relationships and Related Transactions, and Director Independence                      | <u>91</u>   |
| Principal Accounting Fees and Services   | <u>91</u>   |
|  |   |
| Exhibits, Financial Statement Schedules  | <u>92</u>   |
| Form 10-K Summary  | <u>96</u>   |
|  | Risk Factors Unresolved Staff Comments Properties Legal Proceedings Mine Safety Disclosures  Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities Selected Financial Data Management's Discussion and Analysis of Financial Condition and Results of Operations Quantitative and Qualitative Disclosures About Market Risk Financial Statements and Supplementary Data Changes in and Disagreements with Accountants on Accounting and Financial Disclosure Controls and Procedures Other Information  Directors, Executive Officers and Corporate Governance Executive Compensation Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters Certain Relationships and Related Transactions, and Director Independence Principal Accounting Fees and Services  Exhibits, Financial Statement Schedules |

# Signatures

### Trademarks, Trade Names and Service Marks

This report includes some of trademarks, trade names and service marks of Blucora, Inc. (referred to throughout this report as "Blucora," the "Company", "we," "us," or "our"), including Blucora, HD Vest and TaxAct. Each one of these trademarks, trade names or service marks is either (i) our registered trademark, (ii) a trademark for which we have a pending application, (iii) a trade name or service mark for which we claim common law rights or (iv) a registered trademark or application for registration which we have been authorized by a third party to use.

Solely for convenience, the trademarks, service marks and trade names included in this report are without the ®, TM or other applicable symbols, but such references are not intended to indicate, in any way, that we will not assert, to the fullest extent under applicable law, our rights or the rights of the applicable licensors to these trademarks, service marks and trade names. This report may also include additional trademarks, service marks and trade names of others, which are the property of their respective owners. All trademarks, service marks and trade names included in this report are, to our knowledge, the property of their respective owners.

### CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS

This report contains forward-looking statements that involve risks and uncertainties. The statements in this report that are not purely historical are forward-looking statements within the meaning of Section 21E of the Securities Exchange Act of 1934, as amended. Words such as "anticipate," "believe," "plan," "expect," "future," "intend," "may," "will," "should," "estimate," "predict," "potential," "continue," and similar expressions identify forward-looking statements, but the absence of these words does not mean that the statement is not forward-looking. These forward-looking statements include, but are not limited to, statements regarding:

- *our ability to effectively compete within our industry;*
- our ability to attract and retain customers, as well as our ability to provide strong customer service;
- our future capital requirements and the availability of financing, if necessary;
- our ability to meet our current and future debt service obligations, including our ability to maintain compliance with our debt covenants;
- our ability to generate strong investment performance for our customers and the impact of the financial markets on our customers' portfolios;
- political and economic conditions and events that directly or indirectly impact the wealth management and tax preparation industries;
- our ability to attract and retain productive financial advisors;
- our ability to respond to rapid technological changes, including our ability to successfully release new products and services or improve upon
  existing products and services;
- our expectations concerning the revenues we generate from fees associated with the financial products that we distribute;
- our ability to manage leadership and employee transitions;
- our ability to comply with regulations applicable to the wealth management and tax preparation industries, including increased costs associated with new or changing regulations;
- our expectations concerning the benefits that may be derived from our new clearing platform and our investment advisory platform;
- risks associated with the use and implementation of information technology and the effect of security breaches, computer viruses and computer hacking attacks;
- our ability to comply with laws and regulations regarding privacy and protection of user data;
- our ability to maintain our relationships with third party partners, providers, suppliers, vendors, distributors, contractors, financial institutions and licensing partners;
- our beliefs and expectations regarding the seasonality of our business;
- risks associated with litigation;
- our ability to attract and retain qualified employees;
- our assessments and estimates that determine our effective tax rate;
- the impact of new or changing tax legislation on our business and our ability to attract and retain customers;
- our ability to develop, establish and maintain strong brands;
- our ability to protect our intellectual property and the impact of any claim that we have infringed on the intellectual property rights of others; and
- our ability to effectively integrate companies or assets that we acquire.

Forward-looking statements are subject to known and unknown risks, uncertainties, and other factors that may cause our results, levels of activity, performance, achievements, and prospects to be materially different from those expressed or implied by such forward-looking statements. These risks, uncertainties, and other factors include, among others, those identified under Item 1A, "Risk Factors," and elsewhere in this report. You should not rely on forward-looking statements, which speak only as of the date of this Annual Report on Form 10-K. We do not undertake any obligation to update any forward-looking statement to reflect new information, events, or circumstances after the date of this Annual Report on Form 10-K or to reflect the occurrence of unanticipated events, except as required by law.

#### PART I

### ITEM 1. Business

### **General Overview**

Blucora is a Delaware corporation that was founded in 1996, and, through organic growth and strategic acquisitions, we have become a leading provider of technology-enabled financial solutions to consumers, small business owners, and tax professionals. Our products and services in wealth management and tax preparation are offered through HDV Holdings, Inc. and its subsidiaries ("HD Vest") and TaxAct, Inc. and its subsidiary ("TaxAct"), respectively, and help consumers to manage their financial lives.

HD Vest provides financial advisors, who serve as independent contractors through HD Vest's registered broker-dealer, investment adviser and/or insurance subsidiaries, with an integrated platform that includes a broad variety of brokerage, investment advisory and insurance products to assist in making each financial advisor a financial service center for his/her clients. HD Vest generates revenue primarily through securities and insurance commissions, quarterly investment advisory fees based on advisory assets, revenue sharing agreements and other agreements and fees. We regularly review the commissions and fees we charge for these products and services in light of the evolving regulatory and competitive environment in which we operate and as a result of changes in client preferences and needs. We do not offer any proprietary products. As of December 31, 2018, approximately 3,600 advisors with branch offices in all 50 states utilized our HD Vest platform and supported approximately \$42.0 billion of total client assets for almost 350,000 clients.

TaxAct provides affordable digital do-it-yourself ("DDIY") tax preparation solutions for consumers, small business owners and tax professionals through its website www.TaxAct.com. During the year ended December 31, 2018, TaxAct powered approximately 3,700,000 consumer e-files and another 1,800,000 e-files through the 21,000 tax professionals who used TaxAct to prepare and file their taxes or those of their clients. TaxAct generates revenue primarily through its digital service at www.TaxAct.com. TaxAct can be used to file income tax returns in the U.S. federal jurisdiction, various state jurisdictions, and Canada.

Our common stock is listed on the NASDAQ Global Select Market under the symbol "BCOR."

### **Our History**

Blucora was formed in 1996 under the name InfoSpace, Inc. ("InfoSpace"). Over the next two decades, InfoSpace operated a number of digital businesses in search, directory, online commerce, media, and mobile infrastructure markets. In 2008, InfoSpace began principally focusing on internet search services and content (our "Search and Content" business).

In January 2012, InfoSpace acquired TaxAct. In connection with this acquisition, InfoSpace changed its name to Blucora, Inc. in June 2012.

In August 2013, Blucora acquired Monoprice, Inc. ("Monoprice"), an e-commerce company that sold self-branded electronics and accessories to both consumers and businesses (our "E-Commerce" business).

In July 2015, Blucora acquired SimpleTax Software Inc. ("SimpleTax"), a provider of digital tax preparation services for individuals in Canada.

For further detail on these acquisitions, see "Note 4: Business Combinations" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report.

In 2015 we acquired HD Vest and announced our plans to focus on the technology-enabled financial solutions market (the "Strategic Transformation"). The Strategic Transformation refers to our transformation into a technology-enabled financial solutions company comprised of TaxAct and HD Vest and the divestitures of our Search and Content business and our E-Commerce business in 2016. As part of the Strategic Transformation and "One Company" operating model, we relocated our corporate headquarters from Bellevue, Washington to Irving, Texas during 2017.

### **Business Overview**

Through its registered broker-dealer, registered investment adviser, and insurance agency subsidiaries, HD Vest operates the largest U.S. tax-professional-oriented independent broker-dealer, providing wealth management solutions to financial

advisors and their clients nationwide (our "Wealth Management" business or the "Wealth Management segment"). HDV Holdings, Inc. is the parent company of the Wealth Management business and owns all outstanding shares of HD Vest, Inc., which serves as a holding company for our various financial services subsidiaries. Those subsidiaries include HD Vest Investment Securities, Inc. (a registered broker-dealer), HD Vest Advisory Services, Inc. (a registered investment adviser), and HD Vest Insurance Agency, LLC (three insurance agencies domiciled in Texas, Massachusetts, and Montana).

The Tax Preparation business consists of the operations of TaxAct and provides digital tax preparation solutions for consumers, small business owners, and tax professionals through its website www.TaxAct.com (collectively referred to as the "Tax Preparation business" or the "Tax Preparation segment").

We have two reportable segments: the Wealth Management segment, which is comprised of the HD Vest business, and the Tax Preparation segment, which is comprised of the TaxAct business. See "Note 2: Summary of Significant Accounting Policies" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report for additional information on the Wealth Management and Tax Preparation businesses and their revenue. See "Note 3: Segment Information and Revenues" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report for information regarding revenue, operating income, and assets for our Wealth Management and Tax Preparation businesses.

### Wealth Management Business

HD Vest was founded to help tax and accounting professionals integrate financial services into their practices. Unlike traditional independent broker-dealers and/or investment advisers whose client relationships are limited to providing investment advice, most HD Vest advisors have long-standing tax advisory relationships that anchor their wealth management businesses. We believe that tax and accounting professionals, with their existing client relationships and in-depth knowledge of their clients' financial situations, have a competitive advantage and are better positioned than competitors to provide tailored financial solutions that enable clients to meet their goals. HD Vest primarily recruits independent tax professionals or financial advisors who partner with established tax practices and offers specialized training and support, which allows them to join the HD Vest platform as independent financial advisors. HD Vest has designed a learning management system for its advisors with a curriculum that introduces advisors to the investment business and helps them build their practices. The comprehensive training curriculum is administered through numerous outlets, including an annual three-day national sales conference, approximately 600 specialized local training events held annually, and on-demand learning paths.

HD Vest's business model provides an open-architecture investment platform and technology tools to help financial advisors identify investment opportunities for their clients, while the long-standing tax advisory relationships provide a large client base of possible investment clients. This results in an experienced and stable network of financial advisors, who have multiple revenue-generating options to diversify their earnings sources, and have access to HD Vest's innovative Mentor program and Chapter meetings. HD Vest also has a highly experienced home office team that is focused on solutions tailored to the advisor's practice. The home office team provides marketing, practice management, insurance and annuity, wealth management, compliance, succession planning, and other support to our advisors.

### Tax Preparation Business

TaxAct, a top-three provider of digital tax preparation solutions, based upon the number of e-files made in 2018, has leveraged its strong brand, comprehensive suite of tax preparation solutions, and proven digital lead generation capabilities to enable the filing of more than 65 million federal consumer tax returns in the U.S. and Canada since 2000. TaxAct operates as the value player in its market, with a mission to empower people to navigate the complexities of tax preparation with ease and accuracy at a fair price.

In addition to TaxAct's core offerings, TaxAct offers ancillary services such as refund payment transfer, audit defense, stored value cards, gift cards and retirement investment accounts through HD Vest, as well as presenting customers the option to review and take advantage of personalized tax and potential financial savings opportunities offered through third party product providers. TaxAct also has an established brand and reputation that we believe is attractive to customers.

TaxAct had five offerings for consumers for tax year 2017, which is the basis for its 2018 operating results:

- A "free" federal and state edition that handled simple returns;
- · A "basic" offering that offered features for filers with dependents, retirement income, or college expenses;

- A "plus" offering that contained all of the basic offering features in addition to tools to maximize credits and deductions, and enhanced reporting;
- A "self-employed" offering for independent contractors and self-employed filers; and
- A "premium" offering that contained all of the plus offering features in addition to tools for self-employed individuals to maximize credits and deductions

For TaxAct's offerings, state returns can be filed for free for free simple filers, or through the separately-sold state edition. TaxAct also had offerings for small business owners.

TaxAct's professional tax preparer software allows professional tax preparers to prepare and file individual and business returns for their clients. TaxAct offers flexible pricing and packaging options that help tax professionals save money by paying only for the specific services that they need. In addition, the professional tax preparer software includes valuable features that tax professionals count on to maximize their efficiency and productivity, including the option of entering data directly into tax forms, utilizing a question-and-answer interview method to enter data, or easily toggling between the two data entry methods.

# **Growth Strategy**

Our evolving growth strategy for HD Vest and TaxAct includes participating in favorable industry trends and executing growth strategies that we believe will result in customer and advisor retention and growth beyond that of the broader markets in which we operate. Our approach is grounded on the belief that the best way to sustainably grow a business is to earn loyalty based on continuously delivering ever-greater value to target customers and clients.

# **Favorable Industry Trends**

- Wealth Management Industry Trends We believe that HD Vest is and will be the beneficiary of several positive industry trends, including growth
  of investable assets driven by baby boomers' retirement accounts, a continued migration to independent advisor channels, liquidity events and a
  continued shift toward household use of fee-based financial advisors.
- Tax Preparation Industry Trends Tax Act participates in the consumer DDIY tax preparation solutions market, which is the fastest growing segment in the tax preparation industry and is bolstered by a growing millennial population that continues to adopt technology-enabled financial solutions that drive value and ease in their everyday lives, and we believe that tax simplification will drive digital consumer growth.

### **Executing our Growth Strategies**

• Accelerate Growth - There are four key objectives of our growth acceleration strategy:

The first key objective of our growth acceleration strategy is to capture organic growth opportunities by optimizing our HD Vest advisor success and productivity by:

- Prioritizing recruiting, retaining and developing advisors with the highest potential.
- Increasing support for our advisors and focusing on those advisors that are in the best position to grow, some of whom have significant growth potential.
- Focusing on technology and infrastructure upgrades, including leveraging third-party technologies in areas that are not differentiated, and in many cases, are just as good or better than our current solutions.

The second key objective of our growth acceleration strategy is to enhance our technology platform through our recently-completed transition to a new clearing firm by:

• Focusing on new technological capabilities, like highly-integrated business processing, data aggregation and a new client portal. Our goal is to improve advisor efficiency and productivity, which we expect will grow total client assets.

The third key objective of our growth acceleration strategy is to improve end-client penetration by:

• Focusing on opportunities to service more of our advisors' tax clients with wealth management solutions, and increasing the investable assets of those tax clients that are being serviced by our advisors.

The fourth key objective of our growth acceleration strategy is to increase Advisory Assets:

- Advisory Assets are more profitable and predictable than brokerage assets. By focusing on the industry-wide trend of shifting assets from
  brokerage to advisory, and offering new tools and solutions to our advisors, we aim to help our advisors convert assets to advisory.
- Build Tax-Smart Leadership A key element of our business model across both HD Vest and TaxAct is to leverage tax information to enable customers and advisors to better achieve their financial goals and uncover potential opportunities for customers by delivering technology-enabled tax-smart solutions and financial insights.

We have an opportunity to leverage more of our HD Vest experience of providing holistic tax-smart strategies and better outcomes for customers. Some examples include our BluPrint financial assessment, our BluVest - tax-optimized offering and our ability to refer to HD Vest advisors for support or 360-degree tax-smart wealth management.

- Create One Blucora A key objective of our strategy is to continue to enable efficiencies through shared services and expertise across our HD Vest and TaxAct businesses. We believe that building a high-performing organization that attracts, retains, develops and engages the strongest talent will drive a shared purpose and common culture.
- Deliver Results Our goal is to drive continuous improvement by focusing on specific metrics that drive the organization and continue to meet our stated goals and targets.

### Seasonality

Our Tax Preparation business is highly seasonal, with a significant portion of our annual revenue for such services earned in the first four months of our fiscal year. During the third and fourth quarters, the Tax Preparation segment typically reports losses because revenue from the segment is minimal while core operating expenses continue. We anticipate that the seasonal nature of that part of the business will continue in the foreseeable future.

### Competition

We face intense competition in all markets in which our businesses operate. Many of our competitors or potential competitors have substantially greater financial, technical, and marketing resources, larger customer bases, longer operating histories, more developed infrastructures, greater brand recognition, better access to vendors, and more established relationships. Our competitors may be able to adopt more aggressive pricing policies, develop and expand their product and service offerings more rapidly, adapt to new or emerging technologies more quickly, take advantage of acquisitions and other opportunities more readily, achieve greater economies of scale, and devote greater resources to the marketing and sale of their products and services than we can. For our businesses to be successful, we must be competitive in the Wealth Management and Tax Preparation markets, as described in more detail below.

Wealth Management Competition

The wealth management industry is a highly competitive global industry. We and our financial advisors compete directly with a variety of financial institutions, including traditional wirehouses, independent broker-dealers, registered investment advisers, asset managers, banks and insurance companies, and direct distributors such as 1st Global and Cetera Financial Specialists, as well as larger broker-dealers such as Raymond James Financial. Mergers and acquisitions have resulted in consolidation in the wealth management industry. As a result, many of our competitors have greater financial resources, broader and deeper distribution capabilities, and a more comprehensive offering of products and services. We and our financial advisors compete directly with those companies for the provision of products and services to clients, as well as for retention and hiring of financial advisors.

We believe that our competitive position in the wealth management industry is a function of our ability to enable our advisors to offer investment guidance in the context of their clients' tax situations and more specifically to:

• offer high-quality portfolio investment options and competitive product pricing;

- offer a differentiated value proposition (in terms of brand recognition, reputation, and financial advisor payouts) that is sufficient to recruit and retain financial advisors;
- offer products and technology solutions that are attractive to financial advisors and their clients;
- negotiate competitive compensation arrangements with third-parties, including vendors, suppliers, and product sponsors;
- ensure the privacy and security of personal client information submitted to our financial advisors;
- · develop and react to new technology, services, and regulation in the financial services industry; and
- put in place a sufficient support and service network required to support our financial advisors and clients.

#### Tax Preparation Competition

Our TaxAct business operates in a very competitive marketplace. There are many competing software products and digital services. Intuit's TurboTax, H&R Block's and Credit Karma's DDIY consumer products and services have a significant percentage of the software and digital service market. Our TaxAct business must also compete with alternate methods of tax preparation such as storefront tax preparation services, which includes both local tax preparers and large chains such as H&R Block, Liberty Tax, Jackson Hewitt and Credit Karma, and it may also be subject to new market entrants who may take some of our market share. Finally, our TaxAct business faces the risk that state or federal taxing agencies will offer software or systems to provide direct access for individual filers that will reduce the need for TaxAct's software and services.

We believe that our competitive position in the market for tax preparation software and services is a function of our ability to differentiate our brand versus those of competitors by the following:

- offering competitive pricing;
- · continuing to offer reliable, easy-to-use, and accessible software and services that are compelling to consumers;
- offering software that is backed by true financial and tax-expertise;
- ensuring the privacy and security of user data submitted through our products;
- · marketing our software and services in a cost-effective way; and
- offering ancillary services that are attractive to users.

# **Governmental Regulation**

Blucora is a publicly traded company that is subject to Securities and Exchange Commission ("SEC") and NASDAQ Global Select Market rules and regulations regarding public disclosure, financial reporting, internal controls, and corporate governance. Our Wealth Management and Tax Preparation segments are subject to federal and state government requirements, including regulations related to consumer protection, user privacy, security, pricing, taxation, intellectual property, labor, advertising, broker-dealers, securities, investment advisers, asset management, insurance, listing standards, and product and services quality.

Our Wealth Management segment is subject to additional financial industry regulations and supervision, including by the SEC, the Financial Industry Regulatory Authority ("FINRA"), the Department of Labor ("DOL"), state securities and insurance regulators, and other regulatory authorities. Our Wealth Management subsidiary HD Vest Investment Securities, Inc. is a broker-dealer registered with the SEC, a member of FINRA, and a member of the Securities Investor Protection Corporation and the Depository Trust & Clearing Corporation. Broker-dealers and their representatives are subject to rules and regulations covering all aspects of the securities business, including sales and trading practices, use and safekeeping of clients' funds and securities, capital adequacy, recordkeeping and reporting, the conduct of directors, officers, and employees, and general anti-fraud provisions. Broker-dealers and their representatives are also regulated by state securities administrators in those jurisdictions where they do business. Compliance with many of the rules and regulations applicable to us involves a number of risks, because rules and regulations frequently change and are subject to varying interpretations, among other reasons. Regulators make periodic examinations of our broker-dealer operations and review annual, monthly, and other reports on our operations and financial condition. Violations of rules and regulations governing a broker-dealer's actions could result in censure, penalties and fines, the issuance of cease-and-desist orders, the restriction, suspension, or expulsion from the securities industry of such broker-dealer, its representatives or its officers or employees, or other similar adverse consequences.

Our Wealth Management subsidiary, HD Vest Advisory Services, Inc. is registered with the SEC as an investment adviser and is subject to the requirements of the Investment Advisers Act of 1940, as amended (the "Advisers Act"), and the regulations promulgated thereunder. Such requirements relate to, among other things, fiduciary duties to clients, advisory fees, maintaining an effective compliance program, solicitation arrangements, conflicts of interest, advertising, limitations on agency cross and principal transactions between the advisor and advisory clients, recordkeeping and reporting requirements, disclosure requirements, and general anti-fraud provisions. The SEC periodically examines our investment adviser operations and reviews annual, monthly, and other reports on our operations and financial condition. The SEC is authorized to institute proceedings and impose sanctions for violations of the Advisers Act and other federal securities laws, ranging from fines and censure to termination of an investment adviser's registration. Investment adviser representatives also are subject to certain state securities laws and regulations. Failure to comply with the Advisers Act or other federal and state securities laws and regulations, profit disgorgement, fines, or other similar adverse consequences.

Our Wealth Management subsidiaries offer certain products and services subject to the Employee Retirement Income Security Act of 1974, as amended ("ERISA") and Section 4975 of the Internal Revenue Code (the "Code"), and to regulations promulgated under ERISA or the Code, insofar as they provide services with respect to plan clients, or otherwise deal with plan clients that are subject to ERISA or the Code. ERISA imposes certain duties on persons who are "fiduciaries" (as defined in Section 3(21) of ERISA) and prohibits certain transactions involving plans subject to ERISA and fiduciaries or other service providers to such plans. Non-compliance with these provisions may expose an ERISA fiduciary or other service provider to liability under ERISA, which may include monetary penalties as well as equitable remedies for the affected plan. Section 4975 of the Code prohibits certain transactions involving plans (as defined in Section 4975(e)(1), which includes individual retirement accounts and Keogh plans) and service providers, including fiduciaries, to such plans. Section 4975 imposes excise taxes for violations of these prohibitions.

On April 18, 2018, the SEC issued draft rulemaking addressing standards of conduct for broker-dealers and disclosure requirements for broker-dealers and investment advisers. As presently drafted, the SEC's proposed rules would impose a "best interest" standard on broker-dealers and their registered representatives, as well as a new disclosure form (Form CRS) that both broker-dealers and investment advisers would have to give clients before providing them investment advice. The SEC's proposed rules, if adopted in their current form, would heighten the standard of care for broker-dealers when making investment recommendations and would impose disclosure and policy and procedural obligations that could impact the compensation HD Vest and its representatives receive for selling certain types of products, particularly those (such as mutual funds) that offer different compensation across different share classes. The SEC's proposed rules would also limit our ability to use the terms "advisor" or "adviser" when referring publicly to our registered representatives who are not also advisory licensed. Based on comments by SEC Commissioners when the proposed rules were first presented, however, we believe that the SEC's proposed rules may substantially change during the public comment process. In addition, the SEC's final rules may not be issued for many months and, even then, could be the subject of litigation. Accordingly, we cannot predict if and when the SEC will complete any final rulemaking or what the contours of the final rules will be. However, the SEC's final rules could result in additional compliance costs, lesser compensation, and management distraction, all of which could materially impact our business and operations.

Our Tax Preparation segment is subject to federal and state government requirements, including regulations related to the electronic filing of tax returns, the provision of tax preparer assistance, and the use and disclosure of customer information. We are also required to comply with Federal Trade Commission requirements and a variety of state revenue agency standards. In addition, we offer certain other products and services to small businesses and consumers, which are also subject to regulatory requirements. As we expand our products and services, both domestically and internationally, we may become subject to additional government regulation. Further, regulators may adopt new laws or regulations or their interpretation of existing laws or regulations may differ from ours or expand to cover additional products and services. These increased regulatory requirements could impose higher regulatory compliance costs, limitations on our ability to provide some services in some states or countries, and liabilities that might be incurred through lawsuits or regulatory penalties.

On December 22, 2017, President Trump signed the Tax Cuts and Jobs Act into law, which significantly revised the U.S. tax code. The resulting changes became effective beginning with our 2018 calendar year financial reporting period. The primary impacts to us included a reduction of the federal corporate tax rate from 35% to 21% affecting our net deferred tax liabilities, repeal of corporate alternative minimum tax and associated potential refunds of prior paid taxes, and potential deductible limits of certain executive compensation.

We are subject to federal and state laws and government regulations concerning employee safety and health and environmental matters. The Occupational Safety and Health Administration, the Environmental Protection Agency, and other federal and state agencies have the authority to promulgate regulations that may have an impact on our operations.

See the section entitled "Risks Associated With our Businesses" in Part I, Item 1A of this report for additional information regarding governmental regulation of our business and risks related to such regulation.

# Privacy and Security of Customer Information and Transactions

We are also subject to various federal, state and international laws and regulations and to financial institution and healthcare provider requirements relating to the privacy and security of the personal information of customers and employees. In addition, we are subject to laws and regulations that apply to the Internet, behavioral tracking and advertising, mobile applications and messaging, telemarketing, email activities, data hosting and retention, financial and health information, and credit reporting. Additional laws in all of these areas are likely to be passed in the future, which could result in significant limitations on or changes to the ways in which we can collect, use, host, store, or transmit the personal information and data of our customers or employees, communicate with our customers, and deliver products and services, or may significantly increase our compliance costs. As our business expands to new industry segments and new uses of data that are regulated for privacy and security, or to countries outside the United States that have strict data protections laws, our compliance requirements and costs will increase.

Through a privacy policy framework designed to be consistent with globally recognized privacy principles, we comply with United States federal and other country guidelines and practices to help ensure that customers and employees are aware of, and can control, how we use information about them. The TaxAct.com website and its digital products have been certified by TRUSTe, an independent organization that operates a website and digital product privacy certification program representing industry standard practices to address users' and regulators' concerns about online privacy. We also use privacy statements to provide notice to customers of our privacy practices, as well as provide them the opportunity to furnish instructions with respect to use of their personal information. We participate in industry groups whose purpose is to develop or shape industry best practices, and to influence public policy for privacy and security.

To address security concerns, we use security safeguards to help protect the systems and the information customers give to us from loss, misuse and unauthorized alteration. Whenever customers transmit sensitive information, such as credit card information or tax return data, through one of our websites or products, we use industry standards to encrypt the data as it is transmitted to us. We work to protect our systems from unauthorized internal or external access using numerous commercially available computer security products as well as internally developed security procedures and practices.

See the section entitled "Risks Associated With our Businesses" in Part I, Item 1A of this report for additional information regarding risks related to privacy and security of customer information and transactions.

### **Intellectual Property**

Our success depends upon our technology and intellectual property rights. We seek to protect such rights and the value of our corporate brands and reputation through a variety of measures, including: domain name registrations, confidentiality and intellectual property assignment agreements with employees and third parties, protective contractual provisions, and laws regarding copyrights, trademarks, and trade secrets. We hold multiple registered trademarks in the United States and in various foreign countries, and we may apply for additional trademarks as business needs require. We may not be successful in obtaining issuance or registration for such applications or in maintaining existing trademarks. In addition, registered marks may not provide us with any competitive advantages. We may be unable to adequately or cost-effectively protect or enforce our intellectual property rights, and failure to do so could weaken our competitive position and negatively impact our business and financial results. If others claim that our products infringe their intellectual property rights, we may be forced to seek expensive licenses, re-engineer our products, engage in expensive and time-consuming litigation, or stop marketing and licensing our products. See the section entitled "Risks Associated With our Businesses" in Part I Item 1A of this report for additional information regarding protecting and enforcing intellectual property rights by us and third parties against us.

# **Employees**

As of December 31, 2018, we had 529 full-time employees. There is significant competition for qualified personnel in the industries in which we operate, particularly for software development and other technical staff. We believe that our future success will depend in part on our continued ability to hire and retain qualified personnel.

### Company Internet Site and Availability of SEC Filings

Our corporate website is located at www.blucora.com. We make available on that site, as soon as reasonably practicable, our Annual Reports on Form 10-K, Quarterly Reports on Form 10-Q, proxy statements, Current Reports on Form 8-K, other

reports filed with or furnished to the SEC, as well as any amendments to those filings. Our SEC filings, as well as our Code of Ethics and Conduct and other corporate governance documents, can be found in the Investor Relations section of our site and are available free of charge. Amendments to our Code of Ethics and Conduct and any grant of a waiver from a provision of the Code of Ethics and Conduct requiring disclosure under applicable Securities and Exchange Commission rules will be disclosed on our website. Information on our website is not part of this Annual Report on Form 10-K. In addition, the SEC maintains a website at www.sec.gov that contains reports, proxy and information statements, and other information regarding us and other issuers that file electronically with the SEC.

### ITEM 1A. Risk Factors

Our business and future results may be affected by a number of risks and uncertainties that should be considered carefully. In addition, this report also contains forward-looking statements that involve risks and uncertainties. Our actual results could differ materially from those anticipated in such forward-looking statements as a result of certain factors, including the risks described below. The occurrence of one or more of the events listed below could also have a material adverse effect on the Company's business, prospects, results of operations, reputation, financial condition, cash flows or ability to continue current operations without any direct or indirect impairment or disruption, which is referred to throughout these Risk Factors as a "Material Adverse Effect."

# RISKS ASSOCIATED WITH OUR BUSINESSES

# The Tax Preparation and Wealth Management markets are very competitive, and failure to effectively compete could result in a Material Adverse Effect.

Our Tax Preparation business operates in a very competitive marketplace. There are many competing software products and digital services. Intuit's TurboTax and H&R Block's products and services have a significant percentage of the software and digital service market. Our Tax Preparation business must also compete with alternate methods of tax preparation, such as storefront tax preparation services, which includes both local tax preparers and large chains such as H&R Block, Liberty Tax, Jackson Hewitt and Credit Karma, and it may also be subject to new market entrants who may take some of our market share. As digital-do-it-yourself tax preparation continues to be characterized by intense competition, including heavy marketing expenditures, price-based competition, and new entrants, maintaining and growing market share becomes more challenging unless brand relevance, customer experience, and feature/functionality provide meaningful incremental value. If we cannot continue to offer software and services that have quality and ease-of-use that are compelling to consumers, market the software and services in a cost-effective manner, offer ancillary services that are attractive to users, and develop the software and services at a low enough cost to be able to offer them at a competitive price point, it could result in a Material Adverse Effect.

Our Tax Preparation business also faces potential competition from the public sector, where we face the risk of federal and state taxing authorities developing software or other systems to facilitate tax return preparation and electronic filing at no charge to taxpayers, which could reduce the need for TaxAct's software and services. These or similar programs may be introduced or expanded in the future, which may cause us to lose customers and revenue. The Free File Program is currently the sole means by which the U.S. Internal Revenue Service (the "IRS") offers tax software to taxpayers. The Free File Program is a partnership between the IRS and the Free File Alliance, a group of private sector tax preparation companies of which we are a member that has agreed to offer free electronic tax filing services to taxpayers meeting certain income-based guidelines. As part of the program, the IRS has agreed that it will not compete with Free File Alliance companies in providing free, digital tax return preparation and filing services to taxpayers. The Free File Program's continuation depends on a number of factors, including increasing public awareness of and access to the free program, as well as continued government support. The IRS's current agreement with the Free File Alliance is scheduled to expire in October 2021. If the Free File Program is not renewed upon expiration of the agreement or if the Free File Program is terminated, and the IRS enters the software development and return preparation space, the federal government would be a publicly funded direct competitor of us and the U.S. tax services industry as a whole.

The wealth management industry in which our Wealth Management business operates is also highly competitive, and we may not be able to maintain our customers, financial advisors, distribution network, or the terms on which we provide our products and services. Our Wealth Management business competes based on a number of factors, including name recognition, service, the quality of investment advice, investment performance, technology, product offerings and features, price, and perceived financial strength. Competitors in the wealth management industry include broker-dealers, banks, asset managers, insurers, and other financial institutions. Many of these competitors have greater market share, offer a broader range of

products, and have greater financial resources. In addition, over time, certain sectors of the wealth management industry have become considerably more concentrated, as financial institutions involved in a broad range of financial services have been acquired by or merged into other firms. This consolidation could result in our competitors gaining greater resources, and we may experience pressures on our pricing and market share as a result of these factors and as some of our competitors seek to increase market share by reducing prices. In addition, our Wealth Management business seeks to differentiate itself on the basis of offering tax-smart investing advice and solutions. There is no guarantee that this differentiation will be meaningful to our customers and potential customers, or that another competitor will not adopt a similar strategy more effectively. In either case, our ability to compete effectively in the market could be damaged.

Poor service or performance of the financial products we offer or competitive pressures on pricing of such services or products may cause our Wealth Management business customers to withdraw assets on short notice.

Customer service and investment performance are important factors in the success of our Wealth Management business. Strong customer service and investment performance help increase customer retention and generate sales of products and services. In contrast, poor service or investment performance could impair our revenues and earnings, as well as our prospects for growth. Customers can terminate their relationships with us or our financial advisors at will. There can be no assurance as to how future investment performance will compare to that of our competitors, and historical performance is not indicative of future returns. A decline or perceived decline in investment performance, on an absolute or relative basis, could cause a decline in sales of mutual funds and other investment products, an increase in redemptions and the termination of asset management relationships. Such actions may reduce our aggregate amount of advisory assets and reduce management fees. Poor investment performance could also adversely affect our ability to expand the distribution of our products through independent financial advisors.

In addition, the emergence of new financial products or services from others, or competitive pressures on pricing of such services or products, may result in the loss of accounts in our Wealth Management business. We must also monitor the pricing of our services and financial products in relation to competitors and periodically may need to adjust commission and fee rates, interest rates on deposits and margin loans, and other fee structures to remain competitive. Competition from other financial services firms, such as reduced commissions to attract customers or trading volume, direct-to-investor online financial services, or higher deposit interest rates to attract customer cash balances, could adversely impact our business. Customers of our Wealth Management business could also reduce the aggregate amount of their assets managed by us or shift their funds to other types of accounts with different rate structures for any number of reasons, including investment performance, changes in prevailing interest rates, changes in investment preferences, changes in our (or our financial advisors') reputation in the marketplace, changes in customer management or ownership, loss of key investment management personnel and financial market performance. Our customers can withdraw the assets we manage on short notice, making our future customer and revenue base unpredictable. A reduction in advisory assets and the resulting decrease in revenues and earnings could have a Material Adverse Effect.

### Changes in domestic and international economic, political and other factors could have a Material Adverse Effect on our business.

Our Wealth Management business operates in the United States and global financial markets, and our Tax Preparation business offers tax filing services in the federal jurisdiction of the United States, various state jurisdictions and Canada. Accordingly, we are affected by United States and global economic and political conditions that directly and indirectly impact a number of factors in the domestic and global financial markets and economies, which may be detrimental to our operating results. In addition, because the significant majority of our revenue is earned within the United States, economic conditions in the United States have an even greater impact on us than companies with a more diverse international presence.

Domestic and international factors that could affect our business include, but are not limited to, trading levels, investing, origination activity in the securities markets, security and underlying asset valuations, the absolute and relative level and volatility of interest and currency rates, real estate values, the actual and perceived quality of issuers and borrowers, the supply of and demand for loans and deposits, United States and foreign government fiscal and tax policies, United States and foreign government ability, real or perceived, to avoid defaulting on government securities, inflation, decline and stress or recession in the United States and global economies generally, terrorism and armed conflicts, and natural disasters such as weather catastrophes and widespread health emergencies. Furthermore, changes in consumer economic variables, such as the number and size of personal bankruptcy filings, the rate of unemployment, decreases in property values, certain life events, and the level of consumer confidence and consumer debt, may substantially affect consumer loan levels and credit quality.

While United States and global financial markets have, at a macro level, recently experienced growth, uncertainty and potential volatility remain. For instance, on December 6, 2018, the Dow Jones Industrial Average plunged nearly 800 points before recovering rapidly to close the day down 79 points. A period of sustained downturns and/or volatility in the securities markets,

changes in interest rates by the Federal Reserve, a return to increased credit market dislocations, reductions in the value of real estate, and other negative market factors could have a Material Adverse Effect on our business. We could experience a decline in commission revenue from lower trading volumes, a decline in fees from reduced portfolio values of securities managed on behalf of our customers, a reduction in revenue from capital markets and advisory transactions due to reduced activity, increased credit provisions and charge-offs, losses sustained from our customers' and market participants' failure to fulfill their settlement obligations, reduced net interest earnings, and other losses. Periods of reduced revenue and other losses could be accompanied by periods of reduced profitability because certain of our expenses, including, but not limited to, our interest expense on debt, rent, facilities and salary expenses are fixed and, our ability to reduce them over short time periods is limited.

Other more specific trends may also affect our financial condition and results of operations, including, for example: changes in the mix of products preferred by investors that may cause increases or decreases in our fee revenues associated with such products, depending on whether investors gravitate towards or away from such products. The timing of such trends, if any, and their potential impact on our financial condition and results of operations are beyond our control.

Challenging economic times and changes to the Federal or various states' tax code (personal and/or corporate), such as the recent changes passed under the Tax Cuts and Jobs Act, could cause potential new customers not to purchase or to delay purchasing of our products and services, and could cause our existing customers to discontinue purchasing or delay upgrades of our existing products and services, thereby negatively impacting our revenues and future financial results. Poor economic conditions and high unemployment have caused, and could in the future cause, a significant decrease in the number of tax returns filed, which may have a significant effect on the number of tax returns we prepare and file. In addition, weakness in the end-user consumer and small business markets could negatively affect the cash flow of our distributors and resellers who could, in turn, delay paying their obligations to us, which could increase our credit risk exposure and cause delays in our recognition of revenue or future sales to these customers. Any of these events could have a Material Adverse Effect. See "We may be negatively impacted by the recently passed Tax Cuts and Jobs Act or by any future changes in tax laws" for a discussion of additional risks related to changes in the tax code.

Each of these factors could impact customer activity in all of our businesses and have a Material Adverse Effect. In addition, these factors may have an impact on our ability to achieve our strategic objectives and to grow our business.

# If we are unable to attract and retain productive advisors, our financial results will be negatively impacted.

Our Wealth Management business derives a large portion of its revenues from commissions and fees generated by its advisors. Our ability to attract and retain productive advisors has contributed significantly to our growth and success. If we fail to attract new advisors or to retain and motivate our current advisors, our business may suffer. In addition, the wealth management industry in general is experiencing a decline in the number of younger financial advisors entering the industry. We are not immune to that industry trend. If we are unable to replace advisors as they retire, or to assist retiring advisors with transitioning their practices to existing advisors, we could experience a decline in revenue and earnings.

The market for productive advisors is highly competitive, and we devote significant resources to attracting and retaining the most qualified advisors. In attracting and retaining advisors, we compete directly with a variety of financial institutions such as wirehouses, regional broker-dealers, banks, insurance companies and other independent broker-dealers. Financial industry competitors are increasingly offering guaranteed contracts, upfront payments, and greater compensation to attract successful financial advisors. These can be important factors in a current advisor's decision to leave us as well as in a prospective advisor's decision to join us. If we are not successful in retaining highly qualified advisors, we may not be able to recover the expense involved in attracting and training these individuals. There can be no assurance that we will be successful in our efforts to attract and retain the advisors needed to achieve our growth objectives.

Moreover, the costs associated with successfully attracting and retaining advisors could be significant, and there is no assurance that we will generate sufficient revenues from those advisors' business to offset those costs. Designing and implementing new or modified compensation arrangements and equity structures to successfully attract and retain advisors is complicated. Changes to these arrangements could themselves cause instability within our existing investment teams and negatively impact our financial results and ability to grow.

In addition, as some of HD Vest's advisors grow their advisory assets, they may decide to disassociate from HD Vest to establish their own registered investment advisers ("*RIAs*") and take customers and associated assets into those businesses. HD Vest seeks to deter advisors from taking this route by continuously evaluating its technology, product offerings, and service, as well as its advisor compensation, fees, and pay-out policies, to ensure that HD Vest is competitive in the market and attractive to successful advisors. We may not be successful in dissuading such advisors from forming their own RIAs, which could cause a material volume of customer assets to leave HD Vest's platform, which would reduce our revenues and could cause a Material Adverse Effect.

# Rapid growth may place significant demands on our resources.

We have experienced rapid growth since the completion of our Strategic Transformation. Our anticipated future growth will place a substantial demand on our managerial, operational and financial resources due to:

- the need to manage relationships with various strategic partners and other third parties;
- the need to maintain levels of service expected by clients and customers;
- the pressure to deliver our products and services on a timely basis;
- difficulties in hiring and retaining skilled personnel necessary to support our business;
- · pressures for the continued development of our products and financial and information management systems; and
- the possible need to create lines of businesses or departments that do not now exist, and to hire, train, motivate and manage a growing number of staff.

There can be no assurance that we will be able to effectively achieve or manage any future growth. If we have not made adequate allowances for the costs and risks associated with this expansion or if our systems, procedures, or controls are not adequate to support our operations, our business could be harmed and we could experience a Material Adverse Effect.

# Future revenue growth depends upon our ability to adapt to technological change and successfully introduce new and enhanced products and services.

The tax preparation and wealth management industries are characterized by rapidly changing technology, evolving industry and security standards, and frequent new product introductions. Our competitors in these industries offer new and enhanced products and services every year. Consequently, customer expectations are constantly changing. We must successfully innovate and develop or offer new products and features to meet evolving customer needs and demands, while continually updating our technology infrastructure. We must devote significant resources to developing our skills, tools, and capabilities in order to capitalize on existing and emerging technologies. Our inability to quickly and effectively innovate our products, services, and infrastructure could result in a Material Adverse Effect.

We offer our digital tax preparation products and services through our website and through our mobile app. If our customers don't deem our website or our mobile app user friendly or if they deem our competitors' website or mobile app more user friendly or better than ours, our market share could decline, which could have a Material Adverse Effect. In addition, we regularly make upgrades to the technology we use for our tax preparation product that are expected to provide a better user experience and help us to keep existing customers or attract new customers. If our mobile app or the other upgrades we make to the technology we use in our Tax Preparation business are not successful, it could result in wasted development costs or damage to our brands and market share, any of which could have a Material Adverse Effect. We may also encounter problems in connection with our mobile app, and we may need to devote significant resources to the creation, support, and maintenance of new user experiences.

# Our business depends on fees generated from the distribution of financial products and fees earned from management of advisory accounts.

A large portion of our revenues are derived from fees generated from the distribution of financial products, such as mutual funds and variable annuities. Changes in the structure or amount of the fees paid by the sponsors of these products could directly affect our revenues, business and financial condition. In addition, if these products experience losses or increased investor redemptions, we may receive lower fee revenue from the investment management and distribution services we provide on behalf of the mutual funds and annuities. Should issuers of these products leave the market or discontinue offering or paying trail compensation on some or all of their products, our revenues could be negatively impacted. The investment management fees we are paid may also decline over time due to factors such as increased competition, renegotiation of contracts and the introduction of new, lower-priced investment products and services. Changes in market values or in the fee structure of asset management accounts would affect our revenues, business and financial condition. Asset management fees often are primarily comprised of base management and incentive fees, and investment advisers generally are experiencing advisory fee compression due to intense competition. Management fees are primarily based on advisory assets, which are impacted by net inflow/outflow of customer assets and market values. Below-market investment performance by our funds and portfolio managers could result in a loss of managed accounts and could result in reputational damage that might make it more difficult to attract new customers and thus further impact our business and financial condition. If we were to experience the loss of managed accounts, our fee revenue would decline. In addition, in periods of declining market values, our advisory assets may also decline, which would negatively impact our fee revenues and could have a Material Adverse Effect.

Investors in the pooled vehicles that we advise can redeem their investments in those funds at any time without prior notice, which could adversely affect our earnings.

Investors in the mutual funds and some other pooled investment vehicles that we advise may redeem their investments in those funds at any time without prior notice, and investors in other types of pooled vehicles we advise may typically redeem their investments on fairly limited or no prior notice, thereby reducing our advisory assets. These investors may redeem their investments for any number of reasons, including general financial market conditions, the absolute or relative investment performance we have achieved, or their own financial condition and requirements. In a declining stock market, the pace of redemptions could accelerate. Poor investment performance relative to other funds tends to result in decreased purchases and increased redemptions of fund shares.

We have had recent senior leadership transitions, and if we are not effective in managing those transitions, our business could be adversely impacted and we could experience a Material Adverse Effect.

We have had recent senior leadership transition in connection with our Strategic Transition and otherwise and have replaced some of our executive officers and senior leadership team. While many of our executive officers have relevant industry experience, they are new to our Company. Changes in senior management are inherently disruptive and can be difficult to manage, and efforts to implement any new strategic or operating goals may not succeed in the absence of a long-term management team. Periods of transition in senior management are often difficult due to cultural differences that may result from changes in strategy and style and the time required for new executives to gain detailed operational knowledge. These changes could also cause concerns to third parties with whom we do business, and may increase the likelihood of turnover of our employees and, in the case of our Wealth Management business, turnover of advisors. If we are not effective in managing these leadership and employee transitions, our business could be adversely impacted and we could experience a Material Adverse Effect.

Government regulation of our business, including increased regulation or the interpretation of existing laws, rules or regulations, could have a Material Adverse Effect.

We are subject to federal, state, and local laws and regulations that affect our business, such as financial services, data privacy and security requirements, tax, digital content, employment, consumer protection and fraud protection, among others. In addition, there have been significant new regulations and heightened focus by the government on many of the laws and regulations that affect both our Wealth Management and our Tax Preparation businesses. As we expand our products and services and revise our business models, we may become subject to additional government regulation or increased regulatory scrutiny. Regulators may adopt new laws or regulations, or their interpretation of existing laws or regulations may differ from our interpretation or the laws of other jurisdictions in which we operate. If we are found to not be in compliance with certain laws, rules or regulations, it could have a Material Adverse Effect. Increased or new regulatory requirements or changes in the interpretation of existing laws, rules or regulations could, among other things, result in penalties or fines, impose significant limitations on the way we conduct our business, require changes to our business, require certain notifications to customers or employees, restrict our use of personal information, cause our customers to cease utilizing our products or services, make our business more costly, less efficient, or impossible to conduct, require us to modify our current or future products or services in a manner that is detrimental to our business and result in additional compliance costs, which could have a Material Adverse Effect.

The tax preparation industry continues to receive heightened attention from federal and state governments. New legislation, regulation, public policy considerations, changes in the cybersecurity environment, litigation by the government or private entities, or new interpretations of existing laws may result in greater oversight of the tax preparation industry, restrict the types of products and services that we can offer or the prices we can charge, or otherwise cause us to change the way we operate our Tax Preparation business or offer our tax preparation products and services. We may not be able to respond quickly to such regulatory, legislative and other developments, and these changes may in turn increase our cost of doing business and limit our revenue opportunities. In addition, if our practices are not consistent with new interpretations of existing laws, rules or regulations, we may become subject to lawsuits, penalties, fines and other liabilities that did not previously apply. We are also required to comply with Federal Trade Commission (the "FTC") requirements and a variety of state revenue agency standards. Requirements imposed by the FTC or state agencies, including new requirements or their interpretation of existing laws, rules or regulations, could be burdensome on our business, cause us to lose market share due to product changes we are required to implement or may significantly increase the costs of providing those services to our customers and may prevent us from delivering a quality product to our customers in a timely manner and at an acceptable price, all of which could have a Material Adverse Effect. In addition, in our Tax Preparation business, we generate revenue from certain financial products related to our tax preparation software and services. These products include prepaid debit cards or gift cards on which a tax filer may receive his or her tax refund and the ability of certain of our users to have the fees for our services deducted from their tax refund. Any regulation of these p

In addition, we are subject to laws, regulations, and industry rules relating to the collection, use, and security of user data. We expect regulation in this area to increase, and our current data protection policies and practices may not be sufficient and thus may require modification. Numerous jurisdictions have passed, and may in the future pass, new laws related to the use and retention of consumer information, and this area continues to be an area of interest for U.S. federal, state and foreign governmental authorities. These laws may be interpreted and applied inconsistently from jurisdiction to jurisdiction, and our current data protection policies and practices may not be consistent with all of those interpretations and applications. We have incurred, and may continue to incur, significant expenses to comply with privacy and security standards and protocols imposed by law, regulation, industry standards, and contractual obligations. Failure to comply with laws and regulations that protect user data could harm our reputation and could result in a Material Adverse Effect.

Our ability to comply with all applicable laws, rules and regulations and interpretations of such laws, rules and regulations is largely dependent on our establishment and maintenance of compliance, audit, and reporting systems and procedures, as well as our ability to attract and retain qualified compliance, audit, and risk management personnel. While we have adopted systems, policies, and procedures reasonably designed to comply or facilitate compliance with all applicable laws, rules and regulations of such laws, rules and regulations, these systems, policies, and procedures may not be fully effective. There can be no assurance that we will not be subject to investigations, claims, or other actions or proceedings by regulators or third parties with respect to our past or future compliance with applicable laws, rules, and regulations, the outcome of which may have a Material Adverse Effect.

If we fail to comply with applicable laws, rules, regulations and guidance, such failure could have a Material Adverse Effect. See "Our Wealth Management business is subject to extensive regulation, and failure to comply with these regulations could have a Material Adverse Effect" for additional information regarding the regulation of our business.

# Our Wealth Management business is subject to extensive regulation, and failure to comply with these regulations could have a Material Adverse Effect.

Our Wealth Management business is heavily regulated by multiple agencies, including the Securities and Exchange Commission ("SEC"), the Financial Industry Regulatory Authority ("FINRA"), state securities and insurance regulators, and other regulatory authorities. Failure to comply with these regulators' laws, rules, and regulations could result in the restriction of the ongoing conduct or growth, or even liquidation of, parts of our business and otherwise cause a Material Adverse Effect. The regulatory environment in which our Wealth Management business operates is continually evolving, and the level of financial regulation to which we are subject has generally increased in recent years. Among the most significant regulatory changes affecting our Wealth Management business is the Dodd-Frank Wall Street Reform and Consumer Protection Act (the "Dodd-Frank Act"), which mandates broad changes in the supervision and regulations of the wealth management industry. Regulators implementing the Dodd-Frank Act have adopted, proposed to adopt, and may in the future adopt regulations that could impact the manner in which we will market HD Vest products and services, manage HD Vest operations, and interact with regulators. In addition, the Trump Administration has initiated and in some cases completed a broad review of U.S. fiscal laws and regulations. If significant changes are enacted as a result of this review, they could negatively impact our Wealth Management business and cause a Material Adverse Effect.

On April 18, 2018, the SEC issued draft rulemaking addressing standards of conduct for broker-dealers and disclosure requirements for broker-dealers and investment advisers. As presently drafted, the SEC's proposed rules would impose a "best interest" standard on broker-dealers and their registered representatives, as well as a new disclosure form (Form CRS) that both broker-dealers and investment advisers would have to give clients before providing them investment advice. The SEC's proposed rules, if adopted in their current form, would heighten the standard of care for broker-dealers when making investment recommendations and would impose disclosure and policy and procedural obligations that could impact the compensation HD Vest and its representatives receive for selling certain types of products, particularly those (such as mutual funds) that offer different compensation across different share classes. The SEC's proposed rules would also limit our ability to use the terms "advisor" or "adviser" when referring publicly to our registered representatives who are not also advisory licensed. Based on comments by SEC Commissioners when the proposed rules were first presented, however, we believe that the SEC's proposed rules may substantially change during the public comment process. In addition, the SEC's final rules may not be issued for many months and, even then, could be the subject of litigation. Accordingly, we cannot predict if and when the SEC will complete any final rulemaking or what the contours of the final rules will be. However, the SEC's final rules could result in additional compliance costs, lesser compensation, and management distraction, all of which could have a Material Adverse Effect.

Legislatures and securities regulators in certain states in which we do business have enacted (or have considered enacting) their own standard of conduct rules for broker-dealers, insurance agents and investment advisers. To date, the States of Nevada, Connecticut, New Jersey and New York have passed legislation or proposed regulations of this sort. The requirements and scope of these state rules are not uniform. Accordingly, we may have to adopt different policies and procedures in different states, which could create added compliance, supervision and sales costs for our Wealth Management

business. Should more states enact similar legislation or regulation, it could result in material additional compliance costs and could have a Material Adverse Effect.

Our Wealth Management business distributes its products and services through financial advisors who affiliate with us as independent contractors. There can be no assurance that legislative, judicial, or regulatory (including tax) authorities will not introduce proposals or assert interpretations of existing rules and regulations that would change, or at least challenge, the classification of our financial advisors as independent contractors. Although we believe we have properly classified our advisors as independent contractors, the IRS or other U.S. federal or state authorities or similar authorities may determine that we have misclassified our advisors as independent contractors for employment tax or other purposes and, as a result, seek additional taxes from us or attempt to impose fines and penalties, which could have a Material Adverse Effect on our business model, financial condition, and results of operations.

In addition, the SEC and FINRA have extensive rules and regulations with respect to capital requirements. As a registered broker-dealer, our Wealth Management business is subject to Rule 15c3-1 (the "Net Capital Rule") under the Securities Exchange Act of 1934, as amended, and related requirements of self-regulatory organizations, which specify minimum capital requirements that are intended to ensure the general soundness and liquidity of broker-dealers. As a result of the Net Capital Rule, our ability to withdraw capital from our subsidiaries that comprise our Wealth Management business could be restricted, which in turn could limit our ability to repay debt, redeem or purchase shares of our outstanding stock, or pay dividends, which could have a Material Adverse Effect. A large operating loss or charge against net capital could adversely affect our ability to expand or even maintain our present levels of business.

Our Wealth Management business offers products sponsored by third parties, including, but not limited to, mutual funds, insurance, annuities and alternative investments. These products are subject to complex regulations that change frequently. Although we have controls in place to facilitate compliance with such regulations, there can be no assurance that our interpretation of the regulations will be consistent with various regulators' interpretations, that our procedures will be viewed as adequate by regulatory examiners, or that the operating subsidiaries will be deemed to be in compliance with regulatory requirements in all material respects. If products sold by our Wealth Management business do not perform as anticipated due to market factors or otherwise, or if product sponsors become insolvent or are otherwise unable to meet their obligations, this could result in material litigation and regulatory action against us. In addition, we could face liabilities for actual or alleged breaches of legal duties to customers with respect to the suitability of the financial products we make available in our open architecture product platform or the investment advice of our financial advisors.

See "Government regulation of our business, including increased regulation or the interpretation of existing laws, rules or regulations, could have a Material Adverse Effect" for additional information regarding the regulation of our business.

The transition of our Wealth Management business to a new clearing platform may negatively impact our operations and our advisors and the customers of our Wealth Management business.

During the second half of 2018, our Wealth Management business completed the conversion (the "Conversion") of our clearing business to Fidelity Clearing & Custody Solutions ("FCCS"). The Conversion involved significant operational, technological, and logistical effort, since it required all of our Wealth Management business and customer accounts to migrate to FCCS's clearing platform, together with all of the underlying customer data. While the conversion of client assets is complete, we continue to acclimate our advisors and customers to FCCS's technology, product offerings, processes and procedures, which we expect will continue through the first half of 2019.

The movement of business to a new clearing firm is an extremely complex and intensive undertaking, and we have committed a significant amount of human, technological, and financial resources to ensure a successful transition. Given the complexity and magnitude of the transition effort, there can be no guarantee that we will not experience unexpected costs, technological failures, incompatibility of systems or policies, or loss of employees, advisors and customers, which could have a Material Adverse Effect. In addition, during the Conversion, we experienced delays in responding to service requests from our advisors. A recurrence of those delays could result in our advisors becoming unsatisfied and leaving.

We may not realize the financial, operational, and customer-experience benefits that we project over the life of our clearing contract with FCCS. Our cash sweep program under the new clearing firm is a significant component of the anticipated financial benefits of the Conversion. The cash sweep program is subject to interest rate volatility. Should the Federal Reserve not increase interest rates at the pace or to the levels anticipated, we would likely recognize lower revenue from the cash sweep program than expected, potentially in a material amount. In addition, our customers may choose to reduce the amount of cash in their accounts, which would reduce the amount of cash in our sweep program. This would reduce our revenue from the cash sweep program, potentially in a material amount. In addition, as part of the Conversion, we plan to implement policies and pricing intended to encourage the conversion of direct-to-fund assets onto FCCS's clearing platform, but we may not realize the level of conversion of such assets onto FCCS's clearing platform that we anticipate. Should the

number of direct-to-fund assets that convert to FCCS's platform over the life of our agreement with FCCS fall short of expectations, we will likely receive less economic benefit from the new clearing arrangement than we expected, which could be material.

The technology, service and product offerings presented by FCCS may not be accepted by our advisors or customers at the levels we anticipate, and may not provide the level of benefits that we expect even if accepted. If a significant number of our advisors or customers are or become dissatisfied by the different technology, systems, processes, policies and products that FCCS offers and they leave it could have a Material Adverse Effect.

In addition, our Wealth Management business is dependent on the performance, liquidity and continuity of its clearing firm. Should FCCS fail to provide clearing services at the contracted levels for any reason or suffer a liquidity event, or if FCCS significantly changes the products and services it offers, it could result in a Material Adverse Effect.

Simultaneously with the conversion to FCCS, we initiated transition to a new investment advisory platform offered by Envestnet Asset Management, Inc. ("Envestnet"). The Envestnet platform is comprehensive and its implementation will substantially change how our Wealth Management business and its Advisors conduct advisory business. Like the clearing firm conversion, implementation of the Envestnet platform was complex and entailed significant effort and commitment from our employees and contractors. While implementation of the Envestnet platform is complete, we continue to enhance the platform to conform to our expectations and to acclimate our staff and advisors to the new platform's workflows, capabilities and procedures. If we are unable to fully integrate the Envestnet platform with our other systems, this could have a Material Adverse Effect. Similarly, if a significant number of our advisors are or become dissatisfied by the transition to the new Envestnet platform, or by the different technology, systems, processes, and policies it offers, they could leave our Wealth Management business, which could have a Material Adverse Effect.

Our operating systems and network infrastructure are subject to significant and constantly evolving cybersecurity and other technological risks, and the security measures that we have implemented to secure confidential and personal information may be breached; a potential breach may pose risks to the uninterrupted operation of our systems, expose us to mitigation costs, litigation, investigation, fines and penalties by authorities, claims by persons whose information was disclosed, and damage to our reputation.

We collect and retain certain sensitive personal data. Our Tax Preparation and Wealth Management businesses collect, use, and retain large amounts of confidential personal and financial information from their customers, including information regarding income, assets, family members, credit cards, tax returns, login credentials and passwords, bank accounts, social security numbers, and healthcare. Maintaining the integrity of our systems and networks is critical to the success of our business operations, including the retention of our customers and advisors, and to the protection of our proprietary information and our customers' personal information. A major breach of our systems or those of our third-party service providers or partners may have materially negative consequences for our businesses, including possible fines, penalties and damages, reduced demand for our services, harm to our reputation and brands, further regulation and oversight by federal or state agencies, and loss of our ability to provide financial transaction services or accept and process customer credit card orders or tax returns. We may detect, or we may receive notices from customers or public or private agencies that they have detected, vulnerabilities in our servers or our software. The existence of vulnerabilities, even if they do not result in a security breach, may harm customer confidence and require substantial resources to address, and we may not be able to discover or remediate such security vulnerabilities before they are exploited.

In addition, hackers may develop and deploy viruses, worms, and other malicious software programs that can be used to attack our offerings. Although we utilize network and application security measures, internal controls, and physical security procedures to safeguard our systems, there can be no assurance that a security breach, intrusion, or loss or theft of personal information will not occur. Any such incident could cause a Material Adverse Effect and require us to expend significant resources to address these problems, including notification under data privacy regulations. In addition, our employees (including temporary and seasonal employees) and contractors may have access to sensitive and personal information of our customers and employees. While we conduct background checks on our employees and contractors and limit access to systems and data, it is possible that one or more of these individuals may circumvent these controls, resulting in a security breach.

It is also possible that unauthorized access to or disclosure of customer data may occur due to inadequate use of security controls by our customers. Accounts created with weak or recycled passwords could allow cyberattackers to gain access to customer data. Unauthorized persons could gain access to customer accounts if customers do not maintain effective access controls of their systems and software. Further, our customers may choose to use the same user ID and password across multiple products and services unrelated to our products. Such customers' login credentials may be stolen from products offered by third-party service providers unrelated to us and the stolen identity information may be used by a malicious third party to access our products, which could result in disclosure of confidential information.

We rely on third-party vendors to host certain of our sensitive and personal information and data through cloud services. While we conduct due diligence on these third-party partners with respect to their security and business controls, we may not have the ability to effectively monitor or oversee the implementation of these control measures, and, in any event, individuals or third parties may be able to circumvent and/or exploit vulnerabilities that may exist in these security and business controls, resulting in a loss of sensitive and personal customer or employee information and data.

While we maintain cyber liability insurance that provides both third-party liability and first-party liability coverages, this insurance is subject to exclusions and may not be sufficient to protect us against all losses. In addition, the trend toward broad consumer and general public notification of such incidents could exacerbate the harm to our business, financial condition, or results of operations. Even if we successfully protect our technology infrastructure and the confidentiality of sensitive data, we may incur significant expenses in connection with our responses to any such attacks as well as the adoption, implementation, and maintenance of appropriate security measures. We could also suffer harm to our business and reputation if attempted security breaches are publicized. We cannot be certain that advances in criminal capabilities, discovery of new vulnerabilities, attempts to exploit vulnerabilities in our systems, data thefts, physical system or network break-ins, inappropriate access, or other developments will not compromise or breach the technology or other security measures protecting the networks and systems used in connection with our businesses.

Concerns about the current privacy and cybersecurity environment, generally, could deter current and potential customers from adopting our products and services and damage our reputation.

The continued occurrence of cyberattacks and data breaches on governments, businesses and consumers in general, indicates that we operate in an external environment where cyberattacks and data breaches are becoming increasingly common. If the global cybersecurity environment worsens, and there are increased instances of security breaches of third-party offerings where consumers' data and sensitive information is compromised, consumers may be less willing to use online offerings, particularly offerings like ours in which customers often share sensitive financial data. In addition, the increased availability of data obtained as a result of breaches of third-party offerings could make our own products more vulnerable to fraudulent activity. Even if our products are not affected directly by such incidents, they could damage our reputation and deter current and potential customers from adopting our products and services or lead customers to cease using online and connected software products to transact financial business altogether.

In addition, we currently plan to increase our capture and use of user data for marketing purposes. In connection with our use of user data for marketing efforts, concerns may be expressed about whether our products, services, or processes compromise the privacy of users, customers and others. Concerns about our practices with regard to the collection, use, disclosure or security of personal information or other privacy related matters, even if unfounded, could damage the reputation of our business and our brands and adversely affect our operating results.

Stolen identity refund fraud could impede our Tax Preparation customers' ability to timely and successfully file their tax returns and receive their tax refunds, and could diminish customers' perceptions of the security and reliability of our tax preparation products and services, resulting in negative publicity. Increased governmental regulation to attempt to combat that fraud could result in a Material Adverse Effect.

Criminals may utilize stolen information obtained through hacking, phishing, and other means of identity theft in order to electronically file fraudulent federal and state tax returns. As a result, impacted taxpayers must complete additional forms and go through additional steps in order to report to appropriate authorities that their identities have been stolen and their tax returns were filed fraudulently. Though we offer assistance in the refund recovery process, stolen identity refund fraud could impede our customers' ability to timely and successfully file their returns and receive their tax refunds, and could diminish customers' perceptions of the security and reliability of our tax preparation products and services, resulting in negative publicity, despite there having been no breach in the security of our systems. In addition, if stolen identity refund fraud is perpetrated at a material level through our tax preparation products or services, state, federal, or foreign tax authorities may refuse to allow us to continue to process our customers' tax returns electronically. As a result, stolen identity fraud could have a Material Adverse Effect on our Tax Preparation business.

Federal, state, and foreign governmental authorities in jurisdictions in which we operate have taken action, and may take action in the future, in an attempt to combat stolen identity refund fraud, which may require changes to our systems and business practices in ways we cannot anticipate. These actions may have a Material Adverse Effect on our Tax Preparation business.

Complex and evolving U.S. and international laws and regulation regarding privacy and data protection could result in claims, changes to our business practices, penalties, increased cost of operations or otherwise harm our business.

Regulation related to the provision of online services is evolving as federal, state and foreign governments continue to adopt new, or modify existing, laws and regulations addressing data privacy and the collection, processing, storage, transfer and use of data. This includes, for example, the European Union's new regulation, the General Data Protection Regulation, which went into effect on May 25, 2018, and the new California Consumer Protection Act, which will become effective on January 1, 2020. If we are unable to engineer products that meet these evolving requirements or help our customers meet their obligations under these or other new data regulations, we might experience reduced demand for our offerings. Further, penalties for noncompliance with these laws may be significant.

Other governmental authorities throughout the U.S. and around the world are considering similar types of legislative and regulatory proposals concerning data protection. Each of these privacy, security and data protection laws and regulations could impose significant limitations, require changes to our business, require notification to customers or workers of a security breach, restrict our use or storage of personal information, or cause changes in customer purchasing behavior, which may make our business more costly, less efficient or impossible to conduct, and may require us to modify our current or future products or services, which may make customers less likely to purchase our products and may harm our future financial results. Additionally, any actual or alleged noncompliance with these laws and regulations could result in negative publicity and subject us to investigations, claims or other remedies, including demands that we modify or cease existing business practices, and expose us to significant fines, penalties and other damages. We have incurred, and may continue to incur, significant expenses to comply with existing privacy and security standards and protocols imposed by law, regulation, industry standards or contractual obligations.

# If we are unable to develop, manage, and maintain critical third-party business relationships for our Tax Preparation and Wealth Management businesses, it could result in a Material Adverse Effect.

Our Tax Preparation and Wealth Management businesses are dependent on the strength of our business relationships and our ability to continue to develop, maintain, and leverage new and existing relationships. We rely on various third-party partners, including software and service providers, suppliers, vendors, distributors, contractors, financial institutions, and licensing partners, among others, in many areas of these businesses to deliver our services and products. In certain instances, the products or services provided through these third-party relationships may be difficult to replace or substitute, depending on the level of integration of the third party's products or services into, or with, our offerings and/or the general availability of such third party's products and services. In addition, there may be few or no alternative third-party providers or vendors in the market. The failure of third parties to provide acceptable and high-quality products, services, and technologies or to update their products, services, and technologies may result in a disruption to our business operations, which may materially reduce our revenues and profits, cause us to lose customers, and damage our reputation. Alternative arrangements and services may not be available to us on commercially reasonable terms or we may experience business interruptions upon a transition to an alternative partner.

Our Wealth Management business does not offer any proprietary financial products. Instead, it distributes investment and insurance products through distribution agreements with third-party financial institutions, including banks, mutual funds, and insurance companies. These products are sold by our advisors, who are independent contractors. Maintaining and deepening relationships with these unaffiliated distributors and advisors is an important part of our growth strategy because strong third-party distribution arrangements enhance our ability to market our products and increase our advisory assets, revenues, and profitability. There can be no assurance that the distribution and advisor relationships we have established will continue, or that they will continue under existing terms. Our distribution partners and advisors may cease to operate, consolidate, institute cost-cutting efforts, discontinue product sales or compensation streams, or otherwise terminate their relationship with us. Any such reduction in access to third-party distributors and advisors may have a material adverse effect on our ability to market our products and to generate revenue in our Wealth Management segment. In addition, there are risks associated with our third-party clearing firm that we rely on to provide clearing services for our Wealth Management business that are discussed above.

Access to investment and insurance product distribution channels is subject to intense competition due to the large number of competitors and products in the broker-dealer, investment advisory and insurance industries. Relationships with distributors are subject to periodic negotiation that may result in increased distribution costs and/or reductions in the amount of revenue we realize based on sales of particular products or customer assets. In addition, regulatory changes (such as the SEC's proposed "best interest" standard) may negatively impact our revenues and profits related to particular products or services. Any increase in the costs to distribute our products or reduction in the type or amount of products made available for sale, or revenue associated with those products, could have a Material Adverse Effect.

The seasonality of our Tax Preparation business requires a precise development and release schedule and any delays or issues with accuracy or quality may damage our reputation and could result in a Material Adverse Effect.

Our tax preparation software and online service must be ready to launch in final form near the beginning of each calendar year to take advantage of the full tax season. We must update the code for our software and service on a precise schedule each year to account for annual changes in tax laws and regulations and ensure that the software and service are accurate. Delayed and unpredictable changes to federal and state tax laws and regulations can cause an already tight development cycle to become even more challenging. If we are unable to meet this precise schedule and we launch our software and service late, we risk losing customers to our competitors. If we cannot develop our software with a high degree of accuracy and quality, we risk errors in the tax returns that are generated. Such errors could result in loss of reputation, lower customer retention, or legal claims, fees, and payouts related to the warranty on our software and service, which could result in a Material Adverse Effect.

The specialized and highly seasonal nature of our Tax Preparation business presents financial risks and operational challenges, which, if not satisfactorily addressed, could result in a Material Adverse Effect.

Our Tax Preparation business is highly seasonal, with a significant portion of our annual revenue for such services earned in the first four months of our fiscal year. The concentration of our revenue-generating activity during this relatively short period presents a number of challenges for us, including cash and resource management during the last eight months of our fiscal year, when our Tax Preparation business generally operates at a loss and incurs fixed costs of preparing for the upcoming tax season, responding to changes in competitive conditions, including marketing, pricing, and new product offerings, which could affect our position during the tax season, and ensuring optimal uninterrupted operations and service delivery during the tax season. If we experience significant business disruptions during the tax season or if we are unable to satisfactorily address the challenges described above and related challenges associated with a seasonal business, it could result in a Material Adverse Effect.

If our Tax Preparation business fails to process transactions effectively or fails to adequately protect against disputed or potential fraudulent activities, it could have a Material Adverse Effect.

Our Tax Preparation business processes a significant volume and dollar value of transactions on a daily basis, particularly during tax season. Due to the size and volume of transactions that we handle, effective processing systems and controls are essential to ensure that transactions are handled appropriately. Despite our efforts, it is possible that we may make errors or that fraudulent activity may affect our services. In addition to any direct damages and fines that may result from any such problems, which may be substantial, a loss of confidence in our controls may materially harm our business and damage our brand. The systems supporting our Tax Preparation business are comprised of multiple technology platforms, some of which are difficult to scale. If we are unable to effectively manage our systems and processes, we may be unable to process customer data in an accurate, reliable, and timely manner, which could result in a Material Adverse Effect.

# The United States government's inability to agree on a federal budget may adversely impact our operations and financial results.

In the past, the failure of the United States government to timely complete its budget process has resulted in shutdowns of the federal government, including most recently a shutdown that began on December 22, 2018 and lasted until January 25, 2019. During these shutdowns, certain regulatory agencies, such as the Internal Revenue Service and the United States Department of the Treasury, have had to furlough critical employees and cease certain critical activities.

During a prolonged government shutdown, the ability of the Internal Revenue Service to timely review and process tax return filings may be significantly delayed, and representatives of the Internal Revenue Service may be unable to answer crucial taxpayer questions. Even after the shutdown has ended, the Internal Revenue Service may be significantly delayed in processing tax return filings as a result of accumulating a backlog of filings during the shutdown. These may be further exacerbated in years where there are significant changes to existing tax legislation, such as the application of the Tax Legislation to tax return filings related to the 2018 tax year. Any uncertainty surrounding the ability of the Internal Revenue Service to process tax return filings and respond to taxpayer questions could cause our customers not to purchase or to delay purchasing our products and services, thereby negatively impacting our revenues and future financial results, which could result in a Material Adverse Effect.

Our website and transaction management software, data center systems, or the systems of third-party co-location facilities and cloud service providers could fail or become unavailable or otherwise be inadequate, which could materially harm our reputation and/or result in a material loss of revenues and current or potential customers and have a Material Adverse Effect.

Any system interruptions that result in the unavailability or unreliability of our websites, transaction processing systems, or network infrastructure could materially reduce our revenue and impair our ability to properly process transactions. We use both internally developed and third-party systems, including cloud computing and storage systems, for our online

services and certain aspects of transaction processing. Some of our systems are relatively new, and we have some systems that may need updating, which could cause them to be subject to failure or unreliability. Any system unavailability or unreliability may cause unanticipated system disruptions, slower response times, degradation in customer satisfaction, additional expense, or delays in reporting accurate financial information. For example, we have been migrating data to the cloud. This migration has been costly and has diverted some of management's attention and resources in order to ensure a smooth transition to the cloud.

Our data centers and cloud service could be susceptible to damage or disruption, which could have a Material Adverse Effect. Our Tax Preparation and Wealth Management businesses have business continuity plans that include secondary disaster recovery centers, but if their primary data centers fail and those disaster recovery centers do not fully restore the failed environments, our business could suffer. In particular, if such interruption occurs during the tax season, it could have a Material Adverse Effect on our Tax Preparation business.

We regularly invest resources to update and improve our internal information technology systems and software platforms. If we experience prolonged delays or unforeseen difficulties in updating and upgrading our systems and architecture, we may experience outages and may not be able to deliver certain offerings and develop new offerings and enhancements that we need to remain competitive. Such improvements and upgrades are often complex, costly and time consuming. In addition, such improvements can be challenging to integrate with our existing technology systems, or may uncover problems with our existing technology systems. Unsuccessful implementation of hardware or software updates and improvements could result in outages, disruption in our business operations, loss of revenue or damage to our reputation.

Our systems and operations, and those of our third-party service providers and partners, could be damaged or interrupted by fire, flood, earthquakes, other natural disasters, power loss, telecommunications failure, internet breakdown, break-in, human error, software bugs, hardware failures, malicious attacks, computer viruses, computer denial of service attacks, terrorist attacks, or other events beyond our control. Such damage or interruption may affect internal and external systems that we rely upon to provide our services, take and fulfill customer orders, handle customer service requests, and host other products and services. During the period in which services are unavailable, we could be unable or severely limited in our ability to generate revenues, and we may also be exposed to liability from those third parties to whom we provide services. We could face significant losses as a result of these events, and our business interruption insurance may not be adequate to compensate us for all potential losses, which could result in a Material Adverse Effect.

# Current and future litigation or regulatory proceedings or adverse court interpretations of the laws under which the Company operates could have a Material Adverse Effect.

Many aspects of our business involve substantial risks of liability. We are currently subject to lawsuits and are likely to be subject to litigation in the future. In highly volatile markets, the volume of claims and amount of damages sought in litigation and regulatory proceedings against financial institutions have historically increased. Any lawsuits to which we are subject, such as purported class actions, shareholder derivative lawsuits or claims by wealth management customers, could result in substantial expenditures, generate adverse publicity and could significantly impair our business, or force us to cease offering certain products or services. Defense of any lawsuit, even if successful, could require substantial time and attention of our management and could require the expenditure of significant amounts for legal fees and other related costs. In addition, litigation or regulatory proceedings or actions brought by state or federal agencies relating to our products or services may result in additional restrictions on the offering of certain of our products or services. To the extent that any such additional restrictions or legal claims limit our ability to offer such products or services, it could result in a Material Adverse Effect.

# If we are unable to hire, retain, and motivate highly qualified employees, including our key employees, we may not be able to successfully manage our businesses

Our future success depends on our ability to identify, attract, hire, retain, and motivate highly skilled management, technical, sales and marketing, and corporate development personnel, including personnel with experience and expertise in the wealth management, tax preparation, and technology industries to support our new strategic focus. Qualified personnel with experience relevant to our businesses are scarce, and competition to recruit them is intense. If we fail to successfully hire and retain a sufficient number of highly qualified employees, we may have difficulties in supporting or expanding our businesses. Realignments of resources, reductions in workforce, or other operational decisions have created and could continue to create an unstable work environment and may have a negative effect on our ability to hire, retain, and motivate employees.

Our business and operations are substantially dependent on the performance of our key employees. Changes of management or key employees may disrupt operations, which may materially and adversely affect our business and financial results or delay achievement of our business objectives. In addition, if we lose the services of one or more key employees and are unable to recruit and retain a suitable successor with relevant experience, we may not be able to successfully and timely manage our business or achieve our business objectives. There can be no assurance that any retention program we initiate will be successful at retaining employees, including key employees.

We use stock options, restricted stock units, and other equity-based awards to recruit and retain senior-level employees. With respect to those employees to whom we issue such equity-based awards, we face a significant challenge in retaining them if the value of equity-based awards in the aggregate or individually is either not deemed by the employee to be substantial enough or deemed so substantial that the employee leaves after their equity-based awards vest. If our stock price does not increase significantly above the exercise prices of our options, we may need to issue new equity-based awards in order to motivate and retain our key employees. We may undertake or seek stockholder approval to undertake other equity-based programs to retain our employees, which may be viewed as dilutive to our existing stockholders or may increase our compensation costs. There can be no assurance that any such programs, if approved by our stockholders, or any other incentive programs, would be successful in motivating and retaining our employees.

# We may be negatively impacted by the recently passed Tax Cuts and Jobs Act or by any future changes in tax laws.

On December 22, 2017, President Trump signed the Tax Cuts and Jobs Act into law. It is difficult to know at this time how our customers will view the new federal tax laws that were enacted in late 2017 because 2019 will be the first tax season where these laws are in effect. Possible outcomes include a short-term or long-term increase in customers that prefer professional tax advice and preparation services rather than using our software or we may see a change in our how customers value our software services as customers may perceive their tax preparation has become simpler as a result of the new tax laws, which could result in lower demand for our products and could reduce revenue and/or the number of units sold.

Changes in state and federal tax laws require updates to our tax preparation software used in our Tax Preparation business. Such updates are costly and may be time consuming to ensure that they accurately reflect the new laws that are adopted. In addition, further changes in the way that state and federal governments structure their taxation regimes could also cause a Material Adverse Effect on our Tax Preparation business. The introduction of a simplified or flattened federal or state taxation structure may make our services less necessary or attractive to individual filers, which could reduce revenue and the number of units sold. We also face risk from the possibility of increased complexity in taxation structures, which may encourage some of our customers to seek professional tax advice instead of using our software or services. In the event that such changes to tax structures cause us to lose market share or cause a decline in customers, it could cause a Material Adverse Effect.

# Our risk management and conflicts of interest policies and procedures may be ineffective or leave us exposed to unidentified or unanticipated risks.

We are subject to the risks of errors and misconduct by our employees and financial advisors, such as fraud, non-compliance with policies, recommending transactions that are not suitable, and improperly using or disclosing confidential information. Although we have internal controls and other risk-mitigating factors in place, this type of conduct is difficult to detect and deter, and could materially harm our business, results of operations or financial condition. We are further subject to the risk of nonperformance or inadequate performance of contractual obligations by third-party vendors of products and services that are used in our businesses. Management of operational, legal and regulatory risks requires, among other things, policies and procedures to record properly and verify a large number of transactions and events, and these policies and procedures may not be fully effective in mitigating our risk exposure in all market environments or against all types of risk. Insurance and other traditional risk-shifting tools may be held by or available to us in order to manage certain exposures, but they are subject to terms such as deductibles, coinsurance, limits and policy exclusions, as well as the risk of counterparty denial of coverage, default or insolvency.

In our Wealth Management business, prevention and detection of wrongdoing or fraud by our advisors, who are not our employees and tend to be located remotely from our headquarters, present unique challenges. There cannot be any assurance that misconduct by our advisors will not lead to a Material Adverse Effect on our business.

RIAs have fiduciary obligations that require us and our advisors to act in the best interests of our customers and to disclose any material conflicts of interest. Conflicts of interest are under growing scrutiny by U.S. federal and state regulators. Our risk management processes include addressing potential conflicts of interest that arise in our business. Management of potential conflicts of interest has become increasingly complex. A perceived or actual failure to address conflicts of interest adequately could affect our reputation, the willingness of customers to transact business with us or give rise to litigation or regulatory actions, any of which could have a Material Adverse Effect.

# Our business depends on our strong reputation and the value of our brands, which could be negatively impacted by poor performance.

Developing and maintaining awareness of our brands is critical to achieving widespread acceptance of our existing and future products and services and is an important element in attracting new customers. Adverse publicity (whether or not justified) relating to regulatory proceedings or other events or activities attributed to our businesses, our employees, our vendors, or our partners may tarnish our reputation and reduce the value of our brands. Damage to our reputation and loss of

brand equity may reduce demand for our products and services and have a material adverse effect on our future financial results. Such damage also would require additional resources to rebuild our reputation and restore the value of the brands.

If others claim that our services infringe upon their intellectual property rights, we may be forced to seek expensive licenses, reengineer our services, engage in expensive and time-consuming litigation, or stop marketing and licensing our services.

Companies and individuals with rights relating to the technology industry have frequently resorted to litigation regarding intellectual property rights. These parties have in the past made, and may in the future make, claims against us alleging infringement of patents, copyrights, trademarks, trade secrets, or other intellectual property or proprietary rights, or alleging unfair competition or violations of privacy or publicity rights. Responding to any such claims could be time-consuming, result in costly litigation, divert management's attention, cause product or service release delays, or require removal or redesigning of our products or services, payment of damages for infringement, or entry into royalty or licensing agreements. Our technology, services, and products may not be able to withstand any third-party claims or rights against their use. In some cases, the ownership or scope of an entity's or person's rights is unclear. In addition, the ownership or scope of such rights may be altered by changes in the legal landscape, such as through developments in U.S. or international intellectual property laws or regulations or through court, agency, or regulatory board decisions. If a successful claim of infringement were made against us and we could not develop non-infringing technology or content or license the infringed or similar technology or content on a timely and cost-effective basis, our financial condition and results of operations could be materially and adversely affected.

We do not regularly conduct patent searches to determine whether the technology used in our products or services infringes patents held by third parties. Patent searches may not return every issued patent or patent application that may be deemed relevant to a particular product or service. It is therefore difficult to determine, with any level of certainty, whether a particular product or service may be construed as infringing a current or future U.S. or foreign patent.

We rely heavily on our technology and intellectual property, but we may be unable to adequately or cost-effectively protect or enforce our intellectual property rights, thereby weakening our competitive position and negatively impacting our business and financial results. We may have to litigate to enforce our intellectual property rights, which can be time consuming, expensive, and difficult to predict.

To protect our rights related to our services and technology, we rely on a combination of copyright and trademark laws, trade secrets, confidentiality agreements with employees and third parties, and protective contractual provisions. We also rely on laws pertaining to trademarks and domain names to protect the value of our corporate brands and reputation. Despite our efforts to protect our proprietary rights, unauthorized parties may copy aspects of our services or technology, obtain and use information, marks, or technology that we regard as proprietary, or otherwise violate or infringe our intellectual property rights. In addition, it is possible that others could independently develop substantially equivalent intellectual property. Effectively policing the unauthorized use of our services and technology is time-consuming and costly, and the steps taken by us may not prevent misappropriation of our technology or other proprietary assets. If we do not effectively protect our intellectual property, or if others independently develop substantially equivalent intellectual property, our competitive position could be materially weakened.

We may seek to acquire companies or assets that complement our Wealth Management and Tax Preparation businesses, and if we are unsuccessful in completing any such acquisitions on favorable terms or integrating any company acquired it could result in a Material Adverse Effect.

We may seek to acquire companies or assets that complement our Wealth Management and Tax Preparation businesses. There can be no guarantee that any of the opportunities that we evaluate will result in the purchase by us of any business or asset being evaluated, or that, if acquired, we will be able to successfully integrate such acquisition.

If we are successful in our pursuit of any complementary acquisition opportunities, we intend to use available cash, debt and/or equity financing, and/or other capital or ownership structures designed to diversify our capital sources and attract a competitive cost of capital, all of which may change our leverage profile. There are a number of factors that impact our ability to succeed in acquiring the companies and assets we identify, including competition for these companies and assets, sometimes from larger or better-funded competitors. As a result, our success in completing acquisitions is not guaranteed. Our expectation is that, to the extent we are successful, any acquisitions will be additive to our businesses, taking into account potential benefits of operational synergies. However, these new business additions and acquisitions, if any, involve a number of risks and may not achieve our expectations, and, therefore, we could be materially and adversely affected by any such new business additions or acquisitions. There can be no assurance that the short or long-term value of any business or technology that we develop or acquire will be equal to the value of the cash and other consideration that we pay or expenses we incur.

# RISKS RELATED TO OUR FINANCING ARRANGEMENTS

We incurred debt in connection with the repayment of our credit facility used for the acquisition of HD Vest and the redemption of our convertible senior notes and may incur future debt, which may materially and adversely affect our financial condition and future financial results.

In May 2017, we entered into a credit agreement with a syndicate of lenders in order to (a) refinance the credit facilities previously entered into in 2015 to finance the HD Vest acquisition, (b) redeem our convertible notes that were outstanding at the time, and (c) provide a term loan and revolving line of credit for future working capital, capital expenditure and general business purposes (the "*Blucora senior secured credit facilities*"). As of December 31, 2018, we had \$265.0 million of outstanding indebtedness under the term loan, and we had not borrowed any amounts under the revolving credit facility. The final maturity date of the term loan is May 22, 2024. Under the terms of the revolving credit facility, we may borrow up to \$50.0 million.

This borrowing may materially and adversely affect our financial condition and future financial results by, among other things:

- · increasing our vulnerability to downturns in our businesses, to competitive pressures, and to adverse economic and industry conditions;
- requiring the dedication of a portion of our expected cash from operations to service the indebtedness, thereby reducing the amount of expected cash flow available for other purposes, including capital expenditures and complementary acquisitions;
- increasing our interest payment obligations in the event that interest rates rise; and
- · limiting our flexibility in planning for, or reacting to, changes in our businesses and our industries.

The Blucora senior secured credit facilities impose certain restrictions on us, including restrictions on our ability to create liens, incur indebtedness and make investments. In addition, the Blucora senior secured credit facilities include covenants, the breach of which may cause the outstanding indebtedness to be declared immediately due and payable. This borrowing, and our ability to repay it, may also negatively impact our ability to obtain additional financing in the future and may affect the terms of any such financing.

In addition, we or our subsidiaries, may incur additional debt in the future. Any additional debt may result in risks similar to those discussed above or in other risks specific to the credit agreements entered into for those debts.

Existing cash and cash equivalents and cash generated from operations may not be sufficient to meet our anticipated cash needs for servicing debt, working capital, and capital expenditures.

Although we believe that existing cash and cash equivalents and cash generated from operations will be sufficient to meet our anticipated cash needs for servicing debt, working capital, and capital expenditures for at least the next 12 months, the underlying levels of revenues and expenses that we project may not prove to be accurate. As of December 31, 2018, we had \$265.0 million outstanding under our term loan. Servicing this debt will require the dedication of a portion of our expected cash flow from operations, thereby reducing the amount of our cash flow available for other purposes. In addition, our ability to make scheduled payments of the principal of, to pay interest on, or to refinance our indebtedness depends on our future performance, which is subject to economic, financial, competitive, and other factors beyond our control. Our businesses may not continue to generate cash flow from operations in the future sufficient to service our debt and make necessary capital expenditures. If we are unable to generate such cash flow, we may be required to adopt one or more alternatives, such as selling assets, restructuring debt, or obtaining additional equity capital on terms that may be onerous or highly dilutive. Our ability to refinance our indebtedness will depend on the capital markets and our financial condition and results at such time. We may not be able to engage in any of these activities or engage in these activities on desirable terms, which could result in a default on our debt obligations.

In addition, we may evaluate complementary acquisitions of businesses, products, or technologies from time to time. Any such transactions, if completed, may use a significant portion of our cash balances and marketable investments. If we are unable to liquidate our investments when we need liquidity for complementary acquisitions or for other business purposes, we may need to change or postpone such acquisitions or find alternative financing for them. We may seek additional funding through public or private financings, through sales of equity, or through other arrangements. Our ability to raise funds may be materially and adversely affected by a number of factors, including factors beyond our control, such as economic conditions in the markets in which we operate and increased uncertainty in the financial, capital, and credit markets. Adequate funds may not be available when needed or may not be available on favorable terms. If we raise additional funds by issuing equity securities, dilution to existing stockholders may result. Any sale of a substantial amount of our common stock in the public market, either in the initial issuance or in a subsequent resale, could have a material adverse effect on the market price of our common stock. If funding is insufficient at any time in the future, we may be unable, or delayed in our ability, to develop or enhance our

products or services, take advantage of business opportunities, or respond to competitive pressures, any of which could materially harm our business.

### OTHER RISKS

### Our stock price has been highly volatile and such volatility may continue.

The trading price of our common stock has been highly volatile, and such volatility does not always correspond to fluctuations in the market. Between January 1, 2017 and December 31, 2018, our closing stock price ranged from \$14.45 to \$40.25. On February 22, 2019, the closing price of our common stock was \$27.02. Our stock price could decline or fluctuate significantly in response to many factors, including the other risks discussed in this report and the following:

- actual or anticipated variations in quarterly and annual results of operations;
- impairment charges, changes in or loss of material contracts and relationships, dispositions or announcements of complementary acquisitions, or other business developments by us, our partners, or our competitors;
- conditions or trends in the tax preparation or wealth management markets or changes in market share;
- · changes in general conditions in the United States and global economies or financial markets;
- announcements of technological innovations or new services by us or our competitors;
- · changes in financial estimates or recommendations by securities analysts;
- · disclosures of any accounting issues, such as restatements or material weaknesses in internal control over financial reporting;
- equity issuances resulting in the dilution of stockholders;
- · the adoption of new regulations or accounting standards;
- adverse publicity (whether justified or not) with respect to our business; and
- · announcements or publicity relating to litigation or governmental enforcement actions.

In addition, the equities market has experienced extreme price and volume fluctuations, and our stock has been particularly susceptible to such fluctuations. Often, class action litigation has been instituted against companies after periods of volatility in the price of such companies' stock. We have been defendants in such class action litigation in prior periods and could be subject to future litigation, potentially resulting in substantial cost and diversion of management's attention and resources.

# Our financial results may fluctuate, which could cause our stock price to be volatile or decline.

Our financial results have varied on a quarterly basis and are likely to continue to fluctuate in the future. These fluctuations could cause our stock price to be volatile or decline. Many factors could cause our quarterly results to fluctuate materially, including but not limited to:

- the inability of any of our businesses to implement business plans and to meet our expectations;
- the seasonality of our Tax Preparation business and the resulting large quarterly fluctuations in our revenues;
- · variable demand for our services, rapidly evolving technologies and markets, and consumer preferences;
- the level and mix of total client assets and advisory assets, which are subject to fluctuation based on market conditions and customer activity;
- the mix of revenues generated by existing businesses, discontinued operations or other businesses that we develop or acquire;
- changes in interest rates affecting cash sweep revenue;
- volatility in stock markets impacting the value of our advisory assets;
- gains or losses driven by fair value accounting;
- litigation expenses and settlement costs;
- · misconduct by employees and/or HD Vest financial advisors, which is difficult to detect and deter;
- expenses incurred in finding, evaluating, negotiating, consummating, and integrating acquisitions;
- · impairment or negative performance of the many different industries and counterparties we rely on and are exposed to;
- any restructuring charges we may incur;
- any economic downturn, which could result in lower acceptance rates on premium products and services offered by our Wealth Management business and impact the commissions and fee revenues of our financial advisory services;
- new court rulings, or the adoption of new or interpretation of existing laws, rules, or regulations, that adversely affect our business or that otherwise increase our potential liability or compliance costs;
- impairment in the value of long-lived assets or the value of acquired assets, including goodwill, technology, and acquired contracts and relationships; and
- · the effect of changes in accounting principles or standards or in our accounting treatment of revenues or expenses.

For these reasons, among others, you should not rely on period-to-period comparisons of our financial results to forecast our future performance. Furthermore, our fluctuating operating results may fall below the expectations of securities analysts or investors and financial results volatility could make us less attractive to investors, either of which could cause the trading price of our stock to decline.

# Our utilization of our federal net operating loss carryforwards ("NOLs") may be severely limited or potentially eliminated.

As of December 31, 2018, we had federal NOLs of \$454.5 million that will expire primarily between 2020 and 2027, with the majority of them expiring between 2020 and 2024. We are currently able to offset all of our tax liabilities with our federal NOLs, but we may not generate sufficient taxable income in future years to utilize all of our NOLs prior to their expiration. If our federal NOLs expire unused, their full benefit will not be achieved. In addition, in years where our income exceeds our federal NOLs, which we expect to begin occurring in 2022, we will be required to make additional income tax payments.

In addition, if we were to have a change of ownership within the meaning of Section 382 of the Internal Revenue Code (defined as a cumulative change of 50 percentage points or more in the ownership positions of certain stockholders owning five percent or more of a company's common stock over a three-year rolling period), then under certain conditions, the amount of NOLs we could use in any one year could be limited. Our certificate of incorporation imposes certain limited transfer restrictions on our common stock that we expect will assist us in preventing a change of ownership and preserving our NOLs, but there can be no assurance that these restrictions will be sufficient. In addition, other restrictions on our ability to use the NOLs may be triggered by a merger or acquisition, depending on the structure of such a transaction. It is our intention to limit the potential impact of these restrictions, but there can be no guarantee that such efforts will be successful.

If we are unable to use our federal NOLs before they expire, or if the use of this tax benefit is severely limited or eliminated, there could be a material reduction in the amount of after-tax income and cash flow from operations, and it could have an effect on our ability to engage in certain transactions.

# Delaware law and our charter documents may impede or discourage a takeover, which could cause the market price of our shares to decline.

We are a Delaware corporation, and the anti-takeover provisions of Delaware law impose various impediments to the ability of a third party to acquire us, even if a change of control would be beneficial to our existing stockholders. For example, Section 203 of the Delaware General Corporation Law may discourage, delay, or prevent a change in control by prohibiting us from engaging in a business combination with an interested stockholder for a period of three years after the person becomes an interested stockholder. In addition, our certificate of incorporation and bylaws contain provisions that may discourage, delay, or prevent a third party from acquiring us without the consent of our board of directors, even if doing so would be beneficial to our stockholders. Provisions of our charter documents that could have an anti-takeover effect include:

- the classification of our board of directors, which is being phased out between 2017 and 2020, into three groups so that directors serve staggered three-year terms, which may make it difficult for a potential acquirer to gain control of our board of directors;
- the requirement for supermajority approval by stockholders for certain business combinations;
- the ability of our board of directors to authorize the issuance of shares of undesignated preferred stock without a vote by stockholders;
- the ability of our board of directors to amend or repeal our bylaws;
- limitations on the removal of directors;
- limitations on stockholders' ability to call special stockholder meetings;
- advance notice requirements for nominating candidates for election to our board of directors or for proposing matters that can be acted upon by stockholders at stockholder meetings; and
- · certain restrictions in our certificate of incorporation on transfers of our common stock designed to preserve our federal NOLs.

At our 2009 annual meeting, our stockholders approved an amendment to our certificate of incorporation that restricts any person or entity from attempting to transfer our stock, without prior permission from the Board of Directors, to the extent that such transfer would (i) create or result in an individual or entity becoming a five-percent stockholder of our stock, or (ii) increase the stock ownership percentage of any existing five-percent stockholder. This amendment provides that any transfer that violates its provisions shall be null and void and would require the purported transferee to, upon our demand, transfer the shares that exceed the five percent limit to an agent designated by us for the purpose of conducting a sale of such excess shares.

This provision in our certificate of incorporation may make the acquisition of Blucora more expensive to the acquirer and could significantly delay, discourage, or prevent third parties from acquiring Blucora without the approval of our board of directors.

# ITEM 1B. Unresolved Staff Comments

None.

# ITEM 2. Properties

All of our facilities are leased. We believe our properties are suitable and adequate for our present needs.

Our principal corporate office is located in Irving, Texas. The headquarters and data center facility for our HD Vest business, which comprises our Wealth Management segment, are in Irving, Texas, and we have a backup data center for our HD Vest business in Elk Grove, Illinois, as well as access to multiple disaster recovery and data centers across the country through a third party vendor. The headquarters for our TaxAct business, which comprises our Tax Preparation segment, is in Cedar Rapids, Iowa. Our TaxAct business leverages cloud computing for primary and disaster recovery data services.

# ITEM 3. Legal Proceedings

See "Note 11: Commitments and Contingencies" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report for information regarding legal proceedings.

# **ITEM 4. Mine Safety Disclosures**

None.

# PART II

# ITEM 5. Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities

# Market for Our Common Stock

Our common stock trades on the NASDAQ Global Select Market under the symbol "BCOR." On February 22, 2019, the last reported sale price for our common stock on the NASDAQ Global Select Market was \$27.02 per share.

# Holders

As of February 22, 2019, there were 359 holders of record of our common stock. A substantially greater number of holders are beneficial owners whose shares are held of record by banks, brokers, and other financial institutions.

# **Share Repurchases**

There were no stock repurchases in 2018 and 2017.

# ITEM 6. Selected Financial Data

The following data is derived from our audited consolidated financial statements and should be read along with "Management's Discussion and Analysis of Financial Condition and Results of Operations" in Part II Item 7, our consolidated financial statements and notes in Part II Item 8, and the other financial information included elsewhere in this report.

|   |            | Years ended December 31, |          |    |           |       |                  |     |           |    |          |
|---|------------|--------------------------|----------|----|-----------|-------|------------------|-----|-----------|----|----------|
|   |            | 2018 2017 2016 2015      |          |    |           |       |                  |     |           |    | 2014     |
| Consolidated Statements of Operations Data:                     | (1)        |                          |          |    | (In thou  | sands | s, except per sh | are | data)     |    |          |
| Revenue:  |            |                          |          |    |           |       |                  |     |           |    |          |
| Wealth management services revenue                              |            | \$                       | 373,174  | \$ | 348,620   | \$    | 316,546          | \$  | _         | \$ |          |
| Tax preparation services revenue                                |            |                          | 187,282  |    | 160,937   |       | 139,365          |     | 117,708   |    | 103,719  |
| Total revenue   |            |                          | 560,456  |    | 509,557   |       | 455,911          |     | 117,708   |    | 103,719  |
| Operating income (loss)   |            |                          | 67,677   |    | 48,037    |       | 37,117           |     | (4,807)   |    | 4,603    |
| Other loss, net   |            |                          | (15,797) |    | (44,551)  |       | (39,781)         |     | (12,542)  |    | (13,489) |
| Income (loss) from continuing operations before incom-<br>taxes | e          |                          | 51,880   |    | 3,486     |       | (2,664)          |     | (17,349)  |    | (8,886)  |
| Income tax benefit (expense)                                    |            |                          | (311)    |    | 25,890    |       | 1,285            |     | 4,623     |    | 3,342    |
| Income (loss) from continuing operations                        |            |                          | 51,569   |    | 29,376    |       | (1,379)          |     | (12,726)  |    | (5,544)  |
| Discontinued operations, net of income taxes                    | (2)        |                          | _        |    | _         |       | (63,121)         |     | (27,348)  |    | (30,003) |
| Net income (loss)   |            |                          | 51,569   |    | 29,376    |       | (64,500)         |     | (40,074)  |    | (35,547) |
| Net income attributable to noncontrolling interests             |            |                          | (935)    |    | (2,337)   |       | (658)            |     | _         |    | _        |
| Net income (loss) attributable to Blucora, Inc.                 |            | \$                       | 50,634   | \$ | 27,039    | \$    | (65,158)         | \$  | (40,074)  | \$ | (35,547) |
| Net income (loss) per share attributable to Blucora, Inc.       | - basic:   | _                        |          |    |           |       |                  |     |           |    |          |
| Continuing operations   |            | \$                       | 0.94     | \$ | 0.61      | \$    | (0.05)           | \$  | (0.31)    | \$ | (0.13)   |
| Discontinued operations   |            |                          | _        |    |           |       | (1.52)           |     | (0.67)    |    | (0.73)   |
| Basic net income (loss) per share                               |            | \$                       | 0.94     | \$ | 0.61      | \$    | (1.57)           | \$  | (0.98)    | \$ | (0.86)   |
| Weighted average shares outstanding, basic                      |            |                          | 47,394   |    | 44,370    |       | 41,494           |     | 40,959    |    | 41,396   |
| Net income (loss) per share attributable to Blucora, Inc.       | - diluted: |                          |          |    |           |       |                  |     |           |    |          |
| Continuing operations   |            | \$                       | 0.90     | \$ | 0.57      | \$    | (0.05)           | \$  | (0.31)    | \$ | (0.13)   |
| Discontinued operations   |            |                          |          |    | _         |       | (1.52)           |     | (0.67)    |    | (0.73)   |
| Diluted net income (loss) per share                             |            | \$                       | 0.90     | \$ | 0.57      | \$    | (1.57)           | \$  | (0.98)    | \$ | (0.86)   |
| Weighted average shares outstanding, diluted                    |            |                          | 49,381   |    | 47,211    |       | 41,494           |     | 40,959    |    | 41,396   |
| Consolidated Balance Sheet Data:                                | (1)        |                          |          |    |           |       |                  |     |           |    |          |
| Cash, cash equivalents, and investments                         |            | \$                       | 84,524   | \$ | 59,965    | \$    | 58,814           | \$  | 66,774    | \$ | 293,588  |
| Working capital   | (2)(3)(4)  |                          | 82,788   |    | 47,641    |       | 43,480           |     | 174,571   |    | 299,431  |
| Total assets  |            |                          | 997,725  |    | 1,001,671 |       | 1,022,659        |     | 1,299,548 |    | 865,775  |
| Total long-term liabilities                                     | (2)(3)(4)  |                          | 316,905  |    | 390,495   |       | 535,577          |     | 656,122   |    | 311,692  |
| Total stockholders' equity                                      |            |                          | 607,595  |    | 541,387   |       | 417,019          |     | 462,284   |    | 479,025  |

<sup>(1)</sup> On December 31, 2015, we acquired HD Vest. See "Note 4: Business Combinations" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report.

<sup>(2)</sup> On October 14, 2015, we announced plans to divest the Search and Content and E-Commerce businesses. Accordingly, the operating results of these businesses have been presented as discontinued operations for all periods presented, and the related balance sheet data has been classified in its entirety within current assets and current liabilities as of December 31, 2015 but classified within current and long-term assets and liabilities, as appropriate, for prior periods. We sold the Search and Content business and the E-Commerce business on August 9, 2016 and November 17, 2016, respectively.

- (3) During 2016 our Convertible Senior Notes were classified as a long-term liability with an outstanding balance, net of discount and issuance costs, of \$164.2 million. See "Note 10: Debt" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report.
- (4) See "Note 5: Discontinued Operations" and "Note 10: Debt" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report for a discussion of debt activity.

### ITEM 7. Management's Discussion and Analysis of Financial Condition and Results of Operations

You should read the following discussion and analysis in conjunction with the Selected Financial Data and our consolidated financial statements and notes thereto included elsewhere in this report. The following discussion contains forward-looking statements that are subject to risks and uncertainties. See Part I "Cautionary Statement Regarding Forward-Looking Statements" for a discussion of the uncertainties, risks and assumptions associated with those statements. Actual results could differ materially from those discussed in or implied by forward-looking statements as a result of various factors, including those discussed below and elsewhere in this report, particularly in the section titled "Risk Factors."

### **Introduction and Company History**

Blucora operates two businesses: a Wealth Management business and a digital Tax Preparation business. The Wealth Management business consists of the operations of HD Vest, which provides wealth management solutions for financial advisors and their clients. The Tax Preparation business consists of the operations of TaxAct and provides digital tax preparation solutions for consumers, small business owners, and tax professionals.

For a further discussion of Blucora's businesses and history, see "Business" in Part I Item 1 of this report.

### **Recent Developments**

In 2018, we commenced our new clearing services relationship with Fidelity Clearing & Custody Solutions pursuant to an agreement that we executed during the third quarter of 2017. We expect the new clearing relationship to provide tangible benefits to our advisors and customers in the form of improved technology, product offerings and service. We currently expect that this relationship could generate in excess of \$120.0 million of incremental Wealth Management segment income over the 10 years following the conversion to the new platform. In the fourth quarter of 2018, we received approximately \$9.3 million of operating cash flows from incentives from this relationship, which will benefit Wealth Management segment income over the succeeding 10 years and will offset operating expenses.

### Seasonality

Our Tax Preparation segment is highly seasonal, with a significant portion of its annual revenue earned in the first four months of our fiscal year. During the third and fourth quarters, the Tax Preparation segment typically reports losses because revenue from the segment is minimal while core operating expenses continue. We anticipate that the seasonal nature of that part of the business will continue in the foreseeable future.

### RESULTS OF OPERATIONS

# Summary

| (In thousands, except percentages) | <br>Years ended December 31, |     |        |         |      |    |         |  |  |  |  |
|------------------------------------|------------------------------|-----|--------|---------|------|----|---------|--|--|--|--|
|                                    | 2018 Change 2017             |     | Change |         | 2016 |    |         |  |  |  |  |
| Revenue                            | \$<br>560,456                | 10% | \$     | 509,557 | 12%  | \$ | 455,911 |  |  |  |  |
| Operating income                   | \$<br>67,677                 | 41% | \$     | 48,037  | 29%  | \$ | 37,117  |  |  |  |  |

Year ended December 31, 2018 compared with year ended December 31, 2017

Revenue increased approximately \$50.9 million due to increases of \$24.6 million and \$26.3 million in revenue related to our Wealth Management and Tax Preparation businesses, respectively, as discussed in the following "Segment Revenue/Operating Income" section.

Operating income increased approximately \$19.6 million, consisting of the \$50.9 million increase in revenue and offset by a \$31.3 million increase in operating expenses. Key changes in operating expenses were:

- \$22.4 million increase in the Wealth Management segment's operating expenses, primarily due to higher commissions paid to our financial
  advisors, which fluctuated in proportion to the change in underlying commission and advisory revenues earned on client accounts, consulting
  costs, costs incurred in connection with our transition to our new clearing firm, and an increase in stock-based compensation expense related to
  stock options granted to certain HD Vest financial advisors.
- \$12.0 million increase in the Tax Preparation segment's operating expenses, primarily due to higher spend on marketing, particularly offline
  media and digital marketing efforts, an increase in engineering development projects, an increase in consulting expenses primarily related to
  strategic initiatives and an increase in personnel costs, that was primarily related to additional headcount.
- \$3.2 million decrease in corporate-level expense activity, primarily due to lower Strategic Transformation costs, which primarily consisted of severance and other personnel-related costs, offset by higher depreciation due to the abandonment of certain internally-developed software fixed assets and an increase in personnel costs, that was primarily related to additional headcount.

Year ended December 31, 2017 compared with year ended December 31, 2016

Revenue increased approximately \$53.6 million due to increases of \$32.1 million and \$21.6 million in revenue related to our Wealth Management and Tax Preparation businesses, respectively, as discussed in the following "Segment Revenue/Operating Income" section.

Operating income increased approximately \$10.9 million, consisting of the \$53.6 million increase in revenue and offset by a \$42.7 million increase in operating expenses. Key changes in operating expenses were:

- \$27.5 million increase in the Wealth Management segment's operating expenses, primarily due to higher commissions paid to our financial advisors, which fluctuated in proportion to the change in underlying commission and advisory revenues earned on client accounts, and higher net personnel expenses as we continued to standardize employee benefits across our businesses.
- \$15.5 million increase in the Tax Preparation segment's operating expenses, primarily due to higher spending on marketing, higher professional services fees mostly related to marketing and development projects, higher data center costs related to software support and maintenance fees, increases in growth initiative investments, and higher personnel expenses.
- \$0.3 million decrease in corporate-level expense activity, primarily due to lower stock-based compensation costs due to fewer grants in 2017 compared to 2016 and higher expense recognized in 2016 related to grants made to HD Vest employees in 2016 in connection with the HD Vest acquisition, partially offset by decreases within our Tax Preparation business due to prior forfeitures, and lower personnel costs, both offset by Strategic Transformation costs.

# SEGMENT REVENUE/OPERATING INCOME

The revenue and operating income amounts in this section are presented on a basis consistent with accounting principles generally accepted in the U.S. ("GAAP") and include certain reconciling items attributable to our segments. Segment information appearing in "Note 3: Segment Information and Revenues" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report is presented on a basis consistent with our current internal management financial reporting. We do not allocate certain general and administrative costs (including personnel and overhead costs), stock-based compensation, acquisition-related costs, depreciation, amortization of acquired intangible assets, restructuring, other loss, net, and income taxes to segment operating results. Rather, we analyze such general and administrative costs separately under the heading "Corporate-level activity."

We have two reportable segments: Wealth Management and Tax Preparation.

# Wealth Management

| (In thousands, except percentages) | <br>Years ended December 31, |             |    |         |     |      |         |  |  |  |  |  |
|------------------------------------|------------------------------|-------------|----|---------|-----|------|---------|--|--|--|--|--|
|                                    | 2018                         | Change 2017 |    | Change  |     | 2016 |         |  |  |  |  |  |
| Revenue                            | \$<br>373,174                | 7%          | \$ | 348,620 | 10% | \$   | 316,546 |  |  |  |  |  |
| Operating income                   | \$<br>53,053                 | 4%          | \$ | 50,916  | 10% | \$   | 46,296  |  |  |  |  |  |
| Segment margin                     | 14%                          |             |    | 15%     |     |      | 15%     |  |  |  |  |  |

Wealth Management revenue is derived from multiple sources. We track sources of revenue, primary drivers of each revenue source, and recurring revenue. In addition, we focus on several business and key financial metrics in evaluating the success of our business relationships, our resulting financial position and operating performance. A summary of our sources of revenue and business metrics are as follows.

### Sources of revenue

| (In thousands, except per | centages)               |  | Year ended December 31, |         |      |       |    |         |       |     |    |         |
|---------------------------|-------------------------|--|-------------------------|---------|------|-------|----|---------|-------|-----|----|---------|
|                           | Sources of Revenue      | <b>Primary Drivers</b>   | 20                      | 018     | Chai | nge   |    | 2017    | Chang | ge  |    | 2016    |
| Advisor-driven            | Commission              | - Transactions<br>- Asset levels   | \$ 1                    | 164,201 |      | 2 %   | \$ | 160,241 |       | 7%  | \$ | 150,125 |
|                           | Advisory                | - Advisory asset levels  | 1                       | 164,353 |      | 13 %  |    | 145,694 |       | 13% |    | 129,417 |
| Other revenue             | Asset-based             | - Cash balances<br>- Interest rates<br>- Number of accounts<br>- Client asset levels |                         | 31,456  |      | 20 %  |    | 26,297  |       | 16% |    | 22,653  |
| Office Tevenue            | Transaction and fee     | - Account activity - Number of clients - Number of advisors - Number of accounts     |                         | 13,164  |      | (20)% |    | 16,388  |       | 14% |    | 14,351  |
|                           | Total revenue           |  | \$ 3                    | 373,174 |      | 7 %   | \$ | 348,620 |       | 10% | \$ | 316,546 |
|                           | Total recurring revenue |  | \$ 3                    | 303,117 |      | 9 %   | \$ | 277,546 |       | 11% | \$ | 249,310 |
|                           | Recurring revenue rate  |  |                         | 81.2%   |      |       |    | 79.6%   |       |     |    | 78.8%   |

Recurring revenue consists of trailing commissions, advisory fees, fees from cash sweep programs, and certain transaction and fee revenue, all as described further below in *Commission revenue*, *Advisory revenue*, *Asset-based revenue*, and *Transaction and fee revenue*, respectively. Certain recurring revenues are associated with asset balances and fluctuate depending on market values and current interest rates. Accordingly, our recurring revenue can be negatively impacted by adverse external market conditions. However, we believe recurring revenue is meaningful despite these fluctuations because it is not dependent upon transaction volumes or other activity-based revenues, which are more difficult to predict, particularly in declining or volatile markets.

# **Business metrics**

# (In thousands, except percentages and as otherwise indicated)

|                                    | <br>Years ended December 31, |          |    |            |        |    |            |  |  |  |  |  |  |
|------------------------------------|------------------------------|----------|----|------------|--------|----|------------|--|--|--|--|--|--|
|                                    | 2018                         | Change   |    | 2017       | Change |    | 2016       |  |  |  |  |  |  |
| Total Client Assets                | \$<br>42,249,055             | (4)%     | \$ | 44,178,710 | 14 %   | \$ | 38,663,566 |  |  |  |  |  |  |
| Brokerage Assets                   | \$<br>29,693,650             | (4)%     | \$ | 31,648,545 | 12 %   | \$ | 28,266,495 |  |  |  |  |  |  |
| Advisory Assets                    | \$<br>12,555,405             | <u> </u> | \$ | 12,530,165 | 21 %   | \$ | 10,397,071 |  |  |  |  |  |  |
| Percentage of Total Client Assets  | 29.7%                        |          |    | 28.4%      |        |    | 26.9%      |  |  |  |  |  |  |
| Number of advisors (in ones)       | 3,593                        | (10)%    |    | 3,999      | (11)%  |    | 4,472      |  |  |  |  |  |  |
| Advisor-driven revenue per advisor | 23.2                         | 14 %     |    | 20.4       | 25 %   |    | 16.3       |  |  |  |  |  |  |

Total client assets ("total client assets") includes assets that we hold directly or indirectly on behalf of clients under a safekeeping or custody arrangement or for which we provide administrative services for clients. To the extent that we provide more than one service for a client's assets, the value of the asset is only counted once in the total amount of total client assets. Total client assets include advisory assets, non-advisory brokerage accounts, annuities and mutual fund positions held directly with fund companies. These assets are not reported on the consolidated balance sheets. Total client assets were previously reported as "Assets Under Administration" or "AUA."

For the year ended December 31, 2018, total client assets includes \$34.5 million of assets held at our former clearing firm for which we are broker-of-record and whose conversion was administratively delayed.

Advisory assets ("advisory assets") includes external client assets for which we provide investment advisory and management services, typically as a fiduciary under the Investment Advisers Act of 1940. Our compensation for providing such services is typically a fee based on the value of the advisory assets for each advisory client. These assets are not reported on the consolidated balance sheets. Advisory assets were previously reported as "Assets Under Management" or "AUM."

Brokerage assets represents the difference between total client assets and advisory assets.

We have been reducing disengaged advisors who have little to no assets held with us, which has resulted in advisor counts trending down. As we continue to reduce disengaged advisors, the number of advisors could continue to decrease before stabilizing. This decrease has resulted in, and is expected to continue to improve, the growth in advisor-driven revenues per advisor.

Year ended December 31, 2018 compared with year ended December 31, 2017

Wealth Management revenue increased approximately \$24.6 million as a result of the factors discussed with each source of revenue below.

Wealth Management operating income increased approximately \$2.1 million, consisting of the \$24.6 million increase in revenue, offset by a \$22.4 million increase in operating expenses. The increase in Wealth Management operating expenses was primarily due to higher commissions paid to our financial advisors, which fluctuated in proportion to the change in underlying commission and advisory revenues earned on client accounts, consulting costs and costs incurred in connection with our transition to our new clearing firm, and an increase in stock-based compensation expense related to stock options granted to certain HD Vest financial advisors.

Year ended December 31, 2017 compared with year ended December 31, 2016

Wealth Management revenue increased approximately \$32.1 million as a result of the factors discussed with each source of revenue below.

Wealth Management operating income increased approximately \$4.6 million, consisting of the \$32.1 million increase in revenue, offset by a \$27.5 million increase in operating expenses. The increase in Wealth Management operating expenses was primarily due to higher commissions paid to our financial advisors, which fluctuated in proportion to the change in underlying commission and advisory revenues earned on client accounts, and higher net personnel expenses as we continued to standardize employee benefits across our businesses.

Commission revenue: The Wealth Management segment generates two types of commissions: transaction-based sales commissions and trailing commissions. Transaction-based sales commissions, which occur when clients trade securities or purchase investment products, represent gross commissions generated by our financial advisors. The level of transaction-based sales commissions can vary from period-to-period based on the overall economic environment, number of trading days in the reporting period, market volatility, interest rate fluctuations and investment activity of our financial advisors' clients. We earn trailing commissions (a commission or fee that is paid periodically over time) on certain mutual funds and variable annuities held by clients. Trailing commissions are recurring in nature and are based on the market value of investment holdings in trail-eligible assets. Our commission revenue, by product category and by type of commission revenue, was as follows:

| (In thousands)               |          | Years ended December 31, |        |    |         |        |    |         |  |  |  |  |  |
|------------------------------|----------|--------------------------|--------|----|---------|--------|----|---------|--|--|--|--|--|
|                              |          | 2018                     | Change |    | 2017    | Change |    | 2016    |  |  |  |  |  |
| By product category:         |          |                          |        |    | _       |        |    |         |  |  |  |  |  |
| Mutual funds                 | \$       | 87,624                   | 4 %    | \$ | 84,159  | 6%     | \$ | 79,476  |  |  |  |  |  |
| Variable annuities           |          | 51,199                   | —%     |    | 51,385  | 8%     |    | 47,641  |  |  |  |  |  |
| Insurance                    |          | 14,160                   | 8 %    |    | 13,146  | 10%    |    | 11,909  |  |  |  |  |  |
| General securities           |          | 11,218                   | (3)%   |    | 11,551  | 4%     |    | 11,099  |  |  |  |  |  |
| Total commission revenue     | \$       | 164,201                  | 2 %    | \$ | 160,241 | 7%     | \$ | 150,125 |  |  |  |  |  |
| By sales-based and trailing: | <u> </u> |                          |        |    |         |        |    |         |  |  |  |  |  |
| Sales-based                  | \$       | 67,350                   | (1)%   | \$ | 68,199  | 6%     | \$ | 64,452  |  |  |  |  |  |
| Trailing                     |          | 96,851                   | 5 %    |    | 92,042  | 7%     |    | 85,673  |  |  |  |  |  |
| Total commission revenue     | \$       | 164,201                  | 2 %    | \$ | 160,241 | 7%     | \$ | 150,125 |  |  |  |  |  |

In 2018, sales-based commission revenue decreased approximately \$0.8 million, primarily due to increased activity in equities, that was more than offset by decreased activity in mutual funds and alternative investments. General securities include equities, exchange-traded funds, bonds and alternative investments.

In 2018, trailing commission revenue increased approximately \$4.8 million and reflects an increase in the market value of the underlying assets.

In 2017, sales-based commission revenue increased approximately \$3.7 million, primarily due to increased activity in mutual funds, insurance and general securities resulting from overall market performance, portfolio rebalancings, product availability and segment refocusing. General securities include equities, exchange-traded funds, bonds and alternative investments.

In 2017, trailing commission revenue increased approximately \$6.4 million and reflects an increase in the market value of the underlying assets and the impact of new investments.

Advisory revenue: Advisory revenue primarily includes fees charged to clients in advisory accounts where HD Vest is the Registered Investment Adviser ("RIA") and is based on the value of advisory assets. Advisory fees are typically billed to clients quarterly, in advance, and are recognized as revenue ratably during the quarter. The value of the assets in an advisory account on the billing date determines the amount billed and, accordingly, the revenues earned in the following three-month period. The majority of our accounts are billed in advance using values as of the last business day of the prior calendar quarter.

The activity within our advisory assets was as follows:

| (In thousands)                      | Year ended December 31, |            |    |            |    |            |  |
|-------------------------------------|-------------------------|------------|----|------------|----|------------|--|
|                                     |                         | 2018       |    | 2017       |    | 2016       |  |
| Balance, beginning of the period    | \$                      | 12,530,165 | \$ | 10,397,071 | \$ | 9,692,244  |  |
| Net increase in new advisory assets |                         | 957,252    |    | 794,184    |    | 150,701    |  |
| Market impact and other             |                         | (932,012)  |    | 1,338,910  |    | 554,126    |  |
| Balance, end of the period          | \$                      | 12,555,405 | \$ | 12,530,165 | \$ | 10,397,071 |  |

Increases or decreases in advisory assets have a limited impact on advisory fee revenue in the period in which they occur. Rather, increases or decreases in advisory assets are a primary driver of future advisory fee revenue due to advisory fees being billed in advance. Advisory revenue for a particular quarter is predominately driven by the prior quarter-end advisory assets.

In 2018, the increase in advisory revenue of approximately \$18.7 million was primarily due to the increase in the beginning-of-period advisory assets for 2018 compared with 2017.

In 2017, the increase in advisory revenue of approximately \$16.3 million was consistent with the increase in the beginning-of-period advisory assets for 2017 compared with 2016.

Asset-based revenue: Asset-based revenue primarily includes fees from financial product manufacturer sponsorship programs, cash sweep programs and other asset-based revenues, primarily including margin revenues.

In 2018, asset-based revenue increased \$5.2 million, primarily from increased revenues from financial product manufacturer sponsorship programs, higher cash sweep revenues following increases in interest rates and impacts from our transition to our new clearing firm in the third quarter of 2018.

In 2017, asset-based revenue increased \$3.6 million, primarily from higher cash sweep revenues following increases in interest rates.

Transaction and fee revenue: Transaction and fee revenue primarily includes support fees charged to advisors, fees charged for executing certain transactions in client accounts, and other fees related to services provided and other account charges as generally outlined in agreements with financial advisors, clients, and financial institutions.

In 2018, transaction and fee revenue decreased approximately \$3.2 million primarily due to the impact of the adoption of new revenue recognition standards in the first quarter of 2018 and lower technology fees. See "Note 2: Summary of Significant Accounting Policies" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report for additional information concerning the impact of the new revenue recognition standards on our operating results.

In 2017, transaction and fee revenue increased approximately \$2.0 million primarily related to advisor fee increases.

## Tax Preparation

| (In thousands, except percentages) | <br>Years ended December 31, |        |    |         |        |    |         |  |  |  |  |  |
|------------------------------------|------------------------------|--------|----|---------|--------|----|---------|--|--|--|--|--|
|                                    | 2018                         | Change |    | 2017    | Change |    | 2016    |  |  |  |  |  |
| Revenue                            | \$<br>187,282                | 16%    | \$ | 160,937 | 15%    | \$ | 139,365 |  |  |  |  |  |
| Operating income                   | \$<br>87,249                 | 20%    | \$ | 72,921  | 9%     | \$ | 66,897  |  |  |  |  |  |
| Segment margin                     | 47%                          |        |    | 45%     |        |    | 48%     |  |  |  |  |  |

Tax Preparation revenue is derived primarily from the sale of tax preparation digital services, ancillary services, packaged tax preparation software, and arrangements that may include a combination of these items. Ancillary services include tax preparation support services, e-filing services, bank or reloadable pre-paid debit card services, and other value-added services, including tax and wealth management services through our Wealth Management business.

Revenue by category was as follows:

| (In thousands, except percentages) | <br>Years ended December 31, |        |    |         |        |    |         |  |  |  |  |  |
|------------------------------------|------------------------------|--------|----|---------|--------|----|---------|--|--|--|--|--|
|                                    | 2018                         | Change |    | 2017    | Change |    | 2016    |  |  |  |  |  |
| Consumer                           | \$<br>172,207                | 17%    | \$ | 147,084 | 16%    | \$ | 126,289 |  |  |  |  |  |
| Professional                       | 15,075                       | 9%     |    | 13,853  | 6%     |    | 13,076  |  |  |  |  |  |
| Total revenue                      | \$<br>187,282                | 16%    | \$ | 160,937 | 15%    | \$ | 139,365 |  |  |  |  |  |

We measure our consumer tax preparation customers using the number of accepted federal tax e-files made through our software and digital services. We consider the volume of e-files to be an important non-financial metric in measuring the performance of the consumer side of the Tax Preparation business. E-file metrics were as follows:

| (In thousands, except percentages) |       | Years ended December 31, |       |        |       |  |  |  |  |  |  |  |  |
|------------------------------------|-------|--------------------------|-------|--------|-------|--|--|--|--|--|--|--|--|
|                                    | 2018  | Change                   | 2017  | Change | 2016  |  |  |  |  |  |  |  |  |
| Digital e-files                    | 3,539 | (14)%                    | 4,097 | (17)%  | 4,926 |  |  |  |  |  |  |  |  |
| Desktop e-files                    | 159   | (18)%                    | 193   | (21)%  | 244   |  |  |  |  |  |  |  |  |
| Total e-files                      | 3,698 | (14)%                    | 4,290 | (17)%  | 5,170 |  |  |  |  |  |  |  |  |

We participate in the Free File Alliance that is part of an IRS partnership that provides free electronic tax filing services to taxpayers meeting certain income-based guidelines. Free File Alliance e-files are included within digital e-files above.

We measure our professional tax preparer customers using three metrics: the number of accepted federal tax e-files made through our software, the number of units sold, and the number of e-files per unit sold. We consider growth in these areas to be important non-financial metrics in measuring the performance of the professional tax preparer side of the Tax Preparation business. Those metrics were as follows:

| (In thousands, except percentages and as |        | Years ended December 31, |        |        |        |  |  |  |  |  |  |  |  |  |
|--|--------|--------------------------|--------|--------|--------|--|--|--|--|--|--|--|--|--|
| otherwise indicated)                     | 2018   | Change                   | 2017   | Change | 2016   |  |  |  |  |  |  |  |  |  |
| E-files                                  | 1,833  | 3 %                      | 1,774  | 1 %    | 1,755  |  |  |  |  |  |  |  |  |  |
| Units sold (in ones)                     | 20,636 | <b></b> %                | 20,694 | 2 %    | 20,290 |  |  |  |  |  |  |  |  |  |
| E-files per unit sold (in ones)          | 88.8   | 4 %                      | 85.7   | (1)%   | 86.5   |  |  |  |  |  |  |  |  |  |

Year ended December 31, 2018 compared with year ended December 31, 2017

Tax Preparation revenue increased approximately \$26.3 million, primarily due to growth in revenue earned from digital consumer users and increased sales of our professional tax preparer software. Digital consumer revenue grew, despite a decrease in e-files, due to growth in average revenue per user, primarily resulting from price increases, which are expected to continue to be the primary driver of growth in the near future. The decrease in e-files was consistent with our expectations as we continued our multi-year pivot toward more profitable customers. Revenue derived from professional tax preparers increased, despite a minor decrease in the number of professional preparer units sold, primarily due to growth in average revenue per user, primarily resulting from price increases. Revenue from ancillary services, primarily tax refund payment transfer, also grew primarily resulting from price increases.

Tax Preparation operating income increased approximately \$14.3 million, consisting of the \$26.3 million increase in revenue and offset by a \$12.0 million increase in operating expenses. The increase in Tax Preparation segment operating expenses was primarily due to higher spend on marketing, particularly offline media and digital marketing efforts, an increase in engineering development projects, and an increase in consulting expenses primarily related to strategic initiatives.

Year ended December 31, 2017 compared with year ended December 31, 2016

Tax Preparation revenue increased approximately \$21.6 million primarily due to growth in revenue earned from digital consumer users and, to a lesser extent, increased sales of our professional tax preparer software. Digital consumer revenue grew, despite a decrease in e-files, due to growth in average revenue per user, primarily resulting from price increases. Revenue derived from professional tax preparers increased primarily due to an increase in the number of professional preparer units sold.

Tax Preparation operating income increased approximately \$6.0 million, consisting of the \$21.6 million increase in revenue and offset by a \$15.5 million increase in operating expenses. The increase in Tax Preparation segment operating expenses was primarily due to higher spending on marketing, higher professional services fees mostly related to marketing and development projects, higher data center costs related to software support and maintenance fees, increases in growth initiative investments, and higher personnel expenses.

## Corporate-Level Activity

| (In thousands)                             | Years ended December 31, |        |    |         |    |        |        |         |      |        |  |
|--|--------------------------|--------|----|---------|----|--------|--------|---------|------|--------|--|
|  |                          | 2018   |    | Change  |    | 2017   | Change |         | 2016 |        |  |
| Operating expenses                         | \$                       | 20,494 | \$ | (2,413) | \$ | 22,907 | \$     | 3,908   | \$   | 18,999 |  |
| Stock-based compensation                   |                          | 13,253 |    | 1,600   |    | 11,653 |        | (2,475) |      | 14,128 |  |
| Acquisition-related costs                  |                          | _      |    | _       |    | _      |        | (391)   |      | 391    |  |
| Depreciation                               |                          | 5,003  |    | 866     |    | 4,137  |        | (408)   |      | 4,545  |  |
| Amortization of acquired intangible assets |                          | 33,586 |    | (416)   |    | 34,002 |        | (141)   |      | 34,143 |  |
| Restructuring                              |                          | 288    |    | (2,813) |    | 3,101  |        | (769)   |      | 3,870  |  |
| Total corporate-level activity             | \$                       | 72,624 | \$ | (3,176) | \$ | 75,800 | \$     | (276)   | \$   | 76,076 |  |

Certain corporate-level activity is not allocated to our segments, including certain general and administrative costs (including personnel and overhead costs), stock-based compensation, acquisition-related costs, depreciation, amortization of acquired intangible assets, and restructuring. For further detail, refer to segment information appearing in "Note 3: Segment Information and Revenues" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report.

Year ended December 31, 2018 compared with year ended December 31, 2017

Operating expenses included in corporate-level activity decreased primarily due to lower Strategic Transformation costs, which primarily consisted of severance and other personnel-related costs.

Stock-based compensation increased primarily due to activity within our Wealth Management business related to stock options granted to certain HD Vest financial advisors and a decrease in forfeitures from the prior period, partially offset by lower expenses related to the impact of equity award modifications associated with certain individuals impacted by the relocation of our corporate headquarters in 2017.

Depreciation expense increased primarily due to the abandonment of certain internally-developed software fixed assets.

Restructuring expense relates to non-recurring expenses incurred due to the relocation of our corporate headquarters during 2017 from Bellevue, Washington to Irving, Texas. Further detail is provided under the "Operating Expenses - Restructuring" section below.

Year ended December 31, 2017 compared with year ended December 31, 2016

Operating expenses included in corporate-level activity increased primarily due to Strategic Transformation Costs and costs associated with leadership changes at HD Vest. Strategic Transformation Costs primarily related to the relocation cost of our corporate headquarters and are not classified as restructuring. These costs are associated with transitioning of roles such as overlap in staffing and recruiting search fees.

Stock-based compensation decreased primarily due to fewer grants in the current year and higher expense recognized in 2016 related to grants made to HD Vest employees in 2016 that were made in connection with the HD Vest acquisition, partially offset by activity within our Tax Preparation business due to prior forfeitures.

Acquisition-related costs, depreciation and amortization of acquired intangible assets were comparable to 2016.

Restructuring relates to expenses incurred in connection with the relocation of our corporate headquarters in 2017. Further detail is provided under the "Operating Expenses - Restructuring" section of the management's discussion and analysis of financial condition and results of operations below.

## OPERATING EXPENSES

## Cost of Revenue

| (In thousands, except percentages)         | Years ended December 31, |    |        |      |         |        |        |      |         |  |  |  |
|--|--------------------------|----|--------|------|---------|--------|--------|------|---------|--|--|--|
|  | <br>2018                 |    | Change | 2017 |         | Change |        | 2016 |         |  |  |  |
| Wealth management services cost of revenue | \$<br>253,580            | \$ | 17,721 | \$   | 235,859 | \$     | 21,863 | \$   | 213,996 |  |  |  |
| Tax preparation services cost of revenue   | 10,040                   |    | 22     |      | 10,018  |        | 1,650  |      | 8,368   |  |  |  |
| Amortization of acquired technology        | 99                       |    | (96)   |      | 195     |        | (617)  |      | 812     |  |  |  |
| Total cost of revenue                      | \$<br>263,719            | \$ | 17,647 | \$   | 246,072 | \$     | 22,896 | \$   | 223,176 |  |  |  |
| Percentage of revenue                      | 47%                      |    |        |      | 48%     |        |        |      | 49%     |  |  |  |

We record the cost of revenue for sales of services when the related revenue is recognized. Services cost of revenue consists of costs related to our Wealth Management and Tax Preparation businesses, which include commissions to financial advisors, third-party costs, and costs associated with the technical support team and the operation of our data centers. Data center costs include personnel expenses (salaries, stock-based compensation, benefits, and other employee-related costs), the cost of temporary help and contractors, professional services fees (which include technology project consulting fees), software support and maintenance, bandwidth and hosting costs, and depreciation. Cost of revenue also includes the amortization of acquired technology.

Year ended December 31, 2018 compared with year ended December 31, 2017

Wealth management services cost of revenue increased primarily due to higher commissions paid to our financial advisors, which fluctuated in proportion to the change in underlying commission and advisory revenues earned on client accounts.

Tax preparation services cost of revenue was comparable to 2017 as the cost of maintaining our tax preparation platform is somewhat fixed, and does not necessarily vary with revenues.

Year ended December 31, 2017 compared with year ended December 31, 2016

Wealth management services cost of revenue increased primarily due to an increase in commissions paid to our financial advisors, which fluctuated in proportion to the change in underlying commission and advisory revenues earned on client accounts, and higher stock-based compensation costs related to grants to certain HD Vest financial advisors made during 2017 because no comparable grants were made in 2016.

Tax preparation services cost of revenue increased primarily due to an increase in data center costs related to software support and maintenance fees.

Amortization of acquired technology decreased due to amortization expense associated with concluding the useful life of certain TaxAct acquisition-related intangible assets during 2016.

# **Engineering and Technology**

| (In thousands, except percentages) | Years ended December 31, |    |        |    |        |    |        |    |        |  |  |  |
|------------------------------------|--------------------------|----|--------|----|--------|----|--------|----|--------|--|--|--|
|                                    | <br>2018                 |    | Change |    | 2017   |    | Change |    | 2016   |  |  |  |
| Engineering and technology         | \$<br>19,332             | \$ | (282)  | \$ | 19,614 | \$ | 1,834  | \$ | 17,780 |  |  |  |
| Percentage of revenue              | 3%                       |    |        |    | 4%     |    |        |    | 4%     |  |  |  |

Engineering and technology expenses are associated with the research, development, support, and ongoing enhancements of our offerings, which include personnel expenses (salaries, stock-based compensation, benefits, and other employee-related costs), the cost of temporary help and contractors, software support and maintenance, bandwidth and hosting, and professional services fees.

Year ended December 31, 2018 compared with year ended December 31, 2017

Engineering and technology expenses were substantially the same compared to the prior year.

Year ended December 31, 2017 compared with year ended December 31, 2016

Engineering and technology expenses increased primarily due to an increase in consulting and professional services fees, mostly related to Tax Preparation development projects.

## Sales and Marketing

| (In thousands, except percentages) |               |             | Years | ended December 3 | 1, |        |      |        |  |
|------------------------------------|---------------|-------------|-------|------------------|----|--------|------|--------|--|
|                                    | 2018          | Change      |       | 2017 Change      |    |        | 2016 |        |  |
| Sales and marketing                | \$<br>111,361 | \$<br>8,563 | \$    | 102,798          | \$ | 13,438 | \$   | 89,360 |  |
| Percentage of revenue              | 20%           |             |       | 20%              |    |        |      | 20%    |  |

Sales and marketing expenses consist principally of personnel expenses (salaries, stock-based compensation, benefits, and other employee-related costs) and the cost of temporary help and contractors for those engaged in marketing, selling, and sales support operations activities, marketing expenses associated with our HD Vest and TaxAct businesses (which primarily include television, radio, online, text, email, and sponsorship channels), and back office processing support expenses associated with our HD Vest business (occupancy and general office expenses, regulatory fees, and license fees).

Year ended December 31, 2018 compared with year ended December 31, 2017

Sales and marketing expenses increased primarily due to higher spend on marketing, particularly offline media and digital marketing efforts in our Tax Preparation business, increased personnel costs primarily related to additional headcount, consulting costs and costs incurred in connection with our transition to a new clearing firm, which was completed in the third quarter of 2018, higher educational development expenses for our Wealth Management advisors and personnel costs primarily related to increases in headcount, offset by the reclassification of certain regulatory fees following the adoption of ASC 606.

Year ended December 31, 2017 compared with year ended December 31, 2016

Sales and marketing expenses increased primarily due to a \$7.8 million increase in marketing expenses and a \$3.8 million increase in personnel expenses. The increase in marketing expenses was driven by increased marketing in our Tax Preparation business. Personnel expenses increased primarily due to the standardization of employee benefits across our business.

# General and Administrative

| (In thousands, except percentages) |              |             | Years | ended December 3 | 1, |        |              |
|------------------------------------|--------------|-------------|-------|------------------|----|--------|--------------|
|                                    | 2018         | Change      |       | 2017             |    | Change | 2016         |
| General and administrative         | \$<br>60,124 | \$<br>7,456 | \$    | 52,668           | \$ | 5,272  | \$<br>47,396 |
| Percentage of revenue              | 11%          |             |       | 10%              |    |        | 10%          |

General and administrative ("G&A") expenses consist primarily of personnel expenses (salaries, stock-based compensation, benefits, and other employee-related costs), the cost of temporary help and contractors, professional services fees (which include legal, audit, and tax fees), general business development and management expenses, occupancy and general office expenses, business taxes, and insurance expenses.

Year ended December 31, 2018 compared with year ended December 31, 2017

G&A expenses increased primarily due to an increase in consulting expenses primarily related to strategic initiatives, software expenses mainly related to security enhancements and personnel costs primarily related to additional headcount.

Year ended December 31, 2017 compared with year ended December 31, 2016

G&A expenses increased primarily due to a \$4.9 million net increase in personnel expenses, mainly related to Strategic Transformation Costs and costs associated with leadership changes at HD Vest, offset by lower stock-based compensation due to fewer grants in 2017 and higher expense recognized in 2016 related to the timing of grants.

## Depreciation and Amortization of Acquired Intangible Assets

| (In thousands, except percentages)         | Years ended December 31, |    |        |      |        |        |       |      |        |  |  |  |
|--|--------------------------|----|--------|------|--------|--------|-------|------|--------|--|--|--|
|  | 2018                     |    | Change | 2017 |        | Change |       | 2016 |        |  |  |  |
| Depreciation                               | \$<br>4,468              | \$ | 1,008  | \$   | 3,460  | \$     | (421) | \$   | 3,881  |  |  |  |
| Amortization of acquired intangible assets | 33,487                   |    | (320)  |      | 33,807 |        | 476   |      | 33,331 |  |  |  |
| Total                                      | \$<br>37,955             | \$ | 688    | \$   | 37,267 | \$     | 55    | \$   | 37,212 |  |  |  |
| Percentage of revenue                      | <br>7%                   |    |        |      | 7%     |        |       |      | 8%     |  |  |  |

Depreciation of property and equipment includes depreciation of computer equipment and software, office equipment and furniture, and leasehold improvements not recognized in cost of revenue. Amortization of acquired intangible assets primarily includes the amortization of customer relationships, which are amortized over their estimated lives.

Year ended December 31, 2018 compared with year ended December 31, 2017

Depreciation expense increased primarily due to the abandonment of certain internally-developed software fixed assets in 2018.

Year ended December 31, 2017 compared with year ended December 31, 2016

Depreciation and amortization expenses were substantially the same compared to the prior year.

## Restructuring

| (In thousands, except percentages) | <br>Years ended December 31, |    |         |    |       |    |        |    |       |  |  |  |
|------------------------------------|------------------------------|----|---------|----|-------|----|--------|----|-------|--|--|--|
|                                    | <br>2018                     |    | Change  |    | 2017  |    | Change |    | 2016  |  |  |  |
| Restructuring                      | \$<br>288                    | \$ | (2,813) | \$ | 3,101 | \$ | (769)  | \$ | 3,870 |  |  |  |
| Percentage of revenue              | %                            | ,  |         |    | 1%    |    |        |    | 1%    |  |  |  |

In connection with the Strategic Transformation, including the 2017 relocation of our headquarters, we have incurred restructuring costs of approximately \$7.3 million, which includes all costs associated with our non-cancelable operating lease for our former corporate headquarters in Bellevue. While the relocation and the related costs were substantially completed by June 2017, the Company incurred some costs in the year ended December 31, 2018, primarily related to employees who continued to provide service during a portion of 2018.

See "Note 6: Restructuring" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report for additional information on restructuring.

## Other Loss, Net

| (In thousands)  | Years ended December 31, |         |    |          |    |        |        |          |      |        |  |  |
|---|--------------------------|---------|----|----------|----|--------|--------|----------|------|--------|--|--|
|   |                          | 2018    |    | Change   |    | 2017   | Change |          | 2016 |        |  |  |
| Interest expense  | \$                       | 15,610  | \$ | (5,601)  | \$ | 21,211 | \$     | (11,213) | \$   | 32,424 |  |  |
| Loss on debt extinguishment and amortization of debt issuance costs |                          | 2,367   |    | (19,167) |    | 21,534 |        | 18,658   |      | 2,876  |  |  |
| Accretion of debt discounts   |                          | 163     |    | (1,784)  |    | 1,947  |        | (2,743)  |      | 4,690  |  |  |
| Interest income   |                          | (349)   |    | (239)    |    | (110)  |        | (29)     |      | (81)   |  |  |
| Other   |                          | (1,994) |    | (1,963)  |    | (31)   |        | 97       |      | (128)  |  |  |
| Other loss, net   | \$                       | 15,797  | \$ | (28,754) | \$ | 44,551 | \$     | 4,770    | \$   | 39,781 |  |  |

Year ended December 31, 2018 compared with year ended December 31, 2017

The decrease in interest expense relates to lower balances in the Blucora senior secured credit facilities following several prepayments, and the repricing and lowering, in November 2017, of the applicable interest rate margin of the Blucora senior secured credit facilities to 3.0% for Eurodollar Rate loans and 2.0% for ABR loans. See "Note 10: Debt" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report for additional information.

For the year ended December 31, 2017 we had a loss on debt extinguishment related to the prepayment of a portion of the credit facility previously entered into in 2015 for the purpose of financing the HD Vest acquisition (the "TaxAct - HD Vest 2015 credit facility"). In connection with the refinancing through the Blucora senior secured credit facilities that was entered into in May 2017, we paid-off the remaining TaxAct - HD Vest 2015 credit facility and wrote-off the remaining unamortized debt discount and issuance costs.

The decrease in other primarily related to a \$2.1 million gain on the sale of an investment in 2018.

Year ended December 31, 2017 compared with year ended December 31, 2016

The changes in interest expense, loss on debt extinguishment, amortization of debt issuance costs, and accretion of debt discounts primarily related to lower balances in the TaxAct - HD Vest 2015 credit facility and our formerly outstanding Convertible Senior Notes ("the Notes") due to prepayments on a portion of the TaxAct - HD Vest 2015 credit facility in 2017 and 2016 and the redemption of all of the Notes in the second quarter of 2017. In 2017, the applicable interest rate margin of the Blucora senior secured credit facilities was repriced and lowered to 3.0% for Eurodollar Rate loans and 2.0% for ABR loans.

The gain on third party bankruptcy settlement related to amounts received in connection with ongoing distributions from the Lehman Brothers estate, of which we are a creditor.

See "Note 10: Debt" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report for additional information on the "Loss on debt extinguishment and modification expense."

#### Income Taxes

During 2018, we recorded income tax expense of \$0.3 million. Our effective income tax rate differed from the 21% statutory rate in 2018 primarily due to the recognition of previously reserved net operating losses to offset current income tax expense, and the effect of state income taxes. We currently expect to continue to release portions of valuation allowances, which were previously recorded in connection with our net operating losses, to offset future federal income tax liabilities. The majority of these net operating losses will expire, if unutilized, between 2020 and 2024.

During 2017, we recorded an income tax benefit of \$25.9 million. Income tax differed from taxes at the statutory rates primarily due to the January 1, 2017 implementation of Accounting Standards Update ("ASU") 2016-09 on stock-based compensation (see "Note 2: Summary of Significant Accounting Policies" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report for additional information) and the impact of "H.R. 1", formerly known as the Tax Cuts and Jobs Act (the "Tax Legislation"), which President Donald Trump signed into law on December 22, 2017.

The Tax Legislation, which was effective January 1, 2018, significantly revised the U.S. tax code by, among other things, lowering the corporate income tax rate from 35% to 21%. As a result of the reduction in the corporate income tax rate, we re-valued our net deferred tax liabilities during the year ended December 31, 2017. The re-measurement of our deferred tax assets and liabilities resulted in a reduction in the value of our net deferred tax liabilities of approximately \$21.4 million, which was recorded as additional income tax benefit in 2017.

The Tax Legislation also repealed corporate alternative minimum tax ("AMT") for tax years beginning January 1, 2018 and provides that existing AMT credit carryovers are refundable beginning in 2018. We have approximately \$10.9 million of AMT credit carryovers that are expected to be fully refunded by 2022. Additionally, the Tax Legislation made amendments to Section I.R.C. 162(m) of the Code related to the deductibility of certain compensation of executive officers, which may limit the total amount of compensation expense that may be deductible by the Company in future years.

At December 31, 2018, we had gross temporary differences representing future tax deductions of \$628.9 million, which represented deferred tax assets primarily comprised of \$454.3 million of federal net operating loss carryforwards. We have applied a valuation allowance against the net operating loss carryforwards and certain other deferred tax assets. If in the future,

we determine that any additional portion of the deferred tax assets is more likely than not to be realized, we will record a benefit to the income statement. We currently estimate that approximately \$174.7 million and \$132.1 million of federal net operating loss carryforwards will expire in 2020 and 2021, respectively.

During 2016, we recorded an income tax benefit of \$1.3 million. Income tax differed from taxes at the statutory rates primarily due to the domestic manufacturing deduction, offset by non-deductible compensation and state income taxes.

## **Discontinued Operations, Net of Income Taxes**

| (In thousands)                               | Years ended December 31, |    |        |    |      |    |        |    |          |  |  |  |
|--|--------------------------|----|--------|----|------|----|--------|----|----------|--|--|--|
|  | 2018                     |    | Change |    | 2017 |    | Change |    | 2016     |  |  |  |
| Discontinued operations, net of income taxes | \$<br>_                  | \$ | _      | \$ | _    | \$ | 63,121 | \$ | (63,121) |  |  |  |

2016 discontinued operations reflects our former Search and Content and E-Commerce businesses, which were both sold in 2016. See "Note 5: Discontinued Operations" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report for additional information on discontinued operations.

## NON-GAAP FINANCIAL MEASURES

Adjusted EBITDA: We define Adjusted EBITDA as net income (loss) attributable to Blucora, Inc., determined in accordance with GAAP, excluding the effects of stock-based compensation, depreciation and amortization of acquired intangible assets (including acquired technology), restructuring, other loss, net, the impact of noncontrolling interests, income tax expense (benefit), the effects of discontinued operations, net of income taxes, and acquisition-related costs. For purposes of this definition, restructuring costs relate to the move of our corporate headquarters in 2017 and acquisition-related costs include professional services fees and other direct transaction costs and changes in the fair value of contingent consideration liabilities related to acquired companies. The SimpleTax acquisition that was completed in 2015 included contingent consideration, for which the fair value of that liability was revalued in the second quarter of 2016. For further detail, see "Note 10: Debt" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report.

We believe that Adjusted EBITDA provides meaningful supplemental information regarding our performance. We use this non-GAAP financial measure for internal management and compensation purposes, when publicly providing guidance on possible future results, and as a means to evaluate period-to-period comparisons. We believe that Adjusted EBITDA is a common measure used by investors and analysts to evaluate our performance, that it provides a more complete understanding of the results of operations and trends affecting our business when viewed together with GAAP results, and that management and investors benefit from referring to this non-GAAP financial measure. Items excluded from Adjusted EBITDA are significant and necessary components to the operations of our business and, therefore, Adjusted EBITDA should be considered as a supplement to, and not as a substitute for or superior to, GAAP net income (loss). Other companies may calculate Adjusted EBITDA differently and, therefore, our Adjusted EBITDA may not be comparable to similarly titled measures of other companies. A reconciliation of our Adjusted EBITDA to net income (loss) attributable to Blucora, Inc., which we believe to be the most comparable GAAP measure, is presented below:

| (In thousands)  | Years ended December 31, |          |    |          |      |          |  |  |  |
|---|--------------------------|----------|----|----------|------|----------|--|--|--|
|   |                          | 2018     |    | 2017     | 2017 |          |  |  |  |
| Net income (loss) attributable to Blucora, Inc.             | \$                       | 50,634   | \$ | 27,039   | \$   | (65,158) |  |  |  |
| Stock-based compensation                                    |                          | 13,253   |    | 11,653   |      | 14,128   |  |  |  |
| Depreciation and amortization of acquired intangible assets |                          | 38,590   |    | 38,139   |      | 38,688   |  |  |  |
| Restructuring   |                          | 288      |    | 3,101    |      | 3,870    |  |  |  |
| Other loss, net   |                          | 15,797   |    | 44,551   |      | 39,781   |  |  |  |
| Net income attributable to noncontrolling interests         |                          | 935      |    | 2,337    |      | 658      |  |  |  |
| Income tax expense (benefit)                                |                          | 311      |    | (25,890) |      | (1,285)  |  |  |  |
| Discontinued operations, net of income taxes                |                          | _        |    | _        |      | 63,121   |  |  |  |
| Acquisition-related costs                                   |                          | <u> </u> |    |          |      | 391      |  |  |  |
| Adjusted EBITDA   | \$                       | 119,808  | \$ | 100,930  | \$   | 94,194   |  |  |  |

Year ended December 31, 2018 compared with year ended December 31, 2017

The increase in Adjusted EBITDA was primarily due to increases in segment operating income of \$2.1 million and \$14.3 million related to our Wealth Management and Tax Preparation segments, respectively, and a \$2.4 million decrease in corporate operating expenses.

Year ended December 31, 2017 compared with year ended December 31, 2016

The increase in Adjusted EBITDA primarily was due to increases in segment operating income of \$4.6 million and \$6.0 million related to our Wealth Management and Tax Preparation segments, respectively. Offsetting the increase was a \$3.9 million increase in corporate operating expenses primarily related to costs incurred as part of our Strategic Transformation, which related to the relocation cost of our corporate headquarters, and costs associated with transitioning of roles such as overlap in staffing and recruiting search fees.

Non-GAAP net income and non-GAAP net income per share: We define non-GAAP net income as net income (loss) attributable to Blucora, Inc., determined in accordance with GAAP, excluding the effects of discontinued operations, net of income taxes, stock-based compensation, amortization of acquired intangible assets (including acquired technology), accelerated accretion of debt discount on the Notes, gain on the Notes repurchased, accretion and write-off of debt discount and debt issuance costs on previous debt, acquisition-related costs, restructuring costs, the impact of noncontrolling interests, the related cash tax impact of those adjustments, and non-cash income taxes. The write-off of debt discount and debt issuance costs on the terminated Notes and the closed TaxAct - HD Vest 2015 credit facility relates to the debt refinancing that occurred in the second quarter of 2017. We exclude the non-cash portion of income taxes because of our ability to offset a substantial portion of our cash tax liabilities by using deferred tax assets, which primarily consist of U.S. federal net operating losses. The majority of these net operating losses will expire, if unutilized, between 2020 and 2024.

We believe that non-GAAP net income and non-GAAP net income per share provide meaningful supplemental information to management, investors, and analysts regarding our performance and the valuation of our business by excluding items in the statement of operations that we do not consider part of our ongoing operations or have not been, or are not expected to be, settled in cash. Additionally, we believe that non-GAAP net income and non-GAAP net income per share are common measures used by investors and analysts to evaluate our performance and the valuation of our business. Non-GAAP net income and non-GAAP net income per share should be evaluated in light of our financial results prepared in accordance with GAAP and should be considered as a supplement to, and not as a substitute for or superior to, GAAP net income (loss) and GAAP net income (loss) per share. Other companies may calculate these non-GAAP measures differently, and, therefore, our non-GAAP net income and non-GAAP net income per share may not be comparable to similarly titled measures of other companies. Reconciliation of our non-GAAP net income and non-GAAP net income per share to net income (loss) attributable to Blucora, Inc. and net income (loss) per share attributable to Blucora, Inc., respectively, which we believe to be the most comparable GAAP measures, is presented below:

| (In thousands, except per share amounts)  | Years ended December 31, |         |    |          |    |          |  |  |  |  |
|---|--------------------------|---------|----|----------|----|----------|--|--|--|--|
|   | ·                        | 2018    |    | 2017     |    | 2016     |  |  |  |  |
| Net income (loss) attributable to Blucora, Inc.                                   | \$                       | 50,634  | \$ | 27,039   | \$ | (65,158) |  |  |  |  |
| Discontinued operations, net of income taxes                                      |                          | _       |    | _        |    | 63,121   |  |  |  |  |
| Stock-based compensation  |                          | 13,253  |    | 11,653   |    | 14,128   |  |  |  |  |
| Amortization of acquired intangible assets  |                          | 33,586  |    | 34,002   |    | 34,143   |  |  |  |  |
| Accelerated accretion of debt discount on Convertible Senior Notes                |                          | _       |    | _        |    | 1,628    |  |  |  |  |
| Gain on the Convertible Senior Notes repurchased                                  |                          | _       |    | _        |    | (7,724)  |  |  |  |  |
| Accretion and write-off of debt discount and debt issuance costs on previous debt |                          | _       |    | 17,875   |    | 3,666    |  |  |  |  |
| Acquisition-related costs   |                          | _       |    | _        |    | 391      |  |  |  |  |
| Restructuring   |                          | 288     |    | 3,101    |    | 3,870    |  |  |  |  |
| Impact of noncontrolling interests  |                          | 935     |    | 2,337    |    | 658      |  |  |  |  |
| Cash tax impact of adjustments to GAAP net income                                 |                          | (2,257) |    | (6)      |    | 175      |  |  |  |  |
| Non-cash income tax benefit   |                          | (2,403) |    | (26,853) |    | (3,802)  |  |  |  |  |
| Non-GAAP net income   | \$                       | 94,036  | \$ | 69,148   | \$ | 45,096   |  |  |  |  |

| Per diluted share:  |               |         |        |
|---|---------------|---------|--------|
| Net income (loss) attributable to Blucora, Inc. (1)                               | \$<br>0.90 \$ | 0.57 \$ | (1.53) |
| Discontinued operations, net of income taxes                                      | _             | _       | 1.48   |
| Stock-based compensation  | 0.27          | 0.25    | 0.33   |
| Amortization of acquired intangible assets  | 0.68          | 0.72    | 0.80   |
| Accelerated accretion of debt discount on Convertible Senior Notes                | _             | _       | 0.04   |
| Gain on Convertible Senior Notes repurchased                                      | _             | _       | (0.18) |
| Accretion and write-off of debt discount and debt issuance costs on previous debt | _             | 0.37    | 0.09   |
| Acquisition-related costs   | _             | _       | 0.01   |
| Restructuring   | 0.01          | 0.07    | 0.09   |
| Impacts of noncontrolling interests   | 0.14          | 0.05    | 0.02   |
| Cash tax impact of adjustments to GAAP net income                                 | (0.05)        | 0.00    | 0.00   |
| Non-cash income tax benefit   | (0.05)        | (0.57)  | (0.09) |
| Non-GAAP net income   | \$<br>1.90 \$ | 1.46 \$ | 1.06   |
| Weighted average shares outstanding used in computing per diluted share amounts   | <br>49,381    | 47,211  | 42,686 |

<sup>(1)</sup> Any difference in "per diluted share" between this table and the consolidated statements of comprehensive income is due to using different weighted average shares outstanding in the event that there is GAAP net loss but non-GAAP net income and vice versa.

Year ended December 31, 2018 compared with year ended December 31, 2017

The increase in non-GAAP net income was primarily due to increases in segment operating income of \$2.1 million and \$14.3 million related to our Wealth Management and Tax Preparation segments, respectively, a \$7.6 million decrease in interest expense, amortization of debt issuance costs, and accretion of debt discounts, primarily relating to lower balances in the Blucora senior secured credit facilities and the repricing and lowering, in 2017, of the applicable interest rate margin of the Blucora senior secured credit facilities. Further contributing to the increase in non-GAAP net income was a \$1.0 million decrease on debt extinguishment on the Blucora senior secured credit facilities, a \$2.4 million decrease in corporate operating expenses not allocated to the segments and a \$2.1 million gain on the sale of an investment. These increases were offset by a \$4.0 million increase in cash income tax expense, primarily due to lower income tax expense and lower debt extinguishment in 2018.

Year ended December 31, 2017 compared with year ended December 31, 2016

The increase in non-GAAP net income primarily was due to increases in segment operating income of \$4.6 million and \$6.0 million related to our Wealth Management and Tax Preparation segments, respectively. The increase in non-GAAP net income was also due to (i) a \$12.6 million decrease in interest expense, amortization of debt issuance costs, and accretion of debt discounts, mainly related to the TaxAct - HD Vest 2015 credit facility, which was entered into in December 2015 and terminated in the second quarter of 2017, (ii) a \$3.0 million loss on debt extinguishment and modification expense, mainly related to the prepayment of a portion of the TaxAct - HD Vest 2015 credit facility in 2016, (iii) a \$0.4 million decrease in depreciation expense, mainly related to depreciation expense on HD Vest fixed assets, and (iv) a \$1.4 million decrease in cash income tax expense, mainly related to the addition of HD Vest. These increases were offset by a \$3.9 million increase in corporate operating expenses not allocated to the segments primarily related to costs incurred as part of our Strategic Transformation, which related to the relocation cost of our corporate headquarters, and costs associated with transitioning of roles such as overlap in staffing and recruiting search fees.

## LIQUIDITY AND CAPITAL RESOURCES

# Cash and Cash Equivalents

Our principal source of liquidity is our cash and cash equivalents. As of December 31, 2018, we had cash and cash equivalents of \$84.5 million. Our HD Vest broker-dealer subsidiary operates in a highly regulated industry and is subject to various regulatory capital requirements. Failure to meet minimum capital requirements can initiate certain mandatory and possible additional discretionary actions by regulators that, if undertaken, could have substantial monetary and non-monetary

impacts on HD Vest's operations. As of December 31, 2018, HD Vest met all capital adequacy requirements to which it was subject.

We generally invest our excess cash in high quality marketable investments. Recently these investments have included money market funds invested in securities issued by agencies of the U.S. We may invest, from time-to-time, in other vehicles, such as debt instruments issued by the U.S. federal government and its agencies, international governments, municipalities and publicly-held corporations, as well as commercial paper, and insured time deposits with commercial banks. Specific holdings can vary from period to period depending upon our cash requirements. Our financial instrument investments held at December 31, 2018 had minimal default risk and short-term maturities.

Historically, we have financed our operations primarily from cash provided by operating activities. Accordingly, we believe that the cash generated from our operations and the cash and cash equivalents we have on hand will be sufficient to meet our operating, working capital, regulatory capital requirements at our broker-dealer subsidiary, and capital expenditure requirements for at least the next 12 months. However, the underlying levels of revenues and expenses that we project may not prove to be accurate and we may be required to draw on our \$50.0 million revolving credit facility to meet our capital

requirements. For further discussion of the risks to our business related to liquidity, see the paragraph in our Risk Factors (Part I Item 1A of this report) under the heading "Existing cash and cash equivalents, and cash generated from operations may not be sufficient to meet our anticipated cash needs for servicing debt, working capital, and capital expenditures."

#### Use of Cash

We may use our cash and cash equivalents balance in the future on investment in our current businesses, for repayment of debt, for acquiring companies or assets that complement our Wealth Management and Tax Preparation businesses, for stock buybacks, for returning capital to stockholders, or for other utilizations which we deem to be in the best interests of stockholders.

In May 2017, we entered into a credit agreement with a syndicate of lenders for the Blucora senior secured credit facilities in order to (a) refinance the TaxAct - HD Vest 2015 credit facility, (b) redeem our convertible notes that were outstanding at the time, and (c) provide a term loan and revolving line of credit for future working capital, capital expenditure and general business purposes. Consequently, the TaxAct - HD Vest 2015 credit facility was repaid in full with borrowings under the Blucora senior secured credit facilities and the commitments thereunder were terminated. The Blucora senior secured credit facilities in the aggregate committed amount of \$425.0 million consist of a committed \$50.0 million revolving credit facility (including a letter of credit subfacility), and a \$375.0 million term loan facility. The final maturity dates of the revolving credit loan and term loan are May 22, 2022 and May 22, 2024, respectively. In November 2017, the credit facility agreement was amended in order to refinance and reprice the initial term loan, such that the applicable interest rate margin is 3.00% for Eurodollar Rate loans and 2.00% for ABR loans. Depending on Blucora's Consolidated First Lien Net Leverage Ratio (as defined in the credit facility agreement), the applicable interest rate margin on the revolving credit facility is from 2.75% to 3.00% for Eurodollar Rate loans and 1.75% to 2.00% for ABR loans. Obligations under the Blucora senior secured credit facilities are guaranteed by certain of Blucora's subsidiaries and secured by the assets of Blucora and those subsidiaries.

The Blucora senior secured credit facilities include financial and operating covenants with respect to certain ratios, including a net leverage ratio, which are defined further in the credit facility agreement. We were in compliance with these covenants as of December 31, 2018. We initially borrowed \$375.0 million under the term loan and have made prepayments of \$110.0 million towards the term loan since entering into the agreement, of which \$80.0 million was prepaid in the year ended December 31, 2018, such that \$265.0 million was outstanding under the term loan at December 31, 2018. We have not borrowed any amounts under the revolving credit loan and do not have any other debt outstanding. Beginning with the fiscal year ending December 31, 2018, we may be required to make annual prepayments of the term loan in an amount equal to a percentage of our excess cash flow during the applicable fiscal year from 0% to 50%, depending on the Consolidated First Lien Net Leverage Ratio (as defined in the credit facility agreement) for such fiscal year. An excess cash flow payment will not be required for the fiscal year ending 2018. In the past we have used excess cash flows to make debt prepayments, and we currently expect to make further prepayments in 2019. For further detail, see "Note 10: Debt" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report.

Related to the TaxAct - HD Vest 2015 credit facility, we had repayment activity of \$64.0 million during the year ended December 31, 2017 prior to the refinancing.

On July 2, 2015, TaxAct acquired SimpleTax, which included additional consideration of up to C\$4.6 million (with C\$ indicating Canadian dollars and amounting to approximately \$3.7 million based on the acquisition-date exchange rate). The related payments are contingent upon product availability and revenue performance over a three-year period and are expected

to occur annually over that period. The first two payments of \$1.3 million and \$0.9 million were made in the first quarters of 2018 and 2017, respectively, and the remaining payment of \$1.3 million is expected in the first quarter of 2019. For further detail, see "Note 8: Fair Value Measurements" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report.

On October 14, 2015, we announced our Strategic Transformation, which refers to our transformation into a technology-enabled financial solutions company comprised of TaxAct and HD Vest, the divestitures of our Search and Content and E-Commerce businesses, and the relocation of corporate headquarters from Bellevue, Washington to Irving, Texas. See the "Strategic Transformation" subsection in Part I Item 1 for additional detail regarding the related use of cash.

Related to the acquisition of HD Vest, we paid \$613.7 million (after a \$1.8 million working capital adjustment in the first quarter of 2016) in cash, which was funded by a combination of cash on hand and the TaxAct - HD Vest 2015 credit facility. For further detail, see "Note 4: Business Combinations" and "Note 10: Debt" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report.

In connection with the HD Vest acquisition, former management of that business has retained an ownership interest in HD Vest. We are party to put and call arrangements, exercisable beginning in 2019, with respect to those interests. These put and call arrangements allow former HD Vest management to require us to purchase their interests or allow us to acquire such interests, respectively. The redemption amount at December 31, 2018 and December 31, 2017 was \$24.9 million and \$12.4 million, respectively. We expect the December 31, 2018 redemption amount will be paid in 2019, and our cash and cash equivalents balance will decrease by \$24.9 million.

Contractual Obligations and Commitments

Our contractual obligations and commitments are as follows for years ending December 31:

| (In thousands)   | 2019         | 2020         | 2021         | 2022 |        | 2023 |        | Thereafter    | Total         |
|--|--------------|--------------|--------------|------|--------|------|--------|---------------|---------------|
| Operating lease commitments:                           |              |              |              |      |        |      |        |               |               |
| Operating lease obligations                            | \$<br>3,759  | \$<br>3,461  | \$<br>2,245  | \$   | 1,979  | \$   | 1,677  | \$<br>_       | \$<br>13,121  |
| Sublease income  | (1,288)      | (991)        | _            |      | _      |      | _      | _             | (2,279)       |
| Net operating lease commitments                        | 2,471        | 2,470        | 2,245        |      | 1,979  |      | 1,677  | _             | 10,842        |
| Purchase commitments                                   | 10,088       | 7,311        | 5,640        |      | 4,077  |      | 2,444  | 3,325         | 32,885        |
| Debt commitments                                       | _            | _            | _            |      | _      |      | _      | 265,000       | 265,000       |
| Interest payable                                       | 11,654       | 11,686       | 11,654       |      | 11,654 |      | 11,654 | 4,854         | 63,156        |
| Acquisition-related contingent consideration liability | 1,275        | _            | _            |      | _      |      | _      | _             | 1,275         |
| Total  | \$<br>25,488 | \$<br>21,467 | \$<br>19,539 | \$   | 17,710 | \$   | 15,775 | \$<br>273,179 | \$<br>373,158 |

For further detail see "Note 11: Commitments and Contingencies" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report.

Off-balance Sheet Arrangements

We have no off-balance sheet arrangements, other than operating leases.

Unrecognized Tax Benefits

The above table does not reflect unrecognized tax benefits of approximately \$4.7 million, the timing of which is uncertain. For additional discussion on unrecognized tax benefits see "Note 16: Income Taxes" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report.

Cash Flows

Our cash flows were comprised of the following:

| (In thousands)   | Years ended December 31, |          |      |          |    |           |  |  |  |
|--|--------------------------|----------|------|----------|----|-----------|--|--|--|
|  |                          | 2018     | 2017 |          |    | 2016      |  |  |  |
| Net cash provided by operating activities                    | \$                       | 105,548  | \$   | 72,846   | \$ | 85,970    |  |  |  |
| Net cash provided (used) by investing activities             |                          | (7,633)  |      | 2,053    |    | (1,560)   |  |  |  |
| Net cash used by financing activities                        |                          | (74,804) |      | (68,562) |    | (162,001) |  |  |  |
| Net cash provided (used) by continuing operations            |                          | 23,111   |      | 6,337    |    | (77,591)  |  |  |  |
| Net cash provided by discontinued operations                 |                          | _        |      | 1,028    |    | 72,655    |  |  |  |
| Effect of exchange rate changes on cash and cash equivalents |                          | (56)     |      | 78       |    | (26)      |  |  |  |
| Net increase (decrease) in cash and cash equivalents         | \$                       | 23,055   | \$   | 7,443    | \$ | (4,962)   |  |  |  |

Net cash from operating activities: Net cash from the operating activities of continuing operations consists of income (loss) from continuing operations, offset by certain non-cash adjustments, and changes in our working capital.

Net cash provided by operating activities was \$105.5 million, \$72.8 million, and \$86.0 million for the years ended December 31, 2018, 2017, and 2016, respectively. The activity in 2018 included a \$2.6 million working capital contribution and approximately \$103.0 million of income from continuing operations (offset by non-cash adjustments). The working capital contribution was primarily driven by clearing firm conversion incentives and the timing of accurals

The activity in 2017 included a \$(15.3) million working capital contribution and approximately \$88.1 million of income from continuing operations (offset by non-cash adjustments). The working capital contribution was driven by accrued expenses, including accrued interest, and the expected realization (through 2022) of \$10.9 million of repealed corporate AMT credit carryovers, and restructuring activities.

The activity in 2016 included a \$44.8 million working capital contribution and approximately \$41.2 million of non-cash adjustments and a loss from continuing operations. The working capital contribution was driven by accrued expenses and the impact of excess tax benefits from stock-based activity primarily due to utilizing equity net operating loss carryforwards from prior years. In addition, we had placed into escrow \$20.0 million of additional consideration that was contingent upon HD Vest's 2015 earnings performance, and that amount was returned to us in the first quarter of 2016 since it was not achieved (see "Note 4: Business Combinations" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report for additional information). Lastly, the timing of TaxAct's spending on marketing campaigns and the addition of HD Vest provided further working capital contribution during the period.

Net cash from investing activities: Net cash from the investing activities of continuing operations primarily consists of cash outlays for business acquisitions, transactions (purchases of and proceeds from sales and maturities) related to our investments, and purchases of property and equipment. Our investing activities tend to fluctuate from period-to-period primarily based upon the level of acquisition activity.

Net cash provided (used) by investing activities was \$(7.6) million, \$2.1 million, and \$(1.6) million for the years ended December 31, 2018, 2017, and 2016, respectively. The activity in 2018 consisted of \$7.6 million in purchases of property and equipment. The activity in 2017 primarily consisted of net cash inflows on our available-for-sale investments of \$7.1 million offset by \$5.0 million in purchases of property and equipment. The activity in 2016 primarily consisted of \$3.8 million in purchases of property and equipment and payment of the \$1.8 million final working capital adjustment on the HD Vest acquisition, offset by net cash inflows on our available-for-sale investments of \$4.0 million.

Net cash from financing activities: Net cash from the financing activities of continuing operations primarily consists of transactions related to the issuance of debt and stock. Our financing activities tend to fluctuate from period-to-period based upon our financing needs.

Net cash used by financing activities was \$(74.8) million, \$(68.6) million, and \$(162.0) million for the years ended December 31, 2018, 2017, and 2016, respectively.

The activity in 2018 primarily consisted of prepayments of \$80.0 million towards the term loan under the Blucora senior secured credit facilities, \$8.4 million in tax payments from shares withheld for equity awards, and \$1.3 million in contingent

consideration paid related to the 2015 acquisition of SimpleTax. These cash outflows were offset by approximately \$14.9 million in combined proceeds from the issuance of common stock related to stock option exercises and the employee stock purchase plan.

The activity in 2017 primarily consisted of payments of \$290.0 million in connection with the termination of the TaxAct - HD Vest credit facility, \$172.8 million for redemption in full of the outstanding Notes, \$9.1 million in tax payments from shares withheld for equity awards, payment of \$3.2 million on a related party note payable with the former President of HD Vest that arose in connection with the HD Vest acquisition, and \$0.9 million in contingent consideration paid related to the 2015 acquisition of SimpleTax. These cash outflows were offset by approximately \$365.8 million in proceeds from the Blucora senior secured credit facilities that was entered into in May 2017 and \$41.7 million in combined proceeds from the issuance of common stock related to stock option exercises and the employee stock purchase plan.

Net cash used by financing activities was \$162.0 million for the year ended December 31, 2016. The activity in 2016 primarily consisted of payments of \$140.0 million on the TaxAct - HD Vest 2015 credit facility, the \$20.7 million repurchase of the Notes, payment of \$3.2 million on the note payable with related party, and \$1.8 million in tax payments from shares withheld for equity awards. These cash outflows were offset by approximately \$16.0 million in excess tax benefits from stock-based award activity primarily due to utilizing equity net operating loss carryforwards from prior years and \$3.6 million in combined proceeds from the issuance of common stock related to stock option exercises and the employee stock purchase plan.

## **Critical Accounting Policies and Estimates**

This Management's Discussion and Analysis of Financial Condition and Results of Operations and the disclosures included elsewhere in this Annual Report on Form 10-K are based upon our consolidated financial statements, which have been prepared in accordance with GAAP. The preparation of these financial statements requires us to make estimates and assumptions that affect the reported amounts of assets, liabilities, revenues, expenses, and disclosure of contingencies. In some cases, we could have reasonably used different accounting policies and estimates.

The SEC has defined a company's most critical accounting policies as the ones that are the most important to the portrayal of the company's financial condition and results of operations and which require the company to make its most difficult and subjective judgments, often as a result of the need to make estimates of matters that are inherently uncertain. On an ongoing basis, we evaluate the estimates used. We base our estimates on historical experience, current conditions, and on various other assumptions that we believe to be reasonable under the circumstances and, based on information available to us at that time, we make judgments about the carrying values of assets and liabilities that are not readily apparent from other sources as well as identify and assess our accounting treatment with respect to commitments and contingencies. Actual results may differ significantly from these estimates under different assumptions, judgments, or conditions. We believe the following critical accounting policies involve the more significant judgments and estimates used in the preparation of our consolidated financial statements.

Wealth management revenue recognition: Wealth management revenue consists primarily of commission revenue, advisory revenue, asset-based revenue, and transaction and fee revenue. Revenue is recognized upon the transfer of services to customers in an amount that reflects the consideration to which we expect to be entitled in exchange for those services. Payments received in advance of the performance of service are deferred and recognized as revenue when earned. Within the components of Wealth management revenue, commissions revenue is determined through the use of subjective judgments and the use of estimates.

Commissions represent amounts generated by HD Vest's financial advisors for their clients' purchases and sales of securities and various investment products. We generate two types of commissions: transaction-based sales commissions that occur at the point of sale, as well as trailing commissions for which we provide ongoing account support to clients of our financial advisors.

Transaction-based sales commission revenue is recorded on a trade-date basis, which is when our performance obligations in generating the commissions have been substantially completed. Trailing commission revenue is based on a percentage of the current market value of clients' investment holdings in trail-eligible assets and recognized over the period during which services are performed. Since trailing commission revenue is generally paid in arrears, we estimate it based on a number of factors, including market levels and the amount of trailing commission revenues received in prior periods, and also considers historical payout ratios. These estimates are primarily based on historical information and there is not significant judgment involved.

A substantial portion of commission revenue is ultimately paid to financial advisors. Such amounts are recorded as "Commissions and advisory fees payable" on the consolidated balance sheets and "Wealth management services cost of revenue" on the consolidated statements of comprehensive income.

Tax preparation revenue recognition: We derive revenue from the sale of tax preparation digital services, ancillary services, packaged tax preparation software, and multiple element arrangements that may include a combination of these items. Ancillary services primarily include tax preparation support services, and e-filing services. The Company recognizes revenue from the sale of its packaged software when legal title transfers. Legal title transfers generally when its customers download the software from the Internet or, in some cases, when the software ships. Within the components of tax preparation revenue, revenues from bank or reloadable prepaid debit card services and software and/or services that consist of multiple elements are determined through the use of subjective judgments and the use of estimates.

The bank or reloadable prepaid debit card services are offered to taxpayers as an option to receive their tax refunds in the form of a prepaid bank card or to have the fees for the software and/or services purchased by the customers deducted from their refunds. Other value-added service revenue consists of revenue from revenue sharing and royalty arrangements with third party partners. Revenue for these transactions is recognized when the revenue recognition criteria are met; for some arrangements that is upon filing and for other arrangements that is upon our determination of when collectibility is probable.

For software and/or services that consist of multiple elements, we must: (1) determine whether and when each element has been delivered; (2) determine the fair value of each element; and (3) allocate the total price among the various elements based on the relative selling price method. Once we have allocated the total price among the various elements, we recognize revenue when the revenue recognition criteria described above are met for each element. The impact of multiple element arrangements is not material and primarily impacts the timing of revenue recognition over the tax filing season, which is concentrated within the first two quarters of the filing period.

Income taxes: We account for income taxes under the asset and liability method, under which deferred tax assets, including net operating loss carryforwards, and liabilities are determined based on temporary differences between the book and tax bases of assets and liabilities. We periodically evaluate the likelihood of the realization of deferred tax assets and reduce the carrying amount of the deferred tax assets by a valuation allowance to the extent we believe a portion will not be realized. We consider many factors when assessing the likelihood of future realization of the deferred tax assets, including expectations of future taxable income, recent cumulative earnings experience by taxing jurisdiction, and other relevant factors. There is a wide range of possible judgments relating to the valuation of our deferred tax assets.

We record liabilities to address uncertain tax positions that have been taken in previously filed tax returns or that are expected to be taken in a future tax return. The determination for required liabilities is based upon an analysis of each individual tax position, taking into consideration whether it is more likely than not that the tax position, based on technical merits, will be sustained upon examination. The tax benefit to be recognized in the financial statements from such a position is measured as the largest amount of benefit that has a greater than 50% likelihood of being realized upon ultimate settlement with the taxing authority. The difference between the amount recognized and the total tax position is recorded as a liability. The ultimate resolution of these tax positions may be greater or less than the liabilities recorded.

## **Recent Accounting Pronouncements**

See "Note 2: Summary of Significant Accounting Policies" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report.

# **Quarterly Results of Operations (Unaudited)**

The following table presents a summary of our unaudited consolidated results of operations for the eight quarters ended December 31, 2018. The information for each of these quarters has been prepared on a basis consistent with our annual audited consolidated financial statements. You should read this information in conjunction with our consolidated financial statements and notes thereto in Part II Item 8. The operating results for any quarter are not necessarily indicative of results for any future period.

|   | Mar     | ch 31, 2017  | J      | une 30, 2017 | s  | eptember 30, 2017 | D    | ecember 31, 2017    | M     | Iarch 31, 2018    | Ju    | ne 30, 2018 | September 30, 201 |          | De | cember 31, 2018 |
|---|---------|--------------|--------|--------------|----|-------------------|------|---------------------|-------|-------------------|-------|-------------|-------------------|----------|----|-----------------|
| _   |         |              |        |              |    | (In               | thou | sands except per sh | are ( | data and percenta | iges) |             |                   |          |    |                 |
| Revenue:  |         |              |        |              |    |                   |      |                     |       |                   |       |             |                   |          |    |                 |
| Wealth management services revenue                  | \$      | 82,667       | \$     | 85,296       | \$ | 86,809            | \$   | 93,848              | \$    | 92,082            | \$    | 92,015      | \$                | 91,887   | \$ | 97,190          |
| Tax preparation services revenue                    |         | 99,708       |        | 53,866       |    | 3,362             |      | 4,001               |       | 113,883           |       | 65,833      |                   | 3,498    |    | 4,068           |
| Total revenue                                       |         | 182,375      |        | 139,162      |    | 90,171            |      | 97,849              |       | 205,965           |       | 157,848     |                   | 95,385   |    | 101,258         |
| Operating expenses:                                 |         |              |        |              |    |                   |      |                     |       |                   |       |             |                   |          |    |                 |
| Cost of revenue:                                    |         |              |        |              |    |                   |      |                     |       |                   |       |             |                   |          |    |                 |
| Wealth management services cost of revenue          |         | 55,874       |        | 56,963       |    | 59,607            |      | 63,415              |       | 63,064            |       | 62,149      |                   | 62,313   |    | 66,054          |
| Tax preparation services cost of revenue            |         | 3,818        |        | 2,411        |    | 1,314             |      | 2,475               |       | 4,353             |       | 2,459       |                   | 1,370    |    | 1,858           |
| Amortization of acquired technology                 |         | 48           |        | 47           |    | 50                |      | 50                  |       | 50                |       | 49          |                   | _        |    | _               |
| Total cost of revenue                               |         | 59,740       |        | 59,421       |    | 60,971            |      | 65,940              |       | 67,467            |       | 64,657      |                   | 63,683   |    | 67,912          |
| Engineering and technology                          |         | 4,748        |        | 4,242        |    | 5,051             |      | 5,573               |       | 5,131             |       | 4,848       |                   | 4,246    |    | 5,107           |
| Sales and marketing                                 |         | 48,998       |        | 22,296       |    | 13,680            |      | 17,824              |       | 55,253            |       | 23,791      |                   | 15,675   |    | 16,642          |
| General and administrative                          |         | 13,483       |        | 13,715       |    | 12,207            |      | 13,263              |       | 14,866            |       | 15,625      |                   | 13,404   |    | 16,229          |
| Depreciation  |         | 940          |        | 873          |    | 867               |      | 780                 |       | 1,915             |       | 993         |                   | 798      |    | 762             |
| Amortization of other acquired intangible assets    |         | 8,288        |        | 8,289        |    | 8,615             |      | 8,615               |       | 8,307             |       | 8,806       |                   | 8,271    |    | 8,103           |
| Restructuring (1)                                   |         | 2,289        |        | 331          |    | 106               |      | 375                 |       | 289               |       | 2           |                   |          |    | (3)             |
| Total operating expenses                            |         | 138,486      | _      | 109,167      |    | 101,497           |      | 112,370             |       | 153,228           | _     | 118,722     |                   | 106,077  |    | 114,752         |
| Operating income (loss)                             |         | 43,889       | _      | 29,995       |    | (11,326)          | _    | (14,521)            |       | 52,737            |       | 39,126      |                   | (10,692) |    | (13,494)        |
| Other loss, net                                     |         | (9,708)      |        | (24,200)     |    | (5,241)           |      | (5,402)             |       | (5,228)           |       | (2,759)     |                   | (3,863)  |    | (3,947)         |
| Income (loss) before income taxes                   |         | 34,181       |        | 5,795        |    | (16,567)          |      | (19,923)            |       | 47,509            |       | 36,367      |                   | (14,555) |    | (17,441)        |
| Income tax benefit (expense)                        |         | (3,471)      |        | (2,315)      |    | (166)             |      | 31,842              |       | (1,963)           |       | (907)       |                   | 818      |    | 1,741           |
| Net income (loss)                                   | -       | 30,710       |        | 3,480        | _  | (16,733)          |      | 11,919              |       | 45,546            |       | 35,460      |                   | (13,737) |    | (15,700)        |
| Net income attributable to noncontrolling interests |         | (126)        |        | (176)        |    | (164)             |      | (1,871)             |       | (205)             |       | (222)       |                   | (227)    |    | (281)           |
| Net income (loss) attributable to Blucora, Inc.     | \$      | 30,584       | \$     | 3,304        | \$ | (16,897)          | \$   | 10,048              | \$    | 45,341            | \$    | 35,238      | \$                | (13,964) | \$ | (15,981)        |
| Net income (loss) per share att                     | ributal | ble to Bluco | ra, In | ıc.:         | _  |                   |      |                     |       |                   |       |             |                   |          |    |                 |
| Basic   | \$      | 0.73         | \$     | 0.08         | \$ | (0.37)            | \$   | 0.22                | \$    | 0.97              | \$    | 0.75        | \$                | (0.37)   | \$ | (0.38)          |
| Diluted   | \$      | 0.67         | \$     | 0.07         | \$ | (0.37)            | \$   | 0.21                | \$    | 0.93              | \$    | 0.71        | \$                | (0.37)   | \$ | (0.38)          |
| Weighted average shares outst                       | anding  | Σ:           | _      |              | _  |                   | _    |                     | _     |                   |       |             |                   |          |    |                 |
| Basic   |         | 42,145       |        | 43,644       |    | 45,459            |      | 46,231              |       | 46,641            |       | 47,221      |                   | 47,712   |    | 48,002          |
| Diluted   |         | 45,428       |        | 46,937       |    | 45,459            |      | 48,406              |       | 48,665            |       | 49,434      |                   | 47,712   |    | 48,002          |
|   |         |              |        |              |    |                   | 5    | 2                   |       |                   |       |             |                   |          |    |                 |

|  | March 31, 2017 | June 30, 2017 | September 30, 2017 | December 31, 2017 | March 31, 2018 | June 30, 2018 | September 30, 2018 | December 31, 2018 |
|--|----------------|---------------|--------------------|-------------------|----------------|---------------|--------------------|-------------------|
| Revenue:   |                |               |                    |                   |                |               |                    |                   |
| Wealth management services revenue                                 | 45.3 %         | 61.3 %        | 96.3 %             | 95.9 %            | 44.7 %         | 58.3 %        | 96.3 %             | 96.0 %            |
| Tax preparation services   | 54.7           | 38.7          | 3.7                | 4.1               | 55.3           | 41.7          | 3.7                | 4.0               |
| revenue Total revenue  | 100.0          | 100.0         | 100.0              | 100.0             | 100.0          | 100.0         | 100.0              | 100.0             |
| Operating expenses:  | 100.0          | 100.0         | 100.0              | 100.0             | 100.0          | 100.0         | 100.0              | 100.0             |
| Cost of revenue (2):   |                |               |                    |                   |                |               |                    |                   |
| Wealth management services cost of revenue                         | 67.6           | 66.8          | 68.7               | 67.6              | 68.5           | 67.5          | 67.8               | 68.0              |
| Tax preparation<br>services cost of<br>revenue                     | 3.8            | 4.5           | 39.1               | 61.9              | 3.8            | 3.7           | 39.2               | 45.7              |
| Amortization of acquired technology                                | 0.0            | 0.0           | 0.1                | 0.1               | 0.0            | 0.0           | 0.0                | 0.0               |
| Total cost of revenue  | 32.8           | 42.7          | 67.6               | 67.4              | 32.8           | 41.0          | 66.8               | 67.1              |
| Engineering and technology   | 2.6            | 3.0           | 5.6                | 5.7               | 2.5            | 3.1           | 4.5                | 5.0               |
| Sales and marketing  | 26.9           | 16.0          | 15.2               | 18.2              | 26.8           | 15.1          | 16.4               | 16.4              |
| General and administrative   | 7.4            | 9.9           | 13.5               | 13.6              | 7.2            | 9.9           | 14.1               | 16.0              |
| Depreciation   | 0.5            | 0.6           | 1.0                | 0.8               | 0.9            | 0.6           | 0.8                | 0.8               |
| Amortization of other acquired intangible assets                   | 4.5            | 6.0           | 9.6                | 8.8               | 4.0            | 5.6           | 8.7                | 8.0               |
| Restructuring  | 1.3            | 0.2           | 0.1                | 0.4               | 0.1            | _             | _                  | _                 |
| Total operating expenses   | 76.0           | 78.4          | 112.6              | 114.9             | 74.3           | 75.3          | 111.3              | 113.3             |
| Operating income (loss)  | 24.0           | 21.6          | (12.6)             | (14.9)            | 25.7           | 24.7          | (11.3)             | (13.3)            |
| Other loss, net  | (5.3)          | (17.4)        | (5.8)              | (5.5)             | (2.5)          | (1.7)         | (4.0)              | (3.9)             |
| Income (loss) from<br>continuing operations<br>before income taxes | 18.7           | 4.2           | (18.4)             | (20.4)            | 23.2           | 23.0          | (15.3)             | (17.2)            |
| Income tax benefit (expense)                                       | (1.9)          | (1.7)         | (0.2)              | 32.5              | (1.0)          | (0.6)         | 0.9                | 1.7               |
| Net income (loss)  | 16.8           | 2.5           | (18.6)             | 12.1              | 22.2           | 22.4          | (14.4)             | (15.5)            |
| Net income attributable to noncontrolling interests                | (0.1)          | (0.1)         | (0.2)              | (1.9)             | (0.1)          | (0.1)         | (0.2)              | (0.3)             |
| Net income (loss)<br>attributable to Blucora,<br>Inc.              | 16.7 %         | 2.4 %         | (18.8)%            | 10.2 %            | 22.1 %         | 22.3 %        | (14.6)%            | (15.8)%           |

<sup>(1)</sup> In 2017 we relocated our corporate headquarters from Bellevue, Washington to Irving, Texas. In connection with this plan, we incurred restructuring costs. See "Note 6: Restructuring" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report for more information.

<sup>(2) &</sup>quot;Wealth management services cost of revenue" and "Tax preparation services cost of revenue" are calculated based on their respective revenue bases of "Wealth management services revenue" and "Tax Preparation services revenue," respectively. "Total cost of revenue" is calculated based on "Total revenue."

## ITEM 7A. Quantitative and Qualitative Disclosures about Market Risk

We are exposed to financial market risks, including changes in the market values of our marketable debt securities and interest rates.

Financial market risk: We do not invest in financial instruments or their derivatives for trading or speculative purposes. By policy, we limit our credit exposure to any one issuer, other than securities issued by the U.S. federal government and its agencies, and do not have any derivative instruments in our investment portfolio. The three primary goals that guide our investment decisions, with the first being the most important, are: preserve capital, maintain ease of conversion into immediate liquidity, and achieve a rate of return over a pre-determined benchmark. As of December 31, 2018, we were principally invested in money market fund securities. We consider the market value, default, and liquidity risks of our investments to be low at December 31, 2018.

Interest rate risk: At December 31, 2018, our cash equivalent balance of \$23.2 million was held in money market funds. We consider the interest rate risk for our cash equivalent securities held at December 31, 2018 to be low. For further detail on our cash equivalents, see "Note 8: Fair Value Measurements" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report.

In addition, as of December 31, 2018, we had \$265.0 million of debt outstanding under the Blucora senior secured credit facilities, which carries a degree of interest rate risk. This debt has a floating portion of its interest rate tied to the London Interbank Offered Rate ("LIBOR"). For further information on our outstanding debt, see "Note 10: Debt" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report. A hypothetical 100 basis point increase in LIBOR on December 31, 2018 would result in a \$14.6 million increase in our interest expense until the scheduled maturity date in 2024.

The following table provides information about our cash equivalent securities as of December 31, 2018, including principal cash flows for 2019 and thereafter and the related weighted average interest rates. Principal amounts and weighted average interest rates by expected year of maturity are as follows:

| (In thousands, except percentages) | 20           | 19 |       |            | There | after |           | To           | otal |       | Fa | ir Value |
|------------------------------------|--------------|----|-------|------------|-------|-------|-----------|--------------|------|-------|----|----------|
| Money market and other funds       | 23,181       |    | 1.95% |            | _     |       | <u>_%</u> | 23,181       |      | 1.95% |    | 23,181   |
| Cash equivalents                   | \$<br>23,181 |    |       | \$         | _     |       | -         | \$<br>23,181 |      |       | \$ | 23,181   |
|                                    |              |    |       | <i>5</i> 1 |       |       |           |              |      |       |    |          |

# ITEM 8. Financial Statements and Supplementary Data

# INDEX TO CONSOLIDATED FINANCIAL STATEMENTS

|   | Page      |
|---|-----------|
| Report of Independent Registered Public Accounting Firm | <u>56</u> |
| Consolidated Balance Sheets                             | <u>57</u> |
| Consolidated Statements of Comprehensive Income (Loss)  | <u>58</u> |
| Consolidated Statements of Stockholders' Equity         | <u>59</u> |
| Consolidated Statements of Cash Flows                   | <u>60</u> |
| Notes to Consolidated Financial Statements              | <u>61</u> |
| 55  |           |

## REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Stockholders and the Board of Directors of Blucora, Inc.

## **Opinion on the Financial Statements**

We have audited the accompanying consolidated balance sheets of Blucora, Inc. (the Company) as of December 31, 2018 and 2017, the related consolidated statements of comprehensive income (loss), stockholders' equity and cash flows for each of the three years in the period ended December 31, 2018, and the related notes (collectively referred to as the "consolidated financial statements"). In our opinion, the consolidated financial statements present fairly, in all material respects, the financial position of the Company at December 31, 2018 and 2017, and the results of its operations and its cash flows for each of the three years in the period ended December 31, 2018, in conformity with U.S. generally accepted accounting principles.

We also have audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States) (PCAOB), the Company's internal control over financial reporting as of December 31, 2018, based on criteria established in Internal Control-Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission (2013 framework), and our report dated March 1, 2019 expressed an unqualified opinion thereon.

## **Basis for Opinion**

These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on the Company's financial statements based on our audits. We are a public accounting firm registered with the PCAOB and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audits in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement, whether due to error or fraud. Our audits included performing procedures to assess the risks of material misstatement of the financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements. Our audits also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audits provide a reasonable basis for our opinion.

# /s/ Ernst & Young LLP

We have served as the Company's auditor since 2012. Dallas, Texas March 1, 2019

# BLUCORA, INC. CONSOLIDATED BALANCE SHEETS (In thousands, except per share data)

| ASSETS rent assets: Cash and cash equivalents Cash segregated under federal or other regulations Accounts receivable, net of allowance | \$ | 2018      |    | 2017        |
|--|----|-----------|----|-------------|
| rent assets:  Cash and cash equivalents  Cash segregated under federal or other regulations  | \$ |           |    |             |
| Cash and cash equivalents Cash segregated under federal or other regulations   | \$ |           |    |             |
| Cash segregated under federal or other regulations   | \$ |           |    |             |
|  |    | 84,524    | \$ | 59,965      |
| Accounts receivable, net of allowance  |    | 842       |    | 1,371       |
| recounts receivable, net of anowance   |    | 14,977    |    | 10,694      |
| Commissions receivable   |    | 15,562    |    | 16,822      |
| Other receivables  |    | 7,408     |    | 3,180       |
| Prepaid expenses and other current assets, net   |    | 7,755     |    | 7,365       |
| Total current assets   |    | 131,068   |    | 99,397      |
| g-term assets:   |    |           |    |             |
| Property and equipment, net  |    | 12,389    |    | 9,831       |
| Goodwill, net  |    | 548,685   |    | 549,037     |
| Other intangible assets, net   |    | 294,603   |    | 328,205     |
| Other long-term assets   |    | 10,980    |    | 15,201      |
| Total long-term assets   |    | 866,657   |    | 902,274     |
| Total assets   | \$ | 997,725   | \$ | 1,001,671   |
| LIABILITIES AND STOCKHOLDERS' EQUITY   |    |           |    |             |
| rent liabilities:  |    |           |    |             |
| Accounts payable   | \$ | 3,798     | \$ | 4,413       |
| Commissions and advisory fees payable  |    | 15,199    |    | 17,813      |
| Accrued expenses and other current liabilities   |    | 19,026    |    | 19,577      |
| Deferred revenue   |    | 10,257    |    | 9,953       |
| Total current liabilities  |    | 48,280    |    | 51,756      |
| g-term liabilities:  |    |           |    |             |
| Long-term debt, net  |    | 260,390   |    | 338,081     |
| Deferred tax liability, net  |    | 40,394    |    | 43,433      |
| Deferred revenue   |    | 8,581     |    | 804         |
| Other long-term liabilities  |    | 7,540     |    | 8,177       |
| Total long-term liabilities  |    | 316,905   |    | 390,495     |
| Total liabilities  |    | 365,185   |    | 442,251     |
|  |    |           |    |             |
| eemable noncontrolling interests   |    | 24,945    |    | 18,033      |
| amitments and contingencies (Note 11)  |    |           |    |             |
| mitments and contingencies (Note 11)   |    |           |    |             |
| kholders' equity:  |    |           |    |             |
| Common stock, par \$0.0001—authorized shares, 900,000; issued and outstanding shares,  |    | =         |    | 5           |
| 48,044 and 46,366  |    | 1.500.725 |    | 1.555.500   |
| Additional paid-in capital   |    | 1,569,725 |    | 1,555,560   |
| Accumulated deficit  |    | (961,689) |    | (1,014,174) |
| Accumulated other comprehensive loss   |    | (446)     |    | 541 297     |
| Total stockholders' equity   | •  | 607,595   | •  | 541,387     |
| Total liabilities and stockholders' equity  See notes to consolidated financial statements.  | \$ | 997,725   | \$ | 1,001,671   |

# BLUCORA, INC. CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (LOSS) (In thousands, except per share data)

|  |    | 1,       |    |          |          |          |
|--|----|----------|----|----------|----------|----------|
|  |    | 2018     |    | 2017     |          | 2016     |
| Revenue:   |    |          |    |          |          |          |
| Wealth management services revenue                                 | \$ | 373,174  | \$ | 348,620  | \$       | 316,546  |
| Tax preparation services revenue                                   |    | 187,282  |    | 160,937  |          | 139,365  |
| Total revenue  |    | 560,456  |    | 509,557  |          | 455,911  |
| Operating expenses:  |    |          |    |          |          |          |
| Cost of revenue:   |    |          |    |          |          |          |
| Wealth management services cost of revenue                         |    | 253,580  |    | 235,859  |          | 213,996  |
| Tax preparation services cost of revenue                           |    | 10,040   |    | 10,018   |          | 8,368    |
| Amortization of acquired technology                                |    | 99       |    | 195      |          | 812      |
| Total cost of revenue  |    | 263,719  |    | 246,072  |          | 223,176  |
| Engineering and technology   |    | 19,332   |    | 19,614   |          | 17,780   |
| Sales and marketing  |    | 111,361  |    | 102,798  |          | 89,360   |
| General and administrative   |    | 60,124   |    | 52,668   |          | 47,396   |
| Depreciation   |    | 4,468    |    | 3,460    |          | 3,881    |
| Amortization of other acquired intangible assets                   |    | 33,487   |    | 33,807   |          | 33,331   |
| Restructuring  |    | 288      |    | 3,101    |          | 3,870    |
| Total operating expenses   |    | 492,779  |    | 461,520  |          | 418,794  |
| Operating income   |    | 67,677   |    | 48,037   |          | 37,117   |
| Other loss, net  |    | (15,797) |    | (44,551) |          | (39,781) |
| Income (loss) from continuing operations before income taxes       |    | 51,880   |    | 3,486    |          | (2,664)  |
| Income tax benefit (expense)                                       |    | (311)    |    | 25,890   |          | 1,285    |
| Income (loss) from continuing operations                           |    | 51,569   |    | 29,376   |          | (1,379)  |
| Discontinued operations, net of income taxes                       |    | _        |    | _        |          | (63,121) |
| Net income (loss)  |    | 51,569   |    | 29,376   |          | (64,500) |
| Net income attributable to noncontrolling interests                |    | (935)    |    | (2,337)  |          | (658)    |
| Net income (loss) attributable to Blucora, Inc.                    | \$ | 50,634   | \$ | 27,039   | \$       | (65,158) |
| Net income (loss) per share attributable to Blucora, Inc basic*:   |    |          |    |          |          |          |
| Continuing operations  | \$ | 0.94     | \$ | 0.61     | \$       | (0.05)   |
| Discontinued operations  |    | _        |    | _        |          | (1.52)   |
| Basic net income (loss) per share                                  | \$ | 0.94     | \$ | 0.61     | \$       | (1.57)   |
| Net income (loss) per share attributable to Blucora, Inc diluted*: |    |          |    |          |          |          |
| Continuing operations  | \$ | 0.90     | \$ | 0.57     | \$       | (0.05)   |
| Discontinued operations  |    |          | Ψ  |          | •        | (1.52)   |
| Diluted net income (loss) per share                                | \$ | 0.90     | \$ | 0.57     | \$       | (1.57)   |
| Weighted average shares outstanding:                               |    |          | _  |          | <u> </u> | (1.57)   |
| Basic  |    | 47,394   |    | 44,370   |          | 41,494   |
|  |    |          |    |          |          |          |
| Diluted Other comments in come (local)                             |    | 49,381   |    | 47,211   |          | 41,494   |
| Other comprehensive income (loss):                                 | 6  | 51.5(0   | e. | 20.276   | 6        | (64.500) |
| Net income (loss)  | \$ | 51,569   | \$ | 29,376   | \$       | (64,500) |
| Unrealized gain on available-for-sale investments, net of tax      |    | (442)    |    | 276      |          | 9        |
| Foreign currency translation adjustment                            |    | (442)    |    | 376      |          | 137      |
| Other comprehensive income (loss)                                  |    | (442)    |    | 377      |          | 146      |
| Comprehensive income (loss)  |    | 51,127   |    | 29,753   |          | (64,354) |
| Comprehensive income attributable to noncontrolling interests      |    | (935)    |    | (2,337)  |          | (658)    |
| Comprehensive income (loss) attributable to Blucora, Inc.          | \$ | 50,192   | \$ | 27,416   | \$       | (65,012) |

<sup>\*</sup> The 2018 net income (loss) per share amounts include the noncontrolling interest redemption impact discussed further in "Note 2: Summary of Significant Accounting Policies" and in "Note 17: Net Income (Loss) Per Share."

See notes to consolidated financial statements.

# BLUCORA, INC. CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY (In thousands)

|  | Redeemable<br>Noncontrolling | Com          | mon stock | Additional-<br>paid-in | Accumulated    | Accumulated other comprehensive |            |
|--|------------------------------|--------------|-----------|------------------------|----------------|---------------------------------|------------|
|  | Interests                    | Shares       | Amount    | capital                | deficit        | income (loss)                   | Total      |
| Balance as of December 31, 2015  | \$ 15,038                    | 40,954       | \$ 4      | \$ 1,490,405           | \$ (1,027,598) | \$ (527)                        | \$ 462,284 |
| Common stock issued for stock options and restricted stock units             | _                            | 700          | _         | 2,216                  | _              | _                               | 2,216      |
| Common stock issued for employee stock purchase plan                         | _                            | 191          | _         | 1,402                  | _              | _                               | 1,402      |
| Other comprehensive income   | _                            | _            | _         | _                      | _              | 146                             | 146        |
| Stock-based compensation   | _                            | _            | _         | 15,235                 | _              | _                               | 15,235     |
| Tax effect of equity compensation  | _                            | _            | _         | 2,461                  | _              | _                               | 2,461      |
| Tax payments from shares withheld for equity awards                          | _                            | _            | _         | (1,752)                | _              | _                               | (1,752)    |
| Reclassification of equity award to liability award                          | _                            | _            | _         | 185                    | _              | _                               | 185        |
| Net income (loss)  | 658                          | _            | _         | _                      | (65,158)       | _                               | (65,158)   |
| Balance as of December 31, 2016  | 15,696                       | 41,845       | 4         | 1,510,152              | (1,092,756)    | (381)                           | 417,019    |
| Common stock issued for stock options and restricted stock units             | _                            | 4,382        | 1         | 40,271                 | _              | _                               | 40,272     |
| Common stock issued for employee stock purchase plan                         | _                            | 139          | _         | 1,429                  | _              | _                               | 1,429      |
| Other comprehensive income   | _                            | _            | _         | _                      | _              | 377                             | 377        |
| Stock-based compensation and impact of recent ASU                            | _                            | _            | _         | 12,801                 | 51,543         | _                               | 64,344     |
| Tax payments from shares withheld for equity awards                          | _                            | _            | _         | (9,095)                | _              | _                               | (9,095)    |
| Other  | _                            | _            | _         | 2                      | _              | _                               | 2          |
| Net income   | 2,337                        | _            | _         | _                      | 27,039         | _                               | 27,039     |
| Balance as of December 31, 2017  | 18,033                       | 46,366       | 5         | 1,555,560              | (1,014,174)    | (4)                             | 541,387    |
| Common stock issued for stock options and restricted stock units             | _                            | 1,577        | _         | 13,151                 | _              | _                               | 13,151     |
| Common stock issued for employee stock purchase plan                         | _                            | 101          | _         | 2,100                  | _              | _                               | 2,100      |
| Other comprehensive loss   | _                            | _            | _         | _                      | _              | (442)                           | (442)      |
| Stock-based compensation   | _                            | _            | _         | 13,253                 | _              | _                               | 13,253     |
| Tax payments from shares withheld for equity awards                          | _                            | _            | _         | (8,362)                | _              | _                               | (8,362)    |
| Impact of adoption of new accounting guidance related to revenue recognition | -                            | _            | _         | _                      | 1,851          | _                               | 1,851      |
| Net income   | 935                          | _            | _         | _                      | 50,634         | _                               | 50,634     |
| Adjustment of redeemable noncontrolling interests to redemption value        | 5,977                        | _            | _         | (5,977)                | _              | _                               | (5,977)    |
| Balance as of December 31, 2018  | \$ 24,945                    | 48,044       | \$ 5      | \$ 1,569,725           | \$ (961,689)   | \$ (446)                        | \$ 607,595 |
|  |                              | Saa mataa ta | 1:        | Emanaial statement     |                |                                 |            |

See notes to consolidated financial statements.

# BLUCORA, INC. CONSOLIDATED STATEMENTS OF CASH FLOWS (In thousands)

| centing Activities         2018         2017         2018           resting Activities         S 5,1569         \$ 29,376         \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$   | (64,500<br>(63,12)<br>(1,379<br>14,128<br>38,688<br>(362<br>(18,055<br>2,012<br>4,690<br>1,036<br>391 |
|--|---|
| Net immore (loss)         \$ 51,509         \$ 29,376         \$           Less Dissontinued operations, net of income taxes         ————————————————————————————————————  | (63,121)<br>(1,379)<br>14,128<br>38,688<br>(364)<br>(18,055)<br>2,014<br>4,690<br>1,036<br>391        |
| Less: Discontinued operations, net of income taxes         —         —           Net income (loss) from continuing operations         51,569         29,376           Adjustments to reconcile net income (loss) to net cash from operating activities:         Total Control of Con   | (63,121)<br>(1,379)<br>14,128<br>38,688<br>(364)<br>(18,055)<br>2,014<br>4,690<br>1,036<br>391        |
| Net income (loss) from continuing operations         51,569         29,376           Adjustments to reconcile net income (loss) to net cash from operating activities:         13,253         11,653           Stock-based compensation         13,253         11,653           Depreciation and amorization of acquired intangible assets         38,590         38,139           Restructuring (non-cash)         —         1,569           Deferred income taxes         (3,039)         (16,159)           Amorization of premium on investments, net, and debt issuance costs         833         1,099           Accretion of debt discounts         163         1,947           Loss on debt extinguishment and modification expense         1,534         20,445           Revaluation of acquisition-related contingent consideration liability         —         —           Other         7         30         0           Cash provided (used) by changes in operating assets and liabilities:         Use of 4,286         (483)           Accounts receivable         (4,286)         (483)         0           Other receivable         (815)         (869)         0         668)           Other receivable         (815)         (869)         0         10         12.281         1         2.0         1         2.0 </td <td>(1,379<br/>14,128<br/>38,688<br/>(364<br/>(18,055<br/>2,014<br/>4,690<br/>1,036<br/>391</td>   | (1,379<br>14,128<br>38,688<br>(364<br>(18,055<br>2,014<br>4,690<br>1,036<br>391                       |
| Adjustments to reconcile net income (loss) to net cash from operating activities:         13,253         11,653           Stock-based compensation         13,253         11,653           Depreciation and amortization of acquired intangible assets         38,590         38,139           Restructuring (non-cash)         —         1,569           Deferred income taxes         (3,039)         (16,159)           Amortization of premium on investments, net, and debt issuance costs         833         1,099           Accretion of debt discounts         163         1,947           Loss on debt extinguishment and modification expense         1,534         20,445           Revaluation of acquisition-related contingent consideration liability         —         —           Other         72         30           Cash provided (used) by changes in operating assets and liabilities:         S         4,286         (483)           Accounts receivable         (4,286)         (483)         (204)           Other receivables         (3,851)         (204)           Prepaid expenses and other current assets         (815)         (869)           Other long-term assets         (3,851)         (2,281)           Accounts payable         (2,614)         1,226           Oefferred revenue         9,930   | 14,128<br>38,688<br>(36 <sup>2</sup><br>(18,055<br>2,01 <sup>2</sup><br>4,690<br>1,036<br>391         |
| Stock-based compensation         13,253         11,653           Depreciation and amortization of acquired intangible assets         38,590         38,139           Restructuring (non-cash)         —         1,569           Deferred income taxes         (3,09)         (16,159)           Amortization of premium on investments, net, and debt issuance costs         833         1,099           Accretion of debt discounts         163         1,947           Loss on debt extinguishment and modification expense         15,34         20,445           Revaluation of acquisition-related contingent consideration liability         —         —           Other         72         30         Other           Cash provided (used) by changes in operating assets and liabilities:         Use of 4,286         (483)           Commissions receivable         4,286         (483)         (483)           Other receivables         3,851         (204)         (204)           Prepaid expenses and other current assets         (815)         (869)         (165)         (123)           Other long-term assets         3,450         (12,281)         (2,281)         (2,281)         (2,281)         (2,281)         (2,281)         (2,281)         (2,281)         (2,281)         (2,281)         (2,281) <th< td=""><td>38,688<br/>(36<sup>2</sup><br/>(18,055<br/>2,01<sup>4</sup><br/>4,690<br/>1,036<br/>391</td></th<>  | 38,688<br>(36 <sup>2</sup><br>(18,055<br>2,01 <sup>4</sup><br>4,690<br>1,036<br>391                   |
| Depreciation and amortization of acquired intangible assets         38,590         38,139           Restructuring (non-cash)         —         1,569           Deferred income taxes         (3,039)         (16,159)           Amortization of premium on investments, net, and debt issuance costs         833         1,099           Accretion of debt discounts         163         1,947           Loss on debt extinguishment and modification expense         1,534         20,445           Revaluation of acquisition-related contingent consideration liability         —         —           Other         72         30           Cash provided (used) by changes in operating assets and liabilities:         —         —           Accounts receivable         (4,286)         (483)           Other receivables         (3,81)         (204)           Other receivables         (3,81)         (204)           Prepaid expenses and other current assets         (3,15)         (678)           Other long-term assets         (3,15)         (12,281)           Accounts payable         (615)         (12,281)           Commissions and advisory fees payable         (615)         (12,281)           Accrude expenses and other current and long-term liabilities         1,14         1,40 <td< td=""><td>38,688<br/>(36<sup>2</sup><br/>(18,055<br/>2,01<sup>4</sup><br/>4,690<br/>1,036<br/>391</td></td<>  | 38,688<br>(36 <sup>2</sup><br>(18,055<br>2,01 <sup>4</sup><br>4,690<br>1,036<br>391                   |
| Restructuring (non-cash)         —         1,569           Deferred income taxes         (3,039)         (16,159)           Amortization of premium on investments, net, and debt issuance costs         833         1,099           Accretion of debt discounts         163         1,947           Loss on debt extinguishment and modification expense         1,534         20,445           Revaluation of acquisition-related contingent consideration liability         —         —           Other         72         30           Cash provided (used) by changes in operating assets and liabilities:         Secondary (a.286)         (483)           Commissions receivable         (4,286)         (483)           Commissions receivable         (3,851)         (204)           Prepaid expenses and other current assets         (3,851)         (204)           Prepaid expenses and other current assets         3,450         (12,281)           Accounts payable         (615)         (123)           Commissions and advisory fees payable         (2,614)         1,266           Deferred revenue         9,930         (3,248)           Accrued expenses and other current and long-term liabilities         114         1,407           Net cash provided by operating activities         10,548         72,846  | (364<br>(18,055<br>2,014<br>4,690<br>1,036<br>391   |
| Deferred income taxes         (3,039)         (16,159)           Amortization of premium on investments, net, and debt issuance costs         833         1,099           Accretion of debt discounts         163         1,947           Loss on debt extinguishment and modification expense         1,534         20,445           Revaluation of acquisition-related contingent consideration liability         ————————————————————————————————————   | (18,055<br>2,014<br>4,690<br>1,036<br>391   |
| Amortization of premium on investments, net, and debt issuance costs         833         1,099           Accretion of debt discounts         163         1,947           Loss on debt extinguishment and modification expense         1,534         20,445           Revaluation of acquisition-related contingent consideration liability         —         —           Other         72         30           Cash provided (used) by changes in operating assets and liabilities:         —         —           Accounts receivable         (4,286)         (483)           Commissions receivable         1,260         (678)           Other receivables         (3,851)         (204)           Prepaid expenses and other current assets         (815)         (869)           Other long-term assets         (815)         (869)           Other long-term assets         (815)         (12,281)           Accounts payable         (615)         (123)           Commissions and advisory fees payable         (2,614)         1,226           Deferred revenue         9,930         (3,248)           Accrued expenses and other current and long-term liabilities         114         1,407           Net cash provided by operating activities         105,548         72,846           vesting <t< td=""><td>2,014<br/>4,690<br/>1,036<br/>391</td></t<>  | 2,014<br>4,690<br>1,036<br>391  |
| Accretion of debt discounts         163         1,947           Loss on debt extinguishment and modification expense         1,534         20,445           Revaluation of acquisition-related contingent consideration liability         —         —           Other         72         30           Cash provided (used) by changes in operating assets and liabilities:         —         —           Accounts receivable         (4,286)         (483)           Commissions receivable         1,260         (678)           Other receivables         (3,851)         (204)           Prepaid expenses and other current assets         (815)         (869)           Other long-term assets         (815)         (869)           Other long-term assets         (615)         (12,31)           Accounts payable         (615)         (123)           Commissions and advisory fees payable         (2,614)         1,226           Deferred revenue         9,930         (3,248)           Accrued expenses and other current and long-term liabilities         114         1,407           Net cash provided by operating activities         105,548         72,846           vesting Activities         8         72,846           Business acquisitions, net of cash acquired         —  | 4,690<br>1,036<br>391   |
| Loss on debt extinguishment and modification expense         1,534         20,445           Revaluation of acquisition-related contingent consideration liability         —         —           Other         72         30           Cash provided (used) by changes in operating assets and liabilities:         —         —           Accounts receivable         (4,286)         (483)           Commissions receivable         1,260         (678)           Other receivables         (3,851)         (204)           Prepaid expenses and other current assets         (815)         (869)           Other long-term assets         3,450         (12,281)           Accounts payable         (615)         (123)           Commissions and advisory fees payable         (616)         (123)           Deferred revenue         9,930         (3,248)           Accrued expenses and other current and long-term liabilities         114         1,407           Net cash provided by operating activities         105,548         72,846           vesting Activities:         —         —           Business acquisitions, net of cash acquired         —         —           Proceeds from sales of investments         —         —           Proceeds from maturities of investments         —   | 1,036<br>391  |
| Revaluation of acquisition-related contingent consideration liability         — <t< td=""><td>391</td></t<>  | 391   |
| Other         72         30           Cash provided (used) by changes in operating assets and liabilities:         Cash provided (used) by changes in operating assets and liabilities:           Accounts receivable         (4,286)         (483)           Other receivables         (3,851)         (204)           Prepaid expenses and other current assets         (815)         (869)           Other long-term assets         (815)         (12,281)           Accounts payable         (615)         (123)           Commissions and advisory fees payable         (2,614)         1,226           Deferred revenue         9,930         (3,248)           Accrued expenses and other current and long-term liabilities         114         1,407           Net cash provided by operating activities         105,548         72,846           vesting Activities:         8         7         8           Business acquisitions, net of cash acquired         —         —         —           Purchases of property and equipment         (7,633)         (5,039)         Proceeds from sales of investments         —         249           Proceeds from maturities of investments         —         4(409)         Proceeds from sales of investments         —         (409)           Net cash provided (used) by investing activit   | 19  |
| Cash provided (used) by changes in operating assets and liabilities:         (4,286)         (483)           Accounts receivable         1,260         (678)           Other receivables         (3,851)         (204)           Prepaid expenses and other current assets         (815)         (869)           Other long-term assets         (815)         (12,281)           Accounts payable         (615)         (123)           Commissions and advisory fees payable         (2,614)         1,226           Deferred revenue         9,930         (3,248)           Accrued expenses and other current and long-term liabilities         114         1,407           Net cash provided by operating activities         105,548         72,846           vesting Activities:         Testing Activities:         Testing Activities:           Business acquisitions, net of cash acquired         —         —           Proceeds from sales of investments         —         249           Proceeds from maturities of investments         —         7,252           Purchases of investments         —         (409)           Net cash provided (used) by investing activities         (7,633)         2,053  |   |
| Accounts receivable         (4,286)         (483)           Commissions receivable         1,260         (678)           Other receivables         (3,851)         (204)           Prepaid expenses and other current assets         (815)         (869)           Other long-term assets         (31,450)         (12,281)           Accounts payable         (615)         (123)           Commissions and advisory fees payable         (2,614)         1,226           Deferred revenue         9,930         (3,248)           Accrued expenses and other current and long-term liabilities         114         1,407           Net cash provided by operating activities         105,548         72,846           vesting Activities:         Testing Activities         Testing Activities           Business acquisitions, net of cash acquired         —         —           Purchases of property and equipment         (7,633)         (5,039)           Proceeds from sales of investments         —         249           Proceeds from maturities of investments         —         7,252           Purchases of investments         —         (409)           Net cash provided (used) by investing activities         (7,633)         2,053   |   |
| Commissions receivable         1,260         (678)           Other receivables         (3,851)         (204)           Prepaid expenses and other current assets         (815)         (869)           Other long-term assets         3,450         (12,281)           Accounts payable         (615)         (123)           Commissions and advisory fees payable         (2,614)         1,226           Deferred revenue         9,930         (3,248)           Accrued expenses and other current and long-term liabilities         114         1,407           Net cash provided by operating activities         105,548         72,846           vesting Activities:         Business acquisitions, net of cash acquired         —         —           Purchases of property and equipment         (7,633)         (5,039)           Proceeds from sales of investments         —         249           Proceeds from maturities of investments         —         7,252           Purchases of investments         —         4009           Net cash provided (used) by investing activities         (7,633)         2,053   |   |
| Other receivables         (3,851)         (204)           Prepaid expenses and other current assets         (815)         (869)           Other long-term assets         3,450         (12,281)           Accounts payable         (615)         (123)           Commissions and advisory fees payable         (2,614)         1,226           Deferred revenue         9,930         (3,248)           Accrued expenses and other current and long-term liabilities         114         1,407           Net cash provided by operating activities         105,548         72,846           vesting Activities:         Testing Activities:         Testing Activities:           Business acquisitions, net of cash acquired         —         —           Purchases of property and equipment         (7,633)         (5,039)           Proceeds from sales of investments         —         249           Proceeds from maturities of investments         —         7,252           Purchases of investments         —         (409)           Net cash provided (used) by investing activities         (7,633)         2,053   | (2,340  |
| Prepaid expenses and other current assets         (815)         (869)           Other long-term assets         3,450         (12,281)           Accounts payable         (615)         (123)           Commissions and advisory fees payable         (2,614)         1,226           Deferred revenue         9,930         (3,248)           Accrued expenses and other current and long-term liabilities         114         1,407           Net cash provided by operating activities         105,548         72,846           vesting Activities:         -         -           Business acquisitions, net of cash acquired         -         -           Proceeds from sales of investments         -         249           Proceeds from maturities of investments         -         249           Proceeds from maturities of investments         -         7,252           Purchases of investments         -         409)           Net cash provided (used) by investing activities         (7,633)         2,053  | 184   |
| Other long-term assets         3,450         (12,281)           Accounts payable         (615)         (123)           Commissions and advisory fees payable         (2,614)         1,226           Deferred revenue         9,930         (3,248)           Accrued expenses and other current and long-term liabilities         114         1,407           Net cash provided by operating activities         105,548         72,846           vesting Activities:         -         -           Business acquisitions, net of cash acquired         -         -           Purchases of property and equipment         (7,633)         (5,039)           Proceeds from sales of investments         -         249           Proceeds from maturities of investments         -         7,252           Purchases of investments         -         (409)           Net cash provided (used) by investing activities         (7,633)         2,053   | 22,875  |
| Accounts payable       (615)       (123)         Commissions and advisory fees payable       (2,614)       1,226         Deferred revenue       9,930       (3,248)         Accrued expenses and other current and long-term liabilities       114       1,407         Net cash provided by operating activities       105,548       72,846         vesting Activities:       -       -         Business acquisitions, net of cash acquired       -       -         Purchases of property and equipment       (7,633)       (5,039)         Proceeds from sales of investments       -       249         Proceeds from maturities of investments       -       7,252         Purchases of investments       -       (409)         Net cash provided (used) by investing activities       (7,633)       2,053   | 3,741   |
| Commissions and advisory fees payable (2,614) 1,226  Deferred revenue 9,930 (3,248)  Accrued expenses and other current and long-term liabilities 114 1,407  Net cash provided by operating activities 105,548 72,846  vesting Activities:  Business acquisitions, net of cash acquired  Purchases of property and equipment (7,633) (5,039)  Proceeds from sales of investments - 249  Proceeds from maturities of investments - 7,252  Purchases of investments - (409)  Net cash provided (used) by investing activities (7,633) 2,053  | (887  |
| Deferred revenue         9,930         (3,248)           Accrued expenses and other current and long-term liabilities         114         1,407           Net cash provided by operating activities         105,548         72,846           vesting Activities:         -         -           Business acquisitions, net of cash acquired         -         -           Purchases of property and equipment         (7,633)         (5,039)           Proceeds from sales of investments         -         249           Proceeds from maturities of investments         -         7,252           Purchases of investments         -         (409)           Net cash provided (used) by investing activities         (7,633)         2,053  | (153  |
| Accrued expenses and other current and long-term liabilities 114 1,407  Net cash provided by operating activities 72,846  vesting Activities:  Business acquisitions, net of cash acquired  Purchases of property and equipment (7,633) (5,039)  Proceeds from sales of investments - 249  Proceeds from maturities of investments - 7,252  Purchases of investments - (409)  Net cash provided (used) by investing activities (7,633) 2,053   | (395  |
| Net cash provided by operating activities  Pusiness acquisitions, net of cash acquired Purchases of property and equipment Proceeds from sales of investments Proceeds from maturities of investments Purchases of investments Purchases of investments Proceeds from maturities of investments Purchases of investments Purchase | 582   |
| Purchases of investments  Proceeds from maturities of investments  Purchases of investments  Pur | 21,195  |
| Business acquisitions, net of cash acquired——Purchases of property and equipment(7,633)(5,039)Proceeds from sales of investments—249Proceeds from maturities of investments—7,252Purchases of investments—(409)Net cash provided (used) by investing activities(7,633)2,053  | 85,970  |
| Purchases of property and equipment(7,633)(5,039)Proceeds from sales of investments—249Proceeds from maturities of investments—7,252Purchases of investments—(409)Net cash provided (used) by investing activities(7,633)2,053   |   |
| Proceeds from sales of investments—249Proceeds from maturities of investments—7,252Purchases of investments—(409)Net cash provided (used) by investing activities(7,633)2,053  | (1,788  |
| Proceeds from maturities of investments—7,252Purchases of investments—(409)Net cash provided (used) by investing activities(7,633)2,053  | (3,812  |
| Purchases of investments — (409) Net cash provided (used) by investing activities (7,633) 2,053  | _   |
| Net cash provided (used) by investing activities (7,633) 2,053   | 12,807  |
|  | (8,767  |
| nancing Activities:  | (1,560  |
| -  |   |
| Proceeds from credit facility, net of debt issuance costs and debt discount of \$5,913 and \$1,875 in  |   |
| 2017 — 365,836   |   |
| Repurchase of convertible notes — (172,827)  | (20,667   |
|  | (140,000  |
| Repayment of note payable with related party — (3,200)   | (3,200  |
| Proceeds from stock option exercises 12,773 40,271   | 2,216   |
| Proceeds from issuance of stock through employee stock purchase plan 2,100 1,429   | 1,402   |
| Tax payments from shares withheld for equity awards (8,362) (9,095)  | (1,752  |
| Contingent consideration payments for business acquisition (1,315) (946)   | _   |
| Other(30)  |   |
| Net cash used by financing activities (74,804) (68,562)  | (162,001  |
| et cash provided (used) by continuing operations 23,111 6,337  | (77,591   |
| et cash provided by operating activities from discontinued operations — — —  | 14,047  |
| et cash provided by investing activities from discontinued operations  — 1,028   | 83,608  |
| et cash used by financing activities from discontinued operations  — — —   | (25,000   |
| et cash used by financing activities from discontinued operations — 1,028  | 72,655  |
| 1,028  | 12,033  |
| fect of exchange rate changes on cash and cash equivalents (56)  |   |
| et increase (decrease) in cash, cash equivalents, and restricted cash 23,055 7,443   | (26   |
| sh and cash equivalents, beginning of period 62,311 54,868   | (4,962  |

| Cash and cash equivalents, end of period                                | \$               | 85,366 | \$<br>62,311 | \$<br>54,868 |
|---|------------------|--------|--------------|--------------|
| Non-cash investing and financing activities from continuing operations: |                  |        |              |              |
| Cash paid for income taxes from continuing operations                   | \$               | 1,806  | \$<br>1,267  | \$<br>2,012  |
| Cash paid for interest from continuing operations                       | \$               | 15,335 | \$<br>23,316 | \$<br>32,377 |
| See notes to consolidated final   | ncial statements | S.     |              |              |

# BLUCORA, INC. NOTES TO CONSOLIDATED FINANCIAL STATEMENTS Years Ended December 31, 2018, 2017, and 2016

## Note 1: Description of the Business

Blucora, Inc. (the "Company," "Blucora," "we," "our," or "us") operates two businesses: a Wealth Management business and a digital Tax Preparation business. The Wealth Management business consists of the operations of HDV Holdings, Inc. and its subsidiaries ("HD Vest"). HDV Holdings, Inc. is the parent company of the Wealth Management business and owns all outstanding shares of HD Vest, Inc., which serves as a holding company for the various financial services subsidiaries. Those subsidiaries include HD Vest Investment Securities, Inc. (an introducing broker-dealer), H.D. Vest Advisory Services, Inc. (a registered investment adviser), and H.D. Vest Insurance Agency, LLC (an insurance broker) (collectively referred to as the "Wealth Management business" or the "Wealth Management segment"). The Tax Preparation business consists of the operations of TaxAct, Inc. and its subsidiary ("TaxAct") and provides digital tax preparation solutions for consumers, small business owners, and tax professionals through its website www.TaxAct.com (collectively referred to as the "Tax Preparation business" or the "Tax Preparation segment").

Prior to 2017, the Company also operated an internet Search and Content business and an E-Commerce business. The Search and Content business operated through the InfoSpace LLC subsidiary ("InfoSpace") and provided search services to users of its owned and operated and distribution partners' web properties, as well as online content through HowStuffWorks ("HSW"). The E-Commerce business consisted of the operations of Monoprice, Inc. ("Monoprice") and sold self-branded electronics and accessories to both consumers and businesses primarily through its website. The Company completed both divestitures in 2016.

The financial condition, results of operations, cash flows, and the notes to financial statements reflect the Search and Content and E-Commerce businesses as discontinued operations for all periods presented. Except for disclosures related to equity and unless otherwise specified, disclosures in these consolidated financial statements reflect continuing operations.

In 2015 the Company acquired HD Vest and announced its plans to focus on the technology-enabled financial solutions market (the "Strategic Transformation"). The Strategic Transformation refers to our transformation into a technology-enabled financial solutions company comprised of TaxAct and HD Vest and the divestitures of our Search and Content business and our E-Commerce business in 2016. As part of the Strategic Transformation and "One Company" operating model, we relocated our corporate headquarters from Bellevue, Washington to Irving, Texas during 2017.

In connection with the relocation of our corporate headquarters, we incurred restructuring costs of approximately \$7.3 million. These costs are recorded within corporate-level activity for segment purposes. See "Note 6: Restructuring" of the Notes to Consolidated Financial Statements in Part II Item 8 of this report for additional information. We also have incurred costs that do not qualify for restructuring classification, such as recruiting and overlap in personnel expenses as we transitioned positions to Texas ("Strategic Transformation Costs").

Segments: The Company has two reportable segments: the Wealth Management segment, which is the HD Vest business, and the Tax Preparation segment, which is the TaxAct business. Unless the context indicates otherwise, the Company uses the term "Wealth Management" to represent services sold through the HD Vest business and the term "Tax Preparation" to represent services and software sold through the TaxAct business.

Reclassification: The Company reclassified certain amounts on its consolidated statements of cash flows related to excess tax benefits generated from stock-based compensation and restricted cash, both in connection with the implementation of new accounting pronouncements. See the "Recent accounting pronouncements" section of "Note 2: Summary of Significant Accounting Policies" for additional information.

Principles of consolidation: The consolidated financial statements include the accounts of the Company and its subsidiaries. Intercompany accounts and transactions have been eliminated.

Use of estimates: The preparation of financial statements in conformity with accounting principles generally accepted in the United States of America ("GAAP") requires management to make estimates and assumptions that affect the reported amounts of assets, liabilities, revenues, expenses, and disclosure of contingencies. Estimates include those used for impairment of goodwill and other intangible assets, useful lives of other intangible assets, acquisition accounting, valuation of investments, revenue recognition, the estimated allowance for sales returns and doubtful accounts, internally developed software, accrued

## Years Ended December 31, 2018, 2017, and 2016

contingencies, stock option valuation, and valuation allowance for deferred tax assets. Actual amounts may differ from estimates.

Net capital and regulatory requirements: The Company's HD Vest broker-dealer subsidiary operates in a highly regulated industry and is subject to various regulatory capital requirements. Failure to meet minimum capital requirements can initiate certain mandatory and possible additional discretionary actions by regulators that, if undertaken, could have substantial monetary and non-monetary impacts to HD Vest's operations. As of December 31, 2018, HD Vest met all capital adequacy requirements to which it was subject.

Seasonality: Blucora's Tax Preparation segment is highly seasonal, with a significant portion of its annual revenue earned in the first four months of the Company's fiscal year. During the third and fourth quarters, the Tax Preparation segment typically reports losses because revenue from the segment is minimal while core operating expenses continue.

## Note 2: Summary of Significant Accounting Policies

Cash, cash equivalents, restricted cash and cash segregated under federal or other regulations: The following table presents cash, cash equivalents, and restricted cash as reported on the consolidated balance sheets that equal the total amounts on the consolidated statements of cash flows (in thousands):

|  | December 31, |        |    |        |  |
|--|--------------|--------|----|--------|--|
|  |              | 2018   |    | 2017   |  |
| Cash and cash equivalents  | \$           | 84,524 | \$ | 59,965 |  |
| Cash segregated under federal or other regulations                           |              | 842    |    | 1,371  |  |
| Restricted cash included in "Prepaid expenses and other current assets, net" |              | _      |    | 425    |  |
| Restricted cash included in "Other long-term assets"                         |              | _      |    | 550    |  |
| Total cash, cash equivalents, and restricted cash                            | \$           | 85,366 | \$ | 62,311 |  |

Cash segregated under federal and other regulations is held in a separate bank account for the exclusive benefit of the Company's Wealth Management customers. Restricted cash included in prepaid expenses and other current assets, net and other long-term assets represent amounts pledged as collateral for certain of the Company's banking and lease arrangements.

The Company generally invests its available cash in high quality marketable investments. Recently these investments have included money market funds invested in securities issued by agencies of the U.S. government. The Company may invest, from time-to-time, in other vehicles, such as debt instruments issued by the U.S. federal government and its agencies, international governments, municipalities and publicly-held corporations, as well as commercial paper, and insured time deposits with commercial banks. Specific holdings can vary from period to period depending upon the Company's cash requirements. Such investments are reported at fair value on the consolidated balance sheets.

Accounts receivable: Accounts receivable are stated at amounts due from customers net of an allowance for doubtful accounts, which was not material at December 31, 2018 and 2017, respectively.

Property and equipment: Property and equipment are stated at cost. Depreciation is computed under the straight-line method over the following estimated useful lives:

| Computer equipment and software | 3 years                                |
|---------------------------------|--|
| Data center servers             | 3 years                                |
| Internally-developed software   | 3 years                                |
| Office equipment                | 7 years                                |
| Office furniture                | 7 years                                |
| Leasehold improvements          | Shorter of lease term or economic life |

## Years Ended December 31, 2018, 2017, and 2016

The Company capitalizes certain internal-use software development costs, consisting primarily of contractor costs and employee salaries and benefits allocated on a project or product basis. The Company capitalized \$6.5 million, \$3.5 million, and \$1.0 million of internal-use software costs in the years ended December 31, 2018, 2017, and 2016, respectively.

Business combinations and intangible assets including goodwill: The Company accounts for business combinations using the acquisition method. The acquisition-date fair value of total consideration includes cash and contingent consideration. Since the Company is contractually obligated to pay contingent consideration upon the achievement of specified objectives, a contingent consideration liability is recorded at the acquisition date. The Company reviews its assumptions related to the fair value of the contingent consideration liability each reporting period and, if there are material changes, revalues the contingent consideration liability based on the revised assumptions, until such contingency is satisfied through payment upon the achievement of the specified objectives. The change in the fair value of the contingent consideration liability is recognized in "General and administrative" expense on the consolidated statements of comprehensive income for the period in which the fair value changes.

Goodwill is calculated as the excess of the acquisition-date fair value of total consideration over the acquisition-date fair value of net assets, including the amount assigned to identifiable intangible assets, and is assigned to reporting units that are expected to benefit from the synergies of the business combination as of the acquisition date. Reporting units are consistent with reportable segments. Identifiable intangible assets with finite lives are amortized over their useful lives on a straight-line basis, except for advisor relationships which are amortized proportional to expected revenue. Acquisition-related costs, including advisory, legal, accounting, valuation, and other similar costs, are expensed in the periods in which the costs are incurred. The results of operations of acquired businesses are included in the consolidated financial statements from the acquisition date.

Goodwill and intangible assets impairment: The Company evaluates goodwill and indefinite-lived intangible assets for impairment annually, as of November 30, or more frequently when events or circumstances indicate that impairment may have occurred. The Company performed an assessment as of November 30, 2018, and determined that no conditions existed that would make it more likely than not that goodwill and the indefinite-lived assets were impaired.

Definite-lived intangible assets are reviewed for impairment when events or circumstances indicate that the carrying value of an asset or group of assets may not be recoverable. The determination of recoverability is based on an estimate of pre-tax undiscounted future cash flows, using the Company's best estimates of future revenues and operating expenses, expected to result from the use and eventual disposition of the asset or group of assets over the remaining economic life of the primary asset in the asset group. The Company measures the amount of the impairment as the excess of the asset's carrying value over its fair value.

Fair value of financial instruments: The Company measures its cash equivalents and contingent consideration liability at fair value. The Company considers the carrying values of accounts receivable, commissions receivable, other receivables, prepaid expenses, other current assets, accounts payable, commissions and advisory fees payable, accrued expenses, and other current liabilities to approximate fair values primarily due to their short-term natures.

The Company has a contingent consideration liability that is related to the Company's 2015 acquisition of SimpleTax Software Inc. ("SimpleTax"). The Company's contingent consideration liability is classified within Level 3 (see "Note 8: Fair Value Measurements") of the fair value hierarchy because the Company values it utilizing significant inputs not observable in the market. Specifically, the Company has determined the fair value of the contingent consideration liability based on a probability-weighted discounted cash flow analysis, which includes assumptions related to estimating revenues, the probability of payment, and the discount rate. The change in the fair value of the contingent consideration liability is recognized in "General and administrative" expense on the consolidated statements of comprehensive income for the period in which the fair value changes. The Company accounts for contingent consideration in accordance with applicable accounting guidance pertaining to business combinations.

Redeemable noncontrolling interests: Noncontrolling interests that are redeemable at the option of the holder and not solely within the control of the issuer are classified outside of stockholders' equity. In connection with the 2015 acquisition of HD Vest, certain members of the former management of HD Vest retained an ownership interest in that business. The Company is party to put and call arrangements, exercisable beginning in the first quarter of 2019, with respect to these interests. These put and call arrangements allow certain members of HD Vest management to require the Company to purchase their interests or

## Years Ended December 31, 2018, 2017, and 2016

allow the Company to acquire such interests, respectively. These arrangements can be settled for cash within ninety days after the Company files its Annual Report on Form 10-K for the year ended December 31, 2018. The redemption value of the arrangements is based upon several factors, including, among others, the Company's implied enterprise value, implied equity value and certain financial performance measures of the Company. The put arrangements do not meet the definition of a derivative instrument as the put agreements do not provide for net settlement.

To the extent that the redemption value of these interests exceeds the value determined by adjusting the carrying value for the subsidiary's attribution of net income (loss), the value of such interests is adjusted to the redemption value with a corresponding adjustment to additional paid-in capital; this occurred in the third quarter of 2018, and the Company recorded an adjustment of approximately \$6.0 million for the year ended December 31, 2018. The redemption amount of noncontrolling interests at December 31, 2018 was \$24.9 million.

Revenue recognition, general: The Company recognizes revenue when all five revenue recognition criteria have been satisfied: contract(s) with customers have been identified, performance obligations have been identified, transaction prices have been determined, transaction prices have been allocated to the performance obligations, and the performance obligations have been fulfilled by transferring control over the promised services to the customer. Determining whether and when these criteria have been satisfied involves judgment, estimates and assumptions that can have an impact on the timing and amount of revenue that the Company recognizes.

Revenue is recognized net of allowances, which are management's estimates of fees to be paid to a third party service provider for fulfillment of the Company's audit defense services. These fees are not material and generally include an estimate of audit defense fees to be paid, based on an analysis of historical data and contractual terms, and are recorded when revenue is recognized. The Company believes that it can reasonably and reliably estimate fees to the third party service provider in a timely manner.

The Company evaluates whether revenue should be presented on a gross basis, which is the amount that a customer pays for the service or product, or on a net basis, which is the customer payment less amounts the Company pays to suppliers. In making that evaluation, the Company primarily considers whether it acts as the principal or agent in the transaction and whether it controls the services before they are transferred to customers.

Wealth management revenue recognition: Wealth management revenue consists primarily of commission revenue, advisory revenue, asset-based revenue, and transaction and fee revenue. Revenue is recognized upon the transfer of services to customers in an amount that reflects the consideration to which the Company expects to be entitled in exchange for those services. Payments received by the Company in advance of the performance of service are deferred and recognized as revenue when earned. Within the components of wealth management revenue, commission revenue is determined through the use of subjective judgments and the use of estimates.

Commissions represent amounts generated by HD Vest's financial advisors for their clients' purchases and sales of securities and various investment products. The Company generates two types of commissions: transaction-based sales commissions that occur at the point of sale, as well as trailing commissions for which the Company provides ongoing account support to clients of its financial advisors.

The Company records transaction-based sales commission revenue on a trade-date basis, which is when the Company's performance obligations in generating the commissions have been substantially completed. Trailing commission revenue is based on a percentage of the current market value of clients' investment holdings in trail-eligible assets and recognized over the period during which services are performed. Since trailing commission revenue is generally paid in arrears, the Company estimates it based on a number of factors, including stock market index levels and the amount of trailing commission revenues received in prior periods, and also considers historical payout ratios. These estimates are primarily based on historical information and there is not significant judgment involved.

A substantial portion of commission revenue is ultimately paid to financial advisors. The Company records an estimate for transaction-based commissions payable based upon the payout rate of the financial advisor generating the accrued commission revenue. The Company records an estimate for trailing commissions payable based upon historical payout ratios. Such amounts are recorded as "Commissions and advisory fees payable" on the consolidated balance sheets and "Wealth management services cost of revenue" on the consolidated statements of comprehensive income.

## Years Ended December 31, 2018, 2017, and 2016

Tax preparation revenue recognition: The Company derives revenue from the sale of tax preparation digital services, ancillary services, packaged tax preparation software, and multiple element arrangements that may include a combination of these items. Ancillary services primarily include tax preparation support services, and e-filing services. The Company recognizes revenue from the sale of its packaged software when legal title transfers. This is generally when its customers download the software from the Internet or, in some cases, when the software ships. This revenue is recorded in the Tax Preparation segment. Within the components of tax preparation revenue, revenues from bank or reloadable prepaid debit card services and software and/or services that consist of multiple elements are determined through the use of subjective judgments and the use of estimates.

The bank or reloadable prepaid debit card services are offered to taxpayers as an option to receive their tax refunds in the form of a prepaid bank card or to have the fees for the software and/or services purchased by the customers deducted from their refunds. Other value-added service revenue consists of revenue from revenue sharing and royalty arrangements with third party partners. Revenue for these transactions is recognized when the revenue recognition criteria are met; for some arrangements that is upon filing and for other arrangements that is upon the Company's determination of when collectibility is probable.

For software and/or services that consist of multiple elements, the Company must: (1) determine whether and when each element has been delivered; (2) determine the fair value of each element; and (3) allocate the total price among the various elements based on the relative selling price method. Once the Company has allocated the total price among the various elements, it recognizes revenue when the revenue recognition criteria described above are met for each element. The impact of multiple element arrangements is not material and primarily impacts the timing of revenue recognition over the tax filing season, which is concentrated within the first two quarters of the filing period.

Advertising expenses: Costs for advertising are recorded as expense and classified within "Sales and marketing" on the consolidated statements of comprehensive income, when the advertisement appears. Advertising expense totaled \$53.3 million, \$51.7 million, and \$44.0 million for the years ended December 31, 2018, 2017, and 2016, respectively. Prepaid advertising costs were \$0.3 million and \$0.3 million at December 31, 2018 and 2017, respectively.

Stock-based compensation: The Company measures stock-based compensation at the grant date based on the fair value of the award and recognizes it as expense, net of estimated forfeitures, over the vesting or service period, as applicable, of the stock award using the straight-line method. The Company recognizes stock-based compensation over the vesting period for each separately vesting portion of a share-based award as if they were individual share-based awards. The Company estimates forfeitures at the time of grant, based upon historical data, and revises those estimates, if necessary, in subsequent periods if actual forfeitures differ from those estimates.

Income taxes: The Company accounts for income taxes under the asset and liability method, under which deferred tax assets, including net operating loss carryforwards, and liabilities are determined based on temporary differences between the book and tax bases of assets and liabilities. The Company periodically evaluates the likelihood of the realization of deferred tax assets and reduces the carrying amount of the deferred tax assets by a valuation allowance to the extent the Company believes a portion will not be realized. The Company considers many factors when assessing the likelihood of future realization of the deferred tax assets, including expectations of future taxable income, recent cumulative earnings experience by taxing jurisdiction, and other relevant factors. There is a wide range of possible judgments relating to the valuation of the Company's deferred tax assets.

The Company records liabilities to address uncertain tax positions that have been taken in previously filed tax returns or that are expected to be taken in a future tax return. The determination for required liabilities is based upon an analysis of each individual tax position, taking into consideration whether it is more likely than not that the tax position, based on technical merits, will be sustained upon examination. The tax benefit to be recognized in the financial statements from such a position is measured as the largest amount of benefit that has a greater than 50% likelihood of being realized upon ultimate settlement with the taxing authority. The difference between the amount recognized and the total tax position is recorded as a liability. The ultimate resolution of these tax positions may be greater or less than the liabilities recorded. The Company recognizes interest and penalties related to uncertain tax positions in interest expense and general and administrative expense, respectively.

Foreign currency: The financial position and operating results of the Company's foreign operations are consolidated using the local currency as the functional currency. Assets and liabilities recorded in local currencies are translated at the exchange rate on the balance sheet date, while revenues and expenses are translated at the average exchange rate for the applicable

## Years Ended December 31, 2018, 2017, and 2016

period. Translation adjustments resulting from this process are recorded in "Accumulated other comprehensive loss" on the consolidated balance sheets. The gain or loss on foreign currency transactions, calculated as the difference between the historical exchange rate and the exchange rate at the applicable measurement date, are recorded in "Other loss, net" on the consolidated statements of comprehensive income.

Concentration of credit risk: Financial instruments that potentially subject the Company to concentrations of credit risk consist primarily of cash equivalents, short-term investments, trade accounts receivable, and commissions receivable. These instruments are generally unsecured and uninsured.

For cash equivalents, short-term investments, and commissions receivable, the Company attempts to manage exposure to counterparty credit risk by only entering into agreements with major financial institutions and investment sponsors that are expected to be able to fully perform under the terms of the agreement.

Accounts receivable are typically unsecured and are derived from revenues earned from customers primarily located in the United States operating in a variety of geographic areas. The Company performs ongoing credit evaluations of its customers and maintains allowances for potential credit losses.

Geographic revenue information: Almost all of the Company's revenue for 2018, 2017, and 2016 was generated from customers located in the United States.

Recent accounting pronouncements: Changes to GAAP are established by the Financial Accounting Standards Board ("FASB") in the form of accounting standards updates ("ASUs") to the FASB's Accounting Standards Codification ("ASC"). The Company considers the applicability and impact of all recent ASUs. ASUs not listed below were assessed and determined to be either not applicable or are expected to have minimal impact on the Company's consolidated financial position and results of operations. The Company currently is considering, or has recently adopted, ASUs that impact the following areas:

Revenue recognition - In May 2014, the FASB issued guidance codified in ASC 606, "Revenue from Contracts with Customers" ("ASC 606"), which amends the guidance in former ASC 605 "Revenue Recognition." The core principle of the guidance is that an entity should recognize revenue to depict the transfer of promised goods or services to customers in an amount that reflects the consideration to which the entity expects to be entitled in exchange for those goods or services by using a five-step process. This guidance became effective on a retrospective basis--either to each reporting period presented or with the cumulative effect of initially applying this guidance recognized at the date of initial application--for annual reporting periods, including interim reporting periods within those annual reporting periods, beginning after December 15, 2017.

The Company adopted the requirements of the new standard on January 1, 2018, utilizing the modified retrospective transition method. Upon adoption, the Company recognized a \$1.8 million cumulative effect as an adjustment to the opening balance of retained earnings and deferred revenues on the consolidated balance sheets.

As a result of the adoption of ASC 606, the Company now recognizes certain licensing fees on a net basis, which reduced both transaction and fee revenues and operating expenses by \$1.8 million for the year ended December 31, 2018, on the consolidated statements of comprehensive income (loss). Had the Company not adopted ASC 606, total revenues for the year ended December 31, 2018 would have been \$3.3 million higher than reported on the consolidated statements of comprehensive income (loss).

Pursuant to the modified retrospective transition method, prior periods were not retrospectively adjusted, and the Company does not disclose the value of unsatisfied performance obligations for contracts with original expected durations of one year or less.

<u>Leases</u> - In February 2016, the FASB issued guidance codified in ASC 842, "Leases", which supersedes the guidance in ASC 840 "Leases." Under ASC 842, lease assets and liabilities, whether arising from leases that are considered operating or finance (capital) will be recognized on the balance sheet. Enhanced qualitative disclosures also will be required. This guidance is effective on a modified retrospective basis--with various practical expedients related to leases that commenced before the effective date—for annual reporting periods, including interim reporting periods within those annual reporting periods, beginning after December 15, 2018. The Company will adopt ASC 842 on January 1, 2019, for all open leases with a term greater than one year, as of the adoption date, and will elect the hindsight practical expedient in determining its lease terms.

Years Ended December 31, 2018, 2017, and 2016

Based upon the Company's current lease obligations, this is expected to result in between \$6.0 million and \$7.0 million of additional lease assets, between \$8.0 million and \$10.0 million of additional lease liabilities, and an adjustment to the opening balance of retained earnings between \$1.0 million and \$2.0 million. The Company will also reclassify, upon adoption, deferred rent liabilities to reduce the measurement of lease assets.

Stock-based compensation - In March 2016, the FASB issued ASU 2016-09, "Improvements to Employee Shared-Based Payment Accounting." The ASU requires that excess tax benefits and deficiencies be recognized as income tax benefit or expense, rather than as additional paid-in capital. In addition, the ASU requires that excess tax benefits be recorded in the period that shares vest or settle, regardless of whether the benefit reduces taxes payable in the same period. Cash flows related to excess tax benefits will be included as an operating activity, and no longer classified as a financing activity, in the statement of cash flows. This guidance was effective for annual reporting periods, including interim reporting periods within those annual reporting periods, beginning after December 15, 2016. The guidance related to the recognition of excess tax benefits and deficiencies as income tax benefit or expense was effective on a prospective basis, and the guidance related to the timing of excess tax benefit recognition was effective using a modified retrospective transition method with a cumulative-effect adjustment to equity as of the beginning of the period in which the guidance is adopted. The cash flow presentation guidance was effective on a retrospective or prospective basis.

The Company implemented this ASU on January 1, 2017 and recorded a cumulative-effect adjustment of \$51.5 million to credit retained earnings for deferred tax assets related to net operating losses that arose from excess tax benefits, which the Company has deemed realizable. In addition to this:

- At the time of adoption and on a prospective basis, the primary impact of adoption was the recognition of excess tax benefits and deficiencies, including deferred tax assets related to net operating losses that arose from excess tax benefits which the Company has deemed realizable in the income tax provision (rather than in additional paid-in capital). This caused income taxes to differ from taxes at the statutory rates in 2017. For the year ended December 31, 2017, the Company recognized an estimated \$20.1 million decrease to the income tax provision, which resulted in a \$20.1 million increase to income from continuing operations and net income attributable to Blucora, a \$0.45 increase to basic earnings per share, and a \$0.43 increase to diluted earnings per share.
- The Company applied the cash flow presentation guidance on a retrospective basis, restating the consolidated statements of cash flows to present excess tax benefits as an operating activity (rather than a financing activity). For the year ended December 31, 2017, that resulted in an increase to cash provided by operating activities from continuing operations of \$16.0 million and a corresponding decrease to cash used by financing activities from continuing operations. The restatement had no impact on total cash flows for the period presented.

The ASU also clarifies that payments made to tax authorities on an employee's behalf for withheld shares should be presented as a financing activity in the statement of cash flows, allows the repurchase of more of an employee's shares for tax withholding purposes without triggering liability accounting, and provides an accounting policy election to account for forfeitures as they occur. The cash flow presentation requirements for payments made to tax authorities on an employee's behalf had no impact to any periods presented, since such cash flows historically have been presented as a financing activity. The Company is not planning to change tax withholdings and will continue to estimate forfeitures in determining the amount of compensation cost to be recognized in each period.

Measurement of Credit Losses - In June 2016, the FASB issued ASU 2016-03 "Measurement of Credit Losses on Financial Statements" that requires companies to measure credit losses utilizing a methodology that reflects expected credit losses and requires a consideration of a broader range of reasonable and supportable information to inform credit loss estimates. This ASU is effective for fiscal years beginning after December 15, 2019, including those interim periods within those fiscal years. The Company is currently assessing the impact of adopting this ASU, but based on a preliminary assessment, does not expect the adoption of this guidance to have a material impact on its consolidated financial statements and related disclosures.

Statement of cash flows and restricted cash - In November 2016, the FASB issued ASU 2016-18 "Statement of Cash Flows: Restricted Cash" on the classification and presentation of changes in restricted cash on the statement of cash flows. The ASU requires that the statement of cash flows explains the change during the period in the total of cash, cash equivalents, and restricted cash; therefore, the amounts described as restricted cash should be included with cash and cash equivalents when

## Years Ended December 31, 2018, 2017, and 2016

reconciling the beginning and end of period total amounts on the statement of cash flows. This guidance became effective for annual reporting periods, including interim reporting periods within those annual reporting periods, beginning after December 15, 2017. Early adoption is permitted. The guidance is effective on a retrospective basis. The Company elected to early adopt this guidance as of January 1, 2017. The reclassification was not material to the periods presented and had no impact on total cash flows, income from continuing operations, or net income attributable to Blucora for the periods presented.

Share-Based Payments - In June 2018, the FASB issued ASU 2018-07 "Stock Compensation: Improvements to Nonemployee Share-based Payment Accounting" that requires companies to account for share-based payments granted to non-employees similarly to share-based payments granted to employees. This ASU is effective for fiscal years beginning after December 15, 2018, including the interim periods within those fiscal years. Early adoption of this ASU is permitted. In the third quarter of 2018, the Company early adopted the requirements of the new standard effective January 1, 2018, utilizing the alternative adoption method.

The adoption of this ASU had a \$0.3 million cumulative effect on the Company's 2018 results, with a corresponding adjustment to additional paid-in capital (amounts below in thousands, except per share data):

|  | First Quarter |          |    | Second Q |    |          | Quarter |        |
|--|---------------|----------|----|----------|----|----------|---------|--------|
|  |               | Reported |    | Recast   |    | Reported |         | Recast |
| Income statement data:                                     |               |          |    |          |    |          |         |        |
| Wealth management services cost of revenue                 | \$            | 63,067   | \$ | 63,064   | \$ | 62,452   | \$      | 62,149 |
| Operating income (loss)                                    |               | 52,734   |    | 52,737   |    | 38,823   |         | 39,126 |
| Net income (loss)  |               | 45,543   |    | 45,546   |    | 35,157   |         | 35,460 |
| Net income (loss) attributable to Blucora, Inc.            |               | 45,338   |    | 45,341   |    | 34,935   |         | 35,238 |
|  |               |          |    |          |    |          |         |        |
| Net income (loss) per share attributable to Blucora, Inc.: |               |          |    |          |    |          |         |        |
| Basic  | \$            | 0.97     | \$ | 0.97     | \$ | 0.74     | \$      | 0.75   |
|  |               |          |    |          |    |          |         |        |
| Weighted average shares outstanding:                       |               |          |    |          |    |          |         |        |
| Basic  |               | 46,641   |    | 46,641   |    | 47,221   |         | 47,221 |

# Note 3: Segment Information and Revenues

The Company has two reportable segments: the Wealth Management segment and the Tax Preparation segment. The Company's Chief Executive Officer is its chief operating decision maker and reviews financial information presented on a disaggregated basis. This information is used for purposes of allocating resources and evaluating financial performance.

The Company does not allocate certain general and administrative costs (including personnel and overhead costs), stock-based compensation, depreciation, and amortization of intangible assets to the reportable segments. Such amounts are reflected in the table under the heading "Corporate-level activity." In addition, the Company does not allocate other loss, net and income taxes to the reportable segments. The Company does not account for, and does not report to management, its assets or capital expenditures by segment other than goodwill and intangible assets used for impairment analysis purposes.

Information on reportable segments currently presented to the Company's chief operating decision maker and a reconciliation to consolidated net income are presented below (in thousands):

## Years Ended December 31, 2018, 2017, and 2016

|  | <br>Years ended December 31, |    |          |    |          |  |  |
|--|------------------------------|----|----------|----|----------|--|--|
|  | 2018                         |    | 2017     |    | 2016     |  |  |
| Revenue:                                     |                              |    |          |    |          |  |  |
| Wealth Management                            | \$<br>373,174                | \$ | 348,620  | \$ | 316,546  |  |  |
| Tax Preparation                              | <br>187,282                  |    | 160,937  |    | 139,365  |  |  |
| Total revenue                                | 560,456                      |    | 509,557  |    | 455,911  |  |  |
| Operating income:                            |                              |    |          |    |          |  |  |
| Wealth Management                            | 53,053                       |    | 50,916   |    | 46,296   |  |  |
| Tax Preparation                              | 87,249                       |    | 72,921   |    | 66,897   |  |  |
| Corporate-level activity                     | (72,625)                     |    | (75,800) |    | (76,076) |  |  |
| Total operating income                       | 67,677                       |    | 48,037   |    | 37,117   |  |  |
| Other loss, net                              | (15,797)                     |    | (44,551) |    | (39,781) |  |  |
| Income tax benefit (expense)                 | (311)                        |    | 25,890   |    | 1,285    |  |  |
| Discontinued operations, net of income taxes | _                            |    | _        |    | (63,121) |  |  |
| Net income (loss)                            | \$<br>51,569                 | \$ | 29,376   | \$ | (64,500) |  |  |

Revenues by major category within each segment are presented below (in thousands):

|                                 | Years ended December 31, |         |    |         |    |         |  |  |
|---------------------------------|--------------------------|---------|----|---------|----|---------|--|--|
|                                 |                          | 2018    |    | 2017    |    | 2016    |  |  |
| Wealth Management:              |                          |         |    |         |    |         |  |  |
| Commission                      | \$                       | 164,201 | \$ | 160,241 | \$ | 150,125 |  |  |
| Advisory                        |                          | 164,353 |    | 145,694 |    | 129,417 |  |  |
| Asset-based                     |                          | 31,456  |    | 26,297  |    | 22,653  |  |  |
| Transaction and fee             |                          | 13,164  |    | 16,388  |    | 14,351  |  |  |
| Total Wealth Management revenue | \$                       | 373,174 | \$ | 348,620 | \$ | 316,546 |  |  |
| Tax Preparation:                |                          |         |    |         |    |         |  |  |
| Consumer                        | \$                       | 172,207 | \$ | 147,084 | \$ | 126,289 |  |  |
| Professional                    |                          | 15,075  |    | 13,853  |    | 13,076  |  |  |
| Total Tax Preparation revenue   | \$                       | 187,282 | \$ | 160,937 | \$ | 139,365 |  |  |

See "Note 2: Summary of Significant Accounting Policies" for a discussion of the new revenue recognition standard, ASC 606, adopted by the Company on January 1, 2018.

Wealth Management revenue recognition: Wealth Management revenue consists primarily of commission revenue, advisory revenue, asset-based revenue, and transaction and fee revenue. The Company's Wealth Management revenues are earned from customers primarily located in the United States.

Wealth management revenue details are as follows:

Commission revenue - Commission revenue represents amounts generated by the Company's clients' purchases and sales of securities and various investment products. The Company serves as the registered broker/dealer or insurance agent for those trades. The Company generates two types of commission revenues: transaction-based sales commissions that occur on the trade date, which is when the Company's performance obligations have been substantially completed, and trailing commissions, which are paid to the Company (typically in arrears on a quarterly basis) based on the clients' account balance, rather than a per-transaction fee.

Advisory revenue - Advisory revenue includes fees charged to clients in advisory accounts where the Company is the Registered Investment Adviser. These fees are based on the value of assets within these advisory accounts. Advisory revenues

## Years Ended December 31, 2018, 2017, and 2016

are deferred and recognized ratably over the period (typically quarterly) in which the performance obligations, which are defined in ASC 606 as promises to transfer goods or services, have been completed.

Asset-based revenue - Asset-based revenue primarily includes fees from financial product manufacturer sponsorship programs, cash sweep programs and other asset-based revenues, primarily including margin revenues, and are recognized ratably over the period in which services are provided.

Transaction and fee revenue - Transaction and fee revenue primarily includes support fees charged to advisers, which are recognized over time as advisory services are provided, fees charged for executing certain transactions in client accounts, which are recognized on a trade-date basis, and other fees related to services provided and other account charges as generally outlined in agreements with financial advisers, clients, and financial institutions, which are recognized as services are performed or as earned, as applicable.

Details of Wealth Management revenues are:

|                                    | Year ended December 31, 2018 |                               |     |                   |    |         |  |  |
|------------------------------------|------------------------------|-------------------------------|-----|-------------------|----|---------|--|--|
| Wealth Management Segment Revenues | R                            | ecognized Upon<br>Transaction | Rec | ognized Over Time |    | Total   |  |  |
| Commission revenue                 | \$                           | 67,351                        | \$  | 96,850            | \$ | 164,201 |  |  |
| Advisory revenue                   |                              | _                             |     | 164,353           |    | 164,353 |  |  |
| Asset-based revenue                |                              | _                             |     | 31,456            |    | 31,456  |  |  |
| Transaction and fee revenue        |                              | 3,211                         |     | 9,953             |    | 13,164  |  |  |
| Total                              | \$                           | 70,562                        | \$  | 302,612           | \$ | 373,174 |  |  |

Tax Preparation revenue recognition: The Company derives revenue from the sale of Tax Preparation digital services, ancillary services, packaged tax preparation software, and arrangements that may include a combination of these items. Ancillary services include Tax Preparation support services, e-filing services, bank or reloadable pre-paid debit card services, and other value-added services, including enhanced tax and Wealth Management services through HD Vest. The Company's Tax Preparation revenues are earned from customers primarily located in the United States.

Years Ended December 31, 2018, 2017, and 2016

Tax Preparation revenue details are as follows:

Consumer revenue - Consumer revenue includes revenue associated with the Company's digital software products, downloadable or shipped desktop software products, add-on services such as refund payment transfer services, bank or reloadable pre-paid debit card services, gift cards and audit defense services.

Digital revenues include revenues associated with the Company's digital software products sold to customers and businesses primarily for the preparation of individual or business tax returns, and are generally recognized when customers and businesses complete and file returns.

Desktop revenues primarily include revenues from all downloadable or shipped software products and are generally recognized when customers download the software or when the software ships.

Add-on services are revenues related to services such as refund payment transfer services, bank or reloadable pre-paid debit card services, gift cards and audit defense services, and are generally recognized as customers complete and file returns.

Professional revenue - Professional revenues include revenues associated with the Company's desktop software products sold to tax return preparers who utilize the Company's offerings to service end customers and are generally recognized when customers download the software or when the software ships. Professional customers have the option to elect an unlimited e-filing package or a pay-per-return package. As the unlimited e-filing package can be reused, those revenues are recognized over an estimated filing timeline. Revenues from the pay-per-return package are recognized when customers complete and file returns.

Details of Tax Preparation revenues are:

|                                  | Year ended December 31, 2018   |                      |    |         |  |  |  |  |
|----------------------------------|--------------------------------|----------------------|----|---------|--|--|--|--|
| Tax Preparation Segment Revenues | Recognized Upon<br>Transaction | Recognized Over Time |    | Total   |  |  |  |  |
| Consumer                         | \$ 172,207                     | \$                   | \$ | 172,207 |  |  |  |  |
| Professional                     | 12,604                         | 2,471                |    | 15,075  |  |  |  |  |
| Total                            | \$ 184,811                     | \$ 2,471             | \$ | 187,282 |  |  |  |  |

## **Note 4: Business Combinations**

HD Vest: On December 31, 2015, pursuant to a Purchase Agreement dated October 14, 2015, the Company acquired HD Vest for \$613.7 million, including cash acquired of \$38.9 million and after a \$1.8 million final working capital adjustment in the first quarter of 2016. In connection with the acquisition, certain members of HD Vest management rolled over a portion of the proceeds they would have otherwise received at the closing into shares of the acquisition subsidiary through which the Company consummated the purchase of HD Vest. A portion of those shares were sold to the Company in exchange for a promissory note. After giving effect to the rollover shares and related purchase of the rollover shares for the promissory note, the Company indirectly owns 95.52% of HDV Holdings, Inc., with the remaining 4.48% noncontrolling interest held collectively by the rollover management members and subject to put and call arrangements exercisable beginning in 2019.

The acquisition was funded by a combination of cash on hand and the TaxAct - HD Vest 2015 credit facility, under which the Company borrowed \$400.0 million (see "Note 10: Debt").

Certain members of HD Vest management rolled over a portion of the proceeds they would have otherwise received at the acquisition's closing into shares of the acquisition subsidiary through which the Company consummated the purchase of HD Vest. The former President of HD Vest sold a portion of his shares to the Company in exchange for a promissory note. The promissory note was scheduled to be paid over a three-year period with 50% to be paid in year one (\$3.2 million paid in

#### Years Ended December 31, 2018, 2017, and 2016

December 2016), 40% to be paid in 2017, and 10% to be paid in 2018. In December 2017, the Company fully repaid the remaining \$3.2 million outstanding. The note bore interest at a rate of 5% per year.

#### **Note 5: Discontinued Operations**

On October 14, 2015, the Company announced its plans to focus on the technology-enabled financial solutions market, as more fully described in "Note 1: Description of the Business." The Strategic Transformation included plans to divest the Search and Content and E-Commerce businesses. Financial condition, results of operations, cash flows, and the notes to financial statements reflect the Search and Content and E-Commerce businesses as discontinued operations for all periods presented. Amounts in discontinued operations include previously unallocated depreciation, amortization, stock-based compensation, income taxes, and other corporate expenses that were attributable to the Search and Content and E-Commerce businesses.

On November 17, 2016, the Company closed on an agreement with YFC, under which YFC acquired the E-Commerce business for \$40.5 million, which included a working capital adjustment. Of this amount, \$39.5 million was received in the fourth quarter of 2016 and the remaining \$1.0 million was received in the first half of 2017. The Company used all of the proceeds to pay down debt and recognized a loss on sale of the E-Commerce business of approximately \$52.2 million.

On August 9, 2016, the Company closed on an agreement with OpenMail, under which OpenMail acquired substantially all of the assets and assumed certain specified liabilities of the Search and Content business for \$45.2 million. The Company used all of the proceeds to pay down debt and recognized a loss on sale of the Search and Content business of approximately \$21.6 million.

Under a separate agreement, the Company is subleasing to InfoSpace the office space that InfoSpace is using currently. The rent payments and September 2020 termination date are consistent with the underlying non-cancelable operating lease.

Summarized financial information for discontinued operations is as follows (in thousands):

|  | Year ended |                   |  |
|--|------------|-------------------|--|
|  | Γ          | December 31, 2016 |  |
| Major classes of items in net income (loss):                   |            |                   |  |
| Revenues   | \$         | 227,989           |  |
| Operating expenses   |            | (211,395)         |  |
| Other income (loss), net                                       |            | (719)             |  |
| Income (loss) from discontinued operations before income taxes |            | 15,875            |  |
| Loss on sale of discontinued operations before income taxes    |            | (73,800)          |  |
| Discontinued operations, before income taxes                   |            | (57,925)          |  |
| Income tax benefit (expense)                                   |            | (5,196)           |  |
| Discontinued operations, net of income taxes                   | \$         | (63,121)          |  |

Business exit costs: In conjunction with the Strategic Transformation, the Company incurred business exit costs of approximately \$4.5 million, which primarily were recorded in discontinued operations in the fourth quarter of 2015 and the first quarter of 2016.

### **Note 6: Restructuring**

The table below summarizes the activity in the restructuring liability (in thousands), resulting from the relocation of the Company's corporate headquarters to Irving, Texas in 2017. These costs were primarily recorded in "Restructuring" on the consolidated statements of comprehensive income and within corporate-level activity for segment purposes.

#### Years Ended December 31, 2018, 2017, and 2016

|                                 | oyee-Related<br>ination Costs | 7  | Contract<br>Fermination<br>Costs | Fixed Asset<br>Impairments | Stock-Based<br>Compensation | Other Costs | Total       |
|---------------------------------|-------------------------------|----|----------------------------------|----------------------------|-----------------------------|-------------|-------------|
| Balance as of December 31, 2016 | \$<br>4,234                   | \$ | _                                | \$<br>_                    | \$<br>_                     | \$<br>_     | \$<br>4,234 |
| Restructuring charges           | 261                           |    | (241)                            | 1,878                      | 1,148                       | 55          | 3,101       |
| Payments                        | (3,293)                       |    | (535)                            | _                          | _                           | (55)        | (3,883)     |
| Non-cash                        | _                             |    | 1,457                            | (1,878)                    | (1,148)                     | _           | (1,569)     |
| Balance as of December 31, 2017 | \$<br>1,202                   | \$ | 681                              | \$<br>_                    | \$<br>_                     | \$<br>_     | \$<br>1,883 |
| Restructuring charges           | 288                           |    | _                                | _                          | _                           | _           | 288         |
| Payments                        | <br>(1,490)                   |    | (157)                            |                            | <br>                        |             | (1,647)     |
| Balance as of December 31, 2018 | \$<br>_                       | \$ | 524                              | \$<br>_                    | \$<br>_                     | \$<br>_     | \$<br>524   |

Employee-related termination costs primarily include severance benefits, under both ongoing and one-time benefit arrangements. Contract termination costs and fixed asset impairments were incurred in connection with the previous headquarters' operating lease and related fixed assets, which are described further in the next two paragraphs, respectively. Stock-based compensation primarily includes the impact of equity award modifications associated with employment contracts for certain individuals impacted by the relocation, as well as forfeitures that were recorded for severed employees. Other costs include office relocation costs.

The Company has a non-cancelable operating lease that runs through 2020 for its former corporate headquarters in Bellevue, Washington, which the Company occupied until May 2017. In March 2017, the Company agreed to a sublease for the entire Bellevue facility, which was effective June 1, 2017 and expires on September 30, 2020, consistent with the underlying operating lease. Under that sublease agreement, the Company will not recover all of its remaining lease rental obligations (including common area maintenance costs and real estate taxes) and, therefore, recognized a net loss on sublease of \$0.4 million, shown above under contract termination costs.

The Company fully impaired the \$1.9 million carrying value of the leasehold improvements and the office furniture and equipment that would not be fully recovered in connection with this lease.

All of these items were recorded in the first quarter of 2017.

# Note 7: Goodwill and Other Intangible Assets

The following table presents goodwill by reportable segment (in thousands):

|   | Weal | th Management | Т  | ax Preparation | Total         |
|---|------|---------------|----|----------------|---------------|
| Balance as of December 31, 2016         | \$   | 356,041       | \$ | 192,700        | \$<br>548,741 |
| Foreign currency translation adjustment |      |               |    | 296            | 296           |
| Balance as of December 31, 2017         |      | 356,041       |    | 192,996        | 549,037       |
| Foreign currency translation adjustment |      |               |    | (352)          | (352)         |
| Balance as of December 31, 2018         | \$   | 356,041       | \$ | 192,644        | \$<br>548,685 |

# Years Ended December 31, 2018, 2017, and 2016

Intangible assets other than goodwill consisted of the following (in thousands):

|  |   | December 31, 2018 |                             |    |                            |    |         |                             | Dece | ember 31, 2017             | ,  |         |
|--|---|-------------------|-----------------------------|----|----------------------------|----|---------|-----------------------------|------|----------------------------|----|---------|
|  | Weighted Average<br>Amortization<br>Period (months) |                   | Gross<br>carrying<br>amount |    | .ccumulated<br>mortization |    | Net     | Gross<br>carrying<br>amount |      | .ccumulated<br>mortization |    | Net     |
| Definite-lived intangible assets:      |   |                   |                             |    |                            |    |         |                             |      |                            |    |         |
| Advisor relationships                  | 204   | \$                | 240,300                     | \$ | (50,973)                   | \$ | 189,327 | \$<br>240,300               | \$   | (34,211)                   | \$ | 206,089 |
| Customer relationships                 | 13  |                   | 101,686                     |    | (87,811)                   |    | 13,875  | 101,711                     |      | (75,105)                   |    | 26,606  |
| Technology                             | 36  |                   | 43,847                      |    | (38,396)                   |    | 5,451   | 43,895                      |      | (35,452)                   |    | 8,443   |
| Sponsor relationships                  | 180   |                   | 16,500                      |    | (2,750)                    |    | 13,750  | 16,500                      |      | (1,833)                    |    | 14,667  |
| Curriculum                             | 12  |                   | 800                         |    | (600)                      |    | 200     | 800                         |      | (400)                      |    | 400     |
| Total definite-lived intangible assets | 186   |                   | 403,133                     |    | (180,530)                  |    | 222,603 | 403,206                     |      | (147,001)                  |    | 256,205 |
| Indefinite-lived intangible assets:    |   |                   |                             |    |                            |    |         |                             |      |                            |    |         |
| Trade names                            |   |                   | 72,000                      |    | _                          |    | 72,000  | 72,000                      |      | _                          |    | 72,000  |
| Total                                  |   | \$                | 475,133                     | \$ | (180,530)                  | \$ | 294,603 | \$<br>475,206               | \$   | (147,001)                  | \$ | 328,205 |

Amortization expense was as follows (in thousands):

|  | <br>Years ended December 31, |    |        |    |        |  |  |  |
|--|------------------------------|----|--------|----|--------|--|--|--|
|  | 2018                         |    | 2017   |    | 2016   |  |  |  |
| Statement of comprehensive income line item:     |                              |    |        |    |        |  |  |  |
| Cost of revenue                                  | \$<br>99                     | \$ | 195    | \$ | 812    |  |  |  |
| Amortization of other acquired intangible assets | <br>33,487                   |    | 33,807 |    | 33,331 |  |  |  |
| Total  | \$<br>33,586                 | \$ | 34,002 | \$ | 34,143 |  |  |  |

Expected amortization of definite-lived intangible assets held as of December 31, 2018 is as follows (in thousands):

|   | _        | 2019   | <br>2020     | <br>2021     | <br>2022     | <br>2023     | Thereafter | Total      |
|---|----------|--------|--------------|--------------|--------------|--------------|------------|------------|
| Statement of comprehensive income (loss) li | ne item: |        |              |              |              |              |            |            |
| Amortization of other acquired intangible   | e assets | 32.176 | \$<br>19.822 | \$<br>16,992 | \$<br>14.841 | \$<br>14,405 | \$ 124.367 | \$ 222,603 |

#### Years Ended December 31, 2018, 2017, and 2016

#### **Note 8: Fair Value Measurements**

Acquisition-related contingent consideration liability

Total liabilities at fair value

In accordance with ASC 820, "Fair Value Measurements and Disclosures," certain of the Company's assets and liabilities, which are carried at fair value, are classified in one of the following three categories:

- Level 1: Quoted market prices in active markets for identical assets or liabilities.
- Level 2: Observable market-based inputs, other than Level 1, or unobservable inputs that are corroborated by market data.
- Level 3: Unobservable inputs that are not corroborated by market data and reflect the Company's own assumptions.

The fair value hierarchy of the Company's financial assets and liabilities carried at fair value and measured on a recurring basis was as follows (in thousands):

Fair value measurements at the reporting date using

2,689

\$

|  | De | ecember 31, 2018 |    | uoted prices in active<br>arkets using identical<br>assets<br>(Level 1) | Signi | ificant other observable inputs (Level 2) | un   | Significant<br>observable inputs<br>(Level 3) |
|--|----|------------------|----|---|-------|---|------|---|
| Cash equivalents: money market and other funds         | \$ | 23,181           | \$ | 23,181  | \$    | _   | \$   | _   |
| Total assets at fair value                             | \$ | 23,181           | \$ | 23,181  | \$    |   | \$   | _   |
| Acquisition-related contingent consideration liability | \$ | 1,275            | \$ |   | \$    |   | \$   | 1,275   |
| Total liabilities at fair value                        | \$ | 1,275            | \$ | _   | \$    |   | \$   | 1,275   |
|  |    |                  |    | Fair value  | measu | rements at the reporting                  | date | using   |
|  | De | ecember 31, 2017 | •  | uoted prices in active<br>arkets using identical<br>assets<br>(Level 1) | Signi | ificant other observable inputs (Level 2) | un   | Significant<br>observable inputs<br>(Level 3) |
| Cash equivalents: money market and other funds         | \$ | 10,857           | \$ | 10,857  | \$    |   | \$   | _   |
| Total assets at fair value                             | \$ | 10,857           | \$ | 10,857  | \$    | _   | \$   | _   |

2,689

2,689

A reconciliation of Level 3 items measured at fair value on a recurring basis was as follows (in thousands):

\$

|   |    | Years ended December 31, |    |       |  |  |
|---|----|--------------------------|----|-------|--|--|
|   |    | 2018                     |    | 2017  |  |  |
| Acquisition-related contingent consideration liability: | '  | _                        |    |       |  |  |
| Balance at beginning of year                            | \$ | 2,689                    | \$ | 3,421 |  |  |
| Payment   |    | (1,315)                  |    | (946) |  |  |
| Foreign currency transaction (gain) loss                |    | (99)                     |    | 214   |  |  |
| Balance at end of year                                  | \$ | 1,275                    | \$ | 2,689 |  |  |

Cash equivalents are classified within Level 1 of the fair value hierarchy because the Company values them utilizing quoted prices in active markets. Unrealized gains and losses are included in "Accumulated other comprehensive loss" on the consolidated balance sheets, and amounts reclassified out of comprehensive income into net income are determined on the basis of specific identification.

The contingent consideration liability is related to the Company's 2015 acquisition of SimpleTax Software Inc. ("SimpleTax"), and the related payments that began in 2017 and are expected to continue annually through 2019. As of December 31, 2018, the Company was required to an additional amount of \$1.3 million. This liability is included within Level

#### Years Ended December 31, 2018, 2017, and 2016

3 of the fair value hierarchy because the Company values it utilizing inputs not observable in the market. Specifically, the Company has determined the fair value of the contingent consideration liability based on a probability-weighted discounted cash flow analysis, which includes assumptions related to estimating SimpleTax revenues, the probability of payment (100%), and the discount rate (9%).

The change in the fair value of the contingent consideration liability is recognized in "General and administrative" expense on the consolidated statements of comprehensive income for the period in which the fair value changes. As of December 31, 2018, the contingent consideration liability was included in "Accrued expenses and other current liabilities" on the consolidated balance sheets.

#### **Note 9: Balance Sheet Components**

Other receivables consisted of the following (in thousands):

|                         | <br>December 31, |    |       |  |  |
|-------------------------|------------------|----|-------|--|--|
|                         | 2018             |    | 2017  |  |  |
| Income taxes receivable | \$<br>7,243      | \$ | 2,802 |  |  |
| Other receivables       | 165              |    | 378   |  |  |
| Total other receivables | \$<br>7,408      | \$ | 3,180 |  |  |

Prepaid expenses and other current assets, net consisted of the following (in thousands):

|  |      | December 31,   |
|--|------|----------------|
|  | 2018 | 2017           |
| Prepaid expenses                                     | \$   | 7,169 \$ 6,972 |
| Other current assets                                 |      | 586 393        |
| Total prepaid expenses and other current assets, net | \$   | 7,755 \$ 7,365 |

Property and equipment, net consisted of the following (in thousands):

|                                    | December 31, |    |          |  |  |
|------------------------------------|--------------|----|----------|--|--|
|                                    | 2018         |    | 2017     |  |  |
| Internally-developed software      | \$<br>9,220  | \$ | 2,728    |  |  |
| Computer equipment and data center | 5,641        |    | 7,121    |  |  |
| Purchased software                 | 4,214        |    | 4,200    |  |  |
| Leasehold improvements and other   | 3,313        |    | 3,244    |  |  |
| Office furniture                   | 929          |    | 801      |  |  |
| Office equipment                   | 662          |    | 557      |  |  |
|                                    | <br>23,979   |    | 18,651   |  |  |
| Accumulated depreciation           | (13,724)     |    | (12,081) |  |  |
|                                    | 10,255       |    | 6,570    |  |  |
| Capital projects in progress       | 2,134        |    | 3,261    |  |  |
| Total property and equipment, net  | \$<br>12,389 | \$ | 9,831    |  |  |

Total depreciation expense was \$5.0 million, \$4.1 million, and \$4.5 million for the years ended December 31, 2018, 2017, and 2016, respectively.

The net book value of internally-developed software was \$8.0 million and \$4.1 million at December 31, 2018 and 2017, respectively. The Company recorded amortization expense for internally-developed software of \$1.5 million, \$0.9 million, and \$1.0 million for the years ended December 31, 2018, 2017, and 2016, respectively.

#### Years Ended December 31, 2018, 2017, and 2016

Accrued expenses and other current liabilities consisted of the following (in thousands):

|  | <br>December 31, |    |        |  |  |
|--|------------------|----|--------|--|--|
|  | 2018             |    | 2017   |  |  |
| Salaries and related expenses                        | \$<br>13,050     | \$ | 12,451 |  |  |
| Other  | 5,976            |    | 7,126  |  |  |
| Total accrued expenses and other current liabilities | \$<br>19,026     | \$ | 19,577 |  |  |

In 2018, the Company received approximately \$9.3 million of incentives from its new clearing firm provider, which is reported in current and long-term deferred revenue on the consolidated balance sheets. The Company offsets these incentives against operating expenses.

#### Note 10: Debt

The Company's debt consisted of the following (in thousands):

|                                |                     | December | 31, 2018            |                    |                     | Decembe    | r 31, 2017          |                    |
|--------------------------------|---------------------|----------|---------------------|--------------------|---------------------|------------|---------------------|--------------------|
|                                |                     | Unam     | ortized             |                    |                     | Unam       | ortized             |                    |
|                                | Principal<br>amount | Discount | Debt issuance costs | Net carrying value | Principal<br>amount | Discount   | Debt issuance costs | Net carrying value |
| Senior secured credit facility | \$ 265,000          | \$ (970) | \$ (3,640)          | \$ 260,390         | \$ 345,000          | \$ (1,455) | \$ (5,464)          | \$ 338,081         |

Senior secured credit facility: In May 2017, the Company entered into a credit agreement with a syndicate of lenders in order to provide a term loan and revolving line of credit for working capital, capital expenditures and general business purposes (the "Blucora senior secured credit facilities"). The Blucora senior secured credit facilities provide for up to \$425.0 million of borrowings, consisting of a committed \$50.0 million revolving credit facility (including a letter of credit sub-facility) and a \$425.0 million term loan facility that mature in May 22, 2022 and May 22, 2024, respectively. Obligations under the Blucora senior secured credit facilities are guaranteed by certain of Blucora's subsidiaries and secured by the assets of the Company and its subsidiaries. The Blucora senior secured credit facilities include financial and operating covenants, including a consolidated total net leverage ratio, which are set forth in detail in the credit facility agreement. As of December 31, 2018, the Company was in compliance with all of the financial and operating covenants under the credit facility agreement.

Principal payments on the term loan are payable quarterly in an amount equal to 0.25% of the initial outstanding principal. In November 2017, the credit facility agreement was amended in order to refinance and reprice the initial term loan, such that the applicable interest rate margin is 3.00% for Eurodollar Rate loans and 2.00% for ABR loans. During the year ended December 31, 2018, the Company made prepayments of \$80.0 million towards the term loan

Depending on the Company's Consolidated First Lien Net Leverage Ratio (as defined in the credit facility agreement), the applicable interest rate margin on the revolving credit facility is from 2.75% to 3.00% for Eurodollar Rate loans and 1.75% to 2.00% for ABR loans. Interest is payable at the end of each interest period. As of December 31, 2018 the Company had not borrowed any amounts under the revolving credit facility.

The Company also has the right to prepay the term loan or outstanding amounts under the revolving credit facility without any premium or penalty (other than customary Eurodollar breakage costs). Prepayments on the term loan are subject to certain prepayment minimums. Beginning with the fiscal year ending December 31, 2018, the Company may be required to make annual prepayments on the term loan in an amount equal to a percentage of excess cash flow of the Company during the applicable fiscal year from 0% to 50%, depending on the Consolidated First Lien Net Leverage Ratio (as defined in the credit facility agreement) for such fiscal year.

As of December 31, 2018, the term loan facility's principal amount approximated its fair value as it is a variable rate instrument and the current applicable margin approximates current market conditions.

#### Years Ended December 31, 2018, 2017, and 2016

When the Company entered into the Blucora senior secured credit facilities, it terminated the TaxAct - HD Vest 2015 credit facilities (defined below). At that time, the Company performed an analysis by creditor and determined that the refinancing qualified as an extinguishment of the TaxAct - HD Vest 2015 credit facilities and the Notes (each of which are defined below). As a result, the Company recognized a loss on debt extinguishment during the year ended December 31, 2017, which was recorded in "Other loss, net" on the consolidated statements of comprehensive income and consisted of the following (in thousands):

| Loss on debt extinguishment - TaxAct - HD Vest 2015 credit facility | \$<br>9,593  |
|---|--------------|
| Loss on debt extinguishment - Convertible Senior Notes              | 6,715        |
| Total loss on debt extinguishment                                   | \$<br>16,308 |

The amount for the TaxAct - HD Vest 2015 credit facility included the write-off of the remaining unamortized discount and debt issuance costs. For the Notes, the Company allocated the cash paid first to the liability component of the Notes based on the fair value of the redeemed Notes. The fair value was based on a discounted cash flow analysis of the Notes' principal and related interest payments, using a discount rate that approximated the current market rate for similar debt without conversion rights. The difference between the fair value and net carrying value of the repurchased Notes was recognized as a loss and recorded in "Other loss, net" on the consolidated statements of comprehensive income. No amount was allocated to the equity component of the Notes, since the fair value of the liability component exceeded the cash paid.

TaxAct - HD Vest 2015 credit facility: On December 31, 2015, TaxAct and HD Vest entered into an agreement with a syndicate of lenders for the purposes of financing the HD Vest acquisition and providing future working capital flexibility for TaxAct and HD Vest (the "TaxAct - HD Vest 2015 credit facilities"). The credit facility consisted of a \$25.0 million revolving credit loan--which included a letter of credit and swingline loans--and a \$400.0 million term loan for an aggregate \$425.0 million credit facility. The final maturity dates of the revolving credit loan and term loan were December 31, 2020 and December 31, 2022, respectively. Obligations under the credit facility were guaranteed by TaxAct Holdings, Inc. and HD Vest Holdings, Inc. and were secured by the equity of the TaxAct and HD Vest businesses. While Blucora was not a party to the agreement, it had guaranteed the obligations of TaxAct and HD Vest under the credit facility, secured by its equity in TaxAct Holdings, Inc.

TaxAct and HD Vest borrowed \$400.0 million under the term loan and had net repayment activity of \$64.0 million and \$140.0 million during 2017 and 2016, respectively. These repayments resulted in the acceleration of a portion of the unamortized discount and debt issuance costs, which were recorded in "Other loss, net" on the consolidated statements of comprehensive income.

Convertible Senior Notes: On March 15, 2013, the Company issued \$201.3 million aggregate principal amount of its Convertible Senior Notes ("the Notes"), inclusive of the underwriters' exercise in full of their over-allotment option of \$26.3 million. In June 2017, the Company redeemed almost all of the outstanding Notes for cash with proceeds from the senior secured credit facility. The Company received net proceeds from the offering of approximately \$194.8 million after adjusting for debt issuance costs, including the underwriting discount.

During 2016, the Company repurchased \$28.4 million of the Notes' principal for cash of \$20.7 million. The Company allocated the cash paid first to the liability component of the Notes based on the fair value of the repurchased Notes. The fair value was based on a discounted cash flow analysis of the Notes' principal and related interest payments, using a discount rate that approximated the current market rate for similar debt without conversion rights. The difference between the fair value and net carrying value of the repurchased Notes was recognized as a gain, since the Notes were repurchased below par value, and recorded in "Other loss, net" on the consolidated statements of comprehensive income (see "Note 14: Other Loss, Net" for details). No amount was allocated to the equity component of the Notes, since the fair value of the liability component exceeded the cash paid. The repurchase also resulted in the write-down of a portion of the unamortized discount and debt issuance costs, which was also recorded in "Other loss, net" on the consolidated statements of comprehensive income.

#### Years Ended December 31, 2018, 2017, and 2016

The following table sets forth total interest expense related to the Notes (in thousands):

|  | Years end | Years ended December 31, |        |  |  |  |
|--|-----------|--------------------------|--------|--|--|--|
|  | 2017      |                          | 2016   |  |  |  |
| Contractual interest expense (Cash)                | \$ 3,141  | \$                       | 7,619  |  |  |  |
| Amortization of debt issuance costs (Non-cash)     | 401       |                          | 939    |  |  |  |
| Accretion of debt discount (Non-cash)              | 1,567     |                          | 3,666  |  |  |  |
| Total interest expense                             | \$ 5,109  | \$                       | 12,224 |  |  |  |
| Effective interest rate of the liability component | 7.32      | <del></del>              | 7.32%  |  |  |  |

#### Note 11: Commitments and Contingencies

The Company's contractual commitments are as follows for years ending December 31 (in thousands):

|  | 2019         | 2020         | 2021         | 2022         | 2023         | Thereafter    | Total         |
|--|--------------|--------------|--------------|--------------|--------------|---------------|---------------|
| Operating lease commitments:                           |              |              |              |              |              |               |               |
| Operating lease obligations                            | \$<br>3,759  | \$<br>3,461  | \$<br>2,245  | \$<br>1,979  | \$<br>1,677  | \$<br>_       | \$<br>13,121  |
| Sublease income  | (1,288)      | (991)        | _            | _            | _            | _             | (2,279)       |
| Net operating lease commitments                        | 2,471        | 2,470        | 2,245        | 1,979        | 1,677        |               | 10,842        |
| Purchase commitments                                   | 10,088       | 7,311        | 5,640        | 4,077        | 2,444        | 3,325         | 32,885        |
| Debt commitments                                       | _            | _            | _            | _            | _            | 265,000       | 265,000       |
| Interest payable                                       | 11,654       | 11,686       | 11,654       | 11,654       | 11,654       | 4,854         | 63,156        |
| Acquisition-related contingent consideration liability | 1,275        | _            | _            | _            | _            | _             | 1,275         |
| Total  | \$<br>25,488 | \$<br>21,467 | \$<br>19,539 | \$<br>17,710 | \$<br>15,775 | \$<br>273,179 | \$<br>373,158 |

Operating lease commitments: As discussed in "Note 6: Restructuring", the Company has a non-cancelable operating lease that runs through 2020 for its former corporate headquarters in Bellevue, Washington, which the Company occupied until May 2017. In March 2017, the Company agreed to a sublease for the entire Bellevue facility, which was effective June 1, 2017 and expires on September 30, 2020, consistent with the underlying operating lease.

The Company leases office space, and these leases are classified as operating leases. Net rent expense under those operating leases was as follows (in thousands):

|                            | Years ended December 31, |         |    |       |    |       |  |  |
|----------------------------|--------------------------|---------|----|-------|----|-------|--|--|
|                            |                          | 2018    |    | 2017  |    | 2016  |  |  |
| Rent expense               | \$                       | 2,860   | \$ | 2,972 | \$ | 3,793 |  |  |
| Less: sublease rent income |                          | (1,292) |    | (594) |    | (342) |  |  |
| Net rent expense           | \$                       | 1,568   | \$ | 2,378 | \$ | 3,451 |  |  |

*Purchase commitments:* The Company's purchase commitments consist primarily of marketing agreements, commitments with a vendor to provide cloud computation services, commitments to its new clearing firm provider and commitments for advisory support programs.

Debt commitments and interest on Notes: The Company's debt commitments are based upon contractual payment terms and consist of the outstanding principal related to the Blucora senior secured credit facility. For further detail regarding the credit facility, see "Note 10: Debt."

#### Years Ended December 31, 2018, 2017, and 2016

Acquisition-related contingent consideration liability: The contingent consideration liability is related to the Company's acquisition of SimpleTax (see "Note 4: Business Combinations" and "Note 8: Fair Value Measurements"), and the related payments that began in 2017 and are expected to continue annually through 2019.

Off-balance sheet arrangements: The Company has no off-balance sheet arrangements other than operating leases.

Litigation: From time to time, the Company is subject to various legal proceedings or claims that arise in the ordinary course of business. The Company accrues a liability when management believes that it is both probable that a liability has been incurred and the amount of loss can be reasonably estimated. Following is a brief description of the more significant legal proceedings. Although the Company believes that resolving such claims, individually or in aggregate, will not have a material adverse impact on its financial statements, these matters are subject to inherent uncertainties.

On December 12, 2016, a shareholder derivative action was filed by Jeffrey Tilden against the Company, as a nominal defendant, Andrew Snyder, who was a director of the Company at that time, certain companies affiliated with Mr. Snyder, a former officer of the Company, GCA Savvian Advisors, LLC ("GCA Savvian"), and certain other current and former members of the Blucora Board of Directors, in the Superior Court of the State of California in and for the County of San Francisco. The complaint asserted claims for breaches of fiduciary duty against certain current and former directors of the Company related to the Company's share repurchases and the Company's acquisitions of HD Vest and Monoprice. The complaint asserted a claim against GCA Savvian, the Company's financial advisor in connection with the HD Vest acquisition, for aiding and abetting breaches of fiduciary duty. The complaint also asserted a claim for insider trading against Mr. Snyder, a former director of the Company, and certain companies affiliated with Mr. Snyder. The derivative action did not seek monetary damages from the Company. The complaint sought corporate governance reforms, declaratory relief, monetary damages from the other defendants, attorney's fees and prejudgment interest.

On March 10, 2017, the Company filed a motion to dismiss for improper venue as a result of a forum selection provision in the Company's bylaws that required the plaintiff to file his derivative fiduciary duty claims in Delaware. Other defendants also filed motions to quash the summons due to a lack of personal jurisdiction over them. On July 25, 2017, the Court granted the Company's motion to dismiss. The case was stayed by the Court until November 22, 2017 so that Mr. Tilden could file a complaint in Delaware, after which the case was dismissed without further order of the Court.

On November 21, 2017, Mr. Tilden filed a shareholder derivative action in the Delaware Court of Chancery asserting the same claims against the same defendants and seeking the same relief as the San Francisco Superior Court lawsuit. On January 31, 2018, the Company filed a motion to dismiss the Delaware complaint, and a hearing on the motion was held on July 11, 2018. The motion to dismiss was granted on October 26, 2018, and the case has been dismissed with prejudice and without leave to amend.

The Company has entered into indemnification agreements in the ordinary course of business with its officers and directors, and the agreement entered into with GCA Savvian in connection with the acquisition of HD Vest also contained indemnification provisions. Pursuant to these agreements, the Company may be obligated to advance payment of legal fees and costs incurred by the defendants pursuant to the Company's obligations under these indemnification agreements and applicable Delaware law.

### Note 12: Stockholders' Equity

Stock incentive plan: The Company may grant incentive or non-qualified stock options, stock, restricted stock, restricted stock units ("RSUs"), stock appreciation rights and performance shares or performance units to employees, non-employee directors, and consultants.

The Company granted options and RSUs during 2018, 2017 and 2016 under its 2015 Incentive Plan (as amended and restated), as well as options and RSUs during 2016 under its 2016 Inducement Plan. In addition, the Company granted RSUs in 2018 to Directors under its 2018 Long-Term Incentive Plan. Options and RSUs generally vest over a period of one-to-three years, with the majority vesting over three years. Options and RSUs have one-third vesting one year from the date of grant and the remainder vesting ratably thereafter on an annual basis for most awards in 2018 and semi-annually for awards prior to 2018, and options expire seven years from the date of grant. There are a few exceptions to this vesting schedule, which provide for vesting at different rates or based on achievement of performance targets.

#### Years Ended December 31, 2018, 2017, and 2016

The Company issues new shares upon the exercise of options and upon the vesting of RSUs. If an option or RSU is surrendered or otherwise unused, the related shares will continue to be available.

Employee Stock Purchase Plan: The 2016 Employee Stock Purchase Plan ("ESPP") permits eligible employees to contribute up to 15% of their base earnings toward the twice-yearly purchase of Company common stock, subject to an annual maximum dollar amount. The purchase price is the lesser of 85% of the fair market value of common stock on the first day or on the last day of an offering period. An aggregate of 2.4 million shares of common stock are authorized for issuance under the ESPP. Of this amount, 0.8 million shares were available for issuance as of December 31, 2018. The Company issues new shares upon purchase through the ESPP.

Other comprehensive income: The following table provides information about activity in other comprehensive income (in thousands):

|                                 | Unrealized gain (loss) on investments | Foreign currency translation adjustment | Total    |
|---------------------------------|---------------------------------------|---|----------|
| Balance as of December 31, 2015 | \$ (10)                               | \$ (517)                                | \$ (527) |
| Other comprehensive income      | 9                                     | 137                                     | 146      |
| Balance as of December 31, 2016 | (1)                                   | (380)                                   | (381)    |
| Other comprehensive income      | 1                                     | 376                                     | 377      |
| Balance as of December 31, 2017 | _                                     | (4)                                     | (4)      |
| Other comprehensive loss        |                                       | (442)                                   | (442)    |
| Balance as of December 31, 2018 | \$ —                                  | \$ (446)                                | \$ (446) |

#### Note 13: Stock-Based Compensation

A summary of the general terms of stock options and RSUs at December 31, 2018 was as follows:

| Number of shares authorized for awards | 11,958,417 |
|--|------------|
| Options and RSUs outstanding           | 3,310,874  |
| Options and RSUs expected to vest      | 3,105,386  |
| Options and RSUs available for grant   | 6,711,996  |

The following activity occurred under the Company's stock incentive plans:

|   | Options     | •  | Weighted average<br>exercise price | Intrinsic value (in thousands) | Weighted average<br>remaining<br>contractual term (in<br>years) |
|---|-------------|----|------------------------------------|--------------------------------|---|
| Stock options:                                      |             |    |                                    |                                |   |
| Outstanding December 31, 2017                       | 3,805,922   | \$ | 13.13                              |                                |   |
| Granted   | 322,823     | \$ | 24.08                              |                                |   |
| Forfeited   | (199,651)   | \$ | 16.28                              |                                |   |
| Exercised   | (1,481,155) | \$ | 12.62                              |                                |   |
| Outstanding December 31, 2018                       | 2,447,939   | \$ | 14.62                              | \$<br>29,423                   | 4.5   |
| Exercisable December 31, 2018                       | 970,285     | \$ | 11.80                              | \$<br>14,401                   | 4.0   |
| Vested and expected to vest after December 31, 2018 | 2,342,602   | \$ | 14.43                              | \$<br>28,601                   | 4.4   |
|   | 81          |    |                                    |                                |   |

# Years Ended December 31, 2018, 2017, and 2016

|  | Stock units | ighted average<br>t date fair value | Intrinsic value<br>(in thousands) | Weighted average<br>remaining<br>contractual term (in<br>years) |
|--|-------------|-------------------------------------|-----------------------------------|---|
| RSUs:                                    |             |                                     | _                                 |   |
| Outstanding December 31, 2017            | 914,241     | \$<br>12.10                         |                                   |   |
| Granted                                  | 582,285     | \$<br>26.89                         |                                   |   |
| Forfeited                                | (103,684)   | \$<br>17.75                         |                                   |   |
| Vested                                   | (529,907)   | \$<br>11.48                         |                                   |   |
| Outstanding December 31, 2018            | 862,935     | \$<br>21.78                         | \$<br>22,989                      | 1.1   |
| Expected to vest after December 31, 2018 | 762,784     | \$<br>21.65                         | \$<br>20,321                      | 1.1   |

Supplemental information is presented below:

|   | Years ended December 31, |        |    |        |    |       |
|---|--------------------------|--------|----|--------|----|-------|
|   |                          | 2018   |    | 2017   |    | 2016  |
| Stock options:  |                          |        |    |        |    |       |
| Weighted average grant date fair value per share granted  | \$                       | 7.68   | \$ | 6.25   | \$ | 2.46  |
| Total intrinsic value of options exercised (in thousands) | \$                       | 27,759 | \$ | 44,405 | \$ | 437   |
| Total fair value of options vested (in thousands)         | \$                       | 4,142  | \$ | 5,566  | \$ | 7,064 |
| RSUs:   |                          |        |    |        |    |       |
| Weighted average grant date fair value per unit granted   | \$                       | 26.89  | \$ | 18.39  | \$ | 7.82  |
| Total intrinsic value of units vested (in thousands)      | \$                       | 16,452 | \$ | 14,642 | \$ | 5,755 |
| Total fair value of units vested (in thousands)           | \$                       | 6,069  | \$ | 6,469  | \$ | 8,983 |

The Company included the following amounts for stock-based compensation expense, which related to stock options, RSUs, and the ESPP, in the consolidated statements of comprehensive income (in thousands):

|                                | Years ended December 31, |        |    |        |    |        |  |  |
|--------------------------------|--------------------------|--------|----|--------|----|--------|--|--|
|                                |                          | 2018   |    | 2017   |    | 2016   |  |  |
| Cost of revenue                | \$                       | 1,467  | \$ | 774    | \$ | 166    |  |  |
| Engineering and technology     |                          | 766    |    | 984    |    | 1,640  |  |  |
| Sales and marketing            |                          | 2,424  |    | 2,376  |    | 2,548  |  |  |
| General and administrative     |                          | 8,596  |    | 7,519  |    | 9,774  |  |  |
| Restructuring                  |                          | _      |    | 1,148  |    | (364)  |  |  |
| Total in continuing operations |                          | 13,253 |    | 12,801 |    | 13,764 |  |  |
| Discontinued operations        |                          | _      |    | _      |    | 1,471  |  |  |
| Total                          | \$                       | 13,253 | \$ | 12,801 | \$ | 15,235 |  |  |

In 2016, the Company recorded stock-based compensation expense in connection with the corporate headquarters relocation announcement. See "Note 6: Restructuring" for additional information.

#### Years Ended December 31, 2018, 2017, and 2016

To estimate stock-based compensation expense, the Company used the Black-Scholes-Merton valuation method with the following assumptions for stock options granted:

|                         | Y             | Years ended December 31, |               |  |  |  |  |  |
|-------------------------|---------------|--------------------------|---------------|--|--|--|--|--|
|                         | 2018          | 2017                     | 2016          |  |  |  |  |  |
| Risk-free interest rate | 1.82% - 2.54% | 1.2% - 1.94%             | 0.83% - 1.59% |  |  |  |  |  |
| Expected dividend yield | 0%            | 0%                       | 0%            |  |  |  |  |  |
| Expected volatility     | 38% - 42%     | 39% - 45%                | 35% - 45%     |  |  |  |  |  |
| Expected life           | 3.6           | 3.8                      | 3.4           |  |  |  |  |  |

The risk-free interest rate was based on the implied yield available on U.S. Treasury issues with an equivalent remaining term. The Company last paid a dividend in 2008. The expected volatility was based on historical volatility of the Company's stock for the related expected life of the award. The expected life of the award was based on historical experience, including historical post-vesting termination behavior.

As of December 31, 2018, total unrecognized stock-based compensation expense related to unvested stock awards is as follows:

|                                 | Expense<br>thousands) | Weighted average period<br>over which to be recognized<br>(in years) |
|---------------------------------|-----------------------|--|
| Stock options                   | \$<br>3,085           | 1.3  |
| RSUs                            | 8,576                 | 1.8  |
| Total for continuing operations | \$<br>11,661          | 1.7  |

#### Note 14: Other Loss, Net

"Other loss, net" consisted of the following (in thousands):

|   | Years ended December 31, |    |        |      |        |  |  |  |
|---|--------------------------|----|--------|------|--------|--|--|--|
|   | <br>2018 2017            |    |        | 2016 |        |  |  |  |
| Interest expense  | \$<br>15,610             | \$ | 21,211 | \$   | 32,424 |  |  |  |
| Loss on debt extinguishment and amortization of debt issuance costs | 2,367                    |    | 21,534 |      | 2,876  |  |  |  |
| Accretion of debt discounts   | 163                      |    | 1,947  |      | 4,690  |  |  |  |
| Interest income   | (349)                    |    | (110)  |      | (81)   |  |  |  |
| Other   | (1,994)                  |    | (31)   |      | (128)  |  |  |  |
| Other loss, net   | \$<br>15,797             | \$ | 44,551 | \$   | 39,781 |  |  |  |

In 2018, the Company had a \$2.1 million gain on the sale of an investment that is included in "Other" above.

# Note 15: 401(k) Plan

The Company has a 401(k) savings plan covering its employees. Eligible employees may contribute through payroll deductions. The Company may match the employees' 401(k) contributions at the discretion of the Company's Board of Directors. Pursuant to a continuing resolution, the Company has matched a portion of the 401(k) contributions made by its employees. The amount contributed by the Company ranges from 1% to 4% of an employee's salary, depending upon the percentage contributed by the employee. For the years ended December 31, 2018, 2017, and 2016, the Company contributed \$1.9 million, \$1.6 million, and \$1.4 million, respectively, for employees.

#### Years Ended December 31, 2018, 2017, and 2016

#### **Note 16: Income Taxes**

Income (loss) from continuing operations before income taxes consisted of the following (in thousands):

|  | _  | Years ended December 31, |    |       |    |         |  |  |
|--|----|--------------------------|----|-------|----|---------|--|--|
|  |    | 2018                     |    | 2017  |    | 2016    |  |  |
| United States  | \$ | 51,385                   | \$ | 3,293 | \$ | (2,678) |  |  |
| Foreign  |    | 495                      |    | 193   |    | 14      |  |  |
| Income (loss) from continuing operations before income taxes |    | 51,880                   |    | 3,486 |    | (2,664) |  |  |

Income tax expense (benefit) consisted of the following (in thousands):

| <br>Years ended December 31, |  |   |   |  |  |
|------------------------------|--|---|---|--|--|
| 2018                         |  | 2017  |   | 2016   |  |
| <br>                         |  |   |   |  |  |
| \$<br>(42)                   | \$   | 123   | \$  | 14,695   |  |
| 3,230                        |  | 962   |   | 2,048  |  |
| 157                          |  | 122   |   | 27   |  |
| <br>3,345                    |  | 1,207   |   | 16,770   |  |
|                              |  |   |   |  |  |
| (3,035)                      |  | (26,012)  |   | (16,608)   |  |
| 37                           |  | (1,022)   |   | (1,421)  |  |
| <br>(36)                     |  | (63)  |   | (26)   |  |
| <br>(3,034)                  |  | (27,097)  |   | (18,055)   |  |
| \$<br>311                    | \$   | (25,890)  | \$  | (1,285)  |  |
| \$                           | \$ (42)<br>3,230<br>157<br>3,345<br>(3,035)<br>37<br>(36)<br>(3,034) | \$ (42) \$ 3,230 157 3,345  (3,035) 37 (36) (3,034) | 2018         2017           \$ (42) \$ 123           3,230 962           157 122           3,345 1,207           (3,035) (26,012)           37 (1,022)           (36) (63)           (3,034) (27,097) | 2018     2017       \$ (42) \$ 123 \$       3,230 962       157 122       3,345 1,207       (3,035) (26,012)       37 (1,022)       (36) (63)       (3,034) (27,097) |  |

Income tax expense (benefit) differed from the amount computed by applying the statutory federal income tax rate of 21% for 2018, and 35% for 2017 and 2016, as follows (in thousands):

|   | Years ended December 31, |         |    |          |    |         |
|---|--------------------------|---------|----|----------|----|---------|
|   |                          | 2018    |    | 2017     |    | 2016    |
| Income tax expense (benefit) at the statutory federal income tax rate | \$                       | 10,895  | \$ | 1,220    | \$ | (930)   |
| Non-deductible compensation   |                          | 2,796   |    | 283      |    | 249     |
| State income taxes, net of federal benefit                            |                          | 2,014   |    | 582      |    | 454     |
| Uncertain tax positions and audit settlements                         |                          | 473     |    | (321)    |    | (86)    |
| Research and development credit                                       |                          | (552)   |    | _        |    | _       |
| Excess tax benefits of stock-based compensation                       |                          | (6,851) |    | (11,558) |    | _       |
| Valuation allowances  |                          | (8,537) |    | 4,974    |    | 15      |
| Deductible domestic manufacturing costs                               |                          | _       |    | _        |    | (1,225) |
| Tax legislation impact  |                          | _       |    | (21,430) |    | _       |
| Other   |                          | 73      |    | 360      |    | 238     |
| Income tax expense (benefit)  | \$                       | 311     | \$ | (25,890) | \$ | (1,285) |

The tax legislation which became effective January 1, 2018, repealed corporate alternative minimum tax ("AMT") for tax years beginning January 1, 2018 and provides that existing AMT credit carryovers are refundable beginning in 2018. At December 31, 2018 the Company had approximately \$5.5 million and \$5.4 million of AMT credit carryovers reported in "Other receivables" and "Other long-term assets," respectively, on the consolidated balance sheets.

#### Years Ended December 31, 2018, 2017, and 2016

The tax effect of temporary differences and net operating loss carry forwards that gave rise to the Company's deferred tax assets and liabilities were as follows (in thousands):

|   | <br>December 31, |    |           |
|---|------------------|----|-----------|
|   | 2018             |    | 2017      |
| Deferred tax assets:                            |                  |    |           |
| Net operating loss carryforwards                | \$<br>96,602     | \$ | 111,416   |
| Accrued compensation                            | 5,526            |    | 4,586     |
| Deferred revenue                                | 3,700            |    | 1,638     |
| Tax credit carryforwards                        | 552              |    | _         |
| Stock-based compensation                        | 3,971            |    | 3,592     |
| Capital loss                                    | 23,008           |    | 22,579    |
| Other, net                                      | <br>2,143        |    | 3,466     |
| Total gross deferred tax assets                 | 135,502          |    | 147,277   |
| Valuation allowance                             | <br>(100,705)    |    | (109,242) |
| Deferred tax assets, net of valuation allowance | 34,797           |    | 38,035    |
| Deferred tax liabilities:                       |                  |    |           |
| Depreciation and amortization                   | (74,135)         |    | (81,182)  |
| Other, net                                      | <br>(1,056)      |    | (286)     |
| Total gross deferred tax liabilities            | <br>(75,191)     |    | (81,468)  |
| Net deferred tax liabilities                    | \$<br>(40,394)   | \$ | (43,433)  |

At December 31, 2018, the Company evaluated the need for a valuation allowance for certain deferred tax assets based upon its assessment of whether it is more likely than not that the Company will generate sufficient future taxable income necessary to realize the deferred tax benefits. The Company maintains a valuation allowance against its deferred tax assets that are capital in nature to the extent that it is more likely than not that the related deferred tax benefit will not be realized. The Company also has a deferred tax asset related to the net operating losses ("NOLs") that it believes are more likely than not to expire before utilization. If assumptions change and the Company determines it will be able to realize these NOLs, the tax benefit relating to any reversal of the valuation allowance on deferred tax assets as of December 31, 2018 will be recognized as a reduction of income tax expense.

The changes in the valuation allowance for deferred tax assets are shown below (in thousands):

|   | Years ended December 31, |         |      |          |
|---|--------------------------|---------|------|----------|
|   |                          | 2018    | 2017 |          |
| Balance at beginning of year  | \$                       | 109,242 | \$   | 226,813  |
| Decrease in valuation allowance - change in federal income tax rate   |                          | _       |      | (72,482) |
| Decrease in valuation allowance - adoption of ASU 2016-09             |                          | _       |      | (50,203) |
| Increase (decrease) in valuation allowance - current year utilization |                          | (8,597) |      | 4,875    |
| Increase in valuation allowance - other                               |                          | 60      |      | 239      |
| Balance at end of year  | \$                       | 100,705 | \$   | 109,242  |

As of December 31, 2018, the Company's U.S. federal and state net operating loss carryforwards for income tax purposes were \$454.3 million and \$20.8 million, respectively, which primarily related to excess tax benefits for stock-based compensation. Prior to January 1, 2017, when the net operating loss carryforwards related to stock-based compensation were recognized, the income tax benefit of those losses was accounted for as a credit to stockholders' equity on the consolidated balance sheets. Beginning on January 1, 2017, we recognized such income tax benefit on the consolidated financial statements, as further described in the *Recent accounting pronouncements* section of "Note 2: Summary of Significant Accounting Policies." If not utilized, the Company's federal net operating loss carryforwards will expire between 2020 and 2037, with the

#### Years Ended December 31, 2018, 2017, and 2016

majority of them expiring between 2020 and 2024. Additionally, changes in ownership, as defined by Section 382 of the Internal Revenue Code, may limit the amount of net operating loss carryforwards used in any one year.

A reconciliation of the unrecognized tax benefit balances is as follows (in thousands):

|   | Years ended December 31, |        |    |        |    |        |
|---|--------------------------|--------|----|--------|----|--------|
|   | 2018                     |        |    | 2017   |    | 2016   |
| Balance at beginning of year                      | \$                       | 22,625 | \$ | 22,919 | \$ | 21,741 |
| Gross increases for tax positions of prior years  |                          | 516    |    | 93     |    | 331    |
| Gross decreases for tax positions of prior years  |                          | _      |    | (31)   |    | (93)   |
| Gross increases for tax positions of current year |                          | _      |    | _      |    | 997    |
| Settlements with taxing authorities               |                          | _      |    | (66)   |    | (57)   |
| Statute of limitations expirations                |                          | (551)  |    | (290)  |    |        |
| Balance at end of year                            | \$                       | 22,590 | \$ | 22,625 | \$ | 22,919 |

The total amount of unrecognized tax benefits that could affect the Company's effective tax rate if recognized was \$4.7 million and \$4.2 million as of December 31, 2018 and 2017, respectively. The remaining \$17.9 million and \$18.4 million has not been recognized on the consolidated balance sheets as of December 31, 2018 and 2017, respectively, and if recognized, would create a deferred tax asset subject to a valuation allowance. The Company and its subsidiaries file income tax returns in the U.S. federal jurisdiction, various state jurisdictions, and Canada. With few exceptions, the Company is no longer subject to U.S. federal, state and local, or non-U.S. income tax examinations by tax authorities for years before 2014, although NOL carryforwards and tax credit carryforwards from any year are subject to examination and adjustment for at least three years following the year in which they are fully utilized. As of December 31, 2018, no significant adjustments have been proposed relative to the Company's tax positions.

During the years ended December 31, 2018, 2017, and 2016, the Company recognized less than \$0.5 million of interest and penalties related to uncertain tax positions. The Company had approximately \$1.5 million and \$1.1 million accrued for interest and penalties as of December 31, 2018 and 2017, respectively.

# Note 17: Net Income (Loss) Per Share

"Basic net income (loss) per share" is computed using the weighted average number of common shares outstanding during the period. "Diluted net income (loss) per share" is computed using the weighted average number of common shares outstanding plus the number of dilutive potential common shares outstanding during the period. Dilutive potential common shares consist of the incremental common shares issuable upon the exercise of outstanding stock options and the vesting of unvested RSUs. Dilutive potential common shares are excluded from the computation of earnings per share if their effect is antidilutive. The redemption value adjustment of the Company's redeemable noncontrolling interest is deducted from income (loss) (as discussed further in "Note 2: Summary of Significant Accounting Policies").

# Years Ended December 31, 2018, 2017, and 2016

The computation of basic and diluted net income (loss) per share attributable to Blucora, Inc. is as follows (in thousands):

|  | Years ended December 31, |         |    |         |    |          |
|--|--------------------------|---------|----|---------|----|----------|
|  |                          | 2018    |    | 2017    |    | 2016     |
| Numerator:   |                          |         |    |         |    |          |
| Income (loss) from continuing operations   | \$                       | 51,569  | \$ | 29,376  | \$ | (1,379)  |
| Net income attributable to noncontrolling interests  |                          | (935)   |    | (2,337) |    | (658)    |
| Adjustment of redeemable noncontrolling interests*   |                          | (5,977) |    | _       |    | _        |
| Income (loss) from discontinued operations attributable to Blucora, Inc.   |                          | _       |    | _       |    | (63,121) |
| Net income (loss) attributable to Blucora, Inc. shareholders after adjustment of redeemable noncontrolling interests and discontinued operations | \$                       | 44,657  | \$ | 27,039  | \$ | (65,158) |
| Denominator:   |                          |         |    |         |    |          |
| Weighted average common shares outstanding, basic  |                          | 47,394  |    | 44,370  |    | 41,494   |
| Dilutive potential common shares   |                          | 1,987   |    | 2,841   |    | _        |
| Weighted average common shares outstanding, diluted  |                          | 49,381  |    | 47,211  |    | 41,494   |
| Net income (loss) per share attributable to Blucora, Inc basic:  |                          |         |    |         |    |          |
| Continuing operations  | \$                       | 0.94    | \$ | 0.61    | \$ | (0.05)   |
| Discontinued operations  |                          | _       |    | _       |    | (1.52)   |
| Basic net income (loss) per share  | \$                       | 0.94    | \$ | 0.61    | \$ | (1.57)   |
| Net income (loss) per share attributable to Blucora, Inc diluted:  |                          |         |    |         |    |          |
| Continuing operations  | \$                       | 0.90    | \$ | 0.57    | \$ | (0.05)   |
| Discontinued operations  |                          | _       |    | _       |    | (1.52)   |
| Diluted net income (loss) per share  | \$                       | 0.90    | \$ | 0.57    | \$ | (1.57)   |
| Shares excluded  |                          | 354     |    | 1,058   |    | 9,774    |

<sup>\*</sup> See "Note 2: Summary of Significant Accounting Policies" for further discussion of redeemable noncontrolling interests

Shares were excluded from the computation of diluted earnings per common share for these periods because their effect would have been anti-dilutive.

#### ITEM 9. Changes in and Disagreements with Accountants on Accounting and Financial Disclosure

None.

#### ITEM 9A. Controls and Procedures

Evaluation of Disclosure Controls and Procedures

Our management evaluated, with the participation of our Chief Executive Officer and our Chief Financial Officer, the effectiveness of our disclosure controls and procedures as of the end of the period covered by this Annual Report on Form 10-K. Based on this evaluation, our Chief Executive Officer and our Chief Financial Officer have concluded that our disclosure controls and procedures were effective as of December 31, 2018 to ensure that information we are required to disclose in reports that we file or submit under the Securities Exchange Act of 1934 is accumulated and communicated to our management, including our principal executive and principal financial officers, as appropriate to allow timely decisions regarding required disclosure, and that such information is recorded, processed, summarized, and reported within the time periods specified in the Securities and Exchange Commission rules and forms.

Management's Report on Internal Control over Financial Reporting

Our management is responsible for establishing and maintaining adequate internal control over financial reporting, as such term is defined in Exchange Act Rules 13a-15(f). Our internal control over financial reporting is designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions or that the degree of compliance with the policies or procedures may deteriorate.

Under the supervision and with the participation of management, including our Chief Executive Officer and Chief Financial Officer, we conducted an evaluation of the effectiveness of our internal control over financial reporting based on the framework in *Internal Control – Integrated Framework* (2013 framework) issued by the Committee of the Sponsoring Organizations of the Treadway Commission.

Based on our evaluation under the framework in *Internal Control – Integrated Framework* (2013 framework), our management concluded that our internal control over financial reporting was effective as of December 31, 2018.

Ernst & Young LLP has audited the effectiveness of our internal control over financial reporting as of December 31, 2018 and its report is included below.

Changes in Internal Control over Financial Reporting

There was no change in our internal control over financial reporting that occurred during the fourth quarter of 2018 that has materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.

Report of Independent Registered Public Accounting Firm

To the Stockholders and the Board of Directors of Blucora, Inc.

#### Opinion on Internal Control over Financial Reporting

We have audited Blucora, Inc.'s internal control over financial reporting as of December 31, 2018, based on criteria established in Internal Control-Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission (2013 framework) (the COSO criteria). In our opinion, Blucora, Inc. (the Company) maintained, in all material respects, effective internal control over financial reporting as of December 31, 2018, based on the COSO criteria.

We also have audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States) (PCAOB), the consolidated balance sheets of Blucora, Inc. as of December 31, 2018 and 2017, the related consolidated statements of comprehensive income (loss), stockholders' equity and cash flows for each of the three years in the period ended December 31, 2018, and the related notes, and our report dated March 1, 2019 expressed an unqualified opinion thereon.

#### **Basis for Opinion**

The Company's management is responsible for maintaining effective internal control over financial reporting and for its assessment of the effectiveness of internal control over financial reporting included in the accompanying Management's Report on Internal Control over Financial Reporting. Our responsibility is to express an opinion on the Company's internal control over financial reporting based on our audit. We are a public accounting firm registered with the PCAOB and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audit in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether effective internal control over financial reporting was maintained in all material respects.

Our audit included obtaining an understanding of internal control over financial reporting, assessing the risk that a material weakness exists, testing and evaluating the design and operating effectiveness of internal control based on the assessed risk, and performing such other procedures as we considered necessary in the circumstances. We believe that our audit provides a reasonable basis for our opinion.

#### Definition and Limitations of Internal Control Over Financial Reporting

A company's internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company's internal control over financial reporting includes those policies and procedures that (1) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorizations of management and directors of the company; and (3) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

/s/ Ernst & Young LLP

Dallas, Texas March 1, 2019

ITEM 9B. Other Information

None.

#### PART III

As permitted by the rules of the Securities and Exchange Commission, we have omitted certain information from Part III of this Annual Report on Form 10-K. We intend to file a Definitive Proxy Statement (the "*Proxy Statement"*) with the Securities and Exchange Commission relating to our annual meeting of stockholders not later than 120 days after the end of the fiscal year covered by this Annual Report on Form 10-K, and such information is incorporated by reference herein.

### ITEM 10. Directors, Executive Officers and Corporate Governance

The information required in response to this Item 10 is incorporated by reference herein to our Proxy Statement.

#### ITEM 11. Executive Compensation

The information required in response to this Item 11 is incorporated by reference herein to our Proxy Statement.

# ITEM 12. Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters Equity Compensation Plan Information

Our stockholders have approved the Blucora, Inc. 2018 Long-Term Incentive Plan, (the "2018 Plan"), the Blucora, Inc. 2015 Plan, as amended and restated (the "2015 Plan"), the Company's Restated 1996 Flexible Stock Incentive Plan (the "1996 Plan"), and the Blucora, Inc. 2016 Employee Stock Purchase Plan (the "2016 ESPP"). Our Board of Directors adopted the 2016 Inducement Plan (the "Inducement Plan") on January 29, 2016, which did not require stockholder approval under NASDAQ rules. The terms and conditions of the Inducement Plan are substantially similar to those of the 2015 Plan, except that under the Inducement Plan, 2,400,000 shares are authorized for issuance, eligibility is limited to newly hired employees, incentive stock options may not be granted, and the Inducement Plan is not subject to stockholder approval. The 1996 Plan and the 2015 Plan are now terminated, and no additional equity grants may be made under those plans.

See "Note 11: Stockholders' Equity" in the Notes to Consolidated Financial Statements included in Part II, Item 8, for additional information.

The table below sets forth information regarding outstanding awards and shares available for future issuance under the Company's equity compensation plans as of December 31, 2018.

|  | (a)   |     | (b)   |       | (c)   |     |
|--|---|-----|---|-------|---|-----|
| Plan category  | Number of securities to be<br>issued upon exercise of<br>outstanding options,<br>warrants, and rights |     | Weighted-average exercise price<br>of outstanding options,<br>warrants, and rights <sup>(1)</sup> |       | Number of securities remaining<br>available for future issuance under<br>equity compensation plans<br>(excluding securities reflected in<br>column (a)) | r   |
| Equity compensation plans approved by stockholders     | 2,768,128   | (2) | \$  | 15.85 | 6,281,763   | (3) |
| Equity compensation plans not approved by stockholders | 542,682   | (4) | \$  | 9.54  | 1,191,301   | (5) |
| Total  | 3,310,810   |     | \$  | 14.62 | 7,473,064   |     |

- (1) Consists of the weighted-average exercise price of outstanding options, as outstanding restricted stock unit ("RSUs") do not have an exercise price.
- (2) Consists of 1,971,159 shares of common stock issuable upon exercise of outstanding options and 796,969 shares of common stock issuable upon vesting of RSUs under the 2018 Plan, the 2015 Plan, and 1996 Plan.
- (3) Includes 5,520,759 shares available for future grant under the 2018 Plan and 761,004 shares available for future grant under the 2016 ESPP. For the most current offering period that ended on November 14, 2018, 20,340 shares were issued under the 2016 ESPP. The 1996 Plan and the 2015 Plan were terminated for purposes of future grants in May 2015 and June 2018, respectively. Does not include shares available for grant under the 2016 Inducement Plan.
- (4) Consists of 476,780 shares of common stock issuable upon exercise of outstanding options and 65,902 shares of common stock issuable upon vesting of RSUs under the 2016 Inducement Plan.
- (5) Includes shares available for future grant under the 2016 Inducement Plan.

Additional information required in response to this Item 12 is incorporated by reference herein to our Proxy Statement.

# ITEM 13. Certain Relationships and Related Transactions, and Director Independence

The information required in response to this Item 13 is incorporated by reference herein to our Proxy Statement.

#### ITEM 14. Principal Accounting Fees and Services

The information required in response to this Item 14 is incorporated by reference herein to our Proxy Statement.

# PART IV

#### ITEM 15. Exhibits, Financial Statement Schedules

(a)

#### 1. Consolidated Financial Statements

See Index to Consolidated Financial Statements at Item 8 of this report.

#### 2. Financial Statement Schedules

All financial statement schedules required by Item 15(a)(2) have been omitted because they are not applicable or the required information is presented in the Consolidated Financial Statements or Notes thereto.

#### 3. Exhibits

The exhibits required by Item 601 of Regulation S-K are set forth below.

# (b) Exhibits

# INDEX TO EXHIBITS

| Exhibit<br>Number | Exhibit Description  | Form                     | Date of First Filing | Exhibit<br>Number | Filed<br>Herewith |
|-------------------|--|--------------------------|----------------------|-------------------|-------------------|
| 2.1#              | Stock Purchase Agreement by an among HDV Holdings, LLC, Blucora, Inc., Project Baseball Sub, Inc. and HDV Holdings, Inc., dated October 14, 2015                             | 8-K                      | October 14, 2015     | 10.1              |                   |
| 2.2#              | Asset Purchase Agreement between Blucora, Inc., Infospace LLC, OpenMail LLC and InfoSpace Holdings LLC dated July 1, 2016  | 8-K                      | July 5, 2016         | 2.1               |                   |
| 2.3#              | Stock Purchase Agreement between Blucora, Inc., Monoprice Holdings, Inc. and YFC-Boneagle Electric Co., LTD, dated November 14, 2016   | 8-K                      | November 15, 2016    | 2.1               |                   |
| 3.1               | Restated Certificate of Incorporation, as filed with the Secretary of the State of Delaware on August 10, 2012   | 8-K (No. 000-<br>25131)  | August 13, 2012      | 3.1               |                   |
| <u>3.2</u>        | Certificate of Amendment to the Restated Certificate of Incorporation of Blucora, Inc. filed with the Secretary of State of Delaware on June 1, 2017                         | 8-K                      | June 5, 2017         | 3.1               |                   |
| 3.3               | Certificate of Amendment to the Restated Certificate of Incorporation of Blucora, Inc. filed with the Secretary of State of Delaware on June 8, 2018                         | 8-K                      | June 8, 2018         | 3.1               |                   |
| <u>3.4</u>        | Amended and Restated Bylaws of Blucora, Inc.   | 8-K                      | February 28, 2017    | 3.2               |                   |
| 10.1*             | Restated 1996 Flexible Stock Incentive Plan, as amended and restated effective as of June 5, 2012  | S-8 (No. 333-<br>198645) | September 8, 2014    | 99.1              |                   |
| 10.2*             |  | DEF 14A                  | April 25, 2016       | App-              |                   |
|                   | Blucora, Inc. 2015 Incentive Plan, as Amended and Restated   |                          |                      | endix A           |                   |
| 10.3*             | Form of Blucora, Inc. 2015 Incentive Plan Nonqualified Stock Option Grant Notice   | 10-Q                     | July 30, 2015        | 10.2              |                   |
| 10.4*             | Form of Blucora, Inc. 2015 Incentive Plan Restricted Stock Unit Grant Notice   | 10-Q                     | July 30, 2015        | 10.3              |                   |
| 10.5*             | Form of Nonqualified Stock Option Agreement for Executive Officers under the Blucora, Inc. 2015 Incentive Plan, as amended and restated                                      | 8-K                      | February 23, 2018    | 10.2              |                   |
| 10.6*             | Form of Time-Based Restricted Stock Unit Agreement for Executive Officers under the Blucora, Inc. 2015 Incentive Plan, as amended and restated                               | 8-K                      | February 23, 2018    | 10.3              |                   |
| 10.7*             | Form of Performance-Based Restricted Stock Unit Agreement for Executive Officers under the Blucora, Inc. 2015 Incentive Plan, as amended and restated                        | 8-K                      | February 23, 2018    | 10.4              |                   |
| 10.8*             | Form of Nonqualified Stock Option Grant Notice and Agreement for Nonemployee Directors under the Blucora, Inc. 2015 Incentive Plan   | 10-Q                     | April 28, 2016       | 10.3              |                   |
| 10.9*             | Form of Nonqualified Stock Option Grant Notice and Agreement for Nonemployee Chairman of the Board under the Blucora, Inc. 2015 Incentive Plan                               | 10-Q                     | April 28, 2016       | 10.4              |                   |
| 10.10*            | Form of Director Restricted Stock Unit Grant Notice and Award Agreement for Initial Grants to New Directors under the Amended and Restated Blucora, Inc. 2015 Incentive Plan | 10-Q                     | July 27, 2017        | 10.3              |                   |
| 10.11*            | Form of Director Restricted Stock Unit Grant Notice and Award Agreement for Annual Grants to Directors under the Amended and Restated Blucora, Inc. 2015 Incentive Plan      | 10-Q                     | July 27, 2017        | 10.4              |                   |
| 10.12*            | Blucora, Inc. 2018 Long-Term Incentive Plan  | DEF 14A                  | April 19, 2018       | App-<br>endix A   |                   |
|                   |  | 93                       |                      |                   |                   |

| 10.13* | Form of Nonqualified Stock Option Award Agreement for Executive Officers under the Blucora, Inc. 2018 Long-Term Incentive Plan   |       |                   |       | X |
|--------|--|-------|-------------------|-------|---|
| 10.14* | Form of Time-Based Restricted Stock Unit Award Agreement for Executive Officers under the Blucora, Inc. 2018 Long-Term Incentive Plan  |       |                   |       | X |
| 10.15* | Form of Performance-Based Restricted Stock Unit Award<br>Agreement for Executive Officers under the Blucora, Inc. 2018<br>Long-Term Incentive Plan   |       |                   |       | X |
| 10.16* | Form of Director Restricted Stock Unit Grant Notice and Award Agreement for Initial Grants to New Directors under the Blucora, Inc. 2018 Long-Term Incentive Plan                            | 10-Q  | August 1, 2018    | 10.4  |   |
| 10.17* | Form of Director Restricted Stock Unit Grant Notice and Award<br>Agreement for Annual Grants to Directors under Blucora, Inc.<br>2018 Long-Term Incentive Plan                               | 10-Q  | August 1, 2018    | 10.3  |   |
| 10.18* | Blucora, Inc. 2016 Equity Inducement Plan  | S-8   | January 29, 2016  | 99.1  |   |
| 10.19* | Amendment No. 1 to Blucora, Inc. 2016 Inducement Plan  | S-8   | October 14, 2016  | 99.1  |   |
| 10.20* | Amendment No. 2 to the Blucora, Inc. 2016 Inducement Plan  | 8-K   | May 25, 2018      | 10.1  |   |
| 10.21* | Form of Restricted Stock Unit Grant Notice and Award<br>Agreement for Initial Grants to Newly-Hired Executive Officers<br>Under the Blucora, Inc. 2016 Equity Inducement Plan, as<br>amended | 10-Q  | October 31, 2018  | 10.2  |   |
| 10.22* | Form of Blucora, Inc. 2016 Inducement Plan Nonqualified Stock Option Grant Notice  | 10-K  | February 24, 2016 | 10.41 |   |
| 10.23* | Form of Blucora, Inc. 2016 Inducement Plan Restricted Stock Unit Grant Notice  | 10-K  | February 24, 2016 | 10.42 |   |
| 10.25* | Blucora, Inc. 2018 Annual Incentive Plan   | 8-K   | February 23, 2018 | 10.1  |   |
| 10.26* | Employment Agreement between Blucora, Inc. and John S. Clendening dated March 12, 2016   | 8-K   | March 15, 2016    | 10.1  |   |
| 10.27* | First Amendment to Employment Agreement dated September 5, 2017 between Blucora, Inc. and John S. Clendening   | 8-K   | September 5, 2017 | 10.1  |   |
| 10.28* | Employment Agreement by and between Blucora, Inc. and Sanjay Baskaran dated January 12, 2017   | 8-K/A | January 13, 2017  | 10.1  |   |
| 10.29* | Employment Agreement by and between Blucora, Inc., HD Vest, Inc., and Robert D. Oros dated January 22, 2017  | 8-K   | January 23, 2017  | 10.1  |   |
| 10.30* | Employment Agreement by and between Blucora, Inc. and Ann Bruder dated June 19, 2017   | 10-Q  | July 27, 2017     | 10.2  |   |
| 10.31* | Employment Agreement by and between Blucora, Inc. and Davinder Athwal dated February 14, 2018  | 8-K   | February 15, 2018 | 10.1  |   |
| 10.32* | Employment Agreement by and between Blucora, Inc. and Transient Taylor dated September 18, 2018  | 10-Q  | October 31, 2018  | 10.1  |   |
| 10.33* | Employment Agreement by and between Blucora, Inc. and Todd Mackay dated December 24, 2018  |       |                   |       | X |
| 10.34* | Employment Agreement by and between Blucora, Inc. and<br>Curtis Campbell dated October 12, 2018  |       |                   |       | X |
|        | 9  | 4     |                   |       |   |
|        | $\mathbf{q}$   | 4     |                   |       |   |

| 10.35* | Employment Agreement by and between Blucora, Inc. and Mike Hogan dated October 20, 2018  |                          |                   |                 | X              |
|--------|--|--------------------------|-------------------|-----------------|----------------|
| 10.36* | Separation and Release Agreement by and between Blucora, Inc. and Bob Oros dated October 28, 2018  |                          |                   |                 | X              |
| 10.37* | Amended and Restated Employment Agreement, amended and restated effective January 6, 2015 between Blucora, Inc. and Eric M. Emans  | 8-K                      | January 22, 2015  | 10.1            |                |
| 10.38* | Amendment No. 1 to Amended and Restated Employment<br>Agreement by and between Blucora, Inc. and Eric M. Emans<br>dated January 22, 2016   | 8-K                      | January 22, 2016  | 10.1            |                |
| 10.39* | Amendment No. 2 to Amended and Restated Employment<br>Agreement by and between Blucora, Inc. and Eric M. Emans<br>dated January 6, 2015, as amended  | 10-Q                     | October 27, 2016  | 10.4            |                |
| 10.40* | Consulting Agreement, dated October 25, 2017, between Blucora, Inc. and Eric M. Emans  | 10-Q                     | October 26, 2017  | 10.2            |                |
| 10.41  | Credit Agreement, dated May 22, 2017, among Blucora, Inc., as borrower, and most of its direct and indirect domestic subsidiaries, as guarantors, and Credit Suisse AG, Cayman Islands Branch, as administrative agent and collateral agent, and each lender from time to time a party to the Credit Agreement | 8-K                      | May 23, 2017      | 10.1            |                |
| 10.42  | First Amendment, dated November 28, 2017, among Blucora, Inc., as borrower, and most of its direct and indirect domestic subsidiaries, as guarantors, and Credit Suisse AG, Cayman Islands Branch, as administrative agent and collateral agent, and each lender party to the First Amendment                  | 8-K                      | November 29, 2017 | 10.1            |                |
| 10.43  | Office Lease between Blucora, Inc. and Plaza Center Property LLC dated July 19, 2012   | 10-Q (No. 000-<br>25131) | November 1, 2012  | 10.2            |                |
| 10.44  | First Amendment to Office Lease between Blucora, Inc. and Plaza Center Property LLC dated November 5, 2013   | 10-K                     | February 27, 2014 | 10.8            |                |
| 10.45  | Sublease dated April 13, 2017, by and between Blucora, Inc. and Xevo, Inc. related to that certain Office Lease dated July 13, 2012 by and between Blucora, Inc. and KBS SOR Plaza Bellevue, LLC (as successor to Plaza Property Center LLC)   | 10-Q                     | May 4, 2017       | 10.5            |                |
| 10.46  | Lease Agreement, dated January 28, 2008, by and between 2nd Story Software, Inc., PBI Properties, Larry Kane Investments, L.C., and Swati Dandekar for office space located at 1425 60th Street NE, Suite 300, Cedar Rapids, Iowa  | 10-K (No. 000-<br>25131) | March 9, 2012     | 10.13           |                |
| 10.47  | Amendment to Lease Agreement by and between 2nd Story Software, Inc., PBI Properties, Larry Kane Investments, L.C., and Swati Dandekar for office space located at 1425 60th Street NE, Suite 300, Cedar Rapids, Iowa, dated March 14, 2013  | 10-Q                     | May 2, 2013       | 10.5            |                |
| 10.48* | Blucora, Inc., 2016 Employee Stock Purchase Plan   | DEF 14A                  | April 25, 2016    | App-<br>endix B |                |
| 10.49* | Amendment No. 1 to the Blucora, Inc. Employee Stock<br>Purchase Plan   | 10-Q                     | August 1, 2018    | 99.1            |                |
| 10.50* | Blucora, Inc. Non-Employee Director Compensation Policy  | 8-K                      | June 5, 2017      | 10.1            |                |
| 10.51* | Blucora, Inc. Director Tax-Smart Deferral Plan   |                          | ,                 |                 | X              |
| 10.52* | Blucora, Inc. Executive Officer Tax-Smart Deferral Plan  |                          |                   |                 | X              |
| 21.1   | Subsidiaries of the registrant   |                          |                   |                 | X              |
|        |  |                          |                   |                 | · <del>-</del> |
|        |  | 95                       |                   |                 |                |

95

| <u>23.1</u> | Consent of Ernst & Young LLP, Independent Registered Public Accounting Firm   | X |
|-------------|---|---|
| <u>24.1</u> | Power of Attorney (contained on the signature page hereto)  | X |
| 31.1        | Certification of Principal Executive Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002  | X |
| 31.2        | Certification of Principal Financial Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002  | X |
| <u>32.1</u> | Certification of Principal Executive Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002  | X |
| 32.2        | Certification of Principal Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002  | X |
| 101         | The following financial statements from the Company's 10-K for the fiscal year ended December 31, 2017, formatted in XBRL: (i) Consolidated Balance Sheets, (ii) Consolidated Statements of Comprehensive Income, (iii) Consolidated Statements of Stockholders' Equity, (iv) Consolidated Statements of Cash Flows, and (v) Notes to Consolidated Financial Statements | X |

<sup>\*</sup> Indicates a management contract or compensatory plan or arrangement.

#### (c) Financial Statements and Schedules

See Item 15(a) above.

# ITEM 16. Form 10-K Summary

None.

# **SIGNATURES**

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

# BLUCORA, INC.

By: /s/ John S. Clendening

John S. Clendening

Chief Executive Officer and President

Date: March 1, 2019

### POWER OF ATTORNEY

KNOW ALL PERSONS BY THESE PRESENTS, that each person whose signature appears below constitutes and appoints Davinder Athwal and Ann J. Bruder, jointly and severally, his or her attorneys-in-fact, each with the power of substitution, for him or her in any and all capacities to execute any amendments to this Annual Report on Form 10-K, and to file the same, exhibits thereto and other documents in connection therewith, with the Securities and Exchange Commission, hereby ratifying and confirming all that each of said attorneys-in-fact, or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Exchange Act of 1934, this report has been signed below by the following persons on behalf of the registrant and in the capacities indicated and on the dates indicated.

<sup>#</sup> Schedules and exhibits have been omitted pursuant to Item 601(b)(2) of Regulation S-K. Blucora, Inc. hereby undertakes to furnish supplemental copies of any of the omitted schedules and exhibits upon request by the Securities and Exchange Commission.

| <u>Signature</u>                                  | <u>Title</u>  | <u>Date</u>   |
|---|---|---------------|
| /s/ John S. Clendening John S. Clendening         | President, Chief Executive Officer, and Director<br>(Principal Executive Officer) | March 1, 2019 |
| /s/ Davinder Athwal  Davinder Athwal              | Chief Financial Officer<br>(Principal Financial Officer)                          | March 1, 2019 |
| /s/ John Palmer  John Palmer                      | Vice President - Accounting<br>(Principal Accounting Officer)                     | March 1, 2019 |
| /s/ William L. Atwell William L. Atwell           | Chairman and Director   | March 1, 2019 |
| /s/ Steven Aldrich Steven Aldrich                 | Director  | March 1, 2019 |
| /s/ Lance G. Dunn  Lance G. Dunn                  | Director  | March 1, 2019 |
| /s/ E.Carol Hayles E. Carol Hayles                | Director  | March 1, 2019 |
| /s/ H. McIntyre Gardner H. McIntyre Gardner       | Director  | March 1, 2019 |
| /s/ John MacIlwaine  John MacIlwaine              | Director  | March 1, 2019 |
| /s/ Georganne C. Proctor  Georganne C. Proctor    | Director  | March 1, 2019 |
| /s/ Christopher W. Walters Christopher W. Walters | Director  | March 1, 2019 |
| /s/ Mary S. Zappone Mary S. Zappone               | Director  | March 1, 2019 |

# BLUCORA, INC. 2018 LONG-TERM INCENTIVE PLAN

# **NONQUALIFIED STOCK OPTION GRANT NOTICE**

| TO: (the "Participant" or "you")  |
|---|
| FROM: Blucora, Inc., a Delaware corporation (the "Company")   |
| You are hereby granted by the Company a Stock Option (the " <i>Option</i> ") to purchase shares of the Company's Common Stock (" <i>Shares</i> ") pursuant to the Blucora, Inc. 2018 Long-Term Incentive Plan (the " <i>Incentive Plan</i> ").  |
| The Option is subject to all the terms and conditions set forth in this Nonqualified Stock Option Grant Notice (the "Notice of Grant") and in the Stock Option Agreement attached hereto as Exhibit A (the "Agreement") and the Incentive Plan, each of which are incorporated by reference into this Notice of Grant. Capitalized terms that are not defined in the Notice of Grant shall have the meanings given to them in the Agreement, and if not defined in the Agreement, the meanings given to them in the Incentive Plan. |
| Date of Grant:  |
| Option Number:  |
| Number of Shares:   |
| Exercise Price per Share:   |
| Option Expiration Date:   |
| Vesting Commencement Date:  |
| Type of Option: Nonqualified Stock Option   |
| <b>Vesting and Exercisability Schedule:</b> Except as specifically provided in the Agreement and subject to the restrictions and conditions set forth in the Incentive Plan, the Option shall vest and become exercisable as follows:   |
|   |

- (i) one-third (1/3) of the Option (rounded down to the nearest whole Share) shall vest and become exercisable on the first anniversary of the Vesting Commencement Date, provided that you are employed by or providing services to the Company or a Related Company on that date;
- (ii) an additional one-third (1/3) of the Option (rounded down to the nearest whole Share) shall vest and become exercisable on the second anniversary of the Vesting Commencement Date, provided that you are employed by or providing services to the Company or a Related Company on that date; and
- (iii) the remaining one-third (1/3) of the Option shall vest and become exercisable on the third anniversary of the Vesting Commencement Date, provided that you are employed by or providing services to the Company or a Related Company on that date.

Vesting will cease upon your Termination of Service and the unvested portion of the Option will immediately terminate. Notwithstanding the foregoing, upon the occurrence of a Termination of Service due to (i) your death or Total and Permanent Disability, to the extent not already vested, the Option shall become fully vested and exercisable as of the date of such Termination of Service; or (ii) your Retirement on or after the first anniversary of the Date of Grant, to the extent not already vested, the Option shall become fully vested and exercisable as of the date of such Termination of Service. For purposes of this Option, the term "Retirement" shall mean your voluntary Termination of Service on or after your attainment of (i) age sixty (60) and five (5) years of service with the Company or any Related Company, (ii) age fifty-five (55) and ten (10) years of service with the Company; provided, however, that if at any time the Committee determines that your Termination of Service should be a Termination of Service for Cause, then your Termination of Service will no longer be due to your Retirement and the Option shall immediately be forfeited.

Additional Terms/Acknowledgment: You acknowledge and agree that the Notice of Grant and the vesting and exercisability schedule set forth herein do not constitute an express or implied promise of your continued engagement as an employee, officer, director or other service provider for the vesting period, for any period, or at all, and shall not interfere with your right or the Company's right to terminate your employment or service relationship with the Company or its Related Companies at any time, with or without Cause. For purposes of this Option, the term "Cause" shall have the meaning set forth in your Employment Agreement (as defined below), provided that if such Employment Agreement does not define such term or no agreement is then in effect, then it shall mean dishonesty, fraud, serious or willful misconduct, unauthorized use or disclosure of confidential information or trade secrets, or conduct prohibited by law (except minor violations), in each case as determined by the Committee, whose determination shall be conclusive and binding.

**Employment Agreement:** If there is a written employment agreement in effect between you and the Company (the "*Employment Agreement*"), then the Option shall be subject to the terms of such Employment Agreement, so long as such Employment Agreement remains in effect (as it may be amended, supplemented or restated from time to time) and the terms set forth in the Employment Agreement are applicable to the Option.

**Committee Decisions/Interpretations:** You hereby agree to accept as binding, conclusive and final all decisions or interpretations of the Committee upon any questions relating to the Incentive Plan and the Option.

\* \* \* \* \* \*

[Remainder of Page Intentionally Left Blank Signature Page Follows.]

By your signature below or electronic acceptance, you agree that the Notice of Grant, the Agreement and the Employment Agreement (if applicable), and the Incentive Plan, constitute your entire agreement with respect to the Option, and except as set forth therein, may not be modified except by means of a writing signed by the Company and you. This Notice of Grant and Agreement may be executed and/or accepted electronically and/or executed in duplicate counterparts, the production of either of which (including a signature or proof of electronic acceptance) shall be sufficient for all purposes for the proof of the binding terms of this Option.

| BLUCORA, INC.  | PARTICIPANT                       |   |
|--|-----------------------------------|---|
|  |                                   |   |
| By:  | Signature  Date:                  | , |
| Attachments: 1. Stock Option Agreement 2. Incentive Plan |                                   |   |
|  | Signature Page to Notice of Grant |   |

# **EXHIBIT A**

# BLUCORA, INC. 2018 LONG-TERM INCENTIVE PLAN

# STOCK OPTION AGREEMENT

- 1. **Grant.** The Company hereby grants to the Participant listed on the Notice of Grant (the "*Participant*") an Option to purchase the number of Shares and at the exercise price as set forth in the Notice of Grant and subject to the terms and conditions in this Stock Option Agreement (this "*Agreement*") and the Incentive Plan. Unless otherwise defined herein, the capitalized terms used herein shall have the meanings given to them in the Notice of Grant, and if not defined in the Notice of Grant, the meanings given to them in the Incentive Plan.
- 2. **Company's Obligation.** Unless and until the Option vests and is exercised, the Participant will have no right to receive Shares under the Option. Prior to actual distribution of Shares pursuant to any vested and exercised Option, such Option will represent an unsecured obligation of the Company.
- 3. **Vesting and Exercisability.** Subject to Paragraph 4 hereof and to any other relevant Incentive Plan provisions, the Option will vest and become exercisable as provided in the Notice of Grant. Any portion of the Option that is vested may be exercised at any time during the period prior to the date the Option terminates. No partial exercise of the Option may be for less than five percent (5%) of the total number of Shares then available under the Option. In no event shall the Company be required to issue fractional Shares.
- 4. **Termination of Option**. The unvested portion of the Option will terminate automatically and without further notice immediately upon the Participant's Termination of Service (voluntary or involuntary). The vested portion of the Option will terminate automatically and without further notice on the earliest of the dates set forth below:
- a. three (3) months after the Participant's Termination of Service for any reason other than Retirement, death or Total and Permanent Disability;
- b. one (1) year after the Participant's Termination of Service by reason of Retirement, death or Total and Permanent Disability;
- c. immediately upon notification to the Participant of the Participant's Termination of Service for Cause, unless the Committee determines otherwise. If the Participant's employment or service relationship is suspended pending an investigation of whether he or she will be terminated for Cause, all of the Participant's rights under the Option likewise will be suspended during the period of investigation. If any facts that would constitute termination for Cause are discovered after the Participant's Termination of Service, any Option the Participant then holds may be immediately terminated by the Committee; or

d. the Option Expiration Date.

### IT IS THE PARTICIPANT'S RESPONSIBILITY TO BE AWARE OF THE DATE ON WHICH THE OPTION TERMINATES.

- 5. **Leave of Absence.** The effect of a Company-approved leave of absence on the terms and conditions of the Option will be determined by the Committee, subject to applicable laws.
- 6. **Method of Exercise.** The Participant may exercise the Option by giving written notice to the Company, in form and substance satisfactory to the Company, which will state the election to exercise the Option, the date of exercise thereof, and the number of Shares for which the Participant is exercising the Option. The written notice must be accompanied by full payment of the exercise price for the number of Shares that are being purchased (plus any employment tax withholding or other tax payment due with respect to the exercise of the Option).
- 7. Form of Payment of Exercise Price. The Participant may pay the Option exercise price, in whole or in part, (a) in cash; (b) by wire transfer or check acceptable to the Company; (c) if permitted by the Committee, having the Company withhold Shares that would otherwise be issued on exercise of the Option that have an aggregate Fair Market Value equal to the aggregate exercise price of the Shares being purchased under the Option; (d) if permitted by the Committee, tendering (either actually or, so long as the Shares are registered under Section 12(b) or 12(g) of the Exchange Act, by attestation) Shares owned by the Participant that have an aggregate Fair Market Value equal to the aggregate exercise price of the Shares being purchased under the Option; (e) unless the Committee determines otherwise and so long as the Shares are registered under Section 12(b) or 12(g) of the Exchange Act, and to the extent permitted by law, by delivery of a properly executed exercise agreement or notice, together with irrevocable instructions to a brokerage firm designated or approved by the Company to promptly deliver to the Company the aggregate amount of proceeds to pay the Option exercise price; or (e) such other consideration as the Committee may permit.
- 8. **Withholding Taxes.** As a condition to the exercise of any portion of the Option, the Participant must make such arrangements as the Company may require for the satisfaction of any federal, state or local withholding tax obligations that may arise in connection with such exercise. The Company may permit or require the Participant to satisfy all or part of the Participant's tax withholding obligations by (a) paying cash to the Company or a Related Company, as applicable, withhold an amount from any cash amounts otherwise due or to become due from the Company or a Related Company, as applicable, to the Participant; (c) having the Company withhold a number of Shares that would otherwise be issued to the Participant having a Fair Market Value equal to the tax withholding obligations; (d) surrendering a number of Shares the Participant already owns having a Fair Market Value equal to the tax withholding obligations; or (e) any combination of (a), (b), (c) or (d) above. The value of the Shares so withheld or tendered may not exceed the employer's minimum required tax withholding rate.

- 9. Limited Transferability; Who May Exercise. The Option may not be sold, assigned, pledged (as collateral for a loan or as security for the performance of an obligation or for any other purpose) or transferred by the Participant or made subject to attachment or similar proceedings otherwise than by will or by the applicable laws of descent and distribution, except to the extent the Participant designates one or more beneficiaries on a Company-approved form who may exercise the Option after the Participant's death. Notwithstanding the foregoing, the Committee, in its sole discretion, may permit the Participant to assign or transfer the Option, subject to such terms and conditions as specified by the Committee. During the Participant's lifetime only the Participant may exercise the Option. The Option may be exercised by the personal representative of the Participant's estate or the beneficiary thereof following the Participant's death.
- 10. **Regulatory Restrictions on Issuance of Shares** Notwithstanding the other provisions of this Agreement, if at any time the Company determines, in its sole discretion, that the listing, registration or qualification of Shares upon any securities exchange or under any state or federal law, or the consent or approval of any governmental regulatory authority is necessary or desirable as a condition to the issuance of Shares to the Participant (or his or her estate), such issuance will not occur unless and until such listing, registration, qualification, consent or approval will have been effected or obtained free of any conditions not acceptable to the Company. The Company shall be under no obligation to the Participant to register for offering or resale or to qualify for exemption under the Securities Act, or to register or qualify under the laws of any state or foreign jurisdiction, any Shares, security or interest in a security paid or issued under, or created by, the Incentive Plan, or to continue in effect any such registrations or qualifications if made.
- 11. **Participant's Representations**. Notwithstanding any of the provisions hereof, the Participant hereby agrees that the Participant will not exercise the Option, and that the Company will not be obligated to issue any Shares to the Participant if the exercise thereof or the issuance of such Shares shall constitute a violation by the Participant or the Company of any provision of any law or regulation of any governmental authority. Any determination in this connection by the Committee shall be final, binding, and conclusive. The obligations of the Company and the rights of the Participant are subject to all applicable laws, rules, and regulations.
- 12. **Investment Representation**. Unless the Shares are issued to the Participant in a transaction registered under applicable federal and state securities laws, the Participant represents and warrants to the Company that all Shares which may be purchased hereunder will be acquired by the Participant for investment purposes for his or her own account and not with any intent for resale or distribution in violation of federal or state securities laws. Unless the Shares are issued to the Participant in a transaction registered under the applicable federal and state securities laws, at the option of the Company, a stop-transfer order against the Shares may be placed on the official stock books and records of the Company, and a legend indicating that such Shares may not be pledged, sold or otherwise transferred, unless an opinion of counsel is provided (concurred in by counsel for the Company) stating that such transfer is not in violation of any applicable law or regulation, may be stamped on stock certificates to ensure exemption from registration. The

Company may require such other action or agreement by the Participant as may from time to time be necessary to comply with the federal, state and foreign securities laws.

- 13. **Binding Agreement.** Subject to the limitation on the transferability of the Option contained herein, this Agreement will be binding upon and inure to the benefit of the parties hereto and their respective heirs, legatees, legal representatives, successors and assigns.
- 14. **No Stockholder Rights.** Neither the Participant nor any person entitled to exercise the Participant's rights in the event of the Participant's death shall have any of the rights of a stockholder with respect to the Shares subject to the Option unless and until the date of issuance under the Incentive Plan of any such Shares upon the exercise of the Option. Except as otherwise provided in Paragraph 15 hereof, no adjustment shall be made for dividends or other rights for which the record date is prior to the issuance of any Shares subject to the Option. The Participant agrees to execute any documents requested by the Company in connection with the issuance of any Shares.
- 15. **Adjustments.** The number of Shares covered by the Option, and the exercise price thereof, shall be subject to adjustment in accordance with Article 11 of the Incentive Plan.
- 16. **Notices.** Any notice which either party hereto may be required or permitted to give to the other shall be in writing and may be delivered personally, by interoffice mail, by fax, by electronic mail or other electronic means, or via a postal service, postage prepaid, to such electronic mail or postal address and directed to such person as the Company may notify the Participant from time to time; and to the Participant at the Participant's electronic mail or postal address as shown on the records of the Company from time to time, or at such other electronic mail or postal address as the Participant, by notice to the Company, may designate in writing from time to time.
- 17. **Committee Authority; Decisions Conclusive and Binding.** The Participant acknowledges that a copy of the Incentive Plan has been made available for his or her review by the Company, and represents that he or she is familiar with the terms and provisions thereof, and hereby accepts the Award subject to all the terms and provisions thereof. The Committee will have the power to interpret this Agreement, the Notice of Grant and the Incentive Plan, and to adopt such rules for the administration, interpretation and application of the Incentive Plan as are consistent therewith and to interpret or revoke any such rules. The Participant hereby agrees to accept as binding, conclusive, and final all decisions of the Committee upon any questions arising under the Incentive Plan, this Agreement or the Notice of Grant. No member of the Committee will be personally liable for any action, determination or interpretation made in good faith with respect to the Incentive Plan, this Agreement or the Notice of Grant.
- 18. **No Effect on Employment or Service Relationship.** Nothing in the Incentive Plan or any Award granted under the Incentive Plan will be deemed to constitute an employment or service contract or confer or be deemed to confer any right for the Participant to continue in the employ or service of, or to continue any other relationship with, the Company or any Related

Company or limit in any way the right of the Company or any Related Company to terminate the Participant's employment or other service relationship at any time, with or without Cause.

- 19. **No Right to Damages.** The Participant will have no right to bring a claim or to receive damages if the Participant is required to exercise the vested portion of the Option within three (3) months (or one (1) year in the case of Retirement, Total and Permanent Disability or death) of the Participant's Termination of Service or if any portion of the Option is cancelled or expires unexercised. The loss of existing or potential profit in the Option will not constitute an element of damages in the event of the Participant's Termination of Service for any reason even if the termination is in violation of an obligation of the Company or a Related Company to the Participant.
- 20. Claims. The Participant's sole remedy for any Claim shall be against the Company, and the Participant shall not have any claim or right of any nature against any Related Company (including, without limitation, any parent, subsidiary or affiliate of the Company) or any stockholder or existing or former director, officer or employee of the Company or any Related Company. The foregoing individuals and entities (other than the Company) shall be third-party beneficiaries of this Agreement for purposes of enforcing the terms of this Paragraph 20.
- 21. **Covenants and Agreements as Independent Agreements**. Each of the covenants and agreements that is set forth in this Agreement shall be construed as a covenant and agreement independent of any other provision of this Agreement. The existence of any claim or cause of action of the Participant against the Company, whether predicated on this Agreement or otherwise, shall not constitute a defense to the enforcement by the Company of the covenants and agreements that are set forth in this Agreement.
- 22. **Section 409A.** The Option is intended to be exempt from the requirements of Section 409A or to satisfy those requirements, and shall be construed accordingly.
- 23. **Governing Law; Venue.** The validity, interpretation, construction and performance of this Agreement shall be governed by the internal substantive laws of the State of Delaware, without reference to any choice-of-law rules. The Participant irrevocably consents to the nonexclusive jurisdiction and venue of the state and federal courts located in Dallas County, the State of Texas.
- 24. **Recovery of Compensation.** In accordance with Section 6.13 of the Incentive Plan, the Company may recoup all or any portion of any Shares paid to the Participant in connection with the Option, as set forth in the Company's clawback policy, if any, approved by the Board from time to time.
- 25. **Conflicting Terms; Incentive Plan Governs.** This Agreement and the Notice of Grant are subject to all terms and provisions of the Incentive Plan. In the event of a conflict between one or more provisions of this Agreement or the Notice of Grant and one or more provisions of the Incentive Plan, the provisions of the Incentive Plan will govern.

- 26. **Entire Agreement**. This Agreement together with the Notice of Grant and the Incentive Plan supersede any and all other prior understandings and agreements, either oral or in writing, between the parties with respect to the subject matter hereof and constitute the sole and only agreements between the parties with respect to the said subject matter. All prior negotiations and agreements between the parties with respect to the subject matter hereof are merged into this Agreement and the Notice of Grant. Each party to this Agreement and the Notice of Grant acknowledges that no representations, inducements, promises, or agreements, orally or otherwise, have been made by any party or by anyone acting on behalf of any party, which are not embodied in this Agreement, the Notice of Grant or the Incentive Plan and that any agreement, statement, or promise that is not contained in this Agreement, the Notice of Grant or the Incentive Plan shall not be valid or binding or of any force or effect. Notwithstanding anything to the contrary contained in the Notice of Grant, this Agreement or in the Incentive Plan, in the event of any conflict between the terms and conditions of the Option as set forth in the Notice of Grant, this Agreement and in the Incentive Plan, as the case may be, and the terms and conditions of the Employment Agreement, the terms and conditions of the Employment Agreement shall govern unless the conflicting provision in the Notice of Grant, this Agreement or in the Incentive Plan, as the case may be, is more favorable to the Participant; in which case, the provision more favorable to the Participant shall govern; provided, however, that notwithstanding the foregoing, in no event shall any extended exercise period set forth in the Employment Agreement modify or extend the Option Expiration Date as set forth in the Notice of Grant.
- 27. **Legal Construction**. In the event that any one or more of the terms, provisions, or agreements that are contained in this Agreement shall be held by a court of competent jurisdiction to be invalid, illegal, or unenforceable in any respect for any reason, the invalid, illegal, or unenforceable term, provision, or agreement shall not affect any other term, provision, or agreement that is contained in this Agreement, and this Agreement shall be construed in all respects as if the invalid, illegal, or unenforceable term, provision, or agreement had never been contained herein.
- 28. **Headings**. The headings that are used in this Agreement are used for reference and convenience purposes only and do not constitute substantive matters to be considered in construing the terms and provisions of this Agreement.
- 29. **Gender and Number**. Words of any gender used in this Agreement shall be held and construed to include any other gender, and words in the singular number shall be held to include the plural, and vice versa, unless the context requires otherwise.
- 30. **Modification**. No change or modification of this Agreement or the Notice of Grant shall be valid or binding upon the parties unless the change or modification is in writing and signed by the parties; provided, however, that the Company may change or modify this Agreement or the Notice of Grant without the Participant's consent or signature if the Company determines, in its sole discretion, that such change or modification is necessary for purposes of compliance with or exemption from the requirements of Section 409A of the Code or any regulations or other guidance

issued thereunder. Notwithstanding the preceding sentence, the Company may amend the Incentive Plan to the extent permitted by the Incentive Plan.

## BLUCORA, INC. 2018 LONG-TERM INCENTIVE PLAN

# **RESTRICTED STOCK UNIT GRANT NOTICE**

(TIME-BASED RESTRICTED STOCK UNITS)

| TO:                    | (the " <i>Participant</i> " or " <i>you</i> ")   |
|------------------------|--|
| FROM: B                | lucora, Inc., a Delaware corporation (the "Company")   |
| Incentive Pla          | eby granted by the Company a Restricted Stock Unit Award (the " <i>Award</i> ") under the Blucora, Inc. 2018 Long-Term an (the " <i>Incentive Plan</i> "). Each restricted stock unit (an " <i>RSU</i> ") subject to the Award has a notional value equivalent to f the Company's Common Stock for purposes of determining the number of shares of Common Stock (the " <i>Shares</i> ") a Award.   |
| in the Restri          | s subject to all the terms and conditions set forth in this Restricted Stock Unit Grant Notice (the " <b>Notice of Grant</b> ") and cted Stock Unit Agreement attached hereto as <b>Exhibit A</b> (the " <b>Agreement</b> ") and the Incentive Plan, each of which are by reference into this Notice of Grant. Capitalized terms that are not defined in the Notice of Grant shall have the ven to them in the Agreement, and if not defined in the Agreement, the meanings given to them in the Incentive Plan. |
| Date of Gra            | nnt:   |
| Award Num              | nber:  |
| Number of Subject to t |  |
| Vesting Co             | mmencement Date:   |
| •                      | <b>nedule:</b> Except as specifically provided in the Agreement and subject to the restrictions and conditions set forth in the an, the RSUs shall vest as follows:  |
| (i)                    | one-third (1/3) of the RSUs (rounded down to the nearest whole unit) shall vest on the first anniversary of the Vesting Commencement Date, provided that you are employed by or providing services to the Company or a Related Company on that date;   |
|                        |  |
|                        |  |

- (ii) an additional one-third (1/3) of the RSUs (rounded down to the nearest whole unit) shall vest on the second anniversary of the Vesting Commencement Date, provided that you are employed by or providing services to the Company or a Related Company on that date; and
- (iii) the remaining one-third (1/3) of the RSUs shall vest on the third anniversary of the Vesting Commencement Date, provided that you are employed by or providing services to the Company or a Related Company on that date.

Vesting will cease upon your Termination of Service and the unvested portion of the Award will immediately terminate.

Notwithstanding the foregoing, to the extent not already vested, upon the occurrence of a Termination of Service due to (a) your death or Total and Permanent Disability, the RSUs shall become fully vested as of the date of such Termination of Service; or (b) your Retirement on or after the first anniversary of the Date of Grant, the RSUs shall become fully vested as of the date of such Termination of Service. For purposes of this Award, the term "*Retirement*" shall mean your voluntary Termination of Service on or after your attainment of (i) age sixty (60) and five (5) years of service with the Company or any Related Company, (ii) age fifty-five (55) and ten (10) years of service with the Company or any Related Company; *provided, however*, that if at any time the Committee determines that your Termination of Service should be a Termination of Service for Cause (as defined below), then your Termination of Service will no longer be due to your Retirement and all RSUs shall immediately be forfeited. For purposes of this Award, the term "*Cause*" shall have the meaning set forth in your Employment Agreement (as defined below), provided that, if such Employment Agreement does not define such term or no such agreement is then in effect, then it shall mean dishonesty, fraud, serious or willful misconduct, unauthorized use or disclosure of confidential information or trade secrets, or conduct prohibited by law (except minor violations), in each case as determined by the Committee, whose determination shall be conclusive and binding.

**Additional Terms/Acknowledgment:** You acknowledge and agree that the Notice of Grant and the vesting schedule set forth herein do not constitute an express or implied promise of your continued engagement as an employee, officer, director or other service provider for the vesting period, for any period, or at all, and shall not interfere with your right or the Company's right to terminate your employment or service relationship with the Company or its Related Companies at any time, with or without Cause.

**Employment Agreement:** If there is a written employment agreement in effect between you and the Company or a Related Company (the "*Employment Agreement*"), then the Award shall be subject to the terms of such Employment Agreement, so long as such Employment Agreement

remains in effect (as it may be amended, supplemented or restated from time to time) and the terms set forth in the Employment Agreement are applicable to the Award.

**Committee Decisions/Interpretations:** You hereby agree to accept as binding, conclusive and final all decisions or interpretations of the Committee upon any questions relating to the Incentive Plan and the Award.

\* \* \* \* \* \* \*

[Remainder of Page Intentionally Left Blank Signature Page Follows.]

By your signature below or electronic acceptance, you agree that the Notice of Grant, the Agreement, the Incentive Plan, and the Employment Agreement (if applicable), constitute your entire agreement with respect to the Award, and except as set forth therein, may not be modified except by means of a writing signed by the Company and you. This Notice of Grant and Agreement may be executed and/or accepted electronically and/or executed in duplicate counterparts, the production of either of which (including a signature or proof of electronic acceptance) shall be sufficient for all purposes for the proof of the binding terms of this Award.

| BLUCORA, INC.   | PARTICIPANT    |
|---|----------------|
|   |                |
|   |                |
|   | Signature      |
| By:   | Date:          |
| Its:  |                |
|   |                |
| Attachmenter  |                |
| Attachments: 1. Restricted Stock Unit Agreement 2. Incentive Plan |                |
|   |                |
| Signature Page to No  | otice of Grant |

#### **EXHIBIT A**

# BLUCORA, INC. 2018 LONG-TERM INCENTIVE PLAN

#### RESTRICTED STOCK UNIT AGREEMENT

- 1. **Grant.** The Company hereby grants to the Participant listed on the Notice of Grant (the "*Participant*") an Award of RSUs, as set forth in the Notice of Grant and subject to the terms and conditions in this Restricted Stock Unit Agreement (this "*Agreement*") and the Incentive Plan. Unless otherwise defined herein, the capitalized terms used herein shall have the meanings given to them in the Notice of Grant, and if not defined in the Notice of Grant, the meanings given to them in the Incentive Plan.
- 2. **Company's Obligation.** Each RSU represents the right to receive a Share on the vesting date. Unless and until the RSUs vest, the Participant will have no right to receive Shares under such RSUs. Prior to actual distribution of Shares pursuant to any vested RSUs, such RSUs will represent an unsecured obligation of the Company.
- 3. **Vesting Schedule.** Subject to Paragraph 4 hereof and to any other relevant Incentive Plan provisions, the RSUs awarded by this Agreement will vest according to the vesting schedule specified in the Notice of Grant. The effect of a Company approved unpaid leave of absence on the terms and conditions of the RSUs will be determined by the Committee, subject to applicable laws.
- 4. **Forfeiture upon Termination of Service.** Except as provided in the Notice of Grant, if the Participant has a Termination of Service for any or no reason prior to vesting, the unvested RSUs awarded by this Agreement will thereupon be forfeited at no cost to the Company.
- 5. **Payment After Vesting.** Subject to Paragraph 21 hereof, any RSUs that vest in accordance with Paragraph 3 will be paid to the Participant (or in the event of the Participant's death, to his or her estate) in Shares on, or as soon as practicable after, the applicable vesting date (but in any event, within sixty (60) days of the date on which the RSUs vest).
- 6. **Withholding Taxes.** As a condition to the payment of any vested RSUs, the Participant must make such arrangements as the Company may require for the satisfaction of any federal, state or local withholding tax obligations that may arise in connection with such payment. The Company may permit or require the Participant to satisfy all or part of the Participant's tax withholding obligations by (a) paying cash to the Company or a Related Company, as applicable, withhold an amount from any cash amounts otherwise due or to become due from the Company or a Related Company, as applicable, to the Participant; (c) having the Company withhold a number of Shares that would otherwise be issued to the Participant having a Fair Market Value equal to the tax withholding obligations; (d) surrendering a number of Shares the Participant already owns having a Fair Market Value equal

to the tax withholding obligations; or (e) any combination of (a), (b), (c) or (d) above. The value of the Shares so withheld or tendered may not exceed the employer's minimum required tax withholding rate.

- 7. **Payments After Death.** Any distribution or delivery to be made to the Participant under this Agreement will, if the Participant is then deceased, be made to the administrator or executor of the Participant's estate. Any such administrator or executor must furnish the Company with (a) written notice of his or her status as transferee and (b) evidence satisfactory to the Company to establish the validity of the transfer and compliance with any laws or regulations pertaining to said transfer.
- 8. **Rights as Stockholder.** Neither the Participant nor any person claiming under or through the Participant will have any of the rights or privileges of a stockholder of the Company in respect of any Shares deliverable hereunder unless and until the date of issuance of any such Shares under the Incentive Plan. Except as otherwise provided in Paragraph 9, no adjustment shall be made for dividends or other rights for which the record date is prior to the issuance of any Shares subject to the Award. The Participant agrees to execute any documents requested by the Company in connection with the issuance of any Shares.
- 9. **Adjustments.** The number of Shares covered by the Award shall be subject to adjustment in accordance with Article 11 of the Incentive Plan.
- 10. **No Effect on Employment or Service Relationship.** Nothing in the Incentive Plan or any Award granted under the Incentive Plan will be deemed to constitute an employment or service contract or confer or be deemed to confer any right for the Participant to continue in the employ or service of, or to continue any other relationship with, the Company or any Related Company or limit in any way the right of the Company or any Related Company to terminate the Participant's employment or other service relationship at any time, with or without Cause.
- 11. **Notices.** Any notice which either party hereto may be required or permitted to give to the other shall be in writing and may be delivered personally, by interoffice mail, by fax, by electronic mail or other electronic means, or via a postal service, postage prepaid, to such electronic mail or postal address and directed to such person as the Company may notify the Participant from time to time; and to the Participant at the Participant's electronic mail or postal address as shown on the records of the Company from time to time, or at such other electronic mail or postal address as the Participant, by notice to the Company, may designate in writing from time to time.
- 12. **Award Is Not Transferable.** Except to the limited extent provided in Paragraph 7, the Award and the rights and privileges conferred hereby may not be transferred, assigned, pledged (as collateral for a loan or as security for the performance of an obligation or for any other purpose) or hypothecated in any way (whether by operation of law or otherwise) and may not be subject to sale under execution, attachment or similar process. Upon any attempt to transfer, assign, pledge, hypothecate or otherwise dispose of the Award, or any right or privilege conferred hereby, or upon

any attempted sale under any execution, attachment or similar process, the Award and the rights and privileges conferred hereby immediately will become null and void.

- 13. **Binding Agreement.** Subject to the limitation on the transferability of the Award contained herein, this Agreement will be binding upon and inure to the benefit of the parties hereto and their respective heirs, legatees, legal representatives, successors and assigns.
- 14. **Regulatory Restrictions on Issuance of Shares.** Notwithstanding the other provisions of this Agreement, if at any time the Company determines, in its sole discretion, that the listing, registration or qualification of Shares upon any securities exchange or under any state or federal law, or the consent or approval of any governmental regulatory authority is necessary or desirable as a condition to the issuance of Shares to the Participant (or his or her estate), such issuance will not occur unless and until such listing, registration, qualification, consent or approval will have been effected or obtained free of any conditions not acceptable to the Company. The Company shall be under no obligation to the Participant to register for offering or resale or to qualify for exemption under the Securities Act, or to register or qualify under the laws of any state or foreign jurisdiction, any Shares, security or interest in a security paid or issued under, or created by, the Incentive Plan, or to continue in effect any such registrations or qualifications if made.
- 15. **Participant's Representations**. Notwithstanding any of the provisions hereof, the Participant hereby agrees that the Company will not be obligated to issue any Shares to the Participant if the issuance of such Shares shall constitute a violation by the Participant or the Company of any provision of any law or regulation of any governmental authority. Any determination in this connection by the Committee shall be final, binding, and conclusive. The obligations of the Company and the rights of the Participant are subject to all applicable laws, rules, and regulations.
- 16. **Investment Representation**. Unless the Shares are issued to the Participant in a transaction registered under applicable federal and state securities laws, the Participant represents and warrants to the Company that all Shares which may be issued hereunder will be acquired by the Participant for investment purposes for his or her own account and not with any intent for resale or distribution in violation of federal or state securities laws. Unless the Shares are issued to the Participant in a transaction registered under the applicable federal and state securities laws, at the option of the Company, a stop-transfer order against the Shares may be placed on the official stock books and records of the Company, and a legend indicating that such Shares may not be pledged, sold or otherwise transferred, unless an opinion of counsel is provided (concurred in by counsel for the Company) stating that such transfer is not in violation of any applicable law or regulation, may be stamped on stock certificates to ensure exemption from registration. The Company may require such other action or agreement by the Participant as may from time to time be necessary to comply with the federal, state and foreign securities laws.
- 17. **Conflicting Terms; Incentive Plan Governs.** This Agreement and the Notice of Grant are subject to all terms and provisions of the Incentive Plan. In the event of a conflict between one or more provisions of this Agreement or the Notice of Grant and one or more provisions of the Incentive Plan, the provisions of the Incentive Plan will govern.

- 18. Committee Authority; Decisions Conclusive and Binding. The Participant acknowledges that a copy of the Incentive Plan has been made available for his or her review by the Company, and represents that he or she is familiar with the terms and provisions thereof, and hereby accepts the Award subject to all the terms and provisions thereof. The Committee will have the power to interpret this Agreement, the Notice of Grant and the Incentive Plan, and to adopt such rules for the administration, interpretation and application of the Incentive Plan as are consistent therewith and to interpret or revoke any such rules (including, but not limited to, the determination of whether or not any RSUs have vested). The Participant hereby agrees to accept as binding, conclusive, and final all decisions of the Committee upon any questions arising under the Incentive Plan, this Agreement or the Notice of Grant. No member of the Committee will be personally liable for any action, determination or interpretation made in good faith with respect to the Incentive Plan, this Agreement or the Notice of Grant.
- 19. **Claims.** The Participant's sole remedy for any Claim shall be against the Company, and the Participant shall not have any claim or right of any nature against any Related Company (including, without limitation, any parent, subsidiary or affiliate of the Company) or any stockholder or existing or former director, officer or employee of the Company or any Related Company. The foregoing individuals and entities (other than the Company) shall be third-party beneficiaries of this Agreement for purposes of enforcing the terms of this Paragraph 19.
- 20. Covenants and Agreements as Independent Agreements. Each of the covenants and agreements that is set forth in this Agreement shall be construed as a covenant and agreement independent of any other provision of this Agreement. The existence of any claim or cause of action of the Participant against the Company, whether predicated on this Agreement or otherwise, shall not constitute a defense to the enforcement by the Company of the covenants and agreements that are set forth in this Agreement.
- 21. **Section 409A.** The Award is intended to be exempt from or comply with the requirements of Section 409A, and shall be construed accordingly. Notwithstanding any other provision of this Agreement, the Notice of Grant, the Incentive Plan or the Employment Agreement to the contrary, with respect to any payments and benefits to which Section 409A applies, if the Participant is a "specified employee," within the meaning of Section 409A, then to the extent necessary to avoid subjecting the Participant to the imposition of any additional tax under Section 409A, amounts that would otherwise be payable during the sixmonth period immediately following the Participant's "separation from service," within the meaning of Section 409A(a)(2)(A)(i), shall not be paid to the Participant during such period, but shall instead be accumulated and paid to the Participant (or, in the event of the Participant's death, the Participant's estate) in a lump sum on the first business day after the earlier of the date that is six months following the Participant's separation from service or the Participant's death.
- 22. **Governing Law; Venue.** The validity, interpretation, construction and performance of this Agreement shall be governed by the internal substantive laws of the State of Delaware, without reference to any choice-of-law rules. The Participant irrevocably consents to the

nonexclusive jurisdiction and venue of the state and federal courts located in Dallas County, the State of Texas.

- 23. **Recovery of Compensation.** In accordance with Section 6.13 of the Incentive Plan, the Company may recoup all or any portion of any Shares or cash paid to the Participant in connection with the Award, as set forth in the Company's clawback policy, if any, approved by the Board from time to time.
- 24. Entire Agreement; Employment Agreement. This Agreement, together with the Notice of Grant and the Incentive Plan supersede any and all other prior understandings and agreements, either oral or in writing, between the parties with respect to the subject matter hereof and constitute the sole and only agreements between the parties with respect to the said subject matter. All prior negotiations and agreements between the parties with respect to the subject matter hereof are merged into this Agreement and the Notice of Grant. Each party to this Agreement and the Notice of Grant acknowledges that no representations, inducements, promises, or agreements, orally or otherwise, have been made by any party or by anyone acting on behalf of any party, which are not embodied in this Agreement, the Notice of Grant or the Incentive Plan and that any agreement, statement, or promise that is not contained in this Agreement, the Notice of Grant or the Incentive Plan shall not be valid or binding or of any force or effect. Notwithstanding anything to the contrary contained in the Notice of Grant, this Agreement or in the Incentive Plan, in the event of any conflict between the terms and conditions of the Award as set forth in the Notice of Grant, this Agreement and in the Incentive Plan, as the case may be, and the terms and conditions of the Employment Agreement, the terms and conditions of the Employment Agreement shall govern unless the conflicting provision in the Notice of Grant, this Agreement or in the Incentive Plan, as the case may be, is more favorable to the Participant; in which case, the provision more favorable to the Participant shall govern.
- 25. **Legal Construction**. In the event that any one or more of the terms, provisions, or agreements that are contained in this Agreement shall be held by a court of competent jurisdiction to be invalid, illegal, or unenforceable in any respect for any reason, the invalid, illegal, or unenforceable term, provision, or agreement shall not affect any other term, provision, or agreement that is contained in this Agreement, and this Agreement shall be construed in all respects as if the invalid, illegal, or unenforceable term, provision, or agreement had never been contained herein.
- 26. **Headings**. The headings that are used in this Agreement are used for reference and convenience purposes only and do not constitute substantive matters to be considered in construing the terms and provisions of this Agreement.
- 27. **Gender and Number**. Words of any gender used in this Agreement shall be held and construed to include any other gender, and words in the singular number shall be held to include the plural, and vice versa, unless the context requires otherwise.
- 28. **Modification**. No change or modification of this Agreement or the Notice of Grant shall be valid or binding upon the parties unless the change or modification is in writing and signed

by the parties; provided, however, that the Company may change or modify this Agreement or the Notice of Grant without the Participant's consent or signature if the Company determines, in its sole discretion, that such change or modification is necessary for purposes of compliance with or exemption from the requirements of Section 409A of the Code or any regulations or other guidance issued thereunder. Notwithstanding the preceding sentence, the Company may amend the Incentive Plan to the extent permitted by the Incentive Plan.

## BLUCORA, INC. 2018 LONG-TERM INCENTIVE PLAN

# **RESTRICTED STOCK UNIT GRANT NOTICE**

# (PERFORMANCE-BASED RESTRICTED STOCK UNITS)

| TO:(the "Participant" or "you")   |
|---|
| FROM: Blucora, Inc., a Delaware corporation (the "Company")   |
| You are hereby granted by the Company a Restricted Stock Unit Award (the " <i>Award</i> ") under the Blucora, Inc. 2018 Long-Term Incentive Plan (the " <i>Incentive Plan</i> "). Each restricted stock unit (an " <i>RSU</i> ") subject to the Award has a notional value equivalent to one share of the Company's Common Stock for purposes of determining the number of shares of Common Stock (the " <i>Shares</i> " subject to the Award.  |
| The Award is subject to all the terms and conditions set forth in this Restricted Stock Unit Grant Notice (the "Notice of Grant") and in the Restricted Stock Unit Agreement attached hereto as Exhibit A (the "Agreement") and the Incentive Plan, each of which are incorporated by reference into this Notice of Grant. Capitalized terms that are not defined in the Notice of Grant shall have the meanings given to them in the Agreement, and if not defined in the Agreement, the meanings given to them in the Incentive Plan. |
| Date of Grant:  |
| Award Number:   |
| Target Number of RSUs  Subject to the Award: (the "Target RSUs");  provided that the actual number of RSUs that are granted and may be earned is up to 200% of the Target RSUs (or  |
| <b>Vesting Schedule:</b> Except as specifically provided in the Agreement and subject to the restrictions and conditions set forth in the Incentive Plan, the RSUs shall vest on the Vesting Date (as defined on <b>Schedule 1</b> to this Notice of Grant, attached hereto, which is incorporated by reference into this Notice of Grant), based upon the achievement of the performance goals set forth on Schedule 1 (the " <b>Performance Vesting Conditions</b> ").  |
|   |

Additional Terms/Acknowledgment: You acknowledge and agree that the Notice of Grant and the vesting schedule set forth herein do not constitute an express or implied promise of your continued engagement as an employee, officer, director or other service provider for the vesting period, for any period, or at all, and shall not interfere with your right or the Company's right to terminate your employment or service relationship with the Company or its Related Companies at any time, with or without Cause. For purposes of this Award, the term "Cause" shall have the meaning set forth in your Employment Agreement (as defined below), provided that, if such Employment Agreement does not define such term or no such agreement is then in effect, then it shall mean dishonesty, fraud, serious or willful misconduct, unauthorized use or disclosure of confidential information or trade secrets, or conduct prohibited by law (except minor violations), in each case as determined by the Committee, whose determination shall be conclusive and binding.

**Employment Agreement:** If there is a written employment agreement in effect between you and the Company (the "*Employment Agreement*"), then the Award shall be subject to the terms of such Employment Agreement, so long as such Employment Agreement remains in effect (as it may be amended, supplemented or restated from time to time) and the terms set forth in the Employment Agreement are applicable to the Award.

**Committee Decisions/Interpretations:** You hereby agree to accept as binding, conclusive and final all decisions or interpretations of the Committee upon any questions relating to the Incentive Plan and the Award.

\* \* \* \* \* \* \*

[Remainder of Page Intentionally Left Blank Signature Page Follows.]

By your signature below or electronic acceptance, you agree that the Notice of Grant, the Agreement, the Incentive Plan, and the Employment Agreement (if applicable), constitute your entire agreement with respect to the Award, and except as set forth therein, may not be modified except by means of a writing signed by the Company and you. This Notice of Grant and Agreement may be executed and/or accepted electronically and/or executed in duplicate counterparts, the production of either of which (including a signature or proof of electronic acceptance) shall be sufficient for all purposes for the proof of the binding terms of this Award.

| BLUCORA, INC.  | PARTICIPANT                       |  |
|--|-----------------------------------|--|
|  |                                   |  |
|  |                                   |  |
| D  | Signature                         |  |
| By:  | Date:                             |  |
| Its:   |                                   |  |
|  |                                   |  |
| Attachments:   |                                   |  |
| 1. Performance Vesting Conditions  |                                   |  |
| <ul><li>2. Restricted Stock Unit Agreement</li><li>3. Incentive Plan</li></ul> |                                   |  |
|  |                                   |  |
|  |                                   |  |
|  | Signature Page to Notice of Grant |  |

# SCHEDULE 1

# BLUCORA, INC. 2018 LONG-TERM INCENTIVE PLAN PERFORMANCE VESTING CONDITIONS

Schedule 1-1

#### **EXHIBIT A**

# BLUCORA, INC. 2018 LONG-TERM INCENTIVE PLAN

#### RESTRICTED STOCK UNIT AGREEMENT

- 1. **Grant.** The Company hereby grants to the Participant listed on the Notice of Grant (the "*Participant*") an Award of RSUs, as set forth in the Notice of Grant and subject to the terms and conditions in this Restricted Stock Unit Agreement (this "*Agreement*") and the Incentive Plan. Unless otherwise defined herein, the capitalized terms used herein shall have the meanings given to them in the Notice of Grant, and if not defined in the Notice of Grant, the meanings given to them in the Incentive Plan.
- 2. **Company's Obligation.** Each RSU represents the right to receive a Share on the vesting date. Unless and until the RSUs vest, the Participant will have no right to receive Shares under such RSUs. Prior to actual distribution of Shares pursuant to any vested RSUs, such RSUs will represent an unsecured obligation of the Company.
- 3. **Vesting Schedule.** Subject to Paragraph 4 hereof and to any other relevant Incentive Plan provisions, the RSUs awarded by this Agreement will vest according to the vesting schedule specified in the Notice of Grant. The effect of a Company approved unpaid leave of absence on the terms and conditions of the RSUs will be determined by the Committee, subject to applicable laws.
- 4. **Forfeiture upon Termination of Service.** Except as provided in the Notice of Grant, if the Participant has a Termination of Service for any or no reason prior to vesting, the unvested RSUs awarded by this Agreement will thereupon be forfeited at no cost to the Company.
- 5. **Payment After Vesting.** Subject to Paragraph 21 hereof, any RSUs that vest in accordance with Paragraph 3 will be paid to the Participant (or in the event of the Participant's death, to his or her estate) in Shares on, or as soon as practicable after, the applicable vesting date (but in any event, within sixty (60) days of the date on which the RSUs vest).
- 6. **Withholding Taxes**. As a condition to the payment of any vested RSUs, the Participant must make such arrangements as the Company may require for the satisfaction of any federal, state or local withholding tax obligations that may arise in connection with such payment. The Company may permit or require the Participant to satisfy all or part of the Participant's tax withholding obligations by (a) paying cash to the Company or a Related Company, as applicable, withhold an amount from any cash amounts otherwise due or to become due from the Company or a Related Company, as applicable, to the Participant; (c) having the Company withhold a number of Shares that would otherwise be issued to the Participant having a Fair Market Value equal to the tax withholding obligations; (d) surrendering a number of Shares the Participant already owns having a Fair Market Value equal

to the tax withholding obligations; or (e) any combination of (a), (b), (c) or (d) above. The value of the Shares so withheld or tendered may not exceed the employer's minimum required tax withholding rate.

- 7. **Payments After Death.** Any distribution or delivery to be made to the Participant under this Agreement will, if the Participant is then deceased, be made to the administrator or executor of the Participant's estate. Any such administrator or executor must furnish the Company with (a) written notice of his or her status as transferee and (b) evidence satisfactory to the Company to establish the validity of the transfer and compliance with any laws or regulations pertaining to said transfer.
- 8. **Rights as Stockholder.** Neither the Participant nor any person claiming under or through the Participant will have any of the rights or privileges of a stockholder of the Company in respect of any Shares deliverable hereunder unless and until the date of issuance of any such Shares under the Incentive Plan. Except as otherwise provided in Paragraph 9, no adjustment shall be made for dividends or other rights for which the record date is prior to the issuance of any Shares subject to the Award. The Participant agrees to execute any documents requested by the Company in connection with the issuance of any Shares.
- 9. **Adjustments.** The number of Shares covered by the Award shall be subject to adjustment in accordance with Article 11 of the Incentive Plan.
- 10. **No Effect on Employment or Service Relationship.** Nothing in the Incentive Plan or any Award granted under the Incentive Plan will be deemed to constitute an employment or service contract or confer or be deemed to confer any right for the Participant to continue in the employ or service of, or to continue any other relationship with, the Company or any Related Company or limit in any way the right of the Company or any Related Company to terminate the Participant's employment or other service relationship at any time, with or without Cause.
- 11. **Notices.** Any notice which either party hereto may be required or permitted to give to the other shall be in writing and may be delivered personally, by interoffice mail, by fax, by electronic mail or other electronic means, or via a postal service, postage prepaid, to such electronic mail or postal address and directed to such person as the Company may notify the Participant from time to time; and to the Participant at the Participant's electronic mail or postal address as shown on the records of the Company from time to time, or at such other electronic mail or postal address as the Participant, by notice to the Company, may designate in writing from time to time.
- 12. **Award Is Not Transferable.** Except to the limited extent provided in Paragraph 7, the Award and the rights and privileges conferred hereby may not be transferred, assigned, pledged (as collateral for a loan or as security for the performance of an obligation or for any other purpose) or hypothecated in any way (whether by operation of law or otherwise) and may not be subject to sale under execution, attachment or similar process. Upon any attempt to transfer, assign, pledge, hypothecate or otherwise dispose of the Award, or any right or privilege conferred hereby, or upon

any attempted sale under any execution, attachment or similar process, the Award and the rights and privileges conferred hereby immediately will become null and void.

- 13. **Binding Agreement.** Subject to the limitation on the transferability of the Award contained herein, this Agreement will be binding upon and inure to the benefit of the parties hereto and their respective heirs, legatees, legal representatives, successors and assigns.
- 14. **Regulatory Restrictions on Issuance of Shares.** Notwithstanding the other provisions of this Agreement, if at any time the Company determines, in its sole discretion, that the listing, registration or qualification of Shares upon any securities exchange or under any state or federal law, or the consent or approval of any governmental regulatory authority is necessary or desirable as a condition to the issuance of Shares to the Participant (or his or her estate), such issuance will not occur unless and until such listing, registration, qualification, consent or approval will have been effected or obtained free of any conditions not acceptable to the Company. The Company shall be under no obligation to the Participant to register for offering or resale or to qualify for exemption under the Securities Act, or to register or qualify under the laws of any state or foreign jurisdiction, any Shares, security or interest in a security paid or issued under, or created by, the Incentive Plan, or to continue in effect any such registrations or qualifications if made.
- 15. **Participant's Representations**. Notwithstanding any of the provisions hereof, the Participant hereby agrees that the Company will not be obligated to issue any Shares to the Participant if the issuance of such Shares shall constitute a violation by the Participant or the Company of any provision of any law or regulation of any governmental authority. Any determination in this connection by the Committee shall be final, binding, and conclusive. The obligations of the Company and the rights of the Participant are subject to all applicable laws, rules, and regulations.
- 16. **Investment Representation**. Unless the Shares are issued to the Participant in a transaction registered under applicable federal and state securities laws, the Participant represents and warrants to the Company that all Shares which may be issued hereunder will be acquired by the Participant for investment purposes for his or her own account and not with any intent for resale or distribution in violation of federal or state securities laws. Unless the Shares are issued to the Participant in a transaction registered under the applicable federal and state securities laws, at the option of the Company, a stop-transfer order against the Shares may be placed on the official stock books and records of the Company, and a legend indicating that such Shares may not be pledged, sold or otherwise transferred, unless an opinion of counsel is provided (concurred in by counsel for the Company) stating that such transfer is not in violation of any applicable law or regulation, may be stamped on stock certificates to ensure exemption from registration. The Company may require such other action or agreement by the Participant as may from time to time be necessary to comply with the federal, state and foreign securities laws.
- 17. **Conflicting Terms; Incentive Plan Governs.** This Agreement and the Notice of Grant are subject to all terms and provisions of the Incentive Plan. In the event of a conflict between one or more provisions of this Agreement or the Notice of Grant and one or more provisions of the Incentive Plan, the provisions of the Incentive Plan will govern.

- 18. Committee Authority; Decisions Conclusive and Binding. The Participant acknowledges that a copy of the Incentive Plan has been made available for his or her review by the Company, and represents that he or she is familiar with the terms and provisions thereof, and hereby accepts the Award subject to all the terms and provisions thereof. The Committee will have the power to interpret this Agreement, the Notice of Grant and the Incentive Plan, and to adopt such rules for the administration, interpretation and application of the Incentive Plan as are consistent therewith and to interpret or revoke any such rules (including, but not limited to, the determination of whether or not any RSUs have vested). The Participant hereby agrees to accept as binding, conclusive, and final all decisions of the Committee upon any questions arising under the Incentive Plan, this Agreement or the Notice of Grant. No member of the Committee will be personally liable for any action, determination or interpretation made in good faith with respect to the Incentive Plan, this Agreement or the Notice of Grant.
- 19. **Claims.** The Participant's sole remedy for any Claim shall be against the Company, and the Participant shall not have any claim or right of any nature against any Related Company (including, without limitation, any parent, subsidiary or affiliate of the Company) or any stockholder or existing or former director, officer or employee of the Company or any Related Company. The foregoing individuals and entities (other than the Company) shall be third-party beneficiaries of this Agreement for purposes of enforcing the terms of this Paragraph 19.
- 20. Covenants and Agreements as Independent Agreements. Each of the covenants and agreements that is set forth in this Agreement shall be construed as a covenant and agreement independent of any other provision of this Agreement. The existence of any claim or cause of action of the Participant against the Company, whether predicated on this Agreement or otherwise, shall not constitute a defense to the enforcement by the Company of the covenants and agreements that are set forth in this Agreement.
- 21. **Section 409A.** The Award is intended to be exempt from or comply with the requirements of Section 409A, and shall be construed accordingly. Notwithstanding any other provision of this Agreement, the Notice of Grant, the Incentive Plan or the Employment Agreement to the contrary, with respect to any payments and benefits to which Section 409A applies, if the Participant is a "specified employee," within the meaning of Section 409A, then to the extent necessary to avoid subjecting the Participant to the imposition of any additional tax under Section 409A, amounts that would otherwise be payable during the sixmonth period immediately following the Participant's "separation from service," within the meaning of Section 409A(a)(2)(A)(i), shall not be paid to the Participant during such period, but shall instead be accumulated and paid to the Participant (or, in the event of the Participant's death, the Participant's estate) in a lump sum on the first business day after the earlier of the date that is six months following the Participant's separation from service or the Participant's death.
- 22. **Governing Law; Venue.** The validity, interpretation, construction and performance of this Agreement shall be governed by the internal substantive laws of the State of Delaware, without reference to any choice-of-law rules. The Participant irrevocably consents to the

nonexclusive jurisdiction and venue of the state and federal courts located in Dallas County, the State of Texas.

- 23. **Recovery of Compensation.** In accordance with Section 6.13 of the Incentive Plan, the Company may recoup all or any portion of any Shares or cash paid to the Participant in connection with the Award, as set forth in the Company's clawback policy, if any, approved by the Board from time to time.
- 24. Entire Agreement; Employment Agreement. This Agreement, together with the Notice of Grant and the Incentive Plan supersede any and all other prior understandings and agreements, either oral or in writing, between the parties with respect to the subject matter hereof and constitute the sole and only agreements between the parties with respect to the said subject matter. All prior negotiations and agreements between the parties with respect to the subject matter hereof are merged into this Agreement and the Notice of Grant. Each party to this Agreement and the Notice of Grant acknowledges that no representations, inducements, promises, or agreements, orally or otherwise, have been made by any party or by anyone acting on behalf of any party, which are not embodied in this Agreement, the Notice of Grant or the Incentive Plan and that any agreement, statement, or promise that is not contained in this Agreement, the Notice of Grant or the Incentive Plan shall not be valid or binding or of any force or effect. Notwithstanding anything to the contrary contained in the Notice of Grant, this Agreement or in the Incentive Plan, in the event of any conflict between the terms and conditions of the Award as set forth in the Notice of Grant, this Agreement and in the Incentive Plan, as the case may be, and the terms and conditions of the Employment Agreement, the terms and conditions of the Employment Agreement shall govern unless the conflicting provision in the Notice of Grant, this Agreement or in the Incentive Plan, as the case may be, is more favorable to the Participant; in which case, the provision more favorable to the Participant shall govern.
- 25. **Legal Construction**. In the event that any one or more of the terms, provisions, or agreements that are contained in this Agreement shall be held by a court of competent jurisdiction to be invalid, illegal, or unenforceable in any respect for any reason, the invalid, illegal, or unenforceable term, provision, or agreement shall not affect any other term, provision, or agreement that is contained in this Agreement, and this Agreement shall be construed in all respects as if the invalid, illegal, or unenforceable term, provision, or agreement had never been contained herein.
- 26. **Headings**. The headings that are used in this Agreement are used for reference and convenience purposes only and do not constitute substantive matters to be considered in construing the terms and provisions of this Agreement.
- 27. **Gender and Number**. Words of any gender used in this Agreement shall be held and construed to include any other gender, and words in the singular number shall be held to include the plural, and vice versa, unless the context requires otherwise.
- 28. **Modification**. No change or modification of this Agreement or the Notice of Grant shall be valid or binding upon the parties unless the change or modification is in writing and signed

by the parties; provided, however, that the Company may change or modify this Agreement or the Notice of Grant without the Participant's consent or signature if the Company determines, in its sole discretion, that such change or modification is necessary for purposes of compliance with or exemption from the requirements of Section 409A of the Code or any regulations or other guidance issued thereunder. Notwithstanding the preceding sentence, the Company may amend the Incentive Plan to the extent permitted by the Incentive Plan.

#### EMPLOYMENT AGREEMENT

This Employment Agreement (this "*Agreement*") is made and entered into effective as of December 24, 2018, by and between Todd C. Mackay (the "*Executive*") and Blucora, Inc. (the "*Company*").

#### **RECITALS**

WHEREAS, the Company currently employs Executive as the Company's EVP of Business Development (the "*Base Role*"), and effective as of December 24, 2018 (the "*Effective Date*"), the Company desires to employ Executive as the Interim CEO of HD Vest, and the Executive desires to serve in such capacity;

NOW THEREFORE, in consideration of the foregoing, the mutual covenants contained herein, the employment of the Executive by the Company, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

#### 1. Certain Definitions

- (a) "Base Role" is Executive's current role of EVP of Business Development.
- (b) "Base Salary" has the meaning set forth in Section 5(a).
- (c) "Board" means the Board of Directors of the Company.
- (d) "Cause" means, as determined by the Board in its reasonable discretion: (i) the Executive's conviction of, or plea of guilty or nolo contendere to, a misdemeanor involving dishonesty, wrongful taking of property, immoral conduct, bribery or extortion or any felony; (ii) willful material misconduct by the Executive in connection with the business of the Company; (iii) the Executive's continued and willful failure to perform substantially his responsibilities to the Company under this Agreement, after written demand for substantial performance has been given by the Board that specifically identifies how the Executive has not substantially performed his responsibilities; (iv) the Executive's improper disclosure of confidential information or other material breach of this Agreement, including the Confidentiality and Non-Competition Agreement; (v) the Executive's material fraud or dishonesty against the Company; (vi) the Executive's willful and material breach of the Company's written code of conduct and business ethics or other material written policy, procedure or guideline in effect from time to time (provided that the Executive was given access to a copy of such policy, procedure or guideline prior to the alleged breach) relating to personal conduct; or (vii) the Executive's willful attempt to obstruct or willful failure to cooperate with any investigation authorized by the Board or any governmental or self-regulatory entity. Any determination of Cause by the Company shall be made by a resolution approved by a majority of the members of the Board, provided that, with respect to Section 1(c)(iii), the Board must give the Executive notice and 60 days to cure the substantial nonperformance.

- (e) "Change of Control" means the occurrence of any of the following:
- (i) any "person" (as defined in Sections 13(d) and 14(d) of the Exchange Act), excluding for this purpose, (A) the Company or any subsidiary of the Company or (B) any employee benefit plan of the Company or any subsidiary of the Company, or any person or entity organized, appointed or established by the Company for or pursuant to the terms of any such plan that acquires beneficial ownership of voting securities of the Company, is or becomes the "beneficial owner" (as defined in Rule 13d-3 under the Exchange Act), directly or indirectly, of securities of the Company representing more than 50% of the combined voting power of the Company's then outstanding securities;
- (ii) consummation of a reorganization, merger or consolidation of the Company, in each case, unless, following such transaction, all or substantially all the individuals and entities who were the beneficial owners of outstanding voting securities of the Company immediately prior to such transaction beneficially own, directly or indirectly, more than 50% of the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors of the company resulting from such transaction (including, without limitation, a company that, as a result of such transaction, owns the Company or all or substantially all the Company's assets either directly or through one or more subsidiaries) in substantially the same proportions as their ownership immediately prior to such transaction of the outstanding voting securities of the Company;
- (iii) any sale or disposition by the Company, in one transaction or a series of related transactions, of all or substantially all the Company's assets;
- (iv) a "Board Change" which, for purposes of this Agreement, shall have occurred if a majority of the seats on the Board are occupied by individuals who were neither (A) nominated by a majority of the Incumbent Directors nor (B) appointed by directors so nominated ("Incumbent Director" means a member of the Board who has been either (1) nominated by a majority of the directors of the Company then in office or (2) appointed by directors so nominated, but excluding, for this purpose, any such individual whose initial assumption of office occurs as a result of either an actual or threatened election contest (as such terms are used in Rule 14a-11 of Regulation 14A promulgated under the Exchange Act) or other actual or threatened solicitation of proxies or consents by or on behalf of a person other than the Board); or
  - (v) an approval by the stockholders of the Company of a complete liquidation or dissolution of the Company.
  - (f) "Code" means the Internal Revenue Code of 1986, as amended.
  - (g) "Compensation Committee" means the Compensation Committee of the Board.
- (h) "Confidentiality and Non-Competition Agreement" means the Confidentiality and Non-Competition Agreement attached hereto as Exhibit A.

- (i) "Disability" means the Executive's inability to perform his employment duties to the Company hereunder, with or without reasonable accommodation, for 180 days (in the aggregate) in any one-year period as determined by an independent physician selected by the Company.
  - (j) "Exchange Act" means the Securities Exchange Act of 1934, as amended.
- (k) "Good Reason" means the occurrence of any of the following without the Executive's express prior written consent: (i) a material reduction of or to the Executive's Base Role duties, authority, responsibilities (a change in reporting relationship alone does not constitute a material reduction); (ii) a material reduction of the Executive's Base Role Salary; (iii) a material reduction of the Executive's Base Role Target Bonus; (iv) a material reduction in the kind or level of employee benefits to which the Executive is entitled that occur within two months prior to or within 12 months following the consummation of a Change of Control, unless similarly situated employees also experience a reduction; (v) a requirement that the Executive relocate his primary residence more than 50 miles from the greater Denver, Colorado area, or his primary work location more than 50 miles from the greater Dallas/Fort Worth, Texas area, or from any work location to which the Company transfers the Executive during the course of his employment and to which such transfer the Executive has consented; (vi) in connection with a Change of Control, the failure of the Company to assign this Agreement to a successor to the Company or the failure of a successor to the Company to explicitly assume and agree to be bound by this Agreement in a writing delivered to the Executive.

Notwithstanding the foregoing, a Good Reason Termination shall not exist unless (x) the Executive delivers written notice to the Company (the "Good Reason Termination Notice") of the existence of the condition which the Executive believes constitutes a Good Reason Termination within 30 days of the initial existence of such condition (which Good Reason Termination Notice specifically identifies such condition), (y) the Company fails to remedy such condition within 30 days after the date on which it receives such notice (the "Good Reason Termination Cure Period"), and (z) the Executive actually terminates employment within 30 days after the expiration of the Good Reason Termination Cure Period.

- (I) "Interim Role" means Executive's interim role as either Interim General Manager of TaxAct or Interim CEO of HD Vest.
- (m) "Release" means a full release of claims against the Company substantially in the form attached hereto as Exhibit B; provided, however, that notwithstanding the foregoing, such Release is not intended to and will not waive the Executive's rights: (i) to indemnification pursuant to any applicable provision of the Company's Bylaws or Certificate of Incorporation, as amended, pursuant to any written indemnification agreement between the Executive and the Company, or pursuant to applicable law; (ii) to vested benefits or payments specifically to be provided to the Executive under this Agreement or any Company employee benefit plans or policies; or (iii) respecting any claims the Executive may have solely by virtue of the Executive's status as a stockholder of the Company. The Release also shall not include claims that an employee cannot lawfully release through execution of a general release of claims.
- (n) "Section 409A" means Section 409A of the Code and the Treasury Regulations and official guidance issued in respect of Section 409A of the Code.

(o) "Target Bonus" has the meaning set forth in Section 5(b).

#### 2. Duties and Scope of Employment

The Company shall employ the Executive in the Base Role through at least December 31, 2019 and in the Interim Role for so long as Executive's service is required by the Company is such role. In his Interim Role, Executive shall report directly to the Company's President and Chief Executive Officer, and the Company reserves the right to alter such reporting after completion of the Interim Role and Executive's return to the Base Role. The Executive will render such business and professional services in the performance of the Executive's duties, consistent with the Executive's position(s) within the Company, as shall be reasonably assigned to the Executive at any time and from time to time. Upon termination of the Executive's employment for any reason, unless otherwise requested by the Company, the Executive will be deemed to have resigned from all positions held at the Company and its affiliates voluntarily, without any further action by the Executive, as of the end of the Executive's employment, and the Executive, will execute any documents necessary to reflect his resignation.

#### 3. Obligations

While employed hereunder, the Executive will perform his duties ethically, faithfully and to the best of the Executive's ability and in accordance with law and Company policy. The Executive agrees not to actively engage in any other employment, occupation or consulting activity for any direct or indirect remuneration without the express prior written approval of the Company's President and Chief Executive Officer; provided, however, that notwithstanding anything to the contrary in the Confidentiality and Non-Competition Agreement, the Executive may engage in charitable activities so long as such activities do not materially interfere with the Executive's responsibilities to the Company. Executive must receive prior written consent from the President and Chief Executive Officer prior to accepting any role on a board of directors or board of trustees of any public, private or non-profit organization.

#### 4. Agreement Term

Unless earlier terminated as provided herein, the term of this Agreement (the "Agreement Term") shall be for a period commencing on the Effective Date and ending on December 31, 2019, and may be extended thereafter upon the written mutual agreement of the Executive and the Company.

#### 5. Compensation and Benefits

(a) <u>Base Salary</u>. The Company agrees to pay the Executive a base salary (the "*Base Role Salary*") at an annual rate of not less than \$319,300 plus a monthly stipend of \$5,900 during the tenure of the Interim Role, payable in accordance with the regular payroll practices of the Company, but not less frequently than monthly. The Executive's Base Salary shall be subject to annual review by the Board (or a committee thereof).

- (b) Relocation Expenses. Executive shall be entitled to reimbursement for reasonable relocation expenses actually incurred, including purchase and sale transaction expenses, temporary housing costs and related incidental expenses, in an aggregate amount up to \$250,000, grossed up so that such expenses are tax-neutral to Executive ("Relocation Expenses"); provided, however, that such reimbursement of Relocation Expenses shall be subject to the following: (i) if Executive relocates his family to the Dallas/Fort Worth area within 12 months from the Effective Date, Executive will be eligible for reimbursement of Relocation Expenses for a 18-month period following the Effective Date. If Executive resigns his employment with the Company for any reason, or if Executive is terminated by the Company for cause, and such resignation or termination occurs on or before the two-year anniversary of the Effective Date, Executive will repay the Company all amounts paid to Executive as Relocation Expenses.
- (c) Annual Bonus. The Executive shall be eligible to participate in the Company's bonus and other incentive compensation plans and programs for the Company's senior executives at a level commensurate with his position. The Executive shall have the opportunity to earn an annual target bonus (the "Target Bonus") measured against criteria to be determined by the Board (or a committee thereof) of at least 55% of Base Salary for his Base Role, with an additional stipend incentive of 70% for his Interim Role on a pro rata basis (cumulating to 125% of his total Base and Stipend salary on an annualized basis during his Interim Role). Executive's Target Bonus amount for each of 2018 and 2019 will be pro-rated to reflect the number of days of Executive's base role and Interim role tenure in each of 2018 and 2019. The payout of any 2018 and 2019 bonus will occur following the end of the Company's Executive Bonus Plan (the "Plan") year, which is December 31, 2018 and December 31, 2019, respectively; and will be paid in accordance with the terms and conditions of the Plan. The Company reserves the right to change the Plan at any time at its discretion.
- (d) <u>Equity Awards</u>. The Executive will be eligible to participate in the Company's long-term equity incentive programs extended to senior executives of the Company generally at levels commensurate with the Executive's Base Role, which participation and levels shall be determined by the Board (or a committee thereof) in its sole discretion. Notwithstanding the foregoing, on the effective date of his transition to Interim CEO of HD Vest, Executive will receive a stipend award in recognition of his service as Interim General Manager of TaxAct in the amount of \$440,000 worth of shares, subject to a two-year cliff vest. Further, upon successful completion of his service as Interim CEO of HD Vest, Executive will receive a stipend award of equity, similar to that which is typically awarded to the CEO of HD Vest, in recognition of his interim service. Such award shall be made on a pro-rate basis to reflect his actual number of months of interim service and shall contain a two-year cliff vest.
- (e) <u>Benefits</u>. The Executive and his eligible dependents shall be eligible to participate in the employee benefit plans that are available or that become available to other employees of the Company, with the adoption or maintenance of such plans to be in the discretion of the Company, subject in each case to the generally applicable terms and conditions of the plan or program in question and to the determination of any committee administering such plan or program. Such benefits shall include participation in the group medical, life, disability, and retirement plans that are made generally available to employees of the Company, and any supplemental plans available

to senior executives of the Company from time to time. The Company reserves the right to change or terminate its employee benefit plans and programs at any time.

(f) <u>Expenses</u>. The Company shall reimburse the Executive for reasonable business expenses incurred by the Executive in the furtherance of or in connection with the performance of the Executive's duties hereunder, in accordance with the Company's expense reimbursement policy as in effect from time to time.

#### 6. <u>Termination of Employment</u>

- (a) <u>General Provisions</u>. This Agreement and the Executive's employment with the Company may be terminated by either the Executive or the Company at will at any time with or without Cause; provided, however, that the parties' rights and obligations upon such termination during the Agreement Term shall be as set forth in applicable provisions of this Agreement.
- (b) Termination by Company Without Cause or Good Reason Termination Prior to December 31, 2019. In the event of any termination of the Executive's employment with by Company prior to December 31, 2019, (i) the Company shall pay the Executive a severance payment in an amount equal to two times the Executive's Base Role Base Salary in effect as of the date of such termination (the "Termination Date"); (ii) the Company shall pay the Executive any unpaid pro rata bonus at the Target rate of his Base Role compensation as described in Section 5(c), to the extent earned through the Termination Date; and (iii) a lump-sum payment in an amount equal to the monthly COBRA premium in effect under Blucora's group health plan as of your termination date for the coverage in effect under such plan for you (and your spouse and dependent children) on such date, multiplied by 12, which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days following your termination date (but in any event, no later than March 15 of the calendar year immediately following the calendar year that includes your termination date); and (iv) following submission of proper expense reports by the Executive, the Company shall reimburse the Executive for all expenses reasonably and necessarily incurred by the Executive in connection with the business of the Company through the Termination Date (collectively, the "Accrued Obligations"). The Accrued Obligations shall be paid promptly upon termination and within the period of time mandated by applicable law (but, in any event, within 30 days after the Termination Date). The Accrued Obligations paid or provided pursuant to this Section 6(b) shall be in addition to the payments and benefits, if any, to be provided to the Executive upon his termination of employment pursuant to Section 6(c), 6(d), 6(e), or 6(f) as applicable. Except as expressly stated above or as required by law or this Agreement, the Executive shall receive no further compensation in any form other than as set forth in this Section 6(b).
- (c) <u>Termination by Company Without Cause or Good Reason Termination after December 31, 2019</u>. If, other than in connection with a Change of Control as described in Section 6(d), the Executive's employment with the Company is terminated by the Company without Cause or the Executive terminates employment with the Company under circumstances constituting a Good Reason Termination after December 31, 2019, then subject to Section 6(g), the Executive shall receive the following payments and benefits:

- (i) a severance payment in an amount equal to one times the Executive's Base Salary in effect as of the Termination Date (or if the Executive terminates employment under circumstances constituting a Good Reason Termination due to a material reduction of the Executive's Base Salary, in effect immediately prior to such reduction) (less applicable withholding taxes), which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days following the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date), in accordance with Section 14(b)(ii); and
- (ii) a lump-sum payment in an amount equal to twelve months of the COBRA premium in effect under the Company's group health plan as of the Termination Date for the coverage in effect under such plan for the Executive (and the Executive's spouse and dependent children) which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days following the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date), in accordance with Section 14(b)(ii); provided, however, that notwithstanding the foregoing or any other provision in this Agreement to the contrary, the Company (or its successor) may unilaterally amend this Section 6(c)(ii) or eliminate the benefit provided hereunder to the extent it deems necessary to avoid the imposition of excise taxes, penalties or similar charges on the Company or any of its subsidiaries, affiliates or successors, including, without limitation, under Section 4980D of the Code.

Notwithstanding any provision to the contrary in any Company equity compensation plan or any outstanding equity award agreement, if, during the Agreement Term, the Executive terminates employment with the Company under circumstances described in Sections 6(b) or 6(c), there shall be no acceleration of vesting or exercisability of any outstanding equity awards or extension of any option post-termination exercise period.

For the avoidance of doubt, under no circumstances will the Executive be entitled to payments and benefits under both this Section 6(c) and Section 6(d).

- (d) Termination of Employment in Connection With a Change of Control. If the Company terminates the Executive's employment without Cause or the Executive terminates employment with the Company for Good Reason (1) on the day of or during the 12-month period immediately following the consummation of a Change of Control or (2) during the 2-month period prior to the consummation of a Change of Control but at the request of any third party participating in or causing the Change of Control or otherwise in connection with the Change of Control, then subject to Section 6(g) and with respect to clause (2), subject to the consummation of such Change of Control, the Executive shall receive the following payments and benefits:
- (i) a severance payment in an amount equal to one times the Executive's Base Salary then in effect as of the Termination Date and his then current Target Bonus amount (or if the Executive terminates employment for Good Reason due to a material reduction of the Executive's Base Salary or Target Bonus, in effect immediately prior to such reduction) (in each case less applicable withholding taxes), which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days following the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the

Termination Date), in accordance with Section 14(b)(ii). For the avoidance of doubt, if Executive is performing in the Interim Role at the time of such Change of Control, he will also receive the Interim Stipends for both base salary and bonus in conjunction with this paragraph);

- (ii) a lump-sum payment in an amount equal to (A) the monthly COBRA premium in effect under the Company's group health plan as of the Termination Date for the coverage in effect under such plan for the Executive (and the Executive's spouse and dependent children) on such date multiplied by (B) 12 (less applicable withholding taxes), which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days following the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date), in accordance with Section 14(b)(ii); provided, however, that notwithstanding the foregoing or any other provision in this Agreement to the contrary, the Company (or its successor) may unilaterally amend this Section 6(d)(ii) or eliminate the benefit provided hereunder to the extent it deems necessary to avoid the imposition of excise taxes, penalties or similar charges on the Company or any of its subsidiaries, affiliates or successors, including, without limitation, under Section 4980D of the Code; and
- (iii) notwithstanding any provision to the contrary in any applicable equity compensation plan or any outstanding equity award agreement, the treatment of the Executive's outstanding equity awards shall be governed solely by the following provisions: (A) all of the Executive's then-outstanding time-vesting equity awards shall fully vest and all restrictions thereon shall lapse, and (B) to the extent vested (including as a result of the acceleration provided under this Section 6(d)(iii)), all of the Executive's outstanding stock options shall remain exercisable until the first to occur of 12 months following the Termination Date and each such stock option's original expiration date.

If a Change of Control is consummated prior to the expiration of the Agreement Term, this Section 6(d) shall apply to a termination of the Executive's employment by the Company without Cause or by the Executive for Good Reason during the 12-month period immediately following the consummation of the Change of Control even if such 12-month period extends past the expiration of the Agreement Term. Moreover, notwithstanding the expiration of the Agreement Term, if a Change of Control is consummated within two months after the expiration of the Agreement Term, then this Section 6(d) shall apply to a termination of the Executive's employment by the Company without Cause or by the Executive for Good Reason (i) on the day of or during the 12-month period immediately following the consummation of the Change of Control or (ii) during the 2-month period prior to the consummation of the Change of Control but at the request of any third party participating in or causing the Change of Control or otherwise in connection with the Change of Control.

For the avoidance of doubt, the payments and benefits described under this Section 6(d) and the Accrued Obligations shall be the only payments and benefits to which the Executive is entitled in the event that the Executive's employment terminates under this Section 6(d).

(e) <u>Death</u>. In the event of the Executive's death while employed hereunder, and subject to Section 6(g), the Executive's beneficiary (or such other person(s) specified by will or the laws of descent and distribution) shall be entitled to receive a lump-sum payment in an amount equal to three months' Base Salary in effect as of the Termination Date (less applicable withholding taxes),

which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days following the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date), in accordance with Section 14(b)(ii).

- (f) <u>Disability</u>. In the event of the Executive's termination of employment with the Company due to Disability, and subject to Section 6(g), the Executive shall be entitled to receive a lump- sum payment in an amount equal to six months' Base Salary in effect as of the Termination Date (less applicable withholding taxes), which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days following the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date), in accordance with Section 14(b)(ii).
- (g) Release and Other Conditions. The payments and benefits described in Sections 6(c) through 6(f) are expressly conditioned on (i) the Executive (or, in the case of the Executive's death, the Executive's representative) signing and delivering (and not revoking thereafter) a Release to the Company (which, in the case of the Executive's death, also releases any claims by the Executive's estate or survivors), which Release is executed, delivered and effective no later than 60 days following the Termination Date and (ii) the Executive continuing to satisfy any obligations to the Company under this Agreement, the Release and the Confidentiality and Non-Competition Agreement that are incorporated herein by reference, and any other agreement(s) between the Executive and the Company. In the event the Release described in Section 6(g)(i) is not executed, delivered and effective by the 60th day after the Termination Date, none of such payments or benefits shall be provided to the Executive.

#### 7. Section 280G

(a) Amount of Payments and Benefits. Notwithstanding anything to the contrary herein, in the event that the Executive becomes entitled to receive or receives any payments, options, awards or benefits (including, without limitation, the monetary value of any noncash benefits and the accelerated vesting of equity-based awards) under this Agreement or under any other plan, agreement or arrangement with the Company or any person affiliated with the Company (collectively, the "Payments"), that may separately or in the aggregate constitute "parachute payments" within the meaning of Section 280G of the Code and the Treasury Regulations promulgated thereunder (or any similar or successor provision) (collectively, "Section 280G") and it is determined that, but for this Section 7(a), any of the Payments will be subject to any excise tax pursuant to Section 4999 of the Code or any similar or successor provision (the "Excise Tax"), the Company shall pay to the Executive either (i) the full amount of the Payments or (ii) an amount equal to the Payments, reduced by the minimum amount necessary to prevent any portion of the Payments from being an "excess parachute payment" (within the meaning of Section 280G) (the "Capped Payments"), whichever of the foregoing amounts results in the receipt by the Executive, on an after-tax basis, of the greatest amount of Payments notwithstanding that all or some portion of the Payments may be subject to the Excise Tax. For purposes of determining whether the Executive would receive a greater after-tax benefit from the Capped Payments than from receipt of the full amount of the Payments, (i) there shall be taken into account any Excise Tax and all applicable federal, state and local taxes

required to be paid by the Executive in respect of the receipt of such payments and (ii) such payments shall be deemed to be subject to federal income taxes at the highest rate of federal income taxation applicable to individuals that is in effect for the calendar year in which the payments and benefits are to be paid, and state and local income taxes at the highest rate of taxation applicable to individuals in the state and locality of the Executive's residence on the effective date of the relevant transaction described under Section 280G(b) (2)(A)(i) of the Code, net of the maximum reduction in federal income taxes that could be obtained from deduction of such state and local taxes (as determined by assuming that such deduction is subject to the maximum limitation applicable to itemized deductions under Section 68 of the Code and any other limitations applicable to the deduction of state and local income taxes under the Code).

- (b) <u>Computations and Determinations</u>. All computations and determinations called for by this Section 7 shall be made by an independent accounting firm or independent tax counsel appointed by the Company (the "*Tax Counsel*"), and all such computations and determinations shall be conclusive and binding on the Company and the Executive. For purposes of such calculations and determinations, the Tax Counsel may rely on reasonable, good faith interpretations concerning the application of Sections 280G and 4999 of the Code. The Tax Counsel shall submit its determination and detailed supporting calculations to both the Executive and the Company within 15 days after receipt of a notice from either the Company or the Executive that the Executive may receive payments which may be considered "parachute payments." The Company and the Executive shall furnish to the Tax Counsel such information and documents as the Tax Counsel may reasonably request in order to make the computations and determinations called for by this Section 7. The Company shall bear all costs that the Tax Counsel may reasonably incur in connection with the computations and determinations called for by this Section 7.
- (c) <u>Reduction Methodology</u>. In the event that Section 7(a) applies and a reduction is required to be applied to the Payments thereunder, the Payments shall be reduced by the Company in its reasonable discretion in the following order: (i) reduction of any Payments that are subject to Section 409A on a pro-rata basis or such other manner that complies with Section 409A, as determined by the Company, and (ii) reduction of any Payments that are exempt from Section 409A.

### 8. No Impediment to Agreement

The Executive hereby represents to the Company that the Executive is not, as of the date hereof, and will not be, during the Executive's employment with the Company, employed under contract, oral or written, by any other person, firm or entity, and is not and will not be bound by the provisions of any restrictive covenant or confidentiality agreement that would constitute an impediment to, or restriction upon, the Executive's ability to enter this Agreement and to perform the duties of the Executive's employment.

#### 9. Confidentiality and Non-Competition Agreement

The Confidentiality and Non-Competition Agreement is incorporated by reference as if set forth fully herein. The Confidentiality and Non-Competition Agreement shall survive the termination of this Agreement and/or the Executive's employment with the Company.

#### 10. Cooperation

Executive hereby agrees to provide Executive's full cooperation, at the request of the Company, with any of the Released Parties in any and all such lawsuits, investigations or other legal, equitable or business matters or proceedings which involve any matters for which Executive worked on or had responsibility during Executive's employment with the Company. Executive also agrees to be available to the Company and its representatives (including attorneys) to provide general advice or assistance as requested by the Company. This includes but is not limited to testifying (and preparing to testify) as a witness in any proceeding or otherwise providing information or reasonable assistance to the Company in connection with any investigation, claim or suit, and cooperating with the Company regarding any investigation, litigation, claims or other disputed items involving the Company that relate to matters within the knowledge or responsibility of Executive. Specifically, Executive agrees (i) to meet with the Company's representatives, its counsel or other designees at reasonable times and places with respect to any items within the scope of this provision; (ii) to provide truthful testimony regarding same to any court, agency or other adjudicatory body; (iii) to provide the Company with immediate notice of contact or subpoena by any non-governmental adverse party (known to Executive to be adverse to the Company or its interests); and (iv) to not voluntarily assist any such non-governmental adverse party or such non-governmental adverse party's representatives. Executive acknowledges and understands that Executive's obligations of cooperation under this Section are not limited in time and may include, but shall not be limited to, the need for or availability for testimony. Executive shall receive no additional compensation for time spent assisting the Company pursuant to this Section other than the compensation and benefits provided for in this Agreement, provided that Executive shall be entitled to be reimbursed for any reasonable out-of-pocket expenses incurred in fulfilling Executive's obligations pursuant to subsections (i) and (ii) above. Notwithstanding the foregoing, nothing in this Section is intended to interfere with Executive's No Interference rights set forth in Section 1(c) of the Confidentiality and Non-Competition Agreement.

#### 11. Arbitration

(a) Executive agrees that any dispute and/or claim between the Company (including without limitation its officers, directors, employees agents or shareholders and its subsidiaries) and Executive that underlies, relates to and/or results from Executive's employment relationship with the Company or the termination of that relationship or any of the terms of this Agreement, except for any dispute or claim arising from or relating to the Confidentiality and Non-Competition Agreement, that cannot be resolved by mutual agreement of the Company and Executive will be submitted to final, binding arbitration to the maximum extent permitted by law in accordance with the National Rules for the Resolution of Employment Disputes of the American Arbitration Association that are then in effect. This arbitration provision includes, but is not limited to, claims of wrongful discharge, infliction of emotional distress, breach of contract (including breach of this Agreement), breach of any covenant of good faith and fair dealing, and claims of retaliation and/or discrimination in violation of any local, state or federal law. Examples of such laws include Title VII of the Civil Rights Act of 1964; the Age Discrimination in Employment Act of 1967; the Americans with Disabilities Act of 1990; and the Family and Medical Leave Act of 1993, and all amendments to each such law as well as the regulations issued thereunder. This arbitration provision

does not affect the Executive's right to pursue worker's compensation or unemployment compensation benefits for which he may be eligible in accordance with state law, nor does it affect the Executive's right to file and/or to cooperate in the investigation of an administrative charge of discrimination.

- (b) Notwithstanding this arbitration provision, either the Executive or the Company may apply to any court of competent jurisdiction for a temporary restraining order, preliminary injunction, or other interim or conservatory relief, as necessary, without breach of this Agreement and without abridgement of the powers of the arbitrator.
- (c) This arbitration provision does not apply to any dispute or claim arising from or relating to the Confidentiality and Non-Competition Agreement.
- (d) The Company, as further consideration for Executive's agreement to arbitrate covered disputes, agrees to pay for the arbitrator's fees and other costs directly associated with the arbitration that would not otherwise be charged if the parties pursued civil litigation in court.

#### 12. Successors; Personal Services

The services and duties to be performed by the Executive hereunder are personal and may not be assigned or delegated. This Agreement shall be binding upon and inure to the benefit of the Company and its successors and assigns, and the Executive and the Executive's heirs and representatives.

#### 13. Notices

Notices and all other communications contemplated by this Agreement shall be in writing and shall be deemed to have been duly given when personally delivered or when mailed by U.S. registered or certified mail, return receipt requested and postage prepaid. In the case of the Executive, mailed notices shall be addressed to the Executive at the home address the Executive most recently communicated to the Company in writing. In the case of the Company, mailed notices shall be addressed to its corporate headquarters, and all notices shall be directed to the attention of its President and Chief Executive Officer.

#### 14. <u>Section 409A</u>

(a) The parties intend that this Agreement and the payments and benefits provided hereunder be exempt from the requirements of Section 409A, to the maximum extent possible, whether pursuant to the short-term deferral exception described in Treasury Regulation Section 1.409A-1(b)(4), the involuntary separation pay plan exception described in Treasury Regulation Section 1.409A-1(b)(9)(iii), or otherwise. To the extent Section 409A is applicable to this Agreement, the parties intend that this Agreement and any payments and benefits thereunder comply with the deferral, payout and other limitations and restrictions imposed under Section 409A. Notwithstanding anything herein to the contrary, this Agreement shall be interpreted, operated and administered in a manner consistent with such intentions.

- (b) Without limiting the generality of the foregoing, and notwithstanding any other provision of this Agreement to the contrary:
- (i) if the Executive is deemed on the date of termination to be a "specified employee" within the meaning of that term under Section 409A, then with regard to any payment that is considered a "deferral of compensation" under Section 409A payable on account of a "separation from service," such payment shall be made on the date which is the earlier of (A) the date that is six months and one day after the date of such "separation from service" of the Executive and (B) the date of the Executive's death (the "*Delay Period*"), to the extent required under Section 409A. Within ten business days following the expiration of the Delay Period, all payments delayed pursuant to this Section 14(b)(i) (whether they would have otherwise been payable in a single sum or in installments in the absence of such delay) shall be paid to the Executive in a lump sum, and all remaining payments due under this Agreement shall be paid or provided in accordance with the normal payment dates specified for those payments in this Agreement;
- (ii) to the extent that any payments or benefits under this Agreement are conditioned on a Release, if the Release is executed and delivered by the Executive to the Company and becomes irrevocable and effective within the specified 60-day post-termination period, then, subject to Section 14(b)(i) and to the extent not exempt under Section 409A, such payments or benefits shall be made or commence on the first payroll date after the date that is 60 days after the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date). If a payment or benefit under this Agreement is conditioned on a Release and such Release is not executed, delivered and effective by the 60th day after the Termination Date, such payment or benefit shall not be paid or provided to the Executive;
- (iii) all expenses or other reimbursements under this Agreement shall be made on or prior to the last day of the taxable year following the taxable year in which such expenses were incurred by the Executive. No such reimbursement or expenses eligible for reimbursement in any taxable year shall in any way affect the expenses eligible for reimbursement in any other taxable year, and the Executive's right to reimbursement shall not be subject to liquidation or exchange for any other benefit;
- (iv) for purposes of Section 409A, the Executive's right to receive a series of installment payments pursuant to this Agreement shall be treated as a right to receive a series of separate and distinct payments. Whenever a payment under this Agreement specifies a payment period with reference to a number of days (e.g., "payment shall be made within 30 days"), the actual date of payment within the specified period shall be within the sole discretion of the Company;
- (v) in no event shall any payment under this Agreement that constitutes a "deferral of compensation" for purposes of Section 409A be offset by any other payment pursuant to this Agreement or otherwise; and
- (vi) to the extent required for purposes of compliance with Section 409A, termination of employment shall not be deemed to have occurred for purposes of any provision of this Agreement providing for the payment of any amounts or benefits upon or following a termination

of employment unless such termination is also a "separation from service" within the meaning of Section 409A, and for purposes of any such provision of this Agreement, references to a "termination," "termination of employment" or like terms shall mean "separation from service."

(c) The Company and the Executive agree to work together in good faith to consider amendments to this Agreement and to take such reasonable actions that may be necessary, appropriate, or desirable to avoid imposition of additional tax or income recognition on the Executive under Section 409A, in each case to the maximum extent permitted by applicable law. Notwithstanding any provision of this Agreement to the contrary, (i) in no event will the Company be liable for any additional tax, interest or penalty that may be imposed on the Executive by Section 409A or damages for failing to comply with Section 409A and (ii) the Executive acknowledges and agrees that the Executive will not have any claim or right of action against the Company or any of its employees, officers, directors or agents in the event it is determined that any payment or benefit provided hereunder does not comply with Section 409A.

#### 15. Miscellaneous Provisions

- (a) <u>Waiver</u>. No provision of this Agreement shall be modified, waived or discharged unless the modification, waiver or discharge is agreed to in writing and signed by the Executive and by an authorized officer of the Company (other than the Executive). No waiver by either party of any breach of, or of compliance with, any condition or provision of this Agreement by the other party shall be considered a waiver of any other condition or provision or of the same condition or provision at another time.
- (b) Entire Agreement. This Agreement (including exhibits) shall supersede and replace all prior agreements or understandings relating to the subject matter hereof, and no agreements, representations or understandings (whether oral or written or whether express or implied) that are not expressly set forth in this Agreement have been made or entered into by either party with respect to the relevant matters hereof. This Agreement may not be modified except expressly in a writing signed by both parties.
- (c) <u>Disclaimer of Reliance</u>. Except for the specific representations expressly made by the Company in this Agreement, Executive specifically disclaims that Executive is relying upon or has relied upon any communications, promises, statements, inducements, or representation(s) that may have been made, oral or written, regarding the subject matter of this Agreement. Executive represents that Executive relied solely and only on Executive's own judgment in making the decision to enter into this Agreement.
- (d) <u>Choice of Law</u>. The validity, interpretation, construction and performance of this Agreement shall be governed by the internal substantive laws of the State of Texas without reference to any choice of law rules.
- (e) <u>Severability</u>. The invalidity or unenforceability of any provision or provisions of this Agreement shall not affect the validity or enforceability of any other provision hereof, which shall remain in full force and effect.

- (f) No Assignment of Benefits. The rights of any person to payments or benefits under this Agreement shall not be made subject to option or assignment, either by voluntary or involuntary assignment or by operation of law, in respect of bankruptcy, garnishment, attachment or other creditor's process, and any action in violation of this Section 15(f) shall be void.
- (g) No Duty to Mitigate. The Executive shall not be required to mitigate the amount of any payment contemplated by this Agreement, nor shall any such payment be reduced by any earnings that the Executive may receive from any other source.
- (h) Employment Taxes. All payments made pursuant to this Agreement will be subject to withholding of all applicable income, employment and other taxes.
- (i) <u>Assignment by Company</u>. The Company may assign its rights under this Agreement to an affiliate (as defined under the Exchange Act), and an affiliate may assign its rights under this Agreement to another affiliate of the Company or to the Company. In the case of any such assignment, the term "Company" when used in a section of this Agreement shall mean the corporation that actually employs the Executive.
- (j) <u>Counterparts</u>. This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together will constitute one and the same instrument.

[Signature Page Follows]

| BLUCORA, INC.              |          |  |
|----------------------------|----------|--|
|                            |          |  |
| By: /s/ Ann J. Bruder      |          |  |
| Name: Ann J. Bruder        |          |  |
| Title: Chief Legal Officer |          |  |
|                            |          |  |
|                            |          |  |
| EXECUTIVE:                 |          |  |
|                            |          |  |
| /s/ Todd Mackay            | <u> </u> |  |
| Todd C. Mackay             |          |  |
|                            |          |  |
|                            |          |  |
|                            |          |  |
|                            |          |  |
|                            | 16       |  |

IN WITNESS WHEREOF, each of the parties has executed this Agreement, in the case of the Company by its duly authorized

officer, as of the day and year first above written.

## **EXHIBIT A**

# CONFIDENTIALITY AND NON-COMPETITION AGREEMENT

This Confidentiality and Non-Competition ("Agreement") is entered into by and between Blucora, Inc, its subsidiaries, affiliates, successors and/or assigns (the "Company") and Todd C. Mackay ("Executive"). The Effective Date of this Agreement is the date of Executive's execution of this Agreement. The Company and Executive shall be referred to herein individually as a "Party" and collectively as the "Parties."

NOW, THEREFORE, in consideration of the mutual covenants set forth herein, Executive's position with the Company, the Confidential Information (defined below), compensation and benefits provided to Executive, and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, and intending to be legally bound hereby, the Parties agree as follows:

- (a) Confidential Information and Executive's Non-Disclosure Agreement.
- (a) <u>Confidential Information</u>. During Executive's employment with the Company, the Company shall provide Executive Confidential Information (defined below), which is not known to the Company's competitors or within the Company's industry generally, which was developed by the Company over a long period of time and/or at its substantial expense, and which is of great competitive value to the Company. For purposes of this Agreement, "Confidential Information" includes all documents or information, in whatever form or medium, concerning or relating to any of the following: all trade secrets and confidential and proprietary information of or relating to the Company, including, but not limited to: (A) financial models, business records, business plans or processes, strategies (including, without limitation, economic and market research selection and analysis strategies and business development and market segment exploitation strategies), tactics, policies, resolutions, processes, inventions, patents, trademarks, trade secrets, know how, patent or trademark applications and other intellectual property, (B) information regarding litigation or negotiations, (C) any marketing information, sales or product plans, prospects and market research data relating to the business, (D) financial information, cost and performance data and any debt arrangements, equity ownership or securities transaction information, (E) technical information, technical drawings and designs, (F) personnel information, personnel lists, resumes, personnel data, organizational structure, compensation and performance evaluations, (G) customer, consumer, consultants or supplier information, including but not limited to any data regarding any current, prospective or former customers, consumers, consultants or suppliers of Company, (H) information regarding the existence or terms of any agreement or relationship between the Company or any of its subsidiaries or affiliates and any other party, (I) information subject to Section 628 of the Fair Credit Reporting Act and any regulations or guidelines thereunder and (J) any other information of whatever nature, including, without limitation, information which gives to the Company or any of its subsidiaries or affiliates an opportunity to obtain an advantage over its competitors who or which do not have access to such information. Confidential Information, whether prepared or compiled by Executive and/or the Company or furnished to Executive during Executive's employment with the Company, shall be the sole and exclusive property of the Company, and none of such Confidential Information or copies thereof, shall be retained by Executive. Executive agrees not to dispute,

contest, or deny any such ownership rights either during or after Executive's employment with the Company. Executive acknowledges that the Company does not voluntarily disclose Confidential Information, but rather takes precautions to prevent dissemination of Confidential Information beyond those employees such as Executive entrusted with such information. Executive further acknowledges that the Confidential Information: (a) is entrusted to Executive because of Executive's position with the Company; and (b) is of such value and nature as to make it reasonable and necessary for Executive to protect and preserve the confidentiality and secrecy of the Confidential Information. Executive acknowledges and agrees that the Confidential Information is proprietary to and a trade secret of the Company and, as such, is a valuable, special and unique asset of the Company, the unauthorized use or disclosure of which will cause irreparable harm, substantial injury and loss of profits and goodwill to the Company. "Confidential Information" does not include any information which is generally available to and known by the public (other than as a result of Executive's breach of this Agreement or any other agreement or obligation to keep such information confidential).

# (b) Non-Disclosure.

(i) Executive agrees to preserve and protect the confidentiality of all Confidential Information. Executive agrees that during the period of Executive's employment with the Company and at any time thereafter (regardless of the reason for Executive's separation or termination of employment): (A) Executive shall hold all Confidential Information in the strictest confidence, take all reasonable precautions and steps to safeguard all Confidential Information and prevent its wrongful use by or wrongful or inadvertent disclosure or dissemination to any unauthorized person or entity, and follow all policies and procedures of the Company protecting or regarding the Confidential Information; and (B) without prior written authorization of the Company, Executive shall not, directly or indirectly, use for Executive's own account, use in any way or for any other purpose, disclose to anyone, publish, exploit, destroy, copy or remove from the offices of the Company, nor solicit, allow or assist another person or entity to use, disclose, publish, exploit, destroy, copy or remove from the offices of the Company, any Confidential Information or part thereof, except: (1) as permitted in the proper performance of Executive's duties for the Company; (2) as permitted in the ordinary course of the Company's business for the benefit of the Company; or (3) as otherwise permitted or required by law. Executive shall immediately notify the Company if Executive learns of or suspects any actual or potential unauthorized use or disclosure of Confidential Information concerning the Company. Further, the Executive shall not, directly or indirectly, use the Company's Confidential Information to: (1) call upon, solicit business from, attempt to conduct business with, conduct business with, interfere with or divert business away from any customer. client, service provider, supplier or vendor of the Company with whom or which the Company conducted business; and/or (2) recruit, solicit, hire or attempt to recruit, solicit, or hire, directly or by assisting others, any persons employed by the Company. In the event Executive is subpoenaed, served with any legal process or notice, or otherwise requested to produce or divulge, directly or indirectly, any Confidential Information by any entity, agency or person in any formal or informal proceeding including, but not limited to, any interview, deposition, administrative or judicial hearing and/or trial, except where prohibited by law, Executive should immediately notify the Company and deliver a copy of the subpoena, process, notice or other request to the Company as promptly as possible, but under no circumstances more than ten (10) days following Executive's

receipt of same; provided, however, Executive is not required to notify the Company or provide a copy of the subpoena, process, notice or other request where Executive is permitted to make such disclosure of Confidential Information pursuant to this Agreement or applicable law or regulation, as set forth in Section 1(c) and Section 1(d).

- Subject to Section 1(b)(iii), Executive agrees that Executive will not use or disclose any confidential, proprietary or trade secret information belonging to any former employer or third party, and Executive will not bring onto the premises of the Company or onto any Company property, any confidential, proprietary or trade secret information belonging to any former employer or third party without such third party's written consent. Executive acknowledges that that the Company has specifically instructed Executive not to disclose to the Company, use, or induce the Company to use, any confidential, proprietary or trade secret information belonging to any previous employer or others.
- (iii) During Executive's employment, the Company will receive from third parties their confidential and/or proprietary information, subject to a duty on the Company's part to maintain the confidentiality of and to use such information only for certain limited purposes. Executive agrees to hold all such confidential or proprietary information in the strictest confidence and not to disclose it to any person or organization or to use it except as necessary in the course of Executive's employment with the Company and in accordance with the Company's agreement with such third party.
- (iv) Except in the proper performance of Executive's duties and responsibilities, Executive agrees that Executive shall not remove, destroy, deface, damage or delete any Property of the Company. For purposes of this Agreement, the term "Property" means all property or information, in whatever form or media, and all copies thereof whether or not the original was deleted or destroyed, of the Company, including, without limitation, any Confidential Information, software, hardware, including any and all Company-issued equipment, devices, cellular telephones, tablets, computers, laptops, hard drives, keys, access cards, access codes or passwords belonging to the Company, databases, files, records, reports, memoranda, research, plans, proposals, lists, forms, drawings, specifications, notebooks, manuals, correspondence, materials, e-mail, electronic or magnetic recordings or data, and any other physical or electronic documents that Executive receives from or sends to any employee of the Company, that Executive copies from the files or records of the Company, or that Executive otherwise has access to during Executive's employment.
- (c) No Interference. Notwithstanding any other provision of this Agreement, (i) Executive may disclose Confidential Information when required to do so by a court of competent jurisdiction, by any governmental agency having authority over Executive or the business of the Company or by any administrative body or legislative body (including a committee thereof) with jurisdiction to order Executive to divulge, disclose or make accessible such information; and (ii) nothing in this Agreement is intended to interfere with Executive's right to (A) report possible violations of state or federal law or regulation to any governmental or law enforcement agency or entity; (B) make other disclosures that are protected under the whistleblower provisions of state or federal law or regulation; (C) file a claim or charge with any governmental agency or entity; or (D) testify, assist, or participate in an investigation, hearing, or proceeding conducted by any

governmental or law enforcement agency or entity, or any court. For purposes of clarity, in making or initiating any such reports or disclosures or engaging in any of the conduct outlined in subsection (ii) above, Executive may disclose Confidential Information to the extent necessary to such governmental or law enforcement agency or entity or such court, need not seek prior authorization from the Company, and is not required to notify the Company of any such reports, disclosures or conduct.

(d) <u>Defend Trade Secrets Act</u>. Executive is hereby notified in accordance with the Defend Trade Secrets Act of 2016 that Executive will not be held criminally or civilly liable under any federal or state trade secret law for the disclosure of a trade secret that is made in confidence to a federal, state, or local government official, either directly or indirectly, or to an attorney solely for the purpose of reporting or investigating a suspected violation of law, or is made in a complaint or other document that is filed under seal in a lawsuit or other proceeding.

# (e) <u>Inventions</u>.

- (i) Prior Inventions Retained and Licensed. In Exhibit A-1 to this Agreement, Executive has provided a list describing all Inventions (defined below) that Executive: (A) conceived, created, developed, made, reduced to practice or completed, either alone or with others, prior to Executive's employment with the Company; (B) claims a proprietary right or interest in; and (C) does not assign to the Company hereunder (collectively referred to as the "Prior Inventions"). If no such list is attached, Executive represents that there are no such Prior Inventions. Executive understands and agrees that the Company makes no attempt to verify Executive's claim of ownership to any of the Prior Inventions. Executive agrees that Executive shall not incorporate in any work that Executive performs for the Company any Prior Inventions or any of the technology described in any Prior Inventions. Nonetheless, if in the course of Executive's employment with the Company, Executive incorporates Prior Inventions into a product, service, process or machine of the Company, Executive hereby grants and shall be deemed to have granted the Company a nonexclusive, royalty-free, irrevocable, sublicensable, transferable, perpetual, and worldwide license to make, have made, modify, use, import, reproduce, distribute, prepare and have prepared derivative works of, offer to sell, sell and otherwise exploit such Prior Inventions. For purposes of this Agreement, the term "Inventions" means all tangible and intangible materials, work product, information, methods, designs, computer programs, software, databases, formulas, models, prototypes, reports, discoveries, ideas, improvements, know-how, compositions of matter, processes, photographs, drawings, illustrations, sketches, developments, and all related intellectual property, including inventions, original works of authorship, moral rights, mask works, trade secrets and trademarks.
- (ii) <u>Assignment of Inventions</u>. During Executive's employment with the Company and following the termination of Executive's employment for any reason, Executive agrees that Executive shall promptly make full written disclosure to the Company, shall hold in trust for the sole right and benefit of the Company, and hereby assigns and shall be deemed to have assigned to the Company or its designee, all of Executive's right, title, and interest in and to any and all Inventions that have been or may be conceived, created, developed, completed, reduced to practice or otherwise made by Executive, solely or jointly with others, during the period of

Executive's employment with the Company which (A) relate in any manner to the existing or contemplated business, work, or investigations of the Company; (B) are suggested by, result from, or arise out of any work that Executive may do for or on behalf of the Company; (C) result from or arise out of any Confidential Information that may have been disclosed or otherwise made available to Executive as a result of duties assigned to Executive by the Company; or (D) are otherwise made through the use of the time, information, equipment, facilities, supplies or materials of the Company, even if developed, conceived, reduced to practice or otherwise made during other than working hours (collectively referred to as "Company Inventions"). Executive further acknowledges that all original works of authorship that are made by Executive (solely or jointly with others) within the scope of Executive's employment with the Company and that are protectable by copyright are "Works Made for Hire," as that term is defined in the United States Copyright Act. Executive understands and agrees that the decision whether or not to commercialize or market any Company Inventions is within the Company's sole discretion and for the Company's sole benefit, and that no royalty will be due to Executive as a result of the Company's efforts to commercialize or market any such Company Innovation.

- (iii) <u>Maintenance of Records</u>. Executive agrees to keep and maintain adequate and current hard-copy and electronic records of all Company Inventions. The records will be available to and remain the sole property of the Company during Executive's employment with the Company and at all times thereafter.
- (f) Patent and Copyright Registrations. Executive agrees to assist the Company or its designee, at the Company's expense, in every proper way to secure the Company's rights in Company Inventions in any and all countries, including the disclosure to the Company of all pertinent information and data with respect thereto, the execution of all applications, specifications, oaths, assignments, affidavits, and all other instruments which the Company shall deem necessary in order to apply for and obtain such rights and in order to assign and convey to the Company and/or its successors, assigns and nominees, the sole and exclusive rights, title and interest in and to such Company Inventions. Executive further agrees that Executive's obligation to execute or cause to be executed, when it is in Executive's power to do so, any such instrument or papers shall continue after the termination of this Agreement. Executive hereby appoints the General Counsel of the Company as Executive's attorney-in-fact to execute documents on Executive's behalf for this purpose. Executive agrees that this appointment is coupled with an interest and will not be revocable.
- (g) Return of Company Property. Upon request by the Company or upon the termination of Executive's employment for any reason, Executive shall immediately return and deliver to the Company any and all Property, including, without limitation, Confidential Information, software, hardware, including any and all Company-issued equipment, devices, cellular telephones, tablets, computers, laptops, hard drives, keys, access cards, access codes or passwords, databases, files, documents, records, reports, memoranda, research, plans, proposals, lists, papers, books, forms, drawings, specifications, notebooks, manuals, correspondence, materials, e-mail, electronic or magnetic recordings or data, including all copies thereof (in electronic or hard copy format), which belong to the Company or which relate to the Company's business and which are in Executive's possession, custody or control, whether prepared by Executive or others. Executive further agrees that after Executive provides such Property to the Company, Executive will immediately destroy

any information or documents, whether prepared by Executive or others, containing or reflecting any Confidential Information or relating to the business of the Company from any computer, cellular phone or other digital or electronic device in Executive's possession, custody or control, and Executive shall certify such destruction in writing to the Company. Upon request by the Company, Executive shall provide such computer, cellular phone or other digital or electronic device to the Company or the Company's designee for inspection to confirm that such information and documents have been destroyed. If at any time after the termination of Executive's employment for any reason, Executive or the Company determines that Executive has any Property in Executive's possession, custody or control, Executive shall immediately return all such Property, including all copies and portions thereof, to the Company.

Restrictive Covenants. In consideration for (i) the Company's promise to provide Confidential Information; (ii) the substantial economic investment made by the Company in the Confidential Information and goodwill of the Company, and/or the business opportunities disclosed or entrusted to Executive; (iii) access to the customers and clients of the Company; and (iv) the Company's employment of Executive in an executive position and the compensation and other benefits provided by the Company to Executive (including the consideration provided for in the Incentive Retention Letter), to protect the Confidential Information and business goodwill of the Company, Executive agrees to the following restrictive covenants.

(a) Non-Competition. Executive agrees that during the Executive's employment with the Company and for a period of twelve (12) months after the Executive's employment terminates for any reason (the "Restricted Period"), other than in connection with Executive's performance of his duties for the Company, Executive shall not, and shall not use any Confidential Information to, without the prior written consent of an executive officer of the Company, directly or indirectly, either individually or as a principal, partner, stockholder, manager, agent, consultant, contractor, distributor, employee, direct lender, individual investor, or as a director or officer of any corporation or association, or in any other manner or capacity whatsoever, (i) control, manage, operate, establish, take steps to establish, lend money to, invest in, solicit investors for, or otherwise provide capital to, or (ii) become employed by, join, perform services for, consult for, do business with or otherwise engage in any Competing Business (other than any Competing Business Executive has disclosed to the Company in writing on Exhibit A that he is engaged in on the Effective Date of this Agreement) within the Restricted Area (each as defined below) in a role that is substantially similar to any role the Executive has held with the Company during his employment. For purposes of this Agreement, given the scope of Confidential Information to be provided to Executive and job duties to be performed by the Executive, "Restricted Area" means the United States, and any other geographic area for which Executive performed any services or about which Executive received Confidential Information. For purposes of this Agreement, "Competing Business" means H&R Block, Intuit, and any other business, individual, partnership, firm, corporation or other entity that is competing or that is preparing to compete with any aspect of the Company's business, which includes, but is not limited to (a) tax preparation and tax preparation-related products and services provided to consumers and small businesses, and to or through tax professionals; (b) investment and insurance products or services, and related advice and brokerage services, provided to or through tax professionals or in conjunction with tax preparation services, and (c) a tax-optimized RIA or broker-

dealer, and (d) any other business the Company engages in or develops during the Executive's employment with the Company.

- (a) Non-Solicitation. During the Restricted Period, other than in connection with Executive's duties for the Company, Executive shall not, and shall not use any Confidential Information to, directly or indirectly, either as a principal, manager, agent, employee, consultant, officer, director, stockholder, partner, investor or lender or in any other capacity, and whether personally or through other persons, solicit business from, interfere with, or induce to curtail or cancel any business or contracts with the Company, or attempt to solicit business with, interfere with, or induce to curtail or cancel any business or contracts with the Company, or do business with any actual or prospective customer or client of the Company with whom the Company did business or who the Company solicited within the preceding two (2) years, and who or which: (1) Executive contacted, called on, serviced or did business with during Executive's employment with the Company; (2) Executive learned of as a result of Executive's employment with the Company; or (3) about whom Executive received Confidential Information. This restriction applies only to business which is in the scope of services or products provided by the Company.
- (b) <u>Non-Recruitment</u>. During the Restricted Period, other than in connection with Executive's duties for the Company, Executive shall not, on behalf of Executive or on behalf of any other person or entity, directly or indirectly, hire, solicit or recruit, or attempt to hire, solicit or recruit, or encourage to leave or otherwise cease his/her employment or engagement with the Company, any individual who is an employee or independent contractor of the Company or who was an employee or independent contractor of the Company within the twelve (12) month period prior to Executive's separation from employment with the Company.
- (c) Non-Disparagement. Executive agrees that the Company's goodwill and reputation are assets of great value to the Company, which have been obtained and maintained through great costs, time and effort. Therefore, Executive agrees that during Executive's employment and after the termination of Executive's employment, Executive shall not make, publish or otherwise transmit any knowingly false statements, whether written or oral, regarding the Company and its officers, directors, executives, employees, contractors, consultants, products, services, business or business practices. A violation or threatened violation of this Section 2(d) may be enjoined by the courts. The rights afforded the Company under this provision are in addition to any and all rights and remedies otherwise afforded by law. However, nothing in this Section 2(d) restricts or prevents Executive from providing truthful testimony as required by court order or other legal process or is intended to interfere with Executive's rights set forth in Section 1(c).
- (d) <u>Tolling</u>. If Executive violates any of the covenants contained in this Section 2, the Restricted Period applicable to such covenant(s) shall be suspended and shall not run in favor of Executive from the time of the commencement of such violation until the time that Executive cures the violation to the satisfaction of the Company and the period of time in which Executive is in breach shall be added to the Restricted Period applicable to such covenant(s).

Reasonableness. Executive hereby represents to the Company that Executive has read and understands, and agrees to be bound by, the terms of Section 1 and Section 2. Executive

acknowledges that the scope and duration of the restrictions and covenants contained in Section 1 and Section 2 are fair and reasonable in light of (i) the nature and scope of the operations of the Company's business; and (ii) the amount of compensation and Confidential Information (including, without limitation, trade secrets) that Executive is receiving in connection with Executive's employment with the Company and the Incentive Retention Letter. It is the desire and intent of the Parties that the provisions of Section 1 and Section 2 be enforced to the fullest extent permitted under applicable law, whether now or hereafter in effect and therefore, to the extent permitted by applicable law, Executive and the Company hereby waive any provision of applicable law that would render any provision of Section 1 and/or Section 2 invalid or unenforceable.

Remedies. Executive acknowledges that the restrictions and covenants contained in Section 1 and Section 2, in view of the nature of the Company's business and Executive's position with the Company, are reasonable and necessary to protect the Company's legitimate business interests, goodwill and reputation, and that any violation of Section 1 or Section 2 would result in irreparable injury and continuing damage to the Company, and that money damages would not be a sufficient remedy to the Company for any such breach or threatened breach. Therefore, Executive agrees that the Company shall be entitled to a temporary restraining order and injunctive relief restraining Executive from the commission of any breach or threatened breach of Section 1 and/or Section 2, without the necessity of establishing irreparable harm or the posting of a bond, and to recover from Executive damages incurred by the Company as a result of the breach, as well as the Company's attorneys' fees, costs and expenses related to any breach or threatened breach of this Agreement and enforcement of this Agreement. Nothing contained in this Agreement shall be construed as prohibiting the Company from pursuing any other remedies available to it for any breach or threatened breach, including, without limitation, the recovery of money damages, attorneys' fees, and costs. The existence of any claim or cause of action by Executive against the Company, whether predicated on this Agreement or otherwise, shall not constitute a defense to the enforcement by the Company of the restrictions or covenants contained in Section 1 or Section 2, or preclude injunctive relief.

Business Opportunities. Executive, without further compensation, assigns and agrees to assign to the Company and its successors, assigns or designees, all of Executive's right, title and interest in and to all Business Opportunities (defined below), and further acknowledges and agrees that all Business Opportunities constitute the exclusive property of the Company. Executive shall present all Business Opportunities to the Company, and shall not exploit a Business Opportunity. For purposes of this Agreement, "Business Opportunities" means all business ideas, prospects, or proposals pertaining to any aspect of the Company's business and any business the Company prepared to conduct or contemplated conducting during Executive's employment with the Company, which are developed by Executive or originated by any third party and brought to the attention of Executive, together with information relating thereto. For the avoidance of doubt, this Section 5 is not intended to limit or narrow Executive's duties or obligations under federal or state law with respect to corporate opportunities.

Conflicting Activities. Executive agrees that, during Executive's employment with the Company, Executive shall not engage in any employment, consulting relationship, business or other activity that (i) is in any way competitive with the business or proposed business of the Company

(except that Executive may invest less than one percent (1%) of the shares of a company traded on a registered stock exchange); (ii) conflicts with Executive's duty of loyalty, responsibilities or obligations to the Company or interferes with the independent exercise of Executive's judgment in the Company's best interests; or (iii) adversely affects the performance of Executive's job duties and responsibilities with the Company. Executive agrees to not assist any other person or organization in competing with the Company or in preparing to engage in competition with the Company or proposed business of the Company. Executive further agrees that, during Executive's employment with the Company, Executive shall not actively engage in any other employment, occupation or consulting activity for any direct or indirect remuneration without the prior approval of Executive's direct supervisor or the Company's Legal Department. Executive has listed on the Company's Outside Activity Disclosure form, attached hereto as Exhibit B-1, any business activities or ventures with which Executive is involved. If no such list is attached, Executive represents that there are no such outside activities as of the date of this Agreement.

Breach Executive acknowledges that Executive is subject to immediate dismissal by the Company for any breach of this Agreement and that such a dismissal will not relieve Executive from any continuing obligations under this Agreement or from the imposition by a court of any judicial remedies, including, without limitation, money damages and/or injunctive relief for such breach.

Notice. If Executive, in the future, seeks or is offered employment, or any other position or capacity with another company, entity or person, Executive agrees to inform each such company, entity or person of the existence of the restrictions in Section 1 and Section 2. The Company shall be entitled to advise such company, entity or person and third parties of the provisions of Section 1 and Section 2 and to otherwise deal with such company, entity, person or third party to ensure that the provisions of Section 1 and Section 2 are enforced and duly discharged.

Reformation. The Company and Executive agree that in the event any of the terms, provisions, covenants or restrictions contained in this Agreement, or any part thereof, shall be held by any court of competent jurisdiction to be effective in any particular area or jurisdiction only if said term, provision, covenant or restriction is modified to limit its duration or scope, then the court shall have such authority to so reform the term, provision, covenant or restriction and the Parties hereto shall consider such term, provision, covenant or restriction to be amended and modified with respect to that particular area or jurisdiction so as to comply with the order of any such court and, as to all other jurisdictions, the term, provision, covenant or restriction contained herein shall remain in full force and effect as originally written. By agreeing to this contractual modification prospectively at this time, the Company and Executive intend to make Section 1 and Section 2 enforceable under the law or laws of all applicable jurisdictions so that the restrictive covenants in their entirety and this Agreement as prospectively modified shall remain in full force and effect and shall not be rendered void or illegal.

Severability. In the event any court of competent jurisdiction or any foreign, federal, state, county or local government or any other governmental regulatory or administrative agency or authority holds any provision of this Agreement to be invalid, illegal or unenforceable, such invalid, illegal or unenforceable portion(s) shall be limited or excluded from this Agreement to the minimum

extent required, and the remaining provisions shall not be affected or invalidated and shall remain in full force and effect.

Binding Effect of Agreement and Assignment. This Agreement shall be binding upon and inure to the benefit of the Parties hereto and their respective heirs, successors, legal representatives and permitted assigns. Executive may not assign this Agreement to a third party. The Company may assign its rights, together with its obligations hereunder, to any affiliate and/or subsidiary of the Company or any successor thereto or any purchaser of substantially all of the assets of the Company, without Executive's consent and without advance notice.

Survival. Executive agrees that Executive's obligations under this Agreement shall continue in effect after the termination of Executive's employment, regardless of the reason(s) for termination, and whether such termination is voluntary or involuntary.

Waiver. The failure of either Party to insist in any one or more instances upon performance of any terms or conditions of this Agreement shall not be construed as a waiver of future performance of any such term, covenant or condition, but the obligations of either Party with respect thereto shall continue in full force and effect. No waiver of any breach of this Agreement shall be construed to be a waiver as to succeeding breaches and no waiver of any provisions of this Agreement shall constitute a waiver of any other provision of this Agreement. The breach by one party to this Agreement shall not preclude equitable relief, injunctive relief or the obligations in Section 1 or Section 2.

Controlling Law. This Agreement shall be governed by and construed under the laws of the State of Texas, without regard to any applicable conflict of law or choice of law rules.

Venue. Venue of any dispute arising out of, in connection with or in any way related to this Agreement shall be in a state district court of competent jurisdiction in Dallas County, Texas, or the United States District Court for the Northern District of Texas. Executive consents to personal jurisdiction of the state district courts of Dallas County, Texas and to the United States District Court for the Northern District of Texas for any dispute arising out of, in connection with or in any way related to this Agreement, and agrees that Executive shall not challenge personal jurisdiction in such courts. Executive waives any objection that Executive may now or hereafter have to the venue or jurisdiction of any proceeding in such courts or that any such proceeding was brought in an inconvenient forum (and agrees not to plead or claim the same).

WAIVER OF JURY TRIAL. WITH RESPECT TO ANY DISPUTE BETWEEN EMPLOYEE AND THE COMPANY ARISING OUT OF, IN CONNECTION WITH OR IN ANY WAY RELATED TO THIS AGREEMENT, EMPLOYEE AGREES TO RESOLVE SUCH DISPUTE(S) BEFORE A JUDGE WITHOUT A JURY. EMPLOYEE HAS KNOWLEDGE OF THIS PROVISION, AND WILL PROVIDE SERVICES TO THE COMPANY THEREAFTER, HEREBY WAIVING EMPLOYEE'S RIGHT TO TRIAL BY JURY AND AGREES TO HAVE ANY DISPUTE(S) ARISING BETWEEN THE COMPANY AND EMPLOYEE ARISING OUT OF, IN CONNECTION WITH OR IN ANY WAY RELATED TO THIS AGREEMENT RESOLVED BY A JUDGE OF A COMPETENT COURT IN DALLAS COUNTY, TEXAS, SITTING WITHOUT A JURY.

Entire Agreement. This Agreement constitutes the entire agreement between the Parties with respect to the subject matter hereof, and fully supersedes any and all prior and contemporaneous agreements, understandings and/or representations between the Parties, whether oral or written, pertaining to the subject matter of this Agreement; provided, however, Executive's obligations under this Agreement are in addition to Executive's obligations under any applicable law or regulation and the Company's policies and procedures. No oral statements or prior written material not specifically incorporated in this Agreement shall be of any force and effect, and no changes in or additions to this Agreement shall be recognized, unless incorporated in this Agreement by written amendment, such amendment to become effective on the date stipulated in it. Any amendment to this Agreement must be signed by all parties to this Agreement.

Disclaimer of Reliance. Except for the specific representations expressly made by the Company in this Agreement, Executive specifically disclaims that Executive is relying upon or has relied upon any communications, promises, statements, inducements, or representation(s) that may have been made, oral or written, regarding the subject matter of this Agreement. Executive represents that Executive relied solely and only on Executive's own judgment in making the decision to enter into this Agreement.

Voluntary Agreement. Executive (i) acknowledges that Executive has read and understands the terms of this Agreement and believes them to be reasonable, (ii) agrees that the consideration provided by the Company for this Agreement is reasonable, and (iii) is voluntarily executing this Agreement as signified by Executive's signature hereto.

Execution in Multiple Counterparts. This Agreement may be executed in multiple counterparts, whether or not all signatories appear on these counterparts, and each counterpart shall be deemed an original for all purposes.

[Signature Page Follows]

| EXECUTIVE:                                       |                              |
|--|------------------------------|
|  | Printed Name: Todd C. Mackay |
|  | Date:                        |
| THE COMPANY:                                     | Blucora, Inc.:               |
|  | Signature:                   |
|  | Name:                        |
|  | Title:                       |
|  | Date:                        |
|  |                              |
|  |                              |
|  |                              |
|  |                              |
| CONFIDENTIALITY AND NON-SOLICITATION AGREEMENT P | age 12                       |

The signatures below indicate that the Parties have read, understand and will comply with this Agreement.

# EXHIBIT A-1 LIST OF PRIOR INVENTIONS

| Title                                       | Date          | Identifying Number or Brief Description |
|---|---------------|---|
|   |               |   |
|   |               |   |
|   |               |   |
|   |               |   |
|   |               |   |
|   |               |   |
|   |               |   |
|   |               |   |
|   |               |   |
| No Inventions                               |               |   |
| Additional Sheets Attached                  |               |   |
| Maditional Shoots Material                  |               |   |
|   |               |   |
|   |               |   |
| Signature of Employee: /s/ Todd Mackay      |               |   |
|   |               |   |
|   |               |   |
| Print Name of Employee: Todd Mackay         |               |   |
|   |               |   |
|   |               |   |
| Date: December 24, 2018                     |               |   |
|   |               |   |
|   |               |   |
|   |               |   |
|   |               |   |
| CONFIDENTIALITY AND NON COLICITATION AGREEN | MENIT Dogs 12 |   |
| CONFIDENTIALITY AND NON-SOLICITATION AGREEN | ieni rage 13  |   |

# EXHIBIT A-2 OUTSIDE ACTIVITIES

| <u>CONFIDENTIALITY AND NON-SOLICITATION AGREEMENT</u> Pa |
|--|
|--|

#### **EXHIBIT B**

#### GENERAL RELEASE OF ALL CLAIMS

This General Release and Waiver of Claims (this "*Release*") is executed by Todd C. Mackay ("*Executive*") and Blucora, Inc. (the "*Company*") as of the date set forth below, and will become effective as of the "*Effective Date*" as defined below. This Release is in consideration of severance benefits to be paid to Executive by the Company pursuant to the Employment Agreement between Executive and the Company dated as of January \_\_\_\_, 2018 (the "*Employment Agreement*"). Execution of this Release without revocation by Executive will satisfy the requirement, set forth in Section 6(g) of the Employment Agreement, that Executive execute a general release and waiver of claims in order to receive severance benefits pursuant to the Employment Agreement.

# 1. Termination of Employment

Executive acknowledges that his employment with the Company and any of its subsidiaries (collectively, the "Company Group") and any and all appointments he held with any member of the Company Group, whether as officer, director, employee, consultant, agent or otherwise, terminated as of \_\_\_\_\_\_\_ (the "Termination Date"). Effective as of the Termination Date, Executive has not had or exercised or purported to have or exercise any authority to act on behalf of the Company or any other member of the Company Group, nor will Executive have or exercise or purport to have or exercise such authority in the future.

#### 2. Consideration

The Company shall pay Executive the severance benefits pursuant to the Employment Agreement. The Parties agree that but for signing this Release, Executive would not be entitled to the severance benefits set forth in the Employment Agreement. The severance benefits are adequate to make this Release final and binding, and are in addition to payments and benefits to which Executive would otherwise be entitled to as an employee or former employee of the Company.

#### 3. Waiver and Release

(a) Executive, for and on behalf of himself and his heirs and assigns, hereby waives and releases any common law, statutory or other complaints, claims, charges or causes of action arising out of or relating to Executive's employment or termination of employment with, or Executive's serving in any capacity in respect of any member of the Company Group (collectively, "Claims"). The Claims waived and released by this Release include any and all Claims, whether known or unknown, whether in law or in equity, which Executive may now have or ever had against any member of the Company Group or any shareholder, employee, officer, director, agent, attorney, representative, trustee, administrator or fiduciary of any member of the Company Group (collectively, the "Company Releasees") up to and including the date of Executive's execution of this Agreement. The Claims waived and released by this Release include, without limitation, any and all Claims arising out of Executive's employment with the Company Group under, by way of example and not limitation, the Age Discrimination in Employment Act of 1967 ("ADEA", a law which prohibits discrimination on the basis of age against persons age 40 and older), the National

Labor Relations Act, the Civil Rights Act of 1991, the Americans With Disabilities Act of 1990, Title VII of the Civil Rights Act of 1964, the Employee Retirement Income Security Act of 1974, the Family Medical Leave Act, the Securities Act of 1933, the Securities Exchange Act of 1934, and the Texas Labor Code Chapter 21, all as amended, and all other federal, state and local statutes, ordinances, regulations and the common law, and any and all Claims arising out of any express or implied contract, except as described in Paragraphs 2(b) and 2(c) below.

- (b) The waiver and release set forth in this Section 3 is intended to be construed as broadly and comprehensively as applicable law permits. The waiver and release shall not be construed as waiving or releasing any claim or right that as a matter of law cannot be waived or released, including Executive's right to file a charge with the Equal Employment Opportunity Commission or other government agency.
  - (c) Notwithstanding anything else in this Release, Executive does not waive or release claims with respect to:
    - (i) Executive's entitlement, if any, to severance benefits pursuant to the Employment Agreement;
- (ii) vested benefits or payments specifically to be provided to the Executive pursuant to the Employment Agreement or any Company employee benefit plans or policies;
- (iii) indemnification pursuant to any applicable provision of the Company's Bylaws or Certificate of Incorporation, as amended, pursuant to any written indemnification agreement between the Executive and the Company, or pursuant to applicable law;
- (iv) any claims which the Executive may have solely by virtue of the Executive's status as a shareholder of the Company; or
  - (v) unemployment compensation to which Executive may be entitled under applicable law.
- (d) Executive represents and warrants that he is the sole owner of the actual or alleged Claims that are released hereby, that the same have not been assigned, transferred, or disposed of in fact, by operation of law, or in any manner, and that he has the full right and power to grant, execute and deliver the releases, undertakings, and agreements contained herein.
- (e) Subject to Section 4, Executive represents that Executive has not filed any complaints, charges or lawsuits against the Company with any governmental agency or any court based on Claims that are released and waived by this Release.

#### 4. No Interference

Nothing in this Agreement is intended to interfere with Executive's right to report possible violations of federal, state or local law or regulation to any governmental or law enforcement agency or entity, or to make other disclosures that are protected under the whistleblower provisions of federal, state or local law or regulation. Executive further acknowledges that nothing in this

Agreement is intended to interfere with Executive's right to file a claim or charge with, or testify, assist, or participate in an investigation, hearing, or proceeding conducted by, the Equal Employment Opportunity Commission (the "*EEOC*"), any state human rights commission, or any other government agency or entity. However, by executing this Agreement, Executive hereby waives the right to recover any damages or benefits in any proceeding Executive may bring before the EEOC, any state human rights commission, or any other government agency or entity or in any proceeding brought by the EEOC, any state human rights commission, or any other government agency or entity on Employee's behalf with respect to any claim released in this Agreement *except that* Executive may receive bounty money awarded by the U.S. Securities and Exchange Commission pursuant to Section 21F of the Securities Exchange Act of 1934 or any similar provision.

## 5. No Admission of Wrongdoing

This Release shall not be construed as an admission by either party of any wrongful or unlawful act or breach of contract.

# 6. Legal Disclosure.

Subject to Section 4, by signing this Agreement, Executive warrants and represents that Executive has reported to Human Resources or Legal all pending and/or threatened legal proceedings of any kind involving or relating to the Company that Executive became aware of during Executive's tenure with the Company. Executive further warrants and represents that Executive has reported to Human Resources or Legal any alleged violations of law (including alleged securities violations) by the Company that Executive became aware of during Executive's tenure with the Company.

# 7. Binding Agreement; Successors and Assigns

This Release binds Executive's heirs, administrators, representatives, executors, successors, and assigns, and will inure to the benefit of the respective heirs, administrators, representatives, executors, successors, and assigns of any person or entity as to whom the waiver and release set forth in Section 3 applies.

# 8. Other Agreements

This Release does not supersede or modify in any way Executive's continuing obligations pursuant to the Employment Agreement or the Confidentiality and Non-Competition Agreement (Exhibit A thereto) or the dispute resolution provisions of the Employment Agreement.

# 9. Knowing and Voluntary Agreement; Consideration and Revocation Periods

- (a) Executive acknowledges that Executive has been given twenty-one (21) calendar days from the date of receipt of this Release to consider all of the provisions of this Release and that if Executive signs this Release before the 21-day period has ended he knowingly and voluntarily waives some or all of such 21-day period.
- (b) Executive represents that (i) Executive has read this Release carefully, (ii) Executive has hereby been advised by the Company to consult an attorney of his choice and has either done

so or voluntarily chosen not to do so, (iii) Executive fully understands that by signing below he is giving up certain rights which he might otherwise have to sue or assert a claim against any of the Company Releasees, and (iv) Executive has not been forced or pressured in any manner whatsoever to sign this Release, and agrees to all of its terms voluntarily.

(c) Executive shall have seven (7) calendar days from the date of his execution of this Release (the "*Revocation Period*") in which Executive may revoke this Release. Such revocation must be in writing and delivered, prior to the expiration of the Revocation Period, to the attention of the Company's Chief Legal Officer at the Company's then-current headquarters address. If Executive revokes this Release during the Revocation Period, then the Release shall be null and void and without effect.

#### 10. Disclaimer of Reliance

Except for the specific representations expressly made by the Company in the Employment Agreement, Executive specifically disclaims that Executive is relying upon or has relied upon any communications, promises, statements, inducements, or representation(s) that may have been made, oral or written, regarding the subject matter of this Release. Executive represents that Executive relied solely and only on Executive's own judgment in making the decision to enter into this Release.

# 11. Execution in Multiple Counterparts

This Release may be executed by the parties in multiple counterparts, whether or not all signatories appear on these counterparts (including via electronic signatures and exchange of PDF documents via email), each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

## 12. Effective Date

The Effective Date of this Release will be day after the Revocation Period expires without revocation by Executive.

[Signature Page Follows]

EXECUTIVE HAS ELECTED FREELY AND VOLUNTARILY TO EXECUTE THIS RELEASE, TO FULFILL THE PROMISES SET FORTH IN THE EMPLOYMENT AGREEMENT, AND TO RECEIVE THEREBY THE PAYMENT AND OTHER CONSIDERATION DESCRIBED IN THE EMPLOYMENT AGREEMENT. EXECUTIVE UNDERSTANDS THAT, BY SIGNING THIS RELEASE, EXECUTIVE IS AGREEING TO WAIVE AND SETTLE THE RELEASED CLAIMS HEREIN THAT EXECUTIVE HAS OR MIGHT HAVE AGAINST THE COMPANY INCLUDING CLAIMS UNDER THE AGE DISCRIMINATION IN EMPLOYMENT ACT. EXECUTIVE'S SIGNATURE BELOW MEANS THAT EXECUTIVE HAS READ THE RELEASE AND UNDERSTANDS AND AGREES THAT EXECUTIVE IS RELEASING ALL CLAIMS AND AGREES AND CONSENTS TO THE TERMS AND CONDITIONS OF THIS EMPLOYMENT AGREEMENT AND THIS RELEASE.

| EXECUTIVE:   | Signature:                   |
|--------------|------------------------------|
|              |                              |
|              | Printed Name: Todd C. Mackay |
|              | Date:                        |
|              |                              |
| THE COMPANY: | Blucora, Inc.:               |
|              | Signature:                   |
|              | Signature.                   |
|              | Name:                        |
|              | Title:                       |
|              | Date:                        |

#### EMPLOYMENT AGREEMENT

This Employment Agreement (this "Agreement") is made and entered into as of October 12, 2018, by and between Curtis Campbell (the "Executive") and Blucora, Inc. (the "Company").

#### **RECITALS**

WHEREAS, the Company desires to employ the Executive as the President of TaxAct beginning on November 5, 2018 (the "Effective Date"), and the Executive desires to serve in such capacity;

NOW THEREFORE, in consideration of the foregoing, the mutual covenants contained herein, the employment of the Executive by the Company, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

## 1. <u>Certain Definitions</u>

- (a) "Base Salary" has the meaning set forth in Section 5(a).
- (b) "Board" means the Board of Directors of the Company.
- (c) "Cause" means, as determined by the Board in its reasonable discretion: (i) the Executive's conviction of, or plea of guilty or nolo contendere to, a misdemeanor involving dishonesty, wrongful taking of property, immoral conduct, bribery or extortion or any felony; (ii) willful material misconduct by the Executive in connection with the business of the Company; (iii) the Executive's continued and willful failure to perform substantially his responsibilities to the Company under this Agreement, after written demand for substantial performance has been given by the Board that specifically identifies how the Executive has not substantially performed his responsibilities; (iv) the Executive's improper disclosure of confidential information or other material breach of this Agreement, including the Confidentiality and Non-Competition Agreement; (v) the Executive's material fraud or dishonesty against the Company; (vi) the Executive's willful and material breach of the Company's written code of conduct and business ethics or other material written policy, procedure or guideline in effect from time to time (provided that the Executive was given access to a copy of such policy, procedure or guideline prior to the alleged breach) relating to personal conduct; or (vii) the Executive's willful attempt to obstruct or willful failure to cooperate with any investigation authorized by the Board or any governmental or self-regulatory entity. Any determination of Cause by the Company shall be made by a resolution approved by a majority of the members of the Board, provided that, with respect to Section 1(c)(iii), the Board must give the Executive notice and 60 days to cure the substantial nonperformance.
  - (d) "Change of Control" means the occurrence of any of the following:
- (i) any "person" (as defined in Sections 13(d) and 14(d) of the Exchange Act), excluding for this purpose, (A) the Company or any subsidiary of the Company or (B) any employee benefit plan of the Company or any subsidiary of the Company, or any person or entity organized,

appointed or established by the Company for or pursuant to the terms of any such plan that acquires beneficial ownership of voting securities of the Company, is or becomes the "beneficial owner" (as defined in Rule 13d-3 under the Exchange Act), directly or indirectly, of securities of the Company representing more than 50% of the combined voting power of the Company's then outstanding securities;

- (ii) consummation of a reorganization, merger or consolidation of the Company, in each case, unless, following such transaction, all or substantially all the individuals and entities who were the beneficial owners of outstanding voting securities of the Company immediately prior to such transaction beneficially own, directly or indirectly, more than 50% of the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors of the company resulting from such transaction (including, without limitation, a company that, as a result of such transaction, owns the Company or all or substantially all the Company's assets either directly or through one or more subsidiaries) in substantially the same proportions as their ownership immediately prior to such transaction of the outstanding voting securities of the Company;
- (iii) any sale or disposition by the Company, in one transaction or a series of related transactions, of all or substantially all the Company's assets;
- (iv) a "Board Change" which, for purposes of this Agreement, shall have occurred if a majority of the seats on the Board are occupied by individuals who were neither (A) nominated by a majority of the Incumbent Directors nor (B) appointed by directors so nominated ("Incumbent Director" means a member of the Board who has been either (1) nominated by a majority of the directors of the Company then in office or (2) appointed by directors so nominated, but excluding, for this purpose, any such individual whose initial assumption of office occurs as a result of either an actual or threatened election contest (as such terms are used in Rule 14a-11 of Regulation 14A promulgated under the Exchange Act) or other actual or threatened solicitation of proxies or consents by or on behalf of a person other than the Board); or
  - (v) an approval by the stockholders of the Company of a complete liquidation or dissolution of the Company.
  - (e) "Code" means the Internal Revenue Code of 1986, as amended.
  - (f) "Compensation Committee" means the Compensation Committee of the Board.
- (g) "Confidentiality and Non-Competition Agreement" means the Confidentiality and Non-Competition Agreement attached hereto as Exhibit A.
- (h) "Constructive Termination" means the occurrence, on a date that is prior to the two- month period prior to the consummation of a Change of Control or after the 12-month period following the consummation of a Change of Control, of any of the following without the Executive's express prior written consent: (i) a material reduction of or to the Executive's duties, authority or responsibilities (a change in reporting relationship alone does not constitute such a material reduction); (ii) a material reduction by the Company of the Executive's Base Salary, unless similarly situated executives also experience a reduction; or (iii) a requirement that the Executive relocate

his primary work location more than 25 miles from Irving, Texas or from any work location to which the Company transfers the Executive during the course of his employment and to which such transfer the Executive has consented. Notwithstanding the foregoing, a Constructive Termination shall not exist unless (x) the Executive delivers written notice to the Company (the "Constructive Termination Notice") of the existence of the condition which the Executive believes constitutes a Constructive Termination within 30 days of the initial existence of such condition (which Constructive Termination Notice specifically identifies such condition), (y) the Company fails to remedy such condition within 30 days after the date on which it receives such notice (the "Constructive Termination Cure Period"), and (z) the Executive actually terminates employment within 30 days after the expiration of the Constructive Termination Cure Period.

- (i) "Disability" means the Executive's inability to perform his employment duties to the Company hereunder, with or without reasonable accommodation, for 180 days (in the aggregate) in any one-year period as determined by an independent physician selected by the Company.
  - (j) "Exchange Act" means the Securities Exchange Act of 1934, as amended.
- (k) "Good Reason" means the occurrence of any of the following without the Executive's express prior written consent: (i) a material reduction of or to the Executive's duties, authority, responsibilities or reporting relationship; (ii) a material reduction of the Executive's Base Salary; (iii) a material reduction of the Executive's Target Bonus; (iv) a material reduction in the kind or level of employee benefits to which the Executive is entitled that occurs within 12 months following a Change of Control, unless similarly situated employees also experience a reduction; (v) a requirement that the Executive relocate his primary work location more than 25 miles from Irving, Texas or from any work location to which the Company transfers the Executive during the course of his employment and to which such transfer the Executive has consented; (vi) in connection with a Change of Control, the failure of the Company to assign this Agreement to a successor to the Company or the failure of a successor to the Company to explicitly assume and agree to be bound by this Agreement in a writing delivered to the Executive; or (vii) a material breach of this Agreement by the Company.

Notwithstanding the foregoing, termination of employment by the Executive will not be for Good Reason unless (x) the Executive delivers written notice to the Company (the "Good Reason Notice") of the existence of the condition which the Executive believes constitutes Good Reason within 30 days of the initial existence of such condition (which Good Reason Notice specifically identifies such condition), the Company fails to remedy such condition within 30 days after the date on which it receives such notice (the "Good Reason Cure Period"), and (z) the Executive actually terminates employment within 30 days after the expiration of the Good Reason Cure Period.

(I) "Release" means a full release of claims against the Company substantially in the form attached hereto as Exhibit B; provided, however, that notwithstanding the foregoing, such Release is not intended to and will not waive the Executive's rights: (i) to indemnification pursuant to any applicable provision of the Company's Bylaws or Certificate of Incorporation, as amended, pursuant to any written indemnification agreement between the Executive and the Company, or pursuant to applicable law; (ii) to vested benefits or payments specifically to be provided to the Executive under this Agreement or any Company employee benefit plans or policies; or (iii)

respecting any claims the Executive may have solely by virtue of the Executive's status as a stockholder of the Company. The Release also shall not include claims that an employee cannot lawfully release through execution of a general release of claims.

- (m) "Section 409A" means Section 409A of the Code and the Treasury Regulations and official guidance issued in respect of Section 409A of the Code.
  - (n) "Target Bonus" has the meaning set forth in Section 5(b).

# 2. <u>Duties and Scope of Employment</u>

The Company shall employ the Executive in the position of President of TaxAct. The Executive shall report directly to the Company's President and Chief Executive Officer. The Executive will render such business and professional services in the performance of the Executive's duties, consistent with the Executive's position(s) within the Company, as shall be reasonably assigned to the Executive at any time and from time to time by the Company's President and Chief Executive Officer. Upon termination of the Executive's employment for any reason, unless otherwise requested by the President and Chief Executive Officer, the Executive will be deemed to have resigned from all positions held at the Company and its affiliates voluntarily, without any further action by the Executive, as of the end of the Executive's employment, and the Executive, at the President and Chief Executive Officer's request, will execute any documents necessary to reflect his resignation.

# 3. <u>Obligations</u>

While employed hereunder, the Executive will perform his duties ethically, faithfully and to the best of the Executive's ability and in accordance with law and Company policy. The Executive agrees not to actively engage in any other employment, occupation or consulting activity for any direct or indirect remuneration without the express prior written approval of the Company's President and Chief Executive Officer; provided, however, that notwithstanding anything to the contrary in the Confidentiality and Non-Competition Agreement, the Executive may engage in charitable activities so long as such activities do not materially interfere with the Executive's responsibilities to the Company.

# 4. Agreement Term

Unless earlier terminated as provided herein, the term of this Agreement (the "Agreement Term") shall be for a period of three years commencing on the Effective Date, and may be extended thereafter upon the written mutual agreement of the Executive and the Company.

## 5. Compensation and Benefits

(a) <u>Base Salary</u>. The Company agrees to pay the Executive a base salary (the "*Base Salary*") at an annual rate of not less than \$380,000, payable in accordance with the regular payroll practices of the Company, but not less frequently than monthly. The Executive's Base Salary shall be subject to annual review by the Board (or a committee thereof).

- (b) <u>Annual Bonus</u>. The Executive shall be eligible to participate in the Company's bonus and other incentive compensation plans and programs for the Company's senior executives at a level commensurate with his position. The Executive shall have the opportunity to earn an annual target bonus (the "*Target Bonus*") measured against criteria to be determined by the Board (or a committee thereof) of at least 125% of Base Salary. The Executive's Target Bonus amount for 2018 will be pro-rated to reflect the number of days of the Executive's employment in 2018. The payout of any 2018 bonus will occur following the end of the Company's Executive Bonus Plan (the "*Plan*") year, which is December 31, 2018, and will be paid in accordance with the terms and conditions of the Plan. Payment of any bonus pursuant to this Section 5(b) is subject to the Executive's employment by the Company on the date required in the Plan. The Company reserves the right to change the Plan at any time at its discretion.
- (c) Equity Awards. The Executive will be eligible to participate in the Company's long-term equity incentive programs extended to senior executives of the Company generally at levels commensurate with the Executive's position, which participation and levels shall be determined by the Board (or a committee thereof) in its sole discretion. In 2019, the Company's President and Chief Executive Officer shall recommend that the Board (or a committee thereof) grant the Executive, for 2019, equity award(s) with a total aggregate target value of \$950,000 on the date of grant, with such award(s) consisting of a combination of the following types of awards as determined by the Board (or a committee thereof), in its sole discretion: time-based restricted stock units (vesting over three years), performance-based restricted stock units (eligible for vesting based on performance following a three-year performance period), and/or nonqualified stock options (vesting over three years); provided, however, that the Board (or committee thereof) retains the sole discretion to establish the terms and the types of awards that are granted to the Executive with respect to 2019.
- (d) <u>Benefits</u>. The Executive and his eligible dependents shall be eligible to participate in the employee benefit plans that are available or that become available to other employees of the Company, with the adoption or maintenance of such plans to be in the discretion of the Company, subject in each case to the generally applicable terms and conditions of the plan or program in question and to the determination of any committee administering such plan or program. Such benefits shall include participation in the group medical, life, disability, and retirement plans that are made generally available to employees of the Company, and any supplemental plans available to senior executives of the Company from time to time. The Company reserves the right to change or terminate its employee benefit plans and programs at any time.
- (e) <u>Expenses</u>. The Company shall reimburse the Executive for reasonable business expenses incurred by the Executive in the furtherance of or in connection with the performance of the Executive's duties hereunder, in accordance with the Company's expense reimbursement policy as in effect from time to time.
  - (f) Replacement Compensation and Signing Bonus.
- (i) <u>Signing Bonus</u>. The Company will pay Executive a signing bonus of \$350,000, less social security contributions, income tax withholding, and any other applicable deductions, within 30 days following the Effective Date ("Signing Bonus"). If Executive resigns his employment with the Company for any reason, or if Executive is terminated by the Company

for Cause, and such resignation or termination occurs on or before the one-year anniversary of the Effective Date, Executive will repay to the Company the Signing Bonus.

(ii) <u>Initial Equity Award</u>. As a material inducement to the Executive's willingness to accept employment with the Company and to compensate the Executive for equity granted by his prior employer that he forfeited by accepting employment with the Company, on or shortly following the Effective Date, the Executive shall be granted an award of time-based restricted stock units having an aggregate value of \$250,000 on the date of grant that vest on the one-year anniversary of the date of grant so long as Executive is employed by the Company on such date (the "*Initial RSUs*").

The number of Initial RSUs granted to the Executive in accordance with this Section 5(f)(ii) shall be determined by dividing the value of the Initial RSUs being awarded by the closing price of the Company's common stock on the date of grant and rounding down for any fractional shares. The Initial RSUs may be granted under the Blucora, Inc. 2016 Equity Inducement Plan, as amended (the "Inducement Plan"), or the Blucora, Inc. 2018 Long-Term Incentive Plan (the "LTIP") and will vest in accordance with, and have such other terms and conditions as are specified in, the Restricted Stock Unit Grant Notice and Restricted Stock Unit Agreement approved by the Compensation Committee with respect to such award (the "Restricted Stock Unit Agreement") and shall otherwise be subject to the terms and conditions of the Inducement Plan or LTIP and the Restricted Stock Unit Agreement; provided, however, that notwithstanding the foregoing, in the event of a conflict between the terms and conditions of the Restricted Stock Unit Agreement and this Agreement, the terms and conditions of this Agreement shall prevail.

# 6. <u>Termination of Employment</u>

- (a) <u>General Provisions</u>. This Agreement and the Executive's employment with the Company may be terminated by either the Executive or the Company at will at any time with or without Cause; provided, however, that the parties' rights and obligations upon such termination during the Agreement Term shall be as set forth in applicable provisions of this Agreement.
- (b) Any Termination by Company or the Executive. In the event of any termination of the Executive's employment with the Company, whether by the Company or by the Executive, (i) the Company shall pay the Executive any unpaid Base Salary due for periods prior to the date of termination of employment ("Termination Date"); (ii) the Company shall pay the Executive any unpaid bonus compensation pursuant to Section 5(b), to the extent earned through the Termination Date, subject to the terms of the Company's Executive Bonus Plan (or any successor plan thereto); and (iii) following submission of proper expense reports by the Executive, the Company shall reimburse the Executive for all expenses reasonably and necessarily incurred by the Executive in connection with the business of the Company through the Termination Date (collectively, the "Accrued Obligations"). The Accrued Obligations shall be paid promptly upon termination and within the period of time mandated by applicable law (but, in any event, within 30 days after the Termination Date). The Accrued Obligations paid or provided pursuant to this Section 6(b) shall be in addition to the payments and benefits, if any, to be provided to the Executive upon his termination of employment pursuant to Section 6(c), 6(d), 6(e), or 6(f) as applicable. Except as

expressly stated above or as required by law or this Agreement, the Executive shall receive no further compensation in any form other than as set forth in this Section 6(b).

- (c) <u>Termination by Company Without Cause or Constructive Termination</u>. If, other than in connection with a Change of Control as described in Section 6(d), the Executive's employment with the Company is terminated by the Company without Cause or the Executive terminates employment with the Company under circumstances constituting a Constructive Termination, then subject to Section 6(g), the Executive shall receive the following payments and benefits:
- (i) a severance payment in an amount equal to one times the Executive's Base Salary in effect as of the Termination Date (or if the Executive terminates employment under circumstances constituting a Constructive Termination due to a material reduction of the Executive's Base Salary, in effect immediately prior to such reduction) (less applicable withholding taxes), which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days following the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date), in accordance with Section 14(b)(ii); and
- (ii) a lump-sum payment in an amount equal to (A) the monthly COBRA premium in effect under the Company's group health plan as of the Termination Date for the coverage in effect under such plan for the Executive (and the Executive's spouse and dependent children) on such date multiplied by (B) 12 (less applicable withholding taxes), which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days following the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date), in accordance with Section 14(b)(ii); provided, however, that notwithstanding the foregoing or any other provision in this Agreement to the contrary, the Company (or its successor) may unilaterally amend this Section 6(c)(ii) or eliminate the benefit provided hereunder to the extent it deems necessary to avoid the imposition of excise taxes, penalties or similar charges on the Company or any of its subsidiaries, affiliates or successors, including, without limitation, under Section 4980D of the Code.

Notwithstanding any provision to the contrary in any Company equity compensation plan or any outstanding equity award agreement, if, during the Agreement Term, the Executive terminates employment with the Company under circumstances described in this Section 6(c), there shall be no acceleration of vesting or exercisability of any outstanding equity awards or extension of any option post-termination exercise period.

For the avoidance of doubt, under no circumstances will the Executive be entitled to payments and benefits under both this Section 6(c) and Section 6(d).

(d) <u>Termination of Employment in Connection With a Change of Control</u>. If the Company terminates the Executive's employment without Cause or the Executive terminates employment with the Company for Good Reason (1) on the day of or during the 12-month period immediately following the consummation of a Change of Control or (2) during the 2-month period prior to the consummation of a Change of Control but at the request of any third party participating in or causing the Change of Control or otherwise in connection with the Change of Control, then

subject to Section 6(g) and with respect to clause (2), subject to the consummation of such Change of Control, the Executive shall receive the following payments and benefits:

- (i) a severance payment in an amount equal to one times the Executive's Base Salary in effect as of the Termination Date and his then current Target Bonus amount (or if the Executive terminates employment for Good Reason due to a material reduction of the Executive's Base Salary or Target Bonus, in effect immediately prior to such reduction) (in each case less applicable withholding taxes), which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days following the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date), in accordance with Section 14(b)(ii);
- (ii) a lump-sum payment in an amount equal to (A) the monthly COBRA premium in effect under the Company's group health plan as of the Termination Date for the coverage in effect under such plan for the Executive (and the Executive's spouse and dependent children) on such date multiplied by (B) 12 (less applicable withholding taxes), which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days following the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date), in accordance with Section 14(b)(ii); provided, however, that notwithstanding the foregoing or any other provision in this Agreement to the contrary, the Company (or its successor) may unilaterally amend this Section 6(d)(ii) or eliminate the benefit provided hereunder to the extent it deems necessary to avoid the imposition of excise taxes, penalties or similar charges on the Company or any of its subsidiaries, affiliates or successors, including, without limitation, under Section 4980D of the Code; and
- (iii) notwithstanding any provision to the contrary in any applicable equity compensation plan or any outstanding equity award agreement, the treatment of the Executive's outstanding equity awards shall be governed solely by the following provisions: (A) all of the Executive's then-outstanding time-vesting equity awards shall fully vest and all restrictions thereon shall lapse, and (B) to the extent vested (including as a result of the acceleration provided under this Section 6(d)(iii)), all of the Executive's outstanding stock options shall remain exercisable until the first to occur of 12 months following the Termination Date and each such stock option's original expiration date.

If a Change of Control is consummated prior to the expiration of the Agreement Term, this Section 6(d) shall apply to a termination of the Executive's employment by the Company without Cause or by the Executive for Good Reason during the 12-month period immediately following the consummation of the Change of Control even if such 12-month period extends past the expiration of the Agreement Term. Moreover, notwithstanding the expiration of the Agreement Term, if a Change of Control is consummated within two months after the expiration of the Agreement Term, then this Section 6(d) shall apply to a termination of the Executive's employment by the Company without Cause or by the Executive for Good Reason (i) on the day of or during the 12-month period immediately following the consummation of the Change of Control or (ii) during the 2-month period prior to the consummation of the Change of Control but at the request of any third party participating in or causing the Change of Control or otherwise in connection with the Change of Control.

For the avoidance of doubt, the payments and benefits described under this Section 6(d) and the Accrued Obligations shall be the only payments and benefits to which the Executive is entitled in the event that the Executive's employment terminates under this Section 6(d).

- (e) <u>Death</u>. In the event of the Executive's death while employed hereunder, and subject to Section 6(g), the Executive's beneficiary (or such other person(s) specified by will or the laws of descent and distribution) shall be entitled to receive a lump-sum payment in an amount equal to three months' Base Salary in effect as of the Termination Date (less applicable withholding taxes), which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days following the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date), in accordance with Section 14(b)(ii).
- (f) <u>Disability</u>. In the event of the Executive's termination of employment with the Company due to Disability, and subject to Section 6(g), the Executive shall be entitled to receive a lump- sum payment in an amount equal to six months' Base Salary in effect as of the Termination Date (less applicable withholding taxes), which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days following the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date), in accordance with Section 14(b)(ii).
- (g) Release and Other Conditions. The payments and benefits described in Sections 6(c) through 6(f) are expressly conditioned on (i) the Executive (or, in the case of the Executive's death, the Executive's representative) signing and delivering (and not revoking thereafter) a Release to the Company (which, in the case of the Executive's death, also releases any claims by the Executive's estate or survivors), which Release is executed, delivered and effective no later than 60 days following the Termination Date and (ii) the Executive continuing to satisfy any obligations to the Company under this Agreement, the Release and the Confidentiality and Non-Competition Agreement that are incorporated herein by reference, and any other agreement(s) between the Executive and the Company. In the event the Release described in Section 6(g)(i) is not executed, delivered and effective by the 60th day after the Termination Date, none of such payments or benefits shall be provided to the Executive.

#### 7. Section 280G

(a) Amount of Payments and Benefits. Notwithstanding anything to the contrary herein, in the event that the Executive becomes entitled to receive or receives any payments, options, awards or benefits (including, without limitation, the monetary value of any noncash benefits and the accelerated vesting of equity-based awards) under this Agreement or under any other plan, agreement or arrangement with the Company or any person affiliated with the Company (collectively, the "Payments"), that may separately or in the aggregate constitute "parachute payments" within the meaning of Section 280G of the Code and the Treasury Regulations promulgated thereunder (or any similar or successor provision) (collectively, "Section 280G") and it is determined that, but for this Section 7(a), any of the Payments will be subject to any excise tax pursuant to Section 4999 of the Code or any similar or successor provision (the "Excise Tax"), the Company shall pay to the Executive either (i) the full amount of the Payments or (ii) an amount equal to the Payments, reduced

by the minimum amount necessary to prevent any portion of the Payments from being an "excess parachute payment" (within the meaning of Section 280G) (the "Capped Payments"), whichever of the foregoing amounts results in the receipt by the Executive, on an after-tax basis, of the greatest amount of Payments notwithstanding that all or some portion of the Payments may be subject to the Excise Tax. For purposes of determining whether the Executive would receive a greater after-tax benefit from the Capped Payments than from receipt of the full amount of the Payments, (i) there shall be taken into account any Excise Tax and all applicable federal, state and local taxes required to be paid by the Executive in respect of the receipt of such payments and (ii) such payments shall be deemed to be subject to federal income taxes at the highest rate of federal income taxation applicable to individuals that is in effect for the calendar year in which the payments and benefits are to be paid, and state and local income taxes at the highest rate of taxation applicable to individuals in the state and locality of the Executive's residence on the effective date of the relevant transaction described under Section 280G(b)(2)(A)(i) of the Code, net of the maximum reduction in federal income taxes that could be obtained from deduction of such state and local taxes (as determined by assuming that such deduction is subject to the maximum limitation applicable to itemized deductions under Section 68 of the Code and any other limitations applicable to the deduction of state and local income taxes under the Code).

- (b) Computations and Determinations. All computations and determinations called for by this Section 7 shall be made by an independent accounting firm or independent tax counsel appointed by the Company (the "*Tax Counsel*"), and all such computations and determinations shall be conclusive and binding on the Company and the Executive. For purposes of such calculations and determinations, the Tax Counsel may rely on reasonable, good faith interpretations concerning the application of Sections 280G and 4999 of the Code. The Tax Counsel shall submit its determination and detailed supporting calculations to both the Executive and the Company within 15 days after receipt of a notice from either the Company or the Executive that the Executive may receive payments which may be considered "parachute payments." The Company and the Executive shall furnish to the Tax Counsel such information and documents as the Tax Counsel may reasonably request in order to make the computations and determinations called for by this Section 7. The Company shall bear all costs that the Tax Counsel may reasonably incur in connection with the computations and determinations called for by this Section 7.
- (c) <u>Reduction Methodology</u>. In the event that Section 7(a) applies and a reduction is required to be applied to the Payments thereunder, the Payments shall be reduced by the Company in its reasonable discretion in the following order: (i) reduction of any Payments that are subject to Section 409A on a pro-rata basis or such other manner that complies with Section 409A, as determined by the Company, and (ii) reduction of any Payments that are exempt from Section 409A.

#### 8. No Impediment to Agreement

The Executive hereby represents to the Company that the Executive is not, as of the date hereof, and will not be, during the Executive's employment with the Company, employed under contract, oral or written, by any other person, firm or entity, and is not and will not be bound by the provisions of any restrictive covenant or confidentiality agreement that would constitute an

impediment to, or restriction upon, the Executive's ability to enter this Agreement and to perform the duties of the Executive's employment.

# 9. <u>Confidentiality and Non-Competition Agreement</u>

The Confidentiality and Non-Competition Agreement is incorporated by reference as if set forth fully herein. The Confidentiality and Non-Competition Agreement shall survive the termination of this Agreement and/or the Executive's employment with the Company.

## 10. <u>Cooperation</u>

The Executive hereby agrees to provide the Executive's full cooperation, at the request of the Company, with any of the Company Releasees (as defined in the Release) in any and all such lawsuits, investigations or other legal, equitable or business matters or proceedings which involve any matters for which the Executive worked on or had responsibility during the Executive's employment with the Company. The Executive also agrees to be available to the Company and its representatives (including attorneys) to provide general advice or assistance as requested by the Company. This includes but is not limited to testifying (and preparing to testify) as a witness in any proceeding or otherwise providing information or reasonable assistance to the Company in connection with any investigation, claim or suit, and cooperating with the Company regarding any investigation, litigation, claims or other disputed items involving the Company that relate to matters within the knowledge or responsibility of the Executive. Specifically, the Executive agrees (i) to meet with the Company's representatives, its counsel or other designees at reasonable times and places with respect to any items within the scope of this provision; (ii) to provide truthful testimony regarding same to any court, agency or other adjudicatory body; (iii) to provide the Company with immediate notice of contact or subpoena by any non-governmental adverse party (known to the Executive to be adverse to the Company or its interests); and (iv) to not voluntarily assist any such non-governmental adverse party or such non-governmental adverse party's representatives. The Executive acknowledges and understands that the Executive's obligations of cooperation under this Section 10 are not limited in time and may include, but shall not be limited to, the need for or availability for testimony. The Executive shall receive no additional compensation for time spent assisting the Company pursuant to this Section 10 other than the compensation and benefits provided for in this Agreement, provided that the Executive shall be entitled to be reimbursed for any reasonable out-of-pocket expenses incurred in fulfilling the Executive's obligations pursuant to subsections (i) and (ii) above. Notwithstanding the foregoing, nothing in this Section 10 is intended to interfere with the Executive's No Interference rights set forth in Section 1(c) of the Confidentiality and Non-Competition Agreement.

# 11. <u>Arbitration</u>

(a) The Executive agrees that any dispute and/or claim between the Company (including without limitation its officers, directors, employees agents or shareholders and its subsidiaries) and the Executive that underlies, relates to and/or results from the Executive's employment relationship with the Company or the termination of that relationship or any of the terms of this Agreement, except for any dispute or claim arising from or relating to the Confidentiality and Non-Competition Agreement, that cannot be resolved by mutual agreement of the Company and the Executive will

be submitted to final, binding arbitration to the maximum extent permitted by law in accordance with the National Rules for the Resolution of Employment Disputes of the American Arbitration Association that are then in effect. This arbitration provision includes, but is not limited to, claims of wrongful discharge, infliction of emotional distress, breach of contract (including breach of this Agreement), breach of any covenant of good faith and fair dealing, and claims of retaliation and/or discrimination in violation of any local, state or federal law. Examples of such laws include Title VII of the Civil Rights Act of 1964; the Age Discrimination in Employment Act of 1967; the Americans with Disabilities Act of 1990; and the Family and Medical Leave Act of 1993, and all amendments to each such law as well as the regulations issued thereunder. This arbitration provision does not affect the Executive's right to pursue worker's compensation or unemployment compensation benefits for which he may be eligible in accordance with state law, nor does it affect the Executive's right to file and/or to cooperate in the investigation of an administrative charge of discrimination.

- (b) Notwithstanding this arbitration provision, either the Executive or the Company may apply to any court of competent jurisdiction for a temporary restraining order, preliminary injunction, or other interim or conservatory relief, as necessary, without breach of this Agreement and without abridgement of the powers of the arbitrator.
- (c) This arbitration provision does not apply to any dispute or claim arising from or relating to the Confidentiality and Non-Competition Agreement.
- (d) The Company, as further consideration for the Executive's agreement to arbitrate covered disputes, agrees to pay for the arbitrator's fees and other costs directly associated with the arbitration that would not otherwise be charged if the parties pursued civil litigation in court.

## 12. Successors; Personal Services

The services and duties to be performed by the Executive hereunder are personal and may not be assigned or delegated. This Agreement shall be binding upon and inure to the benefit of the Company and its successors and assigns, and the Executive and the Executive's heirs and representatives.

## 13. Notices

Notices and all other communications contemplated by this Agreement shall be in writing and shall be deemed to have been duly given when personally delivered or when mailed by U.S. registered or certified mail, return receipt requested and postage prepaid. In the case of the Executive, mailed notices shall be addressed to the Executive at the home address the Executive most recently communicated to the Company in writing. In the case of the Company, mailed notices shall be addressed to its corporate headquarters, and all notices shall be directed to the attention of its President and Chief Executive Officer.

#### 14. Section 409A

- (a) The parties intend that this Agreement and the payments and benefits provided hereunder be exempt from the requirements of Section 409A, to the maximum extent possible, whether pursuant to the short-term deferral exception described in Treasury Regulation Section 1.409A-1(b)(4), the involuntary separation pay plan exception described in Treasury Regulation Section 1.409A-1(b)(9)(iii), or otherwise. To the extent Section 409A is applicable to this Agreement, the parties intend that this Agreement and any payments and benefits thereunder comply with the deferral, payout and other limitations and restrictions imposed under Section 409A. Notwithstanding anything herein to the contrary, this Agreement shall be interpreted, operated and administered in a manner consistent with such intentions.
- (b) Without limiting the generality of the foregoing, and notwithstanding any other provision of this Agreement to the contrary:
- (i) if the Executive is deemed on the date of termination to be a "specified employee" within the meaning of that term under Section 409A, then with regard to any payment that is considered a "deferral of compensation" under Section 409A payable on account of a "separation from service," such payment shall be made on the date which is the earlier of (A) the date that is six months and one day after the date of such "separation from service" of the Executive and (B) the date of the Executive's death (the "*Delay Period*"), to the extent required under Section 409A. Within ten business days following the expiration of the Delay Period, all payments delayed pursuant to this Section 14(b)(i) (whether they would have otherwise been payable in a single sum or in installments in the absence of such delay) shall be paid to the Executive in a lump sum, and all remaining payments due under this Agreement shall be paid or provided in accordance with the normal payment dates specified for those payments in this Agreement;
- (ii) to the extent that any payments or benefits under this Agreement are conditioned on a Release, if the Release is executed and delivered by the Executive to the Company and becomes irrevocable and effective within the specified 60-day post-termination period, then, subject to Section 14(b)(i) and to the extent not exempt under Section 409A, such payments or benefits shall be made or commence on the first payroll date after the date that is 60 days after the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date). If a payment or benefit under this Agreement is conditioned on a Release and such Release is not executed, delivered and effective by the 60th day after the Termination Date, such payment or benefit shall not be paid or provided to the Executive;
- (iii) all expenses or other reimbursements under this Agreement shall be made on or prior to the last day of the taxable year following the taxable year in which such expenses were incurred by the Executive. No such reimbursement or expenses eligible for reimbursement in any taxable year shall in any way affect the expenses eligible for reimbursement in any other taxable year, and the Executive's right to reimbursement shall not be subject to liquidation or exchange for any other benefit;

- (iv) for purposes of Section 409A, the Executive's right to receive a series of installment payments pursuant to this Agreement shall be treated as a right to receive a series of separate and distinct payments. Whenever a payment under this Agreement specifies a payment period with reference to a number of days (e.g., "payment shall be made within 30 days"), the actual date of payment within the specified period shall be within the sole discretion of the Company;
- (v) in no event shall any payment under this Agreement that constitutes a "deferral of compensation" for purposes of Section 409A be offset by any other payment pursuant to this Agreement or otherwise; and
- (vi) to the extent required for purposes of compliance with Section 409A, termination of employment shall not be deemed to have occurred for purposes of any provision of this Agreement providing for the payment of any amounts or benefits upon or following a termination of employment unless such termination is also a "separation from service" within the meaning of Section 409A, and for purposes of any such provision of this Agreement, references to a "termination," "termination of employment" or like terms shall mean "separation from service."
- (c) The Company and the Executive agree to work together in good faith to consider amendments to this Agreement and to take such reasonable actions that may be necessary, appropriate, or desirable to avoid imposition of additional tax or income recognition on the Executive under Section 409A, in each case to the maximum extent permitted by applicable law. Notwithstanding any provision of this Agreement to the contrary, (i) in no event will the Company be liable for any additional tax, interest or penalty that may be imposed on the Executive by Section 409A or damages for failing to comply with Section 409A and (ii) the Executive acknowledges and agrees that the Executive will not have any claim or right of action against the Company or any of its employees, officers, directors or agents in the event it is determined that any payment or benefit provided hereunder does not comply with Section 409A.

# 15. <u>Miscellaneous Provisions</u>

- (a) <u>Waiver</u>. No provision of this Agreement shall be modified, waived or discharged unless the modification, waiver or discharge is agreed to in writing and signed by the Executive and by an authorized officer of the Company (other than the Executive). No waiver by either party of any breach of, or of compliance with, any condition or provision of this Agreement by the other party shall be considered a waiver of any other condition or provision or of the same condition or provision at another time.
- (b) Entire Agreement. This Agreement (including exhibits) shall supersede and replace all prior agreements or understandings relating to the subject matter hereof, and no agreements, representations or understandings (whether oral or written or whether express or implied) that are not expressly set forth in this Agreement have been made or entered into by either party with respect to the relevant matters hereof. This Agreement may not be modified except expressly in a writing signed by both parties.
- (c) <u>Disclaimer of Reliance</u>. Except for the specific representations expressly made by the Company in this Agreement, the Executive specifically disclaims that the Executive is relying

upon or has relied upon any communications, promises, statements, inducements, or representation(s) that may have been made, oral or written, regarding the subject matter of this Agreement. The Executive represents that the Executive relied solely and only on the Executive's own judgment in making the decision to enter into this Agreement.

- (d) <u>Choice of Law</u>. The validity, interpretation, construction and performance of this Agreement shall be governed by the internal substantive laws of the State of Texas without reference to any choice of law rules.
- (e) <u>Severability</u>. The invalidity or unenforceability of any provision or provisions of this Agreement shall not affect the validity or enforceability of any other provision hereof, which shall remain in full force and effect.
- (f) No Assignment of Benefits. The rights of any person to payments or benefits under this Agreement shall not be made subject to option or assignment, either by voluntary or involuntary assignment or by operation of law, in respect of bankruptcy, garnishment, attachment or other creditor's process, and any action in violation of this Section 15(f) shall be void.
- (g) No Duty to Mitigate. The Executive shall not be required to mitigate the amount of any payment contemplated by this Agreement, nor shall any such payment be reduced by any earnings that the Executive may receive from any other source.
- (h) <u>Employment Taxes</u>. All payments made pursuant to this Agreement will be subject to withholding of all applicable income, employment and other taxes.
- (i) <u>Assignment by Company</u>. The Company may assign its rights under this Agreement to an affiliate (as defined under the Exchange Act), and an affiliate may assign its rights under this Agreement to another affiliate of the Company or to the Company. In the case of any such assignment, the term "Company" when used in a section of this Agreement shall mean the corporation that actually employs the Executive.
- (j) <u>Counterparts</u>. This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together will constitute one and the same instrument.

[Signature Page Follows]

IN WITNESS WHEREOF, each of the parties has executed this Agreement, in the case of the Company by its duly authorized officer, as of the day and year first above written.

# BLUCORA, INC.

By: /s/ John S. Clendening

Name: John S. Clendening

Title: President and Chief Executive Officer

# **EXECUTIVE:**

/s/ Curtis Campbell
Curtis Campbell

#### **EXHIBIT A**

#### CONFIDENTIALITY AND NON-COMPETITION AGREEMENT

This Confidentiality and Non-Competition ("Agreement") is entered into by and between Blucora, Inc., its subsidiaries, affiliates, successors and/or assigns (the "Company") and Curtis Campbell ("Executive"). The Effective Date of this Agreement is the date of Executive's execution of this Agreement. The Company and Executive shall be referred to herein individually as a "Party" and collectively as the "Parties."

NOW, THEREFORE, in consideration of the mutual covenants set forth herein, Executive's position with the Company, the Confidential Information (defined below), compensation and benefits provided to Executive, and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, and intending to be legally bound hereby, the Parties agree as follows:

## 1. <u>Confidential Information and Executive's Non-Disclosure Agreement.</u>

(a) Confidential Information. During Executive's employment with the Company, the Company shall provide Executive with Confidential Information (defined below), which is not known to the Company's competitors or within the Company's industry generally, which was developed by the Company over a long period of time and/or at its substantial expense, and which is of great competitive value to the Company. For purposes of this Agreement, "Confidential Information" includes all documents or information, in whatever form or medium, concerning or relating to any of the following: all trade secrets and confidential and proprietary information of or relating to the Company, including, but not limited to: (A) financial models, business records, business plans or processes, strategies (including, without limitation, economic and market research selection and analysis strategies and business development and market segment exploitation strategies), tactics, policies, resolutions, processes, inventions, patents, trademarks, trade secrets, know how, patent or trademark applications and other intellectual property, (B) information regarding litigation or negotiations, (C) any marketing information, sales or product plans, prospects and market research data relating to the business, (D) financial information, cost and performance data and any debt arrangements, equity ownership or securities transaction information, (E) technical information, technical drawings and designs, (F) personnel information, personnel lists, resumes, personnel data, organizational structure, compensation and performance evaluations, (G) customer, consumer, consultants or supplier information, including but not limited to any data regarding any current, prospective or former customers, consumers, consultants or suppliers of Company, (H) information regarding the existence or terms of any agreement or relationship between the Company or any of its subsidiaries or affiliates and any other party, (I) information subject to Section 628 of the Fair Credit Reporting Act and any regulations or guidelines thereunder and (J) any other information of whatever nature, including, without limitation, information which gives to the Company or any of its subsidiaries or affiliates an opportunity to obtain an advantage over its competitors who or which do not have access to such information. Confidential Information, whether prepared or compiled by Executive and/or the Company or furnished to Executive during Executive's employment with the Company, shall be the sole and exclusive property of the Company, and none of such Confidential Information or copies thereof, shall be retained by Executive.

Executive agrees not to dispute, contest, or deny any such ownership rights either during or after Executive's employment with the Company. Executive acknowledges that the Company does not voluntarily disclose Confidential Information, but rather takes precautions to prevent dissemination of Confidential Information beyond those employees such as Executive entrusted with such information. Executive further acknowledges that the Confidential Information: (a) is entrusted to Executive because of Executive's position with the Company; and (b) is of such value and nature as to make it reasonable and necessary for Executive to protect and preserve the confidentiality and secrecy of the Confidential Information. Executive acknowledges and agrees that the Confidential Information is proprietary to and a trade secret of the Company and, as such, is a valuable, special and unique asset of the Company, the unauthorized use or disclosure of which will cause irreparable harm, substantial injury and loss of profits and goodwill to the Company. "Confidential Information" does not include any information which is generally available to and known by the public (other than as a result of Executive's breach of this Agreement or any other agreement or obligation to keep such information confidential).

#### (b) Non-Disclosure.

(i) Executive agrees to preserve and protect the confidentiality of all Confidential Information. Executive agrees that during the period of Executive's employment with the Company and at any time thereafter (regardless of the reason for Executive's separation or termination of employment): (A) Executive shall hold all Confidential Information in the strictest confidence, take all reasonable precautions and steps to safeguard all Confidential Information and prevent its wrongful use by or wrongful or inadvertent disclosure or dissemination to any unauthorized person or entity, and follow all policies and procedures of the Company protecting or regarding the Confidential Information; and (B) without prior written authorization of the Company, Executive shall not, directly or indirectly, use for Executive's own account, use in any way or for any other purpose, disclose to anyone, publish, exploit, destroy, copy or remove from the offices of the Company, nor solicit, allow or assist another person or entity to use, disclose, publish, exploit, destroy, copy or remove from the offices of the Company, any Confidential Information or part thereof, except: (1) as permitted in the proper performance of Executive's duties for the Company; (2) as permitted in the ordinary course of the Company's business for the benefit of the Company; or (3) as otherwise permitted or required by law. Executive shall immediately notify the Company if Executive learns of or suspects any actual or potential unauthorized use or disclosure of Confidential Information concerning the Company. Further, the Executive shall not, directly or indirectly, use the Company's Confidential Information to: (1) call upon, solicit business from, attempt to conduct business with, conduct business with, interfere with or divert business away from any customer, client, service provider, supplier or vendor of the Company with whom or which the Company conducted business; and/or (2) recruit, solicit, hire or attempt to recruit, solicit, or hire, directly or by assisting others, any persons employed by the Company. In the event Executive is subpoenaed, served with any legal process or notice, or otherwise requested to produce or divulge, directly or indirectly, any Confidential Information by any entity, agency or person in any formal or informal proceeding including, but not limited to, any interview, deposition, administrative or judicial hearing and/or trial, except where prohibited by law, Executive should immediately notify the Company and deliver a copy of the subpoena, process, notice or other request to the Company

as promptly as possible, but under no circumstances more than ten (10) days following Executive's receipt of same; provided, however, Executive is not required to notify the Company or provide a copy of the subpoena, process, notice or other request where Executive is permitted to make such disclosure of Confidential Information pursuant to this Agreement or applicable law or regulation, as set forth in Section 1(c) and Section 1(d).

- (ii) Subject to Section 1(b)(iii), Executive agrees that Executive will not use or disclose any confidential, proprietary or trade secret information belonging to any former employer or third party, and Executive will not bring onto the premises of the Company or onto any Company property, any confidential, proprietary or trade secret information belonging to any former employer or third party without such third party's written consent. Executive acknowledges that that the Company has specifically instructed Executive not to disclose to the Company, use, or induce the Company to use, any confidential, proprietary or trade secret information belonging to any previous employer or others.
- (iii) During Executive's employment, the Company will receive from third parties their confidential and/or proprietary information, subject to a duty on the Company's part to maintain the confidentiality of and to use such information only for certain limited purposes. Executive agrees to hold all such confidential or proprietary information in the strictest confidence and not to disclose it to any person or organization or to use it except as necessary in the course of Executive's employment with the Company and in accordance with the Company's agreement with such third party.
- (iv) Except in the proper performance of Executive's duties and responsibilities, Executive agrees that Executive shall not remove, destroy, deface, damage or delete any Property of the Company. For purposes of this Agreement, the term "*Property*" means all property or information, in whatever form or media, and all copies thereof whether or not the original was deleted or destroyed, of the Company, including, without limitation, any Confidential Information, software, hardware, including any and all Company-issued equipment, devices, cellular telephones, tablets, computers, laptops, hard drives, keys, access cards, access codes or passwords belonging to the Company, databases, files, records, reports, memoranda, research, plans, proposals, lists, forms, drawings, specifications, notebooks, manuals, correspondence, materials, e-mail, electronic or magnetic recordings or data, and any other physical or electronic documents that Executive receives from or sends to any employee of the Company, that Executive copies from the files or records of the Company, or that Executive otherwise has access to during Executive's employment.
- (c) No Interference. Notwithstanding any other provision of this Agreement, (i) Executive may disclose Confidential Information when required to do so by a court of competent jurisdiction, by any governmental agency having authority over Executive or the business of the Company or by any administrative body or legislative body (including a committee thereof) with jurisdiction to order Executive to divulge, disclose or make accessible such information; and (ii) nothing in this Agreement is intended to interfere with Executive's right to (A) report possible violations of state or federal law or regulation to any governmental or law enforcement agency or entity; (B) make other disclosures that are protected under the whistleblower provisions of state or federal law or regulation; (C) file a claim or charge with any governmental agency or entity; or (D)

testify, assist, or participate in an investigation, hearing, or proceeding conducted by any governmental or law enforcement agency or entity, or any court. For purposes of clarity, in making or initiating any such reports or disclosures or engaging in any of the conduct outlined in subsection (ii) above, Executive may disclose Confidential Information to the extent necessary to such governmental or law enforcement agency or entity or such court, need not seek prior authorization from the Company, and is not required to notify the Company of any such reports, disclosures or conduct.

(d) <u>Defend Trade Secrets Act</u>. Executive is hereby notified in accordance with the Defend Trade Secrets Act of 2016 that Executive will not be held criminally or civilly liable under any federal or state trade secret law for the disclosure of a trade secret that is made in confidence to a federal, state, or local government official, either directly or indirectly, or to an attorney solely for the purpose of reporting or investigating a suspected violation of law, or is made in a complaint or other document that is filed under seal in a lawsuit or other proceeding.

## (e) Inventions.

- (i) Prior Inventions Retained and Licensed. In Exhibit A-1 to this Agreement, Executive has provided a list describing all Inventions (defined below) that Executive: (A) conceived, created, developed, made, reduced to practice or completed, either alone or with others, prior to Executive's employment with the Company; (B) claims a proprietary right or interest in; and (C) does not assign to the Company hereunder (collectively referred to as the "Prior Inventions"). If no such list is attached, Executive represents that there are no such Prior Inventions. Executive understands and agrees that the Company makes no attempt to verify Executive's claim of ownership to any of the Prior Inventions. Executive agrees that Executive shall not incorporate in any work that Executive performs for the Company any Prior Inventions or any of the technology described in any Prior Inventions. Nonetheless, if in the course of Executive's employment with the Company, Executive incorporates Prior Inventions into a product, service, process or machine of the Company, Executive hereby grants and shall be deemed to have granted the Company a nonexclusive, royalty-free, irrevocable, sublicensable, transferable, perpetual, and worldwide license to make, have made, modify, use, import, reproduce, distribute, prepare and have prepared derivative works of, offer to sell, sell and otherwise exploit such Prior Inventions. For purposes of this Agreement, the term "Inventions" means all tangible and intangible materials, work product, information, methods, designs, computer programs, software, databases, formulas, models, prototypes, reports, discoveries, ideas, improvements, know-how, compositions of matter, processes, photographs, drawings, illustrations, sketches, developments, and all related intellectual property, including inventions, original works of authorship, moral rights, mask works, trade secrets and trademarks.
- (ii) <u>Assignment of Inventions</u>. During Executive's employment with the Company and following the termination of Executive's employment for any reason, Executive agrees that Executive shall promptly make full written disclosure to the Company, shall hold in trust for the sole right and benefit of the Company, and hereby assigns and shall be deemed to have assigned to the Company or its designee, all of Executive's right, title, and interest in and to any and all Inventions that have been or may be conceived, created, developed, completed, reduced to

practice or otherwise made by Executive, solely or jointly with others, during the period of Executive's employment with the Company which (A) relate in any manner to the existing or contemplated business, work, or investigations of the Company; (B) are suggested by, result from, or arise out of any work that Executive may do for or on behalf of the Company; (C) result from or arise out of any Confidential Information that may have been disclosed or otherwise made available to Executive as a result of duties assigned to Executive by the Company; or (D) are otherwise made through the use of the time, information, equipment, facilities, supplies or materials of the Company, even if developed, conceived, reduced to practice or otherwise made during other than working hours (collectively referred to as "Company Inventions"). Executive further acknowledges that all original works of authorship that are made by Executive (solely or jointly with others) within the scope of Executive's employment with the Company and that are protectable by copyright are "Works Made for Hire," as that term is defined in the United States Copyright Act. Executive understands and agrees that the decision whether or not to commercialize or market any Company Inventions is within the Company's sole discretion and for the Company's sole benefit, and that no royalty will be due to Executive as a result of the Company's efforts to commercialize or market any such Company Invention.

- (iii) <u>Maintenance of Records</u>. Executive agrees to keep and maintain adequate and current hard-copy and electronic records of all Company Inventions. The records will be available to and remain the sole property of the Company during Executive's employment with the Company and at all times thereafter.
- (f) Patent and Copyright Registrations. Executive agrees to assist the Company or its designee, at the Company's expense, in every proper way to secure the Company's rights in Company Inventions in any and all countries, including the disclosure to the Company of all pertinent information and data with respect thereto, the execution of all applications, specifications, oaths, assignments, affidavits, and all other instruments which the Company shall deem necessary in order to apply for and obtain such rights and in order to assign and convey to the Company and/or its successors, assigns and nominees, the sole and exclusive rights, title and interest in and to such Company Inventions. Executive further agrees that Executive's obligation to execute or cause to be executed, when it is in Executive's power to do so, any such instrument or papers shall continue after the termination of this Agreement. Executive hereby appoints the General Counsel of the Company as Executive's attorney-in-fact to execute documents on Executive's behalf for this purpose. Executive agrees that this appointment is coupled with an interest and will not be revocable.
- (g) Return of Company Property. Upon request by the Company or upon the termination of Executive's employment for any reason, Executive shall immediately return and deliver to the Company any and all Property, including, without limitation, Confidential Information, software, hardware, including any and all Company-issued equipment, devices, cellular telephones, tablets, computers, laptops, hard drives, keys, access cards, access codes or passwords, databases, files, documents, records, reports, memoranda, research, plans, proposals, lists, papers, books, forms, drawings, specifications, notebooks, manuals, correspondence, materials, e-mail, electronic or magnetic recordings or data, including all copies thereof (in electronic or hard copy format), which belong to the Company or which relate to the Company's business and which are in Executive's possession, custody or control, whether prepared by Executive or others. Executive further agrees

that after Executive provides such Property to the Company, Executive will immediately destroy any information or documents, whether prepared by Executive or others, containing or reflecting any Confidential Information or relating to the business of the Company from any computer, cellular phone or other digital or electronic device in Executive's possession, custody or control, and Executive shall certify such destruction in writing to the Company. Upon request by the Company, Executive shall provide such computer, cellular phone or other digital or electronic device to the Company or the Company's designee for inspection to confirm that such information and documents have been destroyed. If at any time after the termination of Executive's employment for any reason, Executive or the Company determines that Executive has any Property in Executive's possession, custody or control, Executive shall immediately return all such Property, including all copies and portions thereof, to the Company.

- 2. Restrictive Covenants. In consideration for (i) the Company's promise to provide Confidential Information; (ii) the substantial economic investment made by the Company in the Confidential Information and goodwill of the Company, and/or the business opportunities disclosed or entrusted to Executive; (iii) access to the customers and clients of the Company; and (iv) the Company's employment of Executive in an executive position and the compensation and other benefits provided by the Company to Executive (including the consideration provided for in the Incentive Retention Letter), to protect the Confidential Information and business goodwill of the Company, Executive agrees to the following restrictive covenants.
- (a) Non-Competition. Executive agrees that during the Executive's employment with the Company and for a period of twelve (12) months after the Executive's employment terminates for any reason (the "Restricted Period"), other than in connection with Executive's performance of his duties for the Company, Executive shall not, and shall not use any Confidential Information to, without the prior written consent of an officer of the Company, directly or indirectly, either individually or as a principal, partner, stockholder, manager, agent, consultant, contractor, distributor, employee, lender, investor, or as a director or officer of any corporation or association, or in any other manner or capacity whatsoever, (i) control, manage, operate, establish, take steps to establish, lend money to, invest in, solicit investors for, or otherwise provide capital to, or (ii) become employed by, join, perform services for, consult for, do business with or otherwise engage in any Competing Business within the Restricted Area. For purposes of this Consulting Agreement, given the scope of Confidential Information to be provided to Executive and job duties to be performed by the Executive, "Restricted Area" means the United States, and any other geographic area for which Executive performed any services or about which Executive received Confidential Information. For purposes of this Agreement, "Competing Business" means any business, individual, partnership, firm, corporation or other entity that is competing or that is preparing to compete with any aspect of the Company's business, which includes, but is not limited to (a) tax preparation and tax preparation-related products and services provided to consumers and small businesses, and to or through tax professionals; (b) investment and insurance products or services, and related advice and brokerage services, provided to or through tax professionals or in conjunction with tax preparation services, and (c) any other business the Company engages in or develops during the Executive's employment with the Company.

- (b) Non-Solicitation. During the Restricted Period, other than in connection with Executive's duties for the Company, Executive shall not, and shall not use any Confidential Information to, directly or indirectly, either as a principal, manager, agent, employee, consultant, officer, director, stockholder, partner, investor or lender or in any other capacity, and whether personally or through other persons, solicit business from, interfere with, or induce to curtail or cancel any business or contracts with the Company, or attempt to solicit business with, interfere with, or induce to curtail or cancel any business or contracts with the Company, or do business with any actual or prospective customer or client of the Company with whom the Company did business or who the Company solicited within the preceding two (2) years, and who or which: (1) Executive contacted, called on, serviced or did business with during Executive's employment with the Company; (2) Executive learned of as a result of Executive's employment with the Company; or (3) about whom Executive received Confidential Information. This restriction applies only to business which is in the scope of services or products provided by the Company.
- (c) <u>Non-Recruitment</u>. During the Restricted Period, other than in connection with Executive's duties for the Company, Executive shall not, on behalf of Executive or on behalf of any other person or entity, directly or indirectly, hire, solicit or recruit, or attempt to hire, solicit or recruit, or encourage to leave or otherwise cease his/her employment or engagement with the Company, any individual who is an employee or independent contractor of the Company or who was an employee or independent contractor of the Company within the twelve (12) month period prior to Executive's separation from employment with the Company.
- (d) Non-Disparagement. Executive agrees that the Company's goodwill and reputation are assets of great value to the Company, which have been obtained and maintained through great costs, time and effort. Therefore, Executive agrees that during Executive's employment and after the termination of Executive's employment, Executive shall not make, publish or otherwise transmit any knowingly false statements, whether written or oral, regarding the Company and its officers, directors, executives, employees, contractors, consultants, products, services, business or business practices. A violation or threatened violation of this Section 2(d) may be enjoined by the courts. The rights afforded the Company under this provision are in addition to any and all rights and remedies otherwise afforded by law. However, nothing in this Section 2(d) restricts or prevents Executive from providing truthful testimony as required by court order or other legal process or is intended to interfere with Executive's rights set forth in Section 1(c).
- (e) <u>Tolling</u>. If Executive violates any of the covenants contained in this Section 2, the Restricted Period applicable to such covenant(s) shall be suspended and shall not run in favor of Executive from the time of the commencement of such violation until the time that Executive cures the violation to the satisfaction of the Company and the period of time in which Executive is in breach shall be added to the Restricted Period applicable to such covenant(s).
- 3. **Reasonableness**. Executive hereby represents to the Company that Executive has read and understands, and agrees to be bound by, the terms of Section 1 and Section 2. Executive acknowledges that the scope and duration of the restrictions and covenants contained in Section 1 and Section 2 are fair and reasonable in light of (i) the nature and scope of the operations of the Company's business; and (ii) the amount of compensation and Confidential Information (including,

without limitation, trade secrets) that Executive is receiving in connection with Executive's employment with the Company and the Incentive Retention Letter. It is the desire and intent of the Parties that the provisions of Section 1 and Section 2 be enforced to the fullest extent permitted under applicable law, whether now or hereafter in effect and therefore, to the extent permitted by applicable law, Executive and the Company hereby waive any provision of applicable law that would render any provision of Section 1 and/or Section 2 invalid or unenforceable.

- 4. Remedies. Executive acknowledges that the restrictions and covenants contained in Section 1 and Section 2, in view of the nature of the Company's business and Executive's position with the Company, are reasonable and necessary to protect the Company's legitimate business interests, goodwill and reputation, and that any violation of Section 1 or Section 2 would result in irreparable injury and continuing damage to the Company, and that money damages would not be a sufficient remedy to the Company for any such breach or threatened breach. Therefore, Executive agrees that the Company shall be entitled to a temporary restraining order and injunctive relief restraining Executive from the commission of any breach or threatened breach of Section 1 and/or Section 2, without the necessity of establishing irreparable harm or the posting of a bond, and to recover from Executive damages incurred by the Company as a result of the breach, as well as the Company's attorneys' fees, costs and expenses related to any breach or threatened breach of this Agreement and enforcement of this Agreement. Nothing contained in this Agreement shall be construed as prohibiting the Company from pursuing any other remedies available to it for any breach or threatened breach, including, without limitation, the recovery of money damages, attorneys' fees, and costs. The existence of any claim or cause of action by Executive against the Company, whether predicated on this Agreement or otherwise, shall not constitute a defense to the enforcement by the Company of the restrictions or covenants contained in Section 1 or Section 2, or preclude injunctive relief.
- 5. **Business Opportunities**. Executive, without further compensation, assigns and agrees to assign to the Company and its successors, assigns or designees, all of Executive's right, title and interest in and to all Business Opportunities (defined below), and further acknowledges and agrees that all Business Opportunities constitute the exclusive property of the Company. Executive shall present all Business Opportunities to the Company, and shall not exploit a Business Opportunity. For purposes of this Agreement, "Business Opportunities" means all business ideas, prospects, or proposals pertaining to any aspect of the Company's business and any business the Company prepared to conduct or contemplated conducting during Executive's employment with the Company, which are developed by Executive or originated by any third party and brought to the attention of Executive, together with information relating thereto. For the avoidance of doubt, this Section 5 is not intended to limit or narrow Executive's duties or obligations under federal or state law with respect to corporate opportunities.
- 6. <u>Conflicting Activities</u>. Executive agrees that, during Executive's employment with the Company, Executive shall not engage in any employment, consulting relationship, business or other activity that (i) is in any way competitive with the business or proposed business of the Company (except that Executive may invest less than one percent (1%) of the shares of a company traded on a registered stock exchange); (ii) conflicts with Executive's duty of loyalty, responsibilities or obligations to the Company or interferes with the independent exercise of Executive's judgment

in the Company's best interests; or (iii) adversely affects the performance of Executive's job duties and responsibilities with the Company. Executive agrees to not assist any other person or organization in competing with the Company or in preparing to engage in competition with the Company or proposed business of the Company. Executive further agrees that, during Executive's employment with the Company, Executive shall not actively engage in any other employment, occupation or consulting activity for any direct or indirect remuneration without the prior approval of Executive's direct supervisor or the Company's Legal Department. Executive has listed on the Company's Outside Activity Disclosure form, attached hereto as Exhibit B-1, any business activities or ventures with which Executive is involved. If no such list is attached, Executive represents that there are no such outside activities as of the date of this Agreement.

- 7. **Breach** Executive acknowledges that Executive is subject to immediate dismissal by the Company for any breach of this Agreement and that such a dismissal will not relieve Executive from any continuing obligations under this Agreement or from the imposition by a court of any judicial remedies, including, without limitation, money damages and/or injunctive relief for such breach.
- 8. <u>Notice</u>. If Executive, in the future, seeks or is offered employment, or any other position or capacity with another company, entity or person, Executive agrees to inform each such company, entity or person of the existence of the restrictions in Section 1 and Section 2. The Company shall be entitled to advise such company, entity or person and third parties of the provisions of Section 1 and Section 2 and to otherwise deal with such company, entity, person or third party to ensure that the provisions of Section 1 and Section 2 are enforced and duly discharged.
- 9. **Reformation**. The Company and Executive agree that in the event any of the terms, provisions, covenants or restrictions contained in this Agreement, or any part thereof, shall be held by any court of competent jurisdiction to be effective in any particular area or jurisdiction only if said term, provision, covenant or restriction is modified to limit its duration or scope, then the court shall have such authority to so reform the term, provision, covenant or restriction and the Parties hereto shall consider such term, provision, covenant or restriction to be amended and modified with respect to that particular area or jurisdiction so as to comply with the order of any such court and, as to all other jurisdictions, the term, provision, covenant or restriction contained herein shall remain in full force and effect as originally written. By agreeing to this contractual modification prospectively at this time, the Company and Executive intend to make Section 1 and Section 2 enforceable under the law or laws of all applicable jurisdictions so that the restrictive covenants in their entirety and this Agreement as prospectively modified shall remain in full force and effect and shall not be rendered void or illegal.
- 10. Severability. In the event any court of competent jurisdiction or any foreign, federal, state, county or local government or any other governmental regulatory or administrative agency or authority holds any provision of this Agreement to be invalid, illegal or unenforceable, such invalid, illegal or unenforceable portion(s) shall be limited or excluded from this Agreement to the minimum extent required, and the remaining provisions shall not be affected or invalidated and shall remain in full force and effect.

- 11. Binding Effect of Agreement and Assignment. This Agreement shall be binding upon and inure to the benefit of the Parties hereto and their respective heirs, successors, legal representatives and permitted assigns. Executive may not assign this Agreement to a third party. The Company may assign its rights, together with its obligations hereunder, to any affiliate and/or subsidiary of the Company or any successor thereto or any purchaser of substantially all of the assets of the Company, without Executive's consent and without advance notice.
- 12. <u>Survival</u>. Executive agrees that Executive's obligations under this Agreement shall continue in effect after the termination of Executive's employment, regardless of the reason(s) for termination, and whether such termination is voluntary or involuntary.
- 13. Waiver. The failure of either Party to insist in any one or more instances upon performance of any terms or conditions of this Agreement shall not be construed as a waiver of future performance of any such term, covenant or condition, but the obligations of either Party with respect thereto shall continue in full force and effect. No waiver of any breach of this Agreement shall be construed to be a waiver as to succeeding breaches and no waiver of any provisions of this Agreement shall constitute a waiver of any other provision of this Agreement. The breach by one Party to this Agreement shall not preclude equitable relief, injunctive relief or the obligations in Section 1 or Section 2.
- 14. **Controlling Law.** This Agreement shall be governed by and construed under the laws of the State of Texas, without regard to any applicable conflict of law or choice of law rules.
- 15. Venue. Venue of any dispute arising out of, in connection with or in any way related to this Agreement shall be in a state district court of competent jurisdiction in Dallas County, Texas, or the United States District Court for the Northern District of Texas. Executive consents to personal jurisdiction of the state district courts of Dallas County, Texas and to the United States District Court for the Northern District of Texas for any dispute arising out of, in connection with or in any way related to this Agreement, and agrees that Executive shall not challenge personal jurisdiction in such courts. Executive waives any objection that Executive may now or hereafter have to the venue or jurisdiction of any proceeding in such courts or that any such proceeding was brought in an inconvenient forum (and agrees not to plead or claim the same).
- 16. WAIVER OF JURY TRIAL. WITH RESPECT TO ANY DISPUTE BETWEEN EXECUTIVE AND THE COMPANY ARISING OUT OF, IN CONNECTION WITH OR IN ANY WAY RELATED TO THIS AGREEMENT, EMPLOYEE AGREES TO RESOLVE SUCH DISPUTE(S) BEFORE A JUDGE WITHOUT A JURY. EMPLOYEE HAS KNOWLEDGE OF THIS PROVISION, AND WILL PROVIDE SERVICES TO THE COMPANY THEREAFTER, HEREBY WAIVING EXECUTIVE'S RIGHT TO TRIAL BY JURY AND AGREES TO HAVE ANY DISPUTE(S) ARISING BETWEEN THE COMPANY AND EXECUTIVE ARISING OUT OF, IN CONNECTION WITH OR IN ANY WAY RELATED TO THIS AGREEMENT RESOLVED BY A JUDGE OF A COMPETENT COURT IN DALLAS COUNTY, TEXAS, SITTING WITHOUT A JURY.
- 17. **Entire Agreement**. This Agreement constitutes the entire agreement between the Parties with respect to the subject matter hereof, and fully supersedes any and all prior and

contemporaneous agreements, understandings and/or representations between the Parties, whether oral or written, pertaining to the subject matter of this Agreement; provided, however, Executive's obligations under this Agreement are in addition to Executive's obligations under any applicable law or regulation and the Company's policies and procedures. No oral statements or prior written material not specifically incorporated in this Agreement shall be of any force and effect, and no changes in or additions to this Agreement shall be recognized, unless incorporated in this Agreement by written amendment, such amendment to become effective on the date stipulated in it. Any amendment to this Agreement must be signed by all parties to this Agreement.

- 18. <u>Disclaimer of Reliance</u>. Except for the specific representations expressly made by the Company in this Agreement, Executive specifically disclaims that Executive is relying upon or has relied upon any communications, promises, statements, inducements, or representation(s) that may have been made, oral or written, regarding the subject matter of this Agreement. Executive represents that Executive relied solely and only on Executive's own judgment in making the decision to enter into this Agreement.
- 19. <u>Voluntary Agreement</u>. Executive (i) acknowledges that Executive has read and understands the terms of this Agreement and believes them to be reasonable, (ii) agrees that the consideration provided by the Company for this Agreement is reasonable, and (iii) is voluntarily executing this Agreement as signified by Executive's signature hereto,.
- 20. <u>Execution in Multiple Counterparts</u>. This Agreement may be executed in multiple counterparts, whether or not all signatories appear on these counterparts, and each counterpart shall be deemed an original for all purposes.

[Signature Page Follows]

| The signatures below | indicate that the | Parties have read, | understand and | l will comply with | this Agreement. |
|----------------------|-------------------|--------------------|----------------|--------------------|-----------------|
|                      |                   |                    |                |                    |                 |

| EXECUTIVE:                                      | Signature:                    |  |
|---|-------------------------------|--|
|   | Printed Name: Curtis Campbell |  |
|   | Date:                         |  |
| THE COMPANY:                                    | Blucora, Inc.:                |  |
|   | Signature:                    |  |
|   | Name:                         |  |
|   | Title:                        |  |
|   | Date:                         |  |
|   |                               |  |
|   |                               |  |
| CONFIDENTIALITY AND NON-COMPETITION AGREEMENT P | age 12                        |  |

# EXHIBIT A-1 LIST OF PRIOR INVENTIONS

| Title                                      | Date         | Identifying Number or Brief Description |
|--|--------------|---|
|  |              |   |
|  |              |   |
|  |              |   |
|  |              |   |
|  |              |   |
|  |              |   |
|  |              |   |
| No Inventions  Additional Sheets Attached  |              |   |
| Signature of Employee:                     |              |   |
| Print Name of Employee:                    |              |   |
| Date:                                      |              |   |
| CONFIDENTIALITY AND NON-COMPETITION AGREEM | IENT Page 13 |   |

# EXHIBIT A-2 OUTSIDE ACTIVITIES

#### **EXHIBIT B**

#### GENERAL RELEASE OF ALL CLAIMS

This General Release and Waiver of Claims (this "*Release*") is executed by Curtis Campbell ("*Executive*") and Blucora, Inc. (the "*Company*") as of the date set forth below, and will become effective as of the "*Effective Date*" as defined below. This Release is in consideration of severance benefits to be paid to Executive by the Company pursuant to the Employment Agreement between Executive and the Company dated as of \_\_\_\_\_, 2018 (the "*Employment Agreement*"). Execution of this Release without revocation by Executive will satisfy the requirement, set forth in Section 6(g) of the Employment Agreement, that Executive execute a general release and waiver of claims in order to receive severance benefits pursuant to the Employment Agreement.

## 1. Termination of Employment

Executive acknowledges that his employment with the Company and any of its subsidiaries (collectively, the "Company Group") and any and all appointments he held with any member of the Company Group, whether as officer, director, employee, consultant, agent or otherwise, terminated as of \_\_\_\_\_\_ (the "Termination Date"). Effective as of the Termination Date, Executive has not had or exercised or purported to have or exercise any authority to act on behalf of the Company or any other member of the Company Group, nor will Executive have or exercise or purport to have or exercise such authority in the future.

#### 2. Consideration

The Company shall pay Executive the severance benefits pursuant to the Employment Agreement. The Parties agree that but for signing this Release, Executive would not be entitled to the severance benefits set forth in the Employment Agreement. The severance benefits are adequate to make this Release final and binding, and are in addition to payments and benefits to which Executive would otherwise be entitled to as an employee or former employee of the Company.

#### 3. Waiver and Release

(a) Executive, for and on behalf of himself and his heirs and assigns, hereby waives and releases any common law, statutory or other complaints, claims, charges or causes of action arising out of or relating to Executive's employment or termination of employment with, or Executive's serving in any capacity in respect of any member of the Company Group (collectively, "Claims"). The Claims waived and released by this Release include any and all Claims, whether known or unknown, whether in law or in equity, which Executive may now have or ever had against any member of the Company Group or any shareholder, employee, officer, director, agent, attorney, representative, trustee, administrator or fiduciary of any member of the Company Group (collectively, the "Company Releasees") up to and including the date of Executive's execution of this Agreement. The Claims waived and released by this Release include, without limitation, any and all Claims arising out of Executive's employment with the Company Group under, by way of example and not limitation, the Age Discrimination in Employment Act of 1967 ("ADEA", a law which prohibits discrimination on the basis of age against persons age 40 and older), the National

1

Labor Relations Act, the Civil Rights Act of 1991, the Americans With Disabilities Act of 1990, Title VII of the Civil Rights Act of 1964, the Employee Retirement Income Security Act of 1974, the Family Medical Leave Act, the Securities Act of 1933, the Securities Exchange Act of 1934, and the Texas Labor Code Chapter 21, all as amended, and all other federal, state and local statutes, ordinances, regulations and the common law, and any and all Claims arising out of any express or implied contract, except as described in Paragraphs 2(b) and 2(c) below.

- (b) The waiver and release set forth in this Section 3 is intended to be construed as broadly and comprehensively as applicable law permits. The waiver and release shall not be construed as waiving or releasing any claim or right that as a matter of law cannot be waived or released, including Executive's right to file a charge with the Equal Employment Opportunity Commission or other government agency.
  - (c) Notwithstanding anything else in this Release, Executive does not waive or release claims with respect to:
    - (i) Executive's entitlement, if any, to severance benefits pursuant to the Employment Agreement;
- (ii) vested benefits or payments specifically to be provided to the Executive pursuant to the Employment Agreement or any Company employee benefit plans or policies;
- (iii) indemnification pursuant to any applicable provision of the Company's Bylaws or Certificate of Incorporation, as amended, pursuant to any written indemnification agreement between the Executive and the Company, or pursuant to applicable law;
- (iv) any claims which the Executive may have solely by virtue of the Executive's status as a shareholder of the Company; or
  - (v) unemployment compensation to which Executive may be entitled under applicable law.
- (d) Executive represents and warrants that he is the sole owner of the actual or alleged Claims that are released hereby, that the same have not been assigned, transferred, or disposed of in fact, by operation of law, or in any manner, and that he has the full right and power to grant, execute and deliver the releases, undertakings, and agreements contained herein.
- (e) Subject to Section 4, Executive represents that Executive has not filed any complaints, charges or lawsuits against the Company with any governmental agency or any court based on Claims that are released and waived by this Release.

#### 4. No Interference

Nothing in this Agreement is intended to interfere with Executive's right to report possible violations of federal, state or local law or regulation to any governmental or law enforcement agency or entity, or to make other disclosures that are protected under the whistleblower provisions of federal, state or local law or regulation. Executive further acknowledges that nothing in this

Agreement is intended to interfere with Executive's right to file a claim or charge with, or testify, assist, or participate in an investigation, hearing, or proceeding conducted by, the Equal Employment Opportunity Commission (the "*EEOC*"), any state human rights commission, or any other government agency or entity. However, by executing this Agreement, Executive hereby waives the right to recover any damages or benefits in any proceeding Executive may bring before the EEOC, any state human rights commission, or any other government agency or entity or in any proceeding brought by the EEOC, any state human rights commission, or any other government agency or entity on Executive's behalf with respect to any claim released in this Agreement *except that* Executive may receive bounty money awarded by the U.S. Securities and Exchange Commission pursuant to Section 21F of the Securities Exchange Act of 1934 or any similar provision.

#### 5. No Admission of Wrongdoing

This Release shall not be construed as an admission by either party of any wrongful or unlawful act or breach of contract.

## 6. Legal Disclosure

Subject to Section 4, by signing this Agreement, Executive warrants and represents that Executive has reported to Human Resources or Legal all pending and/or threatened legal proceedings of any kind involving or relating to the Company that Executive became aware of during Executive's tenure with the Company. Executive further warrants and represents that Executive has reported to Human Resources or Legal any alleged violations of law (including alleged securities violations) by the Company that Executive became aware of during Executive's tenure with the Company.

## 7. Binding Agreement; Successors and Assigns

This Release binds Executive's heirs, administrators, representatives, executors, successors, and assigns, and will inure to the benefit of the respective heirs, administrators, representatives, executors, successors, and assigns of any person or entity as to whom the waiver and release set forth in Section 3 applies.

#### 8. Other Agreements

This Release does not supersede or modify in any way Executive's continuing obligations pursuant to the Employment Agreement or the Confidentiality and Non-Competition Agreement (Exhibit A thereto) or the dispute resolution provisions of the Employment Agreement.

## 9. Knowing and Voluntary Agreement; Consideration and Revocation Periods

- (a) Executive acknowledges that Executive has been given twenty-one (21) calendar days from the date of receipt of this Release to consider all of the provisions of this Release and that if Executive signs this Release before the 21-day period has ended he knowingly and voluntarily waives some or all of such 21-day period.
- (b) Executive represents that (i) Executive has read this Release carefully, (ii) Executive has hereby been advised by the Company to consult an attorney of his choice and has either done

so or voluntarily chosen not to do so, (iii) Executive fully understands that by signing below he is giving up certain rights which he might otherwise have to sue or assert a claim against any of the Company Releasees, and (iv) Executive has not been forced or pressured in any manner whatsoever to sign this Release, and agrees to all of its terms voluntarily.

(c) Executive shall have seven (7) calendar days from the date of his execution of this Release (the "*Revocation Period*") in which Executive may revoke this Release. Such revocation must be in writing and delivered, prior to the expiration of the Revocation Period, to the attention of the Company's Chief Legal Officer at the Company's then-current headquarters address. If Executive revokes this Release during the Revocation Period, then the Release shall be null and void and without effect.

#### 10. Disclaimer of Reliance

Except for the specific representations expressly made by the Company in the Employment Agreement, Executive specifically disclaims that Executive is relying upon or has relied upon any communications, promises, statements, inducements, or representation(s) that may have been made, oral or written, regarding the subject matter of this Release. Executive represents that Executive relied solely and only on Executive's own judgment in making the decision to enter into this Release.

## 11. Execution in Multiple Counterparts

This Release may be executed by the parties in multiple counterparts, whether or not all signatories appear on these counterparts (including via electronic signatures and exchange of PDF documents via email), each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

#### 12. Effective Date

The Effective Date of this Release will be day after the Revocation Period expires without revocation by Executive.

[Signature Page Follows]

EXECUTIVE HAS ELECTED FREELY AND VOLUNTARILY TO EXECUTE THIS RELEASE, TO FULFILL THE PROMISES SET FORTH IN THE EMPLOYMENT AGREEMENT, AND TO RECEIVE THEREBY THE PAYMENT AND OTHER CONSIDERATION DESCRIBED IN THE EMPLOYMENT AGREEMENT. EXECUTIVE UNDERSTANDS THAT, BY SIGNING THIS RELEASE, EXECUTIVE IS AGREEING TO WAIVE AND SETTLE THE RELEASED CLAIMS HEREIN THAT EXECUTIVE HAS OR MIGHT HAVE AGAINST THE COMPANY INCLUDING CLAIMS UNDER THE AGE DISCRIMINATION IN EMPLOYMENT ACT. EXECUTIVE'S SIGNATURE BELOW MEANS THAT EXECUTIVE HAS READ THE RELEASE AND UNDERSTANDS AND AGREES THAT EXECUTIVE IS RELEASING ALL CLAIMS AND AGREES AND CONSENTS TO THE TERMS AND CONDITIONS OF THIS EMPLOYMENT AGREEMENT AND THIS RELEASE.

| EXECUTIVE:   | Signature:                    |  |  |
|--------------|-------------------------------|--|--|
|              | Printed Name: Curtis Campbell |  |  |
|              | Date:                         |  |  |
| THE COMPANY: | Blucora, Inc.:                |  |  |
|              | Signature:                    |  |  |
|              | Name:                         |  |  |
|              | Title:                        |  |  |
|              | Date:                         |  |  |

#### EMPLOYMENT AGREEMENT

This Employment Agreement (this "Agreement") is made and entered into as of October 20, 2018, by and between Mike Hogan (the "Executive") and Blucora, Inc. (the "Company").

#### **RECITALS**

WHEREAS, the Company desires to employ the Executive as the President of Tax Innovation for the Company beginning in late October 2018, with the start date of his employment being the effective date of this Agreement (the "*Effective Date*"), and the Executive desires to serve in such capacity;

NOW THEREFORE, in consideration of the foregoing, the mutual covenants contained herein, the employment of the Executive by the Company, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

#### 1. Certain Definitions

- (a) "Base Salary" has the meaning set forth in Section 5(a).
- (b) "Board" means the Board of Directors of the Company.
- (c) "Cause" means, as determined by the Board in its reasonable discretion: (i) the Executive's conviction of, or plea of guilty or nolo contendere to, a misdemeanor involving dishonesty, wrongful taking of property, immoral conduct, bribery or extortion or any felony; (ii) willful material misconduct by the Executive in connection with the business of the Company; (iii) the Executive's continued and willful failure to perform substantially his responsibilities to the Company under this Agreement, after written demand for substantial performance has been given by the Board that specifically identifies how the Executive has not substantially performed his responsibilities; (iv) the Executive's improper disclosure of confidential information or other material breach of this Agreement, including the Confidentiality and Non-Competition Agreement; (v) the Executive's material fraud or dishonesty against the Company; (vi) the Executive's willful and material breach of the Company's written code of conduct and business ethics or other material written policy, procedure or guideline in effect from time to time (provided that the Executive was given access to a copy of such policy, procedure or guideline prior to the alleged breach) relating to personal conduct; or (vii) the Executive's willful attempt to obstruct or willful failure to cooperate with any investigation authorized by the Board or any governmental or self-regulatory entity. Any determination of Cause by the Company shall be made by a resolution approved by a majority of the members of the Board, provided that, with respect to Section 1(c)(iii), the Board must give the Executive notice and 60 days to cure the substantial nonperformance.
  - (d) "Change of Control" means the occurrence of any of the following:
- (i) any "person" (as defined in Sections 13(d) and 14(d) of the Exchange Act), excluding for this purpose, (A) the Company or any subsidiary of the Company or (B) any employee

benefit plan of the Company or any subsidiary of the Company, or any person or entity organized, appointed or established by the Company for or pursuant to the terms of any such plan that acquires beneficial ownership of voting securities of the Company, is or becomes the "beneficial owner" (as defined in Rule 13d-3 under the Exchange Act), directly or indirectly, of securities of the Company representing more than 50% of the combined voting power of the Company's then outstanding securities;

- (ii) consummation of a reorganization, merger or consolidation of the Company, in each case, unless, following such transaction, all or substantially all the individuals and entities who were the beneficial owners of outstanding voting securities of the Company immediately prior to such transaction beneficially own, directly or indirectly, more than 50% of the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors of the company resulting from such transaction (including, without limitation, a company that, as a result of such transaction, owns the Company or all or substantially all the Company's assets either directly or through one or more subsidiaries) in substantially the same proportions as their ownership immediately prior to such transaction of the outstanding voting securities of the Company;
- (iii) any sale or disposition by the Company, in one transaction or a series of related transactions, of all or substantially all the Company's assets;
- (iv) a "Board Change" which, for purposes of this Agreement, shall have occurred if a majority of the seats on the Board are occupied by individuals who were neither (A) nominated by a majority of the Incumbent Directors nor (B) appointed by directors so nominated ("Incumbent Director" means a member of the Board who has been either (1) nominated by a majority of the directors of the Company then in office or (2) appointed by directors so nominated, but excluding, for this purpose, any such individual whose initial assumption of office occurs as a result of either an actual or threatened election contest (as such terms are used in Rule 14a-11 of Regulation 14A promulgated under the Exchange Act) or other actual or threatened solicitation of proxies or consents by or on behalf of a person other than the Board); or
  - (v) an approval by the stockholders of the Company of a complete liquidation or dissolution of the Company.
  - (e) "Code" means the Internal Revenue Code of 1986, as amended.
  - (f) "Compensation Committee" means the Compensation Committee of the Board.
- (g) "Confidentiality and Non-Competition Agreement" means the Confidentiality and Non-Competition Agreement attached hereto as Exhibit A.
- (h) "Constructive Termination" means the occurrence, on a date that is prior to the two- month period prior to the consummation of a Change of Control or after the 12-month period following the consummation of a Change of Control, of any of the following without the Executive's express prior written consent: (i) a material reduction of or to the Executive's duties, authority or responsibilities (a change in reporting relationship alone does not constitute such a material reduction); (ii) a material reduction by the Company of the Executive's Base Salary, unless similarly

situated executives also experience a reduction; or (iii) a requirement that the Executive relocate his primary work location more than 25 miles from Irving, Texas or from any work location to which the Company transfers the Executive during the course of his employment and to which such transfer the Executive has consented. Notwithstanding the foregoing, a Constructive Termination shall not exist unless (x) the Executive delivers written notice to the Company (the "Constructive Termination Notice") of the existence of the condition which the Executive believes constitutes a Constructive Termination within 30 days of the initial existence of such condition (which Constructive Termination Notice specifically identifies such condition), (y) the Company fails to remedy such condition within 30 days after the date on which it receives such notice (the "Constructive Termination Cure Period"), and (z) the Executive actually terminates employment within 30 days after the expiration of the Constructive Termination Cure Period.

- (i) "Disability" means the Executive's inability to perform his employment duties to the Company hereunder, with or without reasonable accommodation, for 180 days (in the aggregate) in any one-year period as determined by an independent physician selected by the Company.
  - (i) "Exchange Act" means the Securities Exchange Act of 1934, as amended.
- (k) "Good Reason" means the occurrence of any of the following without the Executive's express prior written consent: (i) a material reduction of or to the Executive's duties, authority, responsibilities or reporting relationship; (ii) a material reduction of the Executive's Base Salary; (iii) a material reduction of the Executive's Target Bonus; (iv) a material reduction in the kind or level of employee benefits to which the Executive is entitled that occurs within 12 months following a Change of Control, unless similarly situated employees also experience a reduction; (v) a requirement that the Executive relocate his primary work location more than 25 miles from Irving, Texas or from any work location to which the Company transfers the Executive during the course of his employment and to which such transfer the Executive has consented; (vi) in connection with a Change of Control, the failure of the Company to assign this Agreement to a successor to the Company or the failure of a successor to the Company to explicitly assume and agree to be bound by this Agreement in a writing delivered to the Executive; or (vii) a material breach of this Agreement by the Company.

Notwithstanding the foregoing, termination of employment by the Executive will not be for Good Reason unless (x) the Executive delivers written notice to the Company (the "Good Reason Notice") of the existence of the condition which the Executive believes constitutes Good Reason within 30 days of the initial existence of such condition (which Good Reason Notice specifically identifies such condition), the Company fails to remedy such condition within 30 days after the date on which it receives such notice (the "Good Reason Cure Period"), and (z) the Executive actually terminates employment within 30 days after the expiration of the Good Reason Cure Period.

(I) "Release" means a full release of claims against the Company substantially in the form attached hereto as Exhibit B; provided, however, that notwithstanding the foregoing, such Release is not intended to and will not waive the Executive's rights: (i) to indemnification pursuant to any applicable provision of the Company's Bylaws or Certificate of Incorporation, as amended, pursuant to any written indemnification agreement between the Executive and the Company, or pursuant to applicable law; (ii) to vested benefits or payments specifically to be provided to the

Executive under this Agreement or any Company employee benefit plans or policies; or (iii) respecting any claims the Executive may have solely by virtue of the Executive's status as a stockholder of the Company. The Release also shall not include claims that an employee cannot lawfully release through execution of a general release of claims.

- (m) "Section 409A" means Section 409A of the Code and the Treasury Regulations and official guidance issued in respect of Section 409A of the Code.
  - (n) "Target Bonus" has the meaning set forth in Section 5(b).

## 2. <u>Duties and Scope of Employment</u>

The Company shall employ the Executive in the position of President of Tax Innovation for the Company. The Executive shall report directly to the Company's President and Chief Executive Officer. The Executive will render such business and professional services in the performance of the Executive's duties, consistent with the Executive's position(s) within the Company, as shall be reasonably assigned to the Executive at any time and from time to time by the Company's President and Chief Executive Officer. Upon termination of the Executive's employment for any reason, unless otherwise requested by the President and Chief Executive Officer, the Executive will be deemed to have resigned from all positions held at the Company and its affiliates voluntarily, without any further action by the Executive, as of the end of the Executive's employment, and the Executive, at the President and Chief Executive Officer's request, will execute any documents necessary to reflect his resignation.

## 3. Obligations

While employed hereunder, the Executive will perform his duties ethically, faithfully and to the best of the Executive's ability and in accordance with law and Company policy. The Executive agrees not to actively engage in any other employment, occupation or consulting activity for any direct or indirect remuneration without the express prior written approval of the Company's President and Chief Executive Officer; provided, however, that notwithstanding anything to the contrary in the Confidentiality and Non-Competition Agreement, the Executive may engage in charitable activities so long as such activities do not materially interfere with the Executive's responsibilities to the Company.

#### 4. Agreement Term

Unless earlier terminated as provided herein, the term of this Agreement (the "Agreement Term") shall be for a period of three years commencing on the Effective Date, and may be extended thereafter upon the written mutual agreement of the Executive and the Company.

#### 5. Compensation and Benefits

(a) <u>Base Salary</u>. The Company agrees to pay the Executive a base salary (the "*Base Salary*") at an annual rate of not less than \$380,000, payable in accordance with the regular payroll

practices of the Company, but not less frequently than monthly. The Executive's Base Salary shall be subject to annual review by the Board (or a committee thereof).

- (b) <u>Annual Bonus</u>. The Executive shall be eligible to participate in the Company's bonus and other incentive compensation plans and programs for the Company's senior executives at a level commensurate with his position. The Executive shall have the opportunity to earn an annual target bonus (the "*Target Bonus*") measured against criteria to be determined by the Board (or a committee thereof) of at least 120% of Base Salary. The Executive's Target Bonus amount for 2018 will be pro-rated to reflect the number of days of the Executive's employment in 2018. The payout of any 2018 bonus will occur following the end of the Company's Executive Bonus Plan (the "*Plan*") year, which is December 31, 2018, and will be paid in accordance with the terms and conditions of the Plan. Payment of any bonus pursuant to this Section 5(b) is subject to the Executive's employment by the Company on the date required in the Plan. The Company reserves the right to change the Plan at any time at its discretion.
- (c) Equity Awards. The Executive will be eligible to participate in the Company's long-term equity incentive programs extended to senior executives of the Company generally at levels commensurate with the Executive's position, which participation and levels shall be determined by the Board (or a committee thereof) in its sole discretion. In 2019, the Company's President and Chief Executive Officer shall recommend that the Board (or a committee thereof) grant the Executive, for 2019, equity award(s) with a total aggregate target value of \$874,000 on the date of grant, with such award(s) consisting of a combination of the following types of awards as determined by the Board (or a committee thereof), in its sole discretion: time-based restricted stock units (vesting over three years), performance-based restricted stock units (eligible for vesting based on performance following a three-year performance period), and/or nonqualified stock options (vesting over three years); provided, however, that the Board (or committee thereof) retains the sole discretion to establish the terms and the types of awards that are granted to the Executive with respect to 2019.
- (d) <u>Benefits</u>. The Executive and his eligible dependents shall be eligible to participate in the employee benefit plans that are available or that become available to other employees of the Company, with the adoption or maintenance of such plans to be in the discretion of the Company, subject in each case to the generally applicable terms and conditions of the plan or program in question and to the determination of any committee administering such plan or program. Such benefits shall include participation in the group medical, life, disability, and retirement plans that are made generally available to employees of the Company, and any supplemental plans available to senior executives of the Company from time to time. The Company reserves the right to change or terminate its employee benefit plans and programs at any time.
- (e) Expenses. The Company shall reimburse the Executive for reasonable business expenses incurred by the Executive in the furtherance of or in connection with the performance of the Executive's duties hereunder, in accordance with the Company's expense reimbursement policy as in effect from time to time.
- (f) <u>Initial Equity Award</u>. As a material inducement to the Executive's willingness to accept employment with the Company, on or shortly following the Effective Date, the Executive shall be granted an award of time-based restricted stock units having an aggregate value of \$250,000

on the date of grant that ratably over the three-year period from the date of grant (i.e., 1/3 will vest on the first anniversary of the date of grant, and 1/3 will vest on each following anniversary of the date of grant) so long as Executive is employed by the Company on such date (the "*Initial RSUs*").

The number of Initial RSUs granted to the Executive in accordance with this Section 5(f) shall be determined by dividing the value of the Initial RSUs being awarded by the closing price of the Company's common stock on the date of grant and rounding down for any fractional shares. The Initial RSUs may be granted under the Blucora, Inc. 2016 Equity Inducement Plan, as amended (the "Inducement Plan"), or the Blucora, Inc. 2018 Long-Term Incentive Plan (the "LTIP") and will vest in accordance with, and have such other terms and conditions as are specified in, the Restricted Stock Unit Grant Notice and Restricted Stock Unit Agreement approved by the Compensation Committee with respect to such award (the "Restricted Stock Unit Agreement") and shall otherwise be subject to the terms and conditions of the Inducement Plan or LTIP and the Restricted Stock Unit Agreement; provided, however, that notwithstanding the foregoing, in the event of a conflict between the terms and conditions of the Restricted Stock Unit Agreement and this Agreement, the terms and conditions of this Agreement shall prevail.

#### 6. Termination of Employment

- (a) <u>General Provisions</u>. This Agreement and the Executive's employment with the Company may be terminated by either the Executive or the Company at will at any time with or without Cause; provided, however, that the parties' rights and obligations upon such termination during the Agreement Term shall be as set forth in applicable provisions of this Agreement.
- (b) Any Termination by Company or the Executive. In the event of any termination of the Executive's employment with the Company, whether by the Company or by the Executive, (i) the Company shall pay the Executive any unpaid Base Salary due for periods prior to the date of termination of employment ("*Termination Date*"); (ii) the Company shall pay the Executive any unpaid bonus compensation pursuant to Section 5(b), to the extent earned through the Termination Date, subject to the terms of the Company's Executive Bonus Plan (or any successor plan thereto); and (iii) following submission of proper expense reports by the Executive, the Company shall reimburse the Executive for all expenses reasonably and necessarily incurred by the Executive in connection with the business of the Company through the Termination Date (collectively, the "*Accrued Obligations*"). The Accrued Obligations shall be paid promptly upon termination and within the period of time mandated by applicable law (but, in any event, within 30 days after the Termination Date). The Accrued Obligations paid or provided pursuant to this Section 6(b) shall be in addition to the payments and benefits, if any, to be provided to the Executive upon his termination of employment pursuant to Section 6(c), 6(d), 6(e), or 6(f) as applicable. Except as expressly stated above or as required by law or this Agreement, the Executive shall receive no further compensation in any form other than as set forth in this Section 6(b).
- (c) <u>Termination by Company Without Cause or Constructive Termination</u>. If, other than in connection with a Change of Control as described in Section 6(d), the Executive's employment with the Company is terminated by the Company without Cause or the Executive terminates employment with the Company under circumstances constituting a Constructive Termination, then subject to Section 6(g), the Executive shall receive the following payments and benefits:

- (i) a severance payment in an amount equal to one times the Executive's Base Salary in effect as of the Termination Date (or if the Executive terminates employment under circumstances constituting a Constructive Termination due to a material reduction of the Executive's Base Salary, in effect immediately prior to such reduction) (less applicable withholding taxes), which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days following the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date), in accordance with Section 14(b)(ii); and
- (ii) a lump-sum payment in an amount equal to (A) the monthly COBRA premium in effect under the Company's group health plan as of the Termination Date for the coverage in effect under such plan for the Executive (and the Executive's spouse and dependent children) on such date multiplied by (B) 12 (less applicable withholding taxes), which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days following the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date), in accordance with Section 14(b)(ii); provided, however, that notwithstanding the foregoing or any other provision in this Agreement to the contrary, the Company (or its successor) may unilaterally amend this Section 6(c)(ii) or eliminate the benefit provided hereunder to the extent it deems necessary to avoid the imposition of excise taxes, penalties or similar charges on the Company or any of its subsidiaries, affiliates or successors, including, without limitation, under Section 4980D of the Code.

Notwithstanding any provision to the contrary in any Company equity compensation plan or any outstanding equity award agreement, if, during the Agreement Term, the Executive terminates employment with the Company under circumstances described in this Section 6(c), there shall be no acceleration of vesting or exercisability of any outstanding equity awards or extension of any option post-termination exercise period.

For the avoidance of doubt, under no circumstances will the Executive be entitled to payments and benefits under both this Section 6(c) and Section 6(d).

- (d) <u>Termination of Employment in Connection With a Change of Control</u>. If the Company terminates the Executive's employment without Cause or the Executive terminates employment with the Company for Good Reason (1) on the day of or during the 12-month period immediately following the consummation of a Change of Control or (2) during the 2-month period prior to the consummation of a Change of Control but at the request of any third party participating in or causing the Change of Control or otherwise in connection with the Change of Control, then subject to Section 6(g) and with respect to clause (2), subject to the consummation of such Change of Control, the Executive shall receive the following payments and benefits:
- (i) a severance payment in an amount equal to one times the Executive's Base Salary in effect as of the Termination Date and his then current Target Bonus amount (or if the Executive terminates employment for Good Reason due to a material reduction of the Executive's Base Salary or Target Bonus, in effect immediately prior to such reduction) (in each case less applicable withholding taxes), which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days following the Termination Date (but, in any event, by no later

than March 15 of the calendar year immediately following the calendar year that includes the Termination Date), in accordance with Section 14(b)(ii);

- (ii) a lump-sum payment in an amount equal to (A) the monthly COBRA premium in effect under the Company's group health plan as of the Termination Date for the coverage in effect under such plan for the Executive (and the Executive's spouse and dependent children) on such date multiplied by (B) 12 (less applicable withholding taxes), which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days following the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date), in accordance with Section 14(b)(ii); provided, however, that notwithstanding the foregoing or any other provision in this Agreement to the contrary, the Company (or its successor) may unilaterally amend this Section 6(d)(ii) or eliminate the benefit provided hereunder to the extent it deems necessary to avoid the imposition of excise taxes, penalties or similar charges on the Company or any of its subsidiaries, affiliates or successors, including, without limitation, under Section 4980D of the Code; and
- (iii) notwithstanding any provision to the contrary in any applicable equity compensation plan or any outstanding equity award agreement, the treatment of the Executive's outstanding equity awards shall be governed solely by the following provisions: (A) all of the Executive's then-outstanding time-vesting equity awards shall fully vest and all restrictions thereon shall lapse, and (B) to the extent vested (including as a result of the acceleration provided under this Section 6(d)(iii)), all of the Executive's outstanding stock options shall remain exercisable until the first to occur of 12 months following the Termination Date and each such stock option's original expiration date.

If a Change of Control is consummated prior to the expiration of the Agreement Term, this Section 6(d) shall apply to a termination of the Executive's employment by the Company without Cause or by the Executive for Good Reason during the 12-month period immediately following the consummation of the Change of Control even if such 12-month period extends past the expiration of the Agreement Term. Moreover, notwithstanding the expiration of the Agreement Term, if a Change of Control is consummated within two months after the expiration of the Agreement Term, then this Section 6(d) shall apply to a termination of the Executive's employment by the Company without Cause or by the Executive for Good Reason (i) on the day of or during the 12-month period immediately following the consummation of the Change of Control or (ii) during the 2-month period prior to the consummation of the Change of Control but at the request of any third party participating in or causing the Change of Control or otherwise in connection with the Change of Control.

For the avoidance of doubt, the payments and benefits described under this Section 6(d) and the Accrued Obligations shall be the only payments and benefits to which the Executive is entitled in the event that the Executive's employment terminates under this Section 6(d).

(e) <u>Death</u>. In the event of the Executive's death while employed hereunder, and subject to Section 6(g), the Executive's beneficiary (or such other person(s) specified by will or the laws of descent and distribution) shall be entitled to receive a lump-sum payment in an amount equal to three months' Base Salary in effect as of the Termination Date (less applicable withholding taxes), which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days

following the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date), in accordance with Section 14(b)(ii).

- (f) <u>Disability</u>. In the event of the Executive's termination of employment with the Company due to Disability, and subject to Section 6(g), the Executive shall be entitled to receive a lump- sum payment in an amount equal to six months' Base Salary in effect as of the Termination Date (less applicable withholding taxes), which amount shall be payable in a single lump sum on the first payroll date that is at least 60 days following the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date), in accordance with Section 14(b)(ii).
- Release and Other Conditions. The payments and benefits described in Sections 6(c) through 6(f) are expressly conditioned on (i) the Executive (or, in the case of the Executive's death, the Executive's representative) signing and delivering (and not revoking thereafter) a Release to the Company (which, in the case of the Executive's death, also releases any claims by the Executive's estate or survivors), which Release is executed, delivered and effective no later than 60 days following the Termination Date and (ii) the Executive continuing to satisfy any obligations to the Company under this Agreement, the Release and the Confidentiality and Non-Competition Agreement that are incorporated herein by reference, and any other agreement(s) between the Executive and the Company. In the event the Release described in Section 6(g)(i) is not executed, delivered and effective by the 60th day after the Termination Date, none of such payments or benefits shall be provided to the Executive.

#### 7. <u>Section 280G</u>

(a) Amount of Payments and Benefits. Notwithstanding anything to the contrary herein, in the event that the Executive becomes entitled to receive or receives any payments, options, awards or benefits (including, without limitation, the monetary value of any noncash benefits and the accelerated vesting of equity-based awards) under this Agreement or under any other plan, agreement or arrangement with the Company or any person affiliated with the Company (collectively, the "Payments"), that may separately or in the aggregate constitute "parachute payments" within the meaning of Section 280G of the Code and the Treasury Regulations promulgated thereunder (or any similar or successor provision) (collectively, "Section 280G") and it is determined that, but for this Section 7(a), any of the Payments will be subject to any excise tax pursuant to Section 4999 of the Code or any similar or successor provision (the "Excise Tax"), the Company shall pay to the Executive either (i) the full amount of the Payments or (ii) an amount equal to the Payments, reduced by the minimum amount necessary to prevent any portion of the Payments from being an "excess parachute payment" (within the meaning of Section 280G) (the "Capped Payments"), whichever of the foregoing amounts results in the receipt by the Executive, on an after-tax basis, of the greatest amount of Payments notwithstanding that all or some portion of the Payments may be subject to the Excise Tax. For purposes of determining whether the Executive would receive a greater after-tax benefit from the Capped Payments than from receipt of the full amount of the Payments, (i) there shall be taken into account any Excise Tax and all applicable federal, state and local taxes required to be paid by the Executive in respect of the receipt of such payments and (ii) such payments

shall be deemed to be subject to federal income taxes at the highest rate of federal income taxation applicable to individuals that is in effect for the calendar year in which the payments and benefits are to be paid, and state and local income taxes at the highest rate of taxation applicable to individuals in the state and locality of the Executive's residence on the effective date of the relevant transaction described under Section 280G(b)(2)(A)(i) of the Code, net of the maximum reduction in federal income taxes that could be obtained from deduction of such state and local taxes (as determined by assuming that such deduction is subject to the maximum limitation applicable to itemized deductions under Section 68 of the Code and any other limitations applicable to the deduction of state and local income taxes under the Code).

- (b) <u>Computations and Determinations</u>. All computations and determinations called for by this Section 7 shall be made by an independent accounting firm or independent tax counsel appointed by the Company (the "*Tax Counsel*"), and all such computations and determinations shall be conclusive and binding on the Company and the Executive. For purposes of such calculations and determinations, the Tax Counsel may rely on reasonable, good faith interpretations concerning the application of Sections 280G and 4999 of the Code. The Tax Counsel shall submit its determination and detailed supporting calculations to both the Executive and the Company within 15 days after receipt of a notice from either the Company or the Executive that the Executive may receive payments which may be considered "parachute payments." The Company and the Executive shall furnish to the Tax Counsel such information and documents as the Tax Counsel may reasonably request in order to make the computations and determinations called for by this Section 7. The Company shall bear all costs that the Tax Counsel may reasonably incur in connection with the computations and determinations called for by this Section 7.
- (c) <u>Reduction Methodology</u>. In the event that Section 7(a) applies and a reduction is required to be applied to the Payments thereunder, the Payments shall be reduced by the Company in its reasonable discretion in the following order: (i) reduction of any Payments that are subject to Section 409A on a pro-rata basis or such other manner that complies with Section 409A, as determined by the Company, and (ii) reduction of any Payments that are exempt from Section 409A.

## 8. No Impediment to Agreement

The Executive hereby represents to the Company that the Executive is not, as of the date hereof, and will not be, during the Executive's employment with the Company, employed under contract, oral or written, by any other person, firm or entity, and is not and will not be bound by the provisions of any restrictive covenant or confidentiality agreement that would constitute an impediment to, or restriction upon, the Executive's ability to enter this Agreement and to perform the duties of the Executive's employment.

## 9. <u>Confidentiality and Non-Competition Agreement</u>

The Confidentiality and Non-Competition Agreement is incorporated by reference as if set forth fully herein. The Confidentiality and Non-Competition Agreement shall survive the termination of this Agreement and/or the Executive's employment with the Company.

#### 10. <u>Cooperation</u>

The Executive hereby agrees to provide the Executive's full cooperation, at the request of the Company, with any of the Company Releasees (as defined in the Release) in any and all such lawsuits, investigations or other legal, equitable or business matters or proceedings which involve any matters for which the Executive worked on or had responsibility during the Executive's employment with the Company. The Executive also agrees to be available to the Company and its representatives (including attorneys) to provide general advice or assistance as requested by the Company. This includes but is not limited to testifying (and preparing to testify) as a witness in any proceeding or otherwise providing information or reasonable assistance to the Company in connection with any investigation, claim or suit, and cooperating with the Company regarding any investigation, litigation, claims or other disputed items involving the Company that relate to matters within the knowledge or responsibility of the Executive. Specifically, the Executive agrees (i) to meet with the Company's representatives, its counsel or other designees at reasonable times and places with respect to any items within the scope of this provision; (ii) to provide truthful testimony regarding same to any court, agency or other adjudicatory body; (iii) to provide the Company with immediate notice of contact or subpoena by any non-governmental adverse party (known to the Executive to be adverse to the Company or its interests); and (iv) to not voluntarily assist any such non-governmental adverse party or such non-governmental adverse party's representatives. The Executive acknowledges and understands that the Executive's obligations of cooperation under this Section 10 are not limited in time and may include, but shall not be limited to, the need for or availability for testimony. The Executive shall receive no additional compensation for time spent assisting the Company pursuant to this Section 10 other than the compensation and benefits provided for in this Agreement, provided that the Executive shall be entitled to be reimbursed for any reasonable out-of-pocket expenses incurred in fulfilling the Executive's obligations pursuant to subsections (i) and (ii) above. Notwithstanding the foregoing, nothing in this Section 10 is intended to interfere with the Executive's No Interference rights set forth in Section 1(c) of the Confidentiality and Non-Competition Agreement.

#### 11. Arbitration

(a) The Executive agrees that any dispute and/or claim between the Company (including without limitation its officers, directors, employees agents or shareholders and its subsidiaries) and the Executive that underlies, relates to and/or results from the Executive's employment relationship with the Company or the termination of that relationship or any of the terms of this Agreement, except for any dispute or claim arising from or relating to the Confidentiality and Non-Competition Agreement, that cannot be resolved by mutual agreement of the Company and the Executive will be submitted to final, binding arbitration to the maximum extent permitted by law in accordance with the National Rules for the Resolution of Employment Disputes of the American Arbitration Association that are then in effect. This arbitration provision includes, but is not limited to, claims of wrongful discharge, infliction of emotional distress, breach of contract (including breach of this Agreement), breach of any covenant of good faith and fair dealing, and claims of retaliation and/or discrimination in violation of any local, state or federal law. Examples of such laws include Title VII of the Civil Rights Act of 1964; the Age Discrimination in Employment Act of 1967; the Americans with Disabilities Act of 1990; and the Family and Medical Leave Act of 1993, and all

amendments to each such law as well as the regulations issued thereunder. This arbitration provision does not affect the Executive's right to pursue worker's compensation or unemployment compensation benefits for which he may be eligible in accordance with state law, nor does it affect the Executive's right to file and/or to cooperate in the investigation of an administrative charge of discrimination.

- (b) Notwithstanding this arbitration provision, either the Executive or the Company may apply to any court of competent jurisdiction for a temporary restraining order, preliminary injunction, or other interim or conservatory relief, as necessary, without breach of this Agreement and without abridgement of the powers of the arbitrator.
- (c) This arbitration provision does not apply to any dispute or claim arising from or relating to the Confidentiality and Non-Competition Agreement.
- (d) The Company, as further consideration for the Executive's agreement to arbitrate covered disputes, agrees to pay for the arbitrator's fees and other costs directly associated with the arbitration that would not otherwise be charged if the parties pursued civil litigation in court.

#### 12. Successors; Personal Services

The services and duties to be performed by the Executive hereunder are personal and may not be assigned or delegated. This Agreement shall be binding upon and inure to the benefit of the Company and its successors and assigns, and the Executive and the Executive's heirs and representatives.

#### 13. Notices

Notices and all other communications contemplated by this Agreement shall be in writing and shall be deemed to have been duly given when personally delivered or when mailed by U.S. registered or certified mail, return receipt requested and postage prepaid. In the case of the Executive, mailed notices shall be addressed to the Executive at the home address the Executive most recently communicated to the Company in writing. In the case of the Company, mailed notices shall be addressed to its corporate headquarters, and all notices shall be directed to the attention of its President and Chief Executive Officer.

#### 14. Section 409A

(a) The parties intend that this Agreement and the payments and benefits provided hereunder be exempt from the requirements of Section 409A, to the maximum extent possible, whether pursuant to the short-term deferral exception described in Treasury Regulation Section 1.409A-1(b)(4), the involuntary separation pay plan exception described in Treasury Regulation Section 1.409A-1(b)(9)(iii), or otherwise. To the extent Section 409A is applicable to this Agreement, the parties intend that this Agreement and any payments and benefits thereunder comply with the deferral, payout and other limitations and restrictions imposed under Section 409A. Notwithstanding anything herein to the contrary, this Agreement shall be interpreted, operated and administered in a manner consistent with such intentions.

- (b) Without limiting the generality of the foregoing, and notwithstanding any other provision of this Agreement to the contrary:
- (i) if the Executive is deemed on the date of termination to be a "specified employee" within the meaning of that term under Section 409A, then with regard to any payment that is considered a "deferral of compensation" under Section 409A payable on account of a "separation from service," such payment shall be made on the date which is the earlier of (A) the date that is six months and one day after the date of such "separation from service" of the Executive and (B) the date of the Executive's death (the "*Delay Period*"), to the extent required under Section 409A. Within ten business days following the expiration of the Delay Period, all payments delayed pursuant to this Section 14(b)(i) (whether they would have otherwise been payable in a single sum or in installments in the absence of such delay) shall be paid to the Executive in a lump sum, and all remaining payments due under this Agreement shall be paid or provided in accordance with the normal payment dates specified for those payments in this Agreement;
- (ii) to the extent that any payments or benefits under this Agreement are conditioned on a Release, if the Release is executed and delivered by the Executive to the Company and becomes irrevocable and effective within the specified 60-day post-termination period, then, subject to Section 14(b)(i) and to the extent not exempt under Section 409A, such payments or benefits shall be made or commence on the first payroll date after the date that is 60 days after the Termination Date (but, in any event, by no later than March 15 of the calendar year immediately following the calendar year that includes the Termination Date). If a payment or benefit under this Agreement is conditioned on a Release and such Release is not executed, delivered and effective by the 60th day after the Termination Date, such payment or benefit shall not be paid or provided to the Executive;
- (iii) all expenses or other reimbursements under this Agreement shall be made on or prior to the last day of the taxable year following the taxable year in which such expenses were incurred by the Executive. No such reimbursement or expenses eligible for reimbursement in any taxable year shall in any way affect the expenses eligible for reimbursement in any other taxable year, and the Executive's right to reimbursement shall not be subject to liquidation or exchange for any other benefit;
- (iv) for purposes of Section 409A, the Executive's right to receive a series of installment payments pursuant to this Agreement shall be treated as a right to receive a series of separate and distinct payments. Whenever a payment under this Agreement specifies a payment period with reference to a number of days (e.g., "payment shall be made within 30 days"), the actual date of payment within the specified period shall be within the sole discretion of the Company;
- (v) in no event shall any payment under this Agreement that constitutes a "deferral of compensation" for purposes of Section 409A be offset by any other payment pursuant to this Agreement or otherwise; and
- (vi) to the extent required for purposes of compliance with Section 409A, termination of employment shall not be deemed to have occurred for purposes of any provision of this Agreement providing for the payment of any amounts or benefits upon or following a termination

of employment unless such termination is also a "separation from service" within the meaning of Section 409A, and for purposes of any such provision of this Agreement, references to a "termination," "termination of employment" or like terms shall mean "separation from service."

(c) The Company and the Executive agree to work together in good faith to consider amendments to this Agreement and to take such reasonable actions that may be necessary, appropriate, or desirable to avoid imposition of additional tax or income recognition on the Executive under Section 409A, in each case to the maximum extent permitted by applicable law. Notwithstanding any provision of this Agreement to the contrary, (i) in no event will the Company be liable for any additional tax, interest or penalty that may be imposed on the Executive by Section 409A or damages for failing to comply with Section 409A and (ii) the Executive acknowledges and agrees that the Executive will not have any claim or right of action against the Company or any of its employees, officers, directors or agents in the event it is determined that any payment or benefit provided hereunder does not comply with Section 409A.

#### 15. Miscellaneous Provisions

- (a) <u>Waiver</u>. No provision of this Agreement shall be modified, waived or discharged unless the modification, waiver or discharge is agreed to in writing and signed by the Executive and by an authorized officer of the Company (other than the Executive). No waiver by either party of any breach of, or of compliance with, any condition or provision of this Agreement by the other party shall be considered a waiver of any other condition or provision or of the same condition or provision at another time.
- (b) Entire Agreement. This Agreement (including exhibits) shall supersede and replace all prior agreements or understandings relating to the subject matter hereof, and no agreements, representations or understandings (whether oral or written or whether express or implied) that are not expressly set forth in this Agreement have been made or entered into by either party with respect to the relevant matters hereof. This Agreement may not be modified except expressly in a writing signed by both parties.
- (c) <u>Disclaimer of Reliance</u>. Except for the specific representations expressly made by the Company in this Agreement, the Executive specifically disclaims that the Executive is relying upon or has relied upon any communications, promises, statements, inducements, or representation(s) that may have been made, oral or written, regarding the subject matter of this Agreement. The Executive represents that the Executive relied solely and only on the Executive's own judgment in making the decision to enter into this Agreement.
- (d) <u>Choice of Law</u>. The validity, interpretation, construction and performance of this Agreement shall be governed by the internal substantive laws of the State of Texas without reference to any choice of law rules.
- (e) <u>Severability</u>. The invalidity or unenforceability of any provision or provisions of this Agreement shall not affect the validity or enforceability of any other provision hereof, which shall remain in full force and effect.

- (f) No Assignment of Benefits. The rights of any person to payments or benefits under this Agreement shall not be made subject to option or assignment, either by voluntary or involuntary assignment or by operation of law, in respect of bankruptcy, garnishment, attachment or other creditor's process, and any action in violation of this Section 15(f) shall be void.
- (g) No Duty to Mitigate. The Executive shall not be required to mitigate the amount of any payment contemplated by this Agreement, nor shall any such payment be reduced by any earnings that the Executive may receive from any other source.
- (h) <u>Employment Taxes</u>. All payments made pursuant to this Agreement will be subject to withholding of all applicable income, employment and other taxes.
- (i) <u>Assignment by Company</u>. The Company may assign its rights under this Agreement to an affiliate (as defined under the Exchange Act), and an affiliate may assign its rights under this Agreement to another affiliate of the Company or to the Company. In the case of any such assignment, the term "Company" when used in a section of this Agreement shall mean the corporation that actually employs the Executive.
- (j) <u>Counterparts</u>. This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together will constitute one and the same instrument.

[Signature Page Follows]

|         | IN WITNESS WHEREOF,              | each of the parties h | as executed this | Agreement, in the | the case of the | Company by | its duly | authorized |
|---------|----------------------------------|-----------------------|------------------|-------------------|-----------------|------------|----------|------------|
| officer | , as of the day and year first a | bove written.         |                  |                   |                 |            |          |            |

# BLUCORA, INC.

By: /s/ John S. Clendening

Name: John S. Clendening

Title: President and Chief Executive Officer

## **EXECUTIVE:**

/s/ Mike Hogan Mike Hogan

#### **EXHIBIT A**

#### CONFIDENTIALITY AND NON-COMPETITION AGREEMENT

This Confidentiality and Non-Competition ("Agreement") is entered into by and between Blucora, Inc., its subsidiaries, affiliates, successors and/or assigns (the "Company") and Mike Hogan ("Executive"). The Effective Date of this Agreement is the date of Executive's execution of this Agreement. The Company and Executive shall be referred to herein individually as a "Party" and collectively as the "Parties."

NOW, THEREFORE, in consideration of the mutual covenants set forth herein, Executive's position with the Company, the Confidential Information (defined below), compensation and benefits provided to Executive, and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, and intending to be legally bound hereby, the Parties agree as follows:

#### 1. <u>Confidential Information and Executive's Non-Disclosure Agreement.</u>

(a) Confidential Information. During Executive's employment with the Company, the Company shall provide Executive with Confidential Information (defined below), which is not known to the Company's competitors or within the Company's industry generally, which was developed by the Company over a long period of time and/or at its substantial expense, and which is of great competitive value to the Company. For purposes of this Agreement, "Confidential Information" includes all documents or information, in whatever form or medium, concerning or relating to any of the following: all trade secrets and confidential and proprietary information of or relating to the Company, including, but not limited to: (A) financial models, business records, business plans or processes, strategies (including, without limitation, economic and market research selection and analysis strategies and business development and market segment exploitation strategies), tactics, policies, resolutions, processes, inventions, patents, trademarks, trade secrets, know how, patent or trademark applications and other intellectual property, (B) information regarding litigation or negotiations, (C) any marketing information, sales or product plans, prospects and market research data relating to the business, (D) financial information, cost and performance data and any debt arrangements, equity ownership or securities transaction information, (E) technical information, technical drawings and designs, (F) personnel information, personnel lists, resumes, personnel data, organizational structure, compensation and performance evaluations, (G) customer, consumer, consultants or supplier information, including but not limited to any data regarding any current, prospective or former customers, consumers, consultants or suppliers of Company, (H) information regarding the existence or terms of any agreement or relationship between the Company or any of its subsidiaries or affiliates and any other party, (I) information subject to Section 628 of the Fair Credit Reporting Act and any regulations or guidelines thereunder and (J) any other information of whatever nature, including, without limitation, information which gives to the Company or any of its subsidiaries or affiliates an opportunity to obtain an advantage over its competitors who or which do not have access to such information. Confidential Information, whether prepared or compiled by Executive and/or the Company or furnished to Executive during Executive's employment with the Company, shall be the sole and exclusive property of the Company, and none of such Confidential Information or copies thereof, shall be retained by Executive.

Executive agrees not to dispute, contest, or deny any such ownership rights either during or after Executive's employment with the Company. Executive acknowledges that the Company does not voluntarily disclose Confidential Information, but rather takes precautions to prevent dissemination of Confidential Information beyond those employees such as Executive entrusted with such information. Executive further acknowledges that the Confidential Information: (a) is entrusted to Executive because of Executive's position with the Company; and (b) is of such value and nature as to make it reasonable and necessary for Executive to protect and preserve the confidentiality and secrecy of the Confidential Information. Executive acknowledges and agrees that the Confidential Information is proprietary to and a trade secret of the Company and, as such, is a valuable, special and unique asset of the Company, the unauthorized use or disclosure of which will cause irreparable harm, substantial injury and loss of profits and goodwill to the Company. "Confidential Information" does not include any information which is generally available to and known by the public or becomes generally available to and known by the public (other than as a result of Executive's breach of this Agreement or any other agreement or obligation to keep such information confidential).

### (b) Non-Disclosure.

(i) Executive agrees to preserve and protect the confidentiality of all Confidential Information. Executive agrees that during the period of Executive's employment with the Company and at any time thereafter (regardless of the reason for Executive's separation or termination of employment): (A) Executive shall hold all Confidential Information in the strictest confidence, take all reasonable precautions and steps to safeguard all Confidential Information and prevent its wrongful use by or wrongful or inadvertent disclosure or dissemination to any unauthorized person or entity, and follow all policies and procedures of the Company protecting or regarding the Confidential Information; and (B) without prior written authorization of the Company, Executive shall not, directly or indirectly, use for Executive's own account, use in any way or for any other purpose, disclose to anyone, publish, exploit, destroy, copy or remove from the offices of the Company, nor solicit, allow or assist another person or entity to use, disclose, publish, exploit, destroy, copy or remove from the offices of the Company, any Confidential Information or part thereof, except: (1) as permitted in the proper performance of Executive's duties for the Company; (2) as permitted in the ordinary course of the Company's business for the benefit of the Company; or (3) as otherwise permitted or required by law. Executive shall immediately notify the Company if Executive learns of or suspects any actual or potential unauthorized use or disclosure of Confidential Information concerning the Company. Further, the Executive shall not, directly or indirectly, use the Company's Confidential Information to: (1) call upon, solicit business from, attempt to conduct business with, conduct business with, interfere with or divert business away from any customer, client, service provider, supplier or vendor of the Company with whom or which the Company conducted business; and/or (2) recruit, solicit, hire or attempt to recruit, solicit, or hire, directly or by assisting others, any persons employed by the Company. In the event Executive is subpoenaed, served with any legal process or notice, or otherwise requested to produce or divulge, directly or indirectly, any Confidential Information by any entity, agency or person in any formal or informal proceeding including, but not limited to, any interview, deposition, administrative or judicial hearing and/or trial, except where prohibited by law, Executive should immediately notify the Company and deliver a copy of the subpoena, process, notice or other request to the Company

as promptly as possible, but under no circumstances more than ten (10) days following Executive's receipt of same; provided, however, Executive is not required to notify the Company or provide a copy of the subpoena, process, notice or other request where Executive is permitted to make such disclosure of Confidential Information pursuant to this Agreement or applicable law or regulation, as set forth in Section 1(c) and Section 1(d).

- (ii) Subject to Section 1(b)(iii), Executive agrees that Executive will not use or disclose any confidential, proprietary or trade secret information belonging to any former employer or third party, and Executive will not bring onto the premises of the Company or onto any Company property, any confidential, proprietary or trade secret information belonging to any former employer or third party without such third party's written consent. Executive acknowledges that that the Company has specifically instructed Executive not to disclose to the Company, use, or induce the Company to use, any confidential, proprietary or trade secret information belonging to any previous employer or others.
- (iii) During Executive's employment, the Company will receive from third parties their confidential and/or proprietary information, subject to a duty on the Company's part to maintain the confidentiality of and to use such information only for certain limited purposes. Executive agrees to hold all such confidential or proprietary information in the strictest confidence and not to disclose it to any person or organization or to use it except as necessary in the course of Executive's employment with the Company and in accordance with the Company's agreement with such third party.
- (iv) Except in the proper performance of Executive's duties and responsibilities, Executive agrees that Executive shall not remove, destroy, deface, damage or delete any Property of the Company. For purposes of this Agreement, the term "*Property*" means all property or information, in whatever form or media, and all copies thereof whether or not the original was deleted or destroyed, of the Company, including, without limitation, any Confidential Information, software, hardware, including any and all Company-issued equipment, devices, cellular telephones, tablets, computers, laptops, hard drives, keys, access cards, access codes or passwords belonging to the Company, databases, files, records, reports, memoranda, research, plans, proposals, lists, forms, drawings, specifications, notebooks, manuals, correspondence, materials, e-mail, electronic or magnetic recordings or data, and any other physical or electronic documents that Executive receives from or sends to any employee of the Company, that Executive copies from the files or records of the Company, or that Executive otherwise has access to during Executive's employment.
- (c) No Interference. Notwithstanding any other provision of this Agreement, (i) Executive may disclose Confidential Information when required to do so by a court of competent jurisdiction, by any governmental agency having authority over Executive or the business of the Company or by any administrative body or legislative body (including a committee thereof) with jurisdiction to order Executive to divulge, disclose or make accessible such information; and (ii) nothing in this Agreement is intended to interfere with Executive's right to (A) report possible violations of state or federal law or regulation to any governmental or law enforcement agency or entity; (B) make other disclosures that are protected under the whistleblower provisions of state or federal law or regulation; (C) file a claim or charge with any governmental agency or entity; or (D)

testify, assist, or participate in an investigation, hearing, or proceeding conducted by any governmental or law enforcement agency or entity, or any court. For purposes of clarity, in making or initiating any such reports or disclosures or engaging in any of the conduct outlined in subsection (ii) above, Executive may disclose Confidential Information to the extent necessary to such governmental or law enforcement agency or entity or such court, need not seek prior authorization from the Company, and is not required to notify the Company of any such reports, disclosures or conduct.

(d) <u>Defend Trade Secrets Act</u>. Executive is hereby notified in accordance with the Defend Trade Secrets Act of 2016 that Executive will not be held criminally or civilly liable under any federal or state trade secret law for the disclosure of a trade secret that is made in confidence to a federal, state, or local government official, either directly or indirectly, or to an attorney solely for the purpose of reporting or investigating a suspected violation of law, or is made in a complaint or other document that is filed under seal in a lawsuit or other proceeding.

### (e) <u>Inventions</u>.

- (i) Prior Inventions Retained and Licensed. In Exhibit A-1 to this Agreement, Executive has provided a list describing all Inventions (defined below) that Executive: (A) conceived, created, developed, made, reduced to practice or completed, either alone or with others, prior to Executive's employment with the Company; (B) claims a proprietary right or interest in; and (C) does not assign to the Company hereunder (collectively referred to as the "Prior Inventions"). If no such list is attached, Executive represents that there are no such Prior Inventions. Executive understands and agrees that the Company makes no attempt to verify Executive's claim of ownership to any of the Prior Inventions. Executive agrees that Executive shall not incorporate in any work that Executive performs for the Company any Prior Inventions or any of the technology described in any Prior Inventions. Nonetheless, if in the course of Executive's employment with the Company, Executive incorporates Prior Inventions into a product, service, process or machine of the Company, Executive hereby grants and shall be deemed to have granted the Company a nonexclusive, royalty-free, irrevocable, sublicensable, transferable, perpetual, and worldwide license to make, have made, modify, use, import, reproduce, distribute, prepare and have prepared derivative works of, offer to sell, sell and otherwise exploit such Prior Inventions. For purposes of this Agreement, the term "Inventions" means all tangible and intangible materials, work product, information, methods, designs, computer programs, software, databases, formulas, models, prototypes, reports, discoveries, ideas, improvements, know-how, compositions of matter, processes, photographs, drawings, illustrations, sketches, developments, and all related intellectual property, including inventions, original works of authorship, moral rights, mask works, trade secrets and trademarks.
- (ii) Assignment of Inventions. During Executive's employment with the Company and following the termination of Executive's employment for any reason, Executive agrees that Executive shall promptly make full written disclosure to the Company, shall hold in trust for the sole right and benefit of the Company, and hereby assigns and shall be deemed to have assigned to the Company or its designee, all of Executive's right, title, and interest in and to any and all Inventions that have been or may be conceived, created, developed, completed, reduced to

practice or otherwise made by Executive, solely or jointly with others, during the period of Executive's employment with the Company which (A) relate in any manner to the existing or contemplated business, work, or investigations of the Company; (B) are suggested by, result from, or arise out of any work that Executive may do for or on behalf of the Company; (C) result from or arise out of any Confidential Information that may have been disclosed or otherwise made available to Executive as a result of duties assigned to Executive by the Company; or (D) are otherwise made through the use of the time, information, equipment, facilities, supplies or materials of the Company, even if developed, conceived, reduced to practice or otherwise made during other than working hours (collectively referred to as "Company Inventions"). Executive further acknowledges that all original works of authorship that are made by Executive (solely or jointly with others) within the scope of Executive's employment with the Company and that are protectable by copyright are "Works Made for Hire," as that term is defined in the United States Copyright Act. Executive understands and agrees that the decision whether or not to commercialize or market any Company Inventions is within the Company's sole discretion and for the Company's sole benefit, and that no royalty will be due to Executive as a result of the Company's efforts to commercialize or market any such Company Invention.

- (iii) <u>Maintenance of Records</u>. Executive agrees to keep and maintain adequate and current hard-copy and electronic records of all Company Inventions. The records will be available to and remain the sole property of the Company during Executive's employment with the Company and at all times thereafter.
- (f) Patent and Copyright Registrations. Executive agrees to assist the Company or its designee, at the Company's expense, in every proper way to secure the Company's rights in Company Inventions in any and all countries, including the disclosure to the Company of all pertinent information and data with respect thereto, the execution of all applications, specifications, oaths, assignments, affidavits, and all other instruments which the Company shall deem necessary in order to apply for and obtain such rights and in order to assign and convey to the Company and/or its successors, assigns and nominees, the sole and exclusive rights, title and interest in and to such Company Inventions. Executive further agrees that Executive's obligation to execute or cause to be executed, when it is in Executive's power to do so, any such instrument or papers shall continue after the termination of this Agreement. Executive hereby appoints the General Counsel of the Company as Executive's attorney-in-fact to execute documents on Executive's behalf for this purpose. Executive agrees that this appointment is coupled with an interest and will not be revocable.
- (g) Return of Company Property. Upon request by the Company or upon the termination of Executive's employment for any reason, Executive shall immediately return and deliver to the Company any and all Property, including, without limitation, Confidential Information, software, hardware, including any and all Company-issued equipment, devices, cellular telephones, tablets, computers, laptops, hard drives, keys, access cards, access codes or passwords, databases, files, documents, records, reports, memoranda, research, plans, proposals, lists, papers, books, forms, drawings, specifications, notebooks, manuals, correspondence, materials, e-mail, electronic or magnetic recordings or data, including all copies thereof (in electronic or hard copy format), which belong to the Company or which relate to the Company's business and which are in Executive's possession, custody or control, whether prepared by Executive or others. Executive further agrees

that after Executive provides such Property to the Company, Executive will immediately destroy any information or documents, whether prepared by Executive or others, containing or reflecting any Confidential Information or relating to the business of the Company from any computer, cellular phone or other digital or electronic device in Executive's possession, custody or control, and Executive shall certify such destruction in writing to the Company. Upon request by the Company, Executive shall provide such computer, cellular phone or other digital or electronic device to the Company or the Company's designee for inspection to confirm that such information and documents have been destroyed. If at any time after the termination of Executive's employment for any reason, Executive or the Company determines that Executive has any Property in Executive's possession, custody or control, Executive shall immediately return all such Property, including all copies and portions thereof, to the Company.

- 2. Restrictive Covenants. In consideration for (i) the Company's promise to provide Confidential Information; (ii) the substantial economic investment made by the Company in the Confidential Information and goodwill of the Company, and/or the business opportunities disclosed or entrusted to Executive; (iii) access to the customers and clients of the Company; and (iv) the Company's employment of Executive in an executive position and the compensation and other benefits provided by the Company to Executive (including the consideration provided for in the Incentive Retention Letter), to protect the Confidential Information and business goodwill of the Company, Executive agrees to the following restrictive covenants.
- (a) Non-Competition. Executive agrees that during the Executive's employment with the Company and for a period of twelve (12) months after the Executive's employment terminates for any reason (the "Restricted Period"), other than in connection with Executive's performance of his duties for the Company, Executive shall not use any Confidential Information to, without the prior written consent of an officer of the Company, directly or indirectly, either individually or as a principal, partner, stockholder, manager, agent, consultant, contractor, distributor, employee, lender, investor, or as a director or officer of any corporation or association, or in any other manner or capacity whatsoever, (i) control, manage, operate, establish, take steps to establish, lend money to, invest in, solicit investors for, or otherwise provide capital to, or (ii) become employed by, join, perform services for, consult for, do business with or otherwise engage in any Competing Business within the Restricted Area. For purposes of this Consulting Agreement, given the scope of Confidential Information to be provided to Executive and job duties to be performed by the Executive, "Restricted Area" means the United States, and any other geographic area for which Executive performed any services or about which Executive received Confidential Information. For purposes of this Agreement, "Competing Business" means any business, individual, partnership, firm, corporation or other entity that is competing or that is preparing to compete with any aspect of the Company's business, which includes, but is not limited to (a) tax preparation and tax preparation-related products and services provided to consumers and small businesses, and to or through tax professionals; (b) investment and insurance products or services, and related advice and brokerage services, provided to or through tax professionals or in conjunction with tax preparation services, and (c) any other business the Company engages in or develops during the Executive's employment with the Company.

- (b) Non-Solicitation. During the Restricted Period, other than in connection with Executive's duties for the Company, Executive shall not, and shall not use any Confidential Information to, directly or indirectly, either as a principal, manager, agent, employee, consultant, officer, director, stockholder, partner, investor or lender or in any other capacity, and whether personally or through other persons, solicit business from, interfere with, or induce to curtail or cancel any business or contracts with the Company, or attempt to solicit business with, interfere with, or induce to curtail or cancel any business or contracts with the Company, or do business with any actual or prospective customer or client of the Company with whom the Company did business or who the Company solicited within the preceding two (2) years, and who or which: (1) Executive contacted, called on, serviced or did business with during Executive's employment with the Company; or (3) about whom Executive received Confidential Information. This restriction applies only to business which is in the scope of services or products provided by the Company.
- (c) <u>Non-Recruitment</u>. During the Restricted Period, other than in connection with Executive's duties for the Company, Executive shall not, on behalf of Executive or on behalf of any other person or entity, directly or indirectly, hire, solicit or recruit, or attempt to hire, solicit or recruit, or encourage to leave or otherwise cease his/her employment or engagement with the Company, any individual who is an employee or independent contractor of the Company or who was an employee or independent contractor of the Company within the twelve (12) month period prior to Executive's separation from employment with the Company.
- (d) Non-Disparagement. Executive agrees that the Company's goodwill and reputation are assets of great value to the Company, which have been obtained and maintained through great costs, time and effort. Therefore, Executive agrees that during Executive's employment and after the termination of Executive's employment, Executive shall not make, publish or otherwise transmit any knowingly false statements, whether written or oral, regarding the Company and its officers, directors, executives, employees, contractors, consultants, products, services, business or business practices. A violation or threatened violation of this Section 2(d) may be enjoined by the courts. The rights afforded the Company under this provision are in addition to any and all rights and remedies otherwise afforded by law. However, nothing in this Section 2(d) restricts or prevents Executive from providing truthful testimony as required by court order or other legal process or is intended to interfere with Executive's rights set forth in Section 1(c).
- (e) <u>Tolling</u>. If Executive violates any of the covenants contained in this Section 2, the Restricted Period applicable to such covenant(s) shall be suspended and shall not run in favor of Executive from the time of the commencement of such violation until the time that Executive cures the violation to the satisfaction of the Company and the period of time in which Executive is in breach shall be added to the Restricted Period applicable to such covenant(s).
- 3. **Reasonableness**. Executive hereby represents to the Company that Executive has read and understands, and agrees to be bound by, the terms of Section 1 and Section 2. Executive acknowledges that the scope and duration of the restrictions and covenants contained in Section 1 and Section 2 are fair and reasonable in light of (i) the nature and scope of the operations of the Company's business; and (ii) the amount of compensation and Confidential Information (including,

without limitation, trade secrets) that Executive is receiving in connection with Executive's employment with the Company and the Incentive Retention Letter. It is the desire and intent of the Parties that the provisions of Section 1 and Section 2 be enforced to the fullest extent permitted under applicable law, whether now or hereafter in effect and therefore, to the extent permitted by applicable law, Executive and the Company hereby waive any provision of applicable law that would render any provision of Section 1 and/or Section 2 invalid or unenforceable.

- 4. Remedies. Executive acknowledges that the restrictions and covenants contained in Section 1 and Section 2, in view of the nature of the Company's business and Executive's position with the Company, are reasonable and necessary to protect the Company's legitimate business interests, goodwill and reputation, and that any violation of Section 1 or Section 2 would result in irreparable injury and continuing damage to the Company, and that money damages would not be a sufficient remedy to the Company for any such breach or threatened breach. Therefore, Executive agrees that the Company shall be entitled to a temporary restraining order and injunctive relief restraining Executive from the commission of any breach or threatened breach of Section 1 and/or Section 2, without the necessity of establishing irreparable harm or the posting of a bond, and to recover from Executive damages incurred by the Company as a result of the breach, as well as the Company's attorneys' fees, costs and expenses related to any breach or threatened breach of this Agreement and enforcement of this Agreement. Nothing contained in this Agreement shall be construed as prohibiting the Company from pursuing any other remedies available to it for any breach or threatened breach, including, without limitation, the recovery of money damages, attorneys' fees, and costs. The existence of any claim or cause of action by Executive against the Company, whether predicated on this Agreement or otherwise, shall not constitute a defense to the enforcement by the Company of the restrictions or covenants contained in Section 1 or Section 2, or preclude injunctive relief.
- 5. <u>Business Opportunities</u>. Executive, without further compensation, assigns and agrees to assign to the Company and its successors, assigns or designees, all of Executive's right, title and interest in and to all Business Opportunities (defined below), and further acknowledges and agrees that all Business Opportunities constitute the exclusive property of the Company. Executive shall present all Business Opportunities to the Company, and shall not exploit a Business Opportunity. For purposes of this Agreement, "Business Opportunities" means all business ideas, prospects, or proposals pertaining to any aspect of the Company's business and any business the Company prepared to conduct or contemplated conducting during Executive's employment with the Company, which are developed by Executive or originated by any third party and brought to the attention of Executive, together with information relating thereto. For the avoidance of doubt, this Section 5 is not intended to limit or narrow Executive's duties or obligations under federal or state law with respect to corporate opportunities.
- 6. <u>Conflicting Activities</u>. Executive agrees that, during Executive's employment with the Company, Executive shall not engage in any employment, consulting relationship, business or other activity that (i) is in any way competitive with the business or proposed business of the Company (except that Executive may invest less than one percent (1%) of the shares of a company traded on a registered stock exchange); (ii) conflicts with Executive's duty of loyalty, responsibilities or obligations to the Company or interferes with the independent exercise of Executive's judgment

in the Company's best interests; or (iii) adversely affects the performance of Executive's job duties and responsibilities with the Company. Executive agrees to not assist any other person or organization in competing with the Company or in preparing to engage in competition with the Company or proposed business of the Company. Executive further agrees that, during Executive's employment with the Company, Executive shall not actively engage in any other employment, occupation or consulting activity for any direct or indirect remuneration without the prior approval of Executive's direct supervisor or the Company's Legal Department. Executive has listed on the Company's Outside Activity Disclosure form, attached hereto as Exhibit B-1, any business activities or ventures with which Executive is involved. If no such list is attached, Executive represents that there are no such outside activities as of the date of this Agreement.

- 7. **Breach** Executive acknowledges that Executive is subject to immediate dismissal by the Company for any breach of this Agreement and that such a dismissal will not relieve Executive from any continuing obligations under this Agreement or from the imposition by a court of any judicial remedies, including, without limitation, money damages and/or injunctive relief for such breach.
- 8. Notice. If Executive, in the future, seeks or is offered employment, or any other position or capacity with another company, entity or person, Executive agrees to inform each such company, entity or person of the existence of the restrictions in Section 1 and Section 2. The Company shall be entitled to advise such company, entity or person and third parties of the provisions of Section 1 and Section 2 and to otherwise deal with such company, entity, person or third party to ensure that the provisions of Section 1 and Section 2 are enforced and duly discharged.
- 9. **Reformation**. The Company and Executive agree that in the event any of the terms, provisions, covenants or restrictions contained in this Agreement, or any part thereof, shall be held by any court of competent jurisdiction to be effective in any particular area or jurisdiction only if said term, provision, covenant or restriction is modified to limit its duration or scope, then the court shall have such authority to so reform the term, provision, covenant or restriction and the Parties hereto shall consider such term, provision, covenant or restriction to be amended and modified with respect to that particular area or jurisdiction so as to comply with the order of any such court and, as to all other jurisdictions, the term, provision, covenant or restriction contained herein shall remain in full force and effect as originally written. By agreeing to this contractual modification prospectively at this time, the Company and Executive intend to make Section 1 and Section 2 enforceable under the law or laws of all applicable jurisdictions so that the restrictive covenants in their entirety and this Agreement as prospectively modified shall remain in full force and effect and shall not be rendered void or illegal.
- 10. Severability. In the event any court of competent jurisdiction or any foreign, federal, state, county or local government or any other governmental regulatory or administrative agency or authority holds any provision of this Agreement to be invalid, illegal or unenforceable, such invalid, illegal or unenforceable portion(s) shall be limited or excluded from this Agreement to the minimum extent required, and the remaining provisions shall not be affected or invalidated and shall remain in full force and effect.

- 11. **Binding Effect of Agreement and Assignment**. This Agreement shall be binding upon and inure to the benefit of the Parties hereto and their respective heirs, successors, legal representatives and permitted assigns. Executive may not assign this Agreement to a third party. The Company may assign its rights, together with its obligations hereunder, to any affiliate and/or subsidiary of the Company or any successor thereto or any purchaser of substantially all of the assets of the Company, without Executive's consent and without advance notice.
- 12. **Survival**. Executive agrees that Executive's obligations under this Agreement shall continue in effect after the termination of Executive's employment, regardless of the reason(s) for termination, and whether such termination is voluntary or involuntary.
- 13. Waiver. The failure of either Party to insist in any one or more instances upon performance of any terms or conditions of this Agreement shall not be construed as a waiver of future performance of any such term, covenant or condition, but the obligations of either Party with respect thereto shall continue in full force and effect. No waiver of any breach of this Agreement shall be construed to be a waiver as to succeeding breaches and no waiver of any provisions of this Agreement shall constitute a waiver of any other provision of this Agreement. The breach by one Party to this Agreement shall not preclude equitable relief, injunctive relief or the obligations in Section 1 or Section 2.
- 14. **Controlling Law.** This Agreement shall be governed by and construed under the laws of the State of Texas, without regard to any applicable conflict of law or choice of law rules.
- 15. <u>Venue</u>. Venue of any dispute arising out of, in connection with or in any way related to this Agreement shall be in a state district court of competent jurisdiction in Dallas County, Texas, or the United States District Court for the Northern District of Texas. Executive consents to personal jurisdiction of the state district courts of Dallas County, Texas and to the United States District Court for the Northern District of Texas for any dispute arising out of, in connection with or in any way related to this Agreement, and agrees that Executive shall not challenge personal jurisdiction in such courts. Executive waives any objection that Executive may now or hereafter have to the venue or jurisdiction of any proceeding in such courts or that any such proceeding was brought in an inconvenient forum (and agrees not to plead or claim the same).
- 16. WAIVER OF JURY TRIAL. WITH RESPECT TO ANY DISPUTE BETWEEN EXECUTIVE AND THE COMPANY ARISING OUT OF, IN CONNECTION WITH OR IN ANY WAY RELATED TO THIS AGREEMENT, EMPLOYEE AGREES TO RESOLVE SUCH DISPUTE(S) BEFORE A JUDGE WITHOUT A JURY. EMPLOYEE HAS KNOWLEDGE OF THIS PROVISION, AND WILL PROVIDE SERVICES TO THE COMPANY THEREAFTER, HEREBY WAIVING EXECUTIVE'S RIGHT TO TRIAL BY JURY AND AGREES TO HAVE ANY DISPUTE(S) ARISING BETWEEN THE COMPANY AND EXECUTIVE ARISING OUT OF, IN CONNECTION WITH OR IN ANY WAY RELATED TO THIS AGREEMENT RESOLVED BY A JUDGE OF A COMPETENT COURT IN DALLAS COUNTY, TEXAS, SITTING WITHOUT A JURY.
- 17. **Entire Agreement**. This Agreement constitutes the entire agreement between the Parties with respect to the subject matter hereof, and fully supersedes any and all prior and

contemporaneous agreements, understandings and/or representations between the Parties, whether oral or written, pertaining to the subject matter of this Agreement; provided, however, Executive's obligations under this Agreement are in addition to Executive's obligations under any applicable law or regulation and the Company's policies and procedures. No oral statements or prior written material not specifically incorporated in this Agreement shall be of any force and effect, and no changes in or additions to this Agreement shall be recognized, unless incorporated in this Agreement by written amendment, such amendment to become effective on the date stipulated in it. Any amendment to this Agreement must be signed by all parties to this Agreement.

- 18. <u>Disclaimer of Reliance</u>. Except for the specific representations expressly made by the Company in this Agreement, Executive specifically disclaims that Executive is relying upon or has relied upon any communications, promises, statements, inducements, or representation(s) that may have been made, oral or written, regarding the subject matter of this Agreement. Executive represents that Executive relied solely and only on Executive's own judgment in making the decision to enter into this Agreement.
- 19. **Voluntary Agreement**. Executive (i) acknowledges that Executive has read and understands the terms of this Agreement and believes them to be reasonable, (ii) agrees that the consideration provided by the Company for this Agreement is reasonable, and (iii) is voluntarily executing this Agreement as signified by Executive's signature hereto,.
- 20. **Execution in Multiple Counterparts**. This Agreement may be executed in multiple counterparts, whether or not all signatories appear on these counterparts, and each counterpart shall be deemed an original for all purposes.

[Signature Page Follows]

| The signatures below indicate that the Parties have read, understand and will comply with this Agreement. |
|---|
|   |

| EXECUTIVE:                                       | Signature:               |
|--|--------------------------|
|  | Printed Name: Mike Hogan |
|  | Date:                    |
| THE COMPANY:                                     | Blucora, Inc.:           |
|  | Signature:               |
|  | Name:                    |
|  | Title:                   |
|  | Date:                    |
|  |                          |
|  |                          |
|  |                          |
| CONFIDENTIALITY AND NON-COMPETITION AGREEMENT PA | age 12                   |

# EXHIBIT A-1 LIST OF PRIOR INVENTIONS

| Title   | Date        | Identifying Number or Brief Description |
|---|-------------|---|
|   |             |   |
|   |             |   |
|   |             |   |
|   |             |   |
|   |             |   |
|   |             |   |
| No Inventions Additional Sheets Attached Signature of Employee: |             |   |
| Print Name of Employee:   |             |   |
| Date:   |             |   |
| CONFIDENTIALITY AND NON-COMPETITION AGREEMI                     | ENT Page 13 |   |

# EXHIBIT A-2 OUTSIDE ACTIVITIES

### **EXHIBIT B**

### GENERAL RELEASE OF ALL CLAIMS

This General Release and Waiver of Claims (this "*Release*") is executed by Mike Hogan ("*Executive*") and Blucora, Inc. (the "*Company*") as of the date set forth below, and will become effective as of the "*Effective Date*" as defined below. This Release is in consideration of severance benefits to be paid to Executive by the Company pursuant to the Employment Agreement between Executive and the Company dated as of \_\_\_\_\_, 2018 (the "*Employment Agreement*"). Execution of this Release without revocation by Executive will satisfy the requirement, set forth in Section 6(g) of the Employment Agreement, that Executive execute a general release and waiver of claims in order to receive severance benefits pursuant to the Employment Agreement.

### 1. Termination of Employment

Executive acknowledges that his employment with the Company and any of its subsidiaries (collectively, the "Company Group") and any and all appointments he held with any member of the Company Group, whether as officer, director, employee, consultant, agent or otherwise, terminated as of \_\_\_\_\_\_ (the "Termination Date"). Effective as of the Termination Date, Executive has not had or exercised or purported to have or exercise any authority to act on behalf of the Company or any other member of the Company Group, nor will Executive have or exercise or purport to have or exercise such authority in the future.

### 2. Consideration

The Company shall pay Executive the severance benefits pursuant to the Employment Agreement. The Parties agree that but for signing this Release, Executive would not be entitled to the severance benefits set forth in the Employment Agreement. The severance benefits are adequate to make this Release final and binding, and are in addition to payments and benefits to which Executive would otherwise be entitled to as an employee or former employee of the Company.

### 3. Waiver and Release

(a) Executive, for and on behalf of himself and his heirs and assigns, hereby waives and releases any common law, statutory or other complaints, claims, charges or causes of action arising out of or relating to Executive's employment or termination of employment with, or Executive's serving in any capacity in respect of any member of the Company Group (collectively, "Claims"). The Claims waived and released by this Release include any and all Claims, whether known or unknown, whether in law or in equity, which Executive may now have or ever had against any member of the Company Group or any shareholder, employee, officer, director, agent, attorney, representative, trustee, administrator or fiduciary of any member of the Company Group (collectively, the "Company Releasees") up to and including the date of Executive's execution of this Agreement. The Claims waived and released by this Release include, without limitation, any and all Claims arising out of Executive's employment with the Company Group under, by way of example and not limitation, the Age Discrimination in Employment Act of 1967 ("ADEA", a law which prohibits discrimination on the basis of age against persons age 40 and older), the National

1

Labor Relations Act, the Civil Rights Act of 1991, the Americans With Disabilities Act of 1990, Title VII of the Civil Rights Act of 1964, the Employee Retirement Income Security Act of 1974, the Family Medical Leave Act, the Securities Act of 1933, the Securities Exchange Act of 1934, and the Texas Labor Code Chapter 21, all as amended, and all other federal, state and local statutes, ordinances, regulations and the common law, and any and all Claims arising out of any express or implied contract, except as described in Paragraphs 2(b) and 2(c) below.

- (b) The waiver and release set forth in this Section 3 is intended to be construed as broadly and comprehensively as applicable law permits. The waiver and release shall not be construed as waiving or releasing any claim or right that as a matter of law cannot be waived or released, including Executive's right to file a charge with the Equal Employment Opportunity Commission or other government agency.
  - (c) Notwithstanding anything else in this Release, Executive does not waive or release claims with respect to:
    - (i) Executive's entitlement, if any, to severance benefits pursuant to the Employment Agreement;
- (ii) vested benefits or payments specifically to be provided to the Executive pursuant to the Employment Agreement or any Company employee benefit plans or policies;
- (iii) indemnification pursuant to any applicable provision of the Company's Bylaws or Certificate of Incorporation, as amended, pursuant to any written indemnification agreement between the Executive and the Company, or pursuant to applicable law;
- (iv) any claims which the Executive may have solely by virtue of the Executive's status as a shareholder of the Company; or
  - (v) unemployment compensation to which Executive may be entitled under applicable law.
- (d) Executive represents and warrants that he is the sole owner of the actual or alleged Claims that are released hereby, that the same have not been assigned, transferred, or disposed of in fact, by operation of law, or in any manner, and that he has the full right and power to grant, execute and deliver the releases, undertakings, and agreements contained herein.
- (e) Subject to Section 4, Executive represents that Executive has not filed any complaints, charges or lawsuits against the Company with any governmental agency or any court based on Claims that are released and waived by this Release.

### 4. No Interference

Nothing in this Agreement is intended to interfere with Executive's right to report possible violations of federal, state or local law or regulation to any governmental or law enforcement agency or entity, or to make other disclosures that are protected under the whistleblower provisions of federal, state or local law or regulation. Executive further acknowledges that nothing in this

Agreement is intended to interfere with Executive's right to file a claim or charge with, or testify, assist, or participate in an investigation, hearing, or proceeding conducted by, the Equal Employment Opportunity Commission (the "*EEOC*"), any state human rights commission, or any other government agency or entity. However, by executing this Agreement, Executive hereby waives the right to recover any damages or benefits in any proceeding Executive may bring before the EEOC, any state human rights commission, or any other government agency or entity or in any proceeding brought by the EEOC, any state human rights commission, or any other government agency or entity on Executive's behalf with respect to any claim released in this Agreement *except that* Executive may receive bounty money awarded by the U.S. Securities and Exchange Commission pursuant to Section 21F of the Securities Exchange Act of 1934 or any similar provision.

# 5. No Admission of Wrongdoing

This Release shall not be construed as an admission by either party of any wrongful or unlawful act or breach of contract.

### 6. Legal Disclosure

Subject to Section 4, by signing this Agreement, Executive warrants and represents that Executive has reported to Human Resources or Legal all pending and/or threatened legal proceedings of any kind involving or relating to the Company that Executive became aware of during Executive's tenure with the Company. Executive further warrants and represents that Executive has reported to Human Resources or Legal any alleged violations of law (including alleged securities violations) by the Company that Executive became aware of during Executive's tenure with the Company.

### 7. Binding Agreement; Successors and Assigns

This Release binds Executive's heirs, administrators, representatives, executors, successors, and assigns, and will inure to the benefit of the respective heirs, administrators, representatives, executors, successors, and assigns of any person or entity as to whom the waiver and release set forth in Section 3 applies.

# 8. Other Agreements

This Release does not supersede or modify in any way Executive's continuing obligations pursuant to the Employment Agreement or the Confidentiality and Non-Competition Agreement (Exhibit A thereto) or the dispute resolution provisions of the Employment Agreement.

### 9. Knowing and Voluntary Agreement; Consideration and Revocation Periods

- (a) Executive acknowledges that Executive has been given twenty-one (21) calendar days from the date of receipt of this Release to consider all of the provisions of this Release and that if Executive signs this Release before the 21-day period has ended he knowingly and voluntarily waives some or all of such 21-day period.
- (b) Executive represents that (i) Executive has read this Release carefully, (ii) Executive has hereby been advised by the Company to consult an attorney of his choice and has either done

so or voluntarily chosen not to do so, (iii) Executive fully understands that by signing below he is giving up certain rights which he might otherwise have to sue or assert a claim against any of the Company Releasees, and (iv) Executive has not been forced or pressured in any manner whatsoever to sign this Release, and agrees to all of its terms voluntarily.

(c) Executive shall have seven (7) calendar days from the date of his execution of this Release (the "*Revocation Period*") in which Executive may revoke this Release. Such revocation must be in writing and delivered, prior to the expiration of the Revocation Period, to the attention of the Company's Chief Legal Officer at the Company's then-current headquarters address. If Executive revokes this Release during the Revocation Period, then the Release shall be null and void and without effect.

### 10. Disclaimer of Reliance

Except for the specific representations expressly made by the Company in the Employment Agreement, Executive specifically disclaims that Executive is relying upon or has relied upon any communications, promises, statements, inducements, or representation(s) that may have been made, oral or written, regarding the subject matter of this Release. Executive represents that Executive relied solely and only on Executive's own judgment in making the decision to enter into this Release.

## 11. Execution in Multiple Counterparts

This Release may be executed by the parties in multiple counterparts, whether or not all signatories appear on these counterparts (including via electronic signatures and exchange of PDF documents via email), each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

### 12. Effective Date

The Effective Date of this Release will be day after the Revocation Period expires without revocation by Executive.

[Signature Page Follows]

EXECUTIVE HAS ELECTED FREELY AND VOLUNTARILY TO EXECUTE THIS RELEASE, TO FULFILL THE PROMISES SET FORTH IN THE EMPLOYMENT AGREEMENT, AND TO RECEIVE THEREBY THE PAYMENT AND OTHER CONSIDERATION DESCRIBED IN THE EMPLOYMENT AGREEMENT. EXECUTIVE UNDERSTANDS THAT, BY SIGNING THIS RELEASE, EXECUTIVE IS AGREEING TO WAIVE AND SETTLE THE RELEASED CLAIMS HEREIN THAT EXECUTIVE HAS OR MIGHT HAVE AGAINST THE COMPANY INCLUDING CLAIMS UNDER THE AGE DISCRIMINATION IN EMPLOYMENT ACT. EXECUTIVE'S SIGNATURE BELOW MEANS THAT EXECUTIVE HAS READ THE RELEASE AND UNDERSTANDS AND AGREES THAT EXECUTIVE IS RELEASING ALL CLAIMS AND AGREES AND CONSENTS TO THE TERMS AND CONDITIONS OF THIS EMPLOYMENT AGREEMENT AND THIS RELEASE.

| EXECUTIVE:   | Signature:               |
|--------------|--------------------------|
|              | Printed Name: Mike Hogan |
|              | Date:                    |
| THE COMPANY: | Blucora, Inc.:           |
|              | Signature:               |
|              | Name:                    |
|              | Title:                   |
|              | Date:                    |

### SEPARATION AND RELEASE AGREEMENT

This Separation and Release Agreement ("Agreement") is entered into by and between Blucora, Inc. (the "Company"), H.D. Vest, Inc., ("HD Vest") and Robert D. Oros ("Executive"), effective as the date both Parties execute this Agreement ("Effective Date"). The Company and Executive are referred to herein individually, as a "Party" and collectively, as the "Parties." Defined terms not defined in this Agreement have the meaning set forth in the Employment Agreement

WHEREAS, Executive has been employed by the Company as its Chief Executive Officer ("CEO");

WHEREAS, Executive executed an Employment Agreement with the Company effective as of January 22, 2017 ("Employment Agreement");

**WHEREAS**, Executive voluntarily resigns his employment and officer position effective November 15, 2018 (the "Separation Date"):

**WHEREAS**, Executive agrees to make himself available following the Separation Date to provide transition services to the Company through March 1, 2019 (the "*Transition Period*");

WHEREAS, the Parties desire to set forth Executive's separation benefits and obligations and to finally, fully and completely resolve all matters arising from or during Executive's employment and separation from employment, any benefits, bonuses and compensation connected with such employment and all other disputes and matters that the Parties may have for any reason; and

**NOW, THEREFORE**, in consideration of the premises and mutual covenants and agreements hereinafter set forth, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereto agree as follows:

### **End of Executive's Employment and Transition.**

- (a) The Parties agree that Executive's employment with the Company shall terminate on the Separation Date. All of Executive's positions with the Company, including all officer positions, shall terminate effective as of the Separation Date. Executive shall execute all documents and take such further steps as may be required to effectuate such termination(s). Executive shall not perform any work except as set forth in this Agreement, and shall not make any representations or execute any documents, or take any other actions, on behalf of the Company as of the Separation Date. The Company shall pay Executive for all base salary earned but unpaid through the Separation Date, and shall reimburse Executive for all reasonable business expenses incurred and submitted in accordance with the Company's policies.
- (b) Notwithstanding the foregoing, Executive agrees to cooperate fully and provide assistance during the Transition Period, as requested by the Company, in the orderly transitioning of Executive's duties and responsibilities to such other persons as the Company shall designate and agrees to thoroughly and diligently perform those duties and actions which are necessary or appropriate to cause such orderly transition. Executive agrees to: (i) fully inform the Company and Executive's successors of all activities in which Executive was involved prior to the Separation Date and of the status of any projects; (ii) transfer or otherwise make available to Executive's successors or others designated by the Company to the extent possible, all of Executive's knowledge and experience regarding Executive's duties; (iii) accomplish a smooth transition of Executive's responsibilities to Executive's successors; (iv) comply with the Company's and HD Vest's codes of conduct and employee handbooks, and this Agreement, (v) not take

any action contrary to the goodwill, reputation, and ongoing business of the Company including not making any disparaging comments regarding the Company or its officers, directors, executives, shareholders or employees; and (vi) take all steps necessary to maintain, and in no way act to hinder, the foregoing duties (collectively, the "*Transition Services*"). Executive acknowledges and agrees that Executive shall continue to receive his base salary during the Transition Period as compensation for the Transition Services but shall receive no other compensation other than the compensation and benefits provided for in this Agreement. Should Executive provide the Transition Services to the satisfaction of the Company throughout the term of the Transition Period, Executive will be entitled to vest previously granted equity, which has vesting dates during the Transition Period. Further, the Company and Executive agree that, should Executive accept a full-time role which (a) does not violate Section 7 of this Agreement or the terms of the Surviving Agreement (as defined below), (b) has a start date of January 1, 2019 or later, and (c) for which no public announcement in any way related to Executive's acceptance of the role or employment with the employer is made before January 1, 2019, Executive may terminate the Transition Period with fifteen days' notice to the Company. If Executive terminates the Transition Period pursuant to this Section 1(b) prior to the date of vesting of any equity awards that occurs during the Transition Period, Executive will forfeit the vesting of equity which would have otherwise vested in the Transition Period.

- 2. Consideration. Provided that Executive complies and remains in continued compliance with this Agreement and the Surviving Agreement and does not revoke this Agreement under Section 17, the Company agrees to waive Executive's obligation in Exhibit C of the Employment Agreement, Commuting, Relocation and Other Expenses, to repay to the Company all Moving/Commuting Expenses paid by the Company or reimbursed to the Executive. Executive acknowledges and agrees that, but for this Agreement, Executive is not otherwise entitled to the consideration set forth in this Section 2. In the event Executive fails to timely execute this Agreement, or revokes this Agreement in accordance with Section 17 below, or breaches the terms of this Agreement, Executive must fully repay to the Company all Moving/Commuting Expenses as set forth in Exhibit C of the Employment Agreement. Other than the consideration provided for in this Agreement, Executive shall not be entitled to any additional compensation, bonuses, severance pay, payments, grants, options or benefits under the Employment Agreement, any other agreement or any benefit plan, long term incentive plan, short term incentive plan, severance pay plan or bonus or incentive program established by the Company.
- 3. Release. In consideration of the promises of the Company provided herein, including, the consideration provided for in Section 2 and other consideration provided for in this Agreement, that being good and valuable consideration, the receipt, adequacy and sufficiency of which Executive acknowledges, Executive, on Executive's own behalf and on behalf of Executive's agents, administrators, representatives, executors, successors, heirs, devisees and assigns (collectively, the "Executive Releasing Parties") hereby fully and forever waives, releases, extinguishes and discharges the Company, HD Vest their shareholders, their affiliates, subsidiaries and each of their respective past, present and future parents, owners, officers, directors, shareholders, members, executives, employees, consultants, independent contractors, partners, agents, attorneys, advisers, insurers, fiduciaries, employee benefit plans, representatives, successors and assigns (each, a "Company Released Party" and collectively, the "Company Released Parties"), jointly and severally, from any and all claims, rights, demands, debts, obligations, losses, causes of action, suits, controversies, setoffs, affirmative defenses, counterclaims, third party actions, damages, penalties, costs, expenses, attorneys' fees, liabilities and indemnities of any kind or nature whatsoever (collectively, the "Claims"), whether known or unknown, suspected or unsuspected, accrued or unaccrued, whether at law, equity, administrative, statutory or otherwise, and whether for injunctive relief, back pay, front pay, fringe benefits, reinstatement, reemployment, compensatory damages, punitive damages, or any other kind of damages, which any of Executive Releasing Parties have, had or may have against any of the Company Released Parties relating to or arising out of any matter arising on or before the date this Agreement is

executed by Executive. Such released Claims include, without limitation, all Claims arising from or relating to Executive's employment with the Company or the termination of that employment relationship or any circumstances related thereto, or any other matter, cause or thing whatsoever, including without limitation all Claims arising at law or equity or sounding in contract (express or implied) or tort, Claims arising by statute, common law or otherwise, Claims arising under any federal, state, county or local laws, of any jurisdiction, including Claims for wrongful discharge, libel, slander, breach of express or implied contract or implied covenant of good faith and fair dealing, Claims for alleged fraud, concealment, negligence, negligent misrepresentation, promissory estoppel, quantum meruit, intentional or negligent infliction of emotional distress, violation of public policy, and Claims for discrimination, retaliation, sexual harassment and Claims arising under any laws that prohibit age, sex, sexual orientation, race, national origin, color, disability, religion, veteran, workers' compensation or any other form of discrimination, harassment, or retaliation, including, without limitation, Claims under the Age Discrimination in Employment Act of 1967, as amended, the Americans with Disabilities Act of 1990, as amended, the Rehabilitation Act of 1973, Title VII of the Civil Rights Act of 1964, as amended, 42 U.S.C. §1981, the Civil Rights Act of 1991, the Civil Rights Act of 1866 and/or 1871, the Equal Pay Act of 1963, the Lilly Ledbetter Fair Pay Act of 2009, the Fair Labor Standards Act, the Employee Retirement Income Security Act of 1974, as amended, the Family and Medical Leave Act of 1993, the Occupational Safety and Health Act, the Employee Polygraph Protection Act, the Uniformed Services Employment and Reemployment Rights Act, the Worker Adjustment and Retraining Notification Act, the Genetic Information Nondiscrimination Act, the National Labor Relations Act, the Labor Management Relations Act, the Immigration Reform and Control Act, any statute or laws of the State of Texas (including but not limited to the Texas Labor Code), the State of Massachusettes (including the Fair Employment Practices Act), any other t federal or state laws, any other federal, state, local, municipal or common law whistleblower, discrimination or anti-retaliation statute law or ordinance, and any other Claims arising under state or federal law, as well as any expenses, costs or attorneys' fees. Except as required by law, Executive agrees that Executive will not commence, maintain, initiate, or prosecute, or cause, encourage, assist, volunteer, advise or cooperate with any other person to commence, maintain, initiate or prosecute, any action, lawsuit, proceeding, charge, petition, complaint or Claim before any court, agency or tribunal against the Company or any of the Company Released Parties arising from, concerned with, or otherwise relating to, in whole or in part, Executive's employment, the terms and conditions of Executive's employment, or Executive's separation from employment with the Company or any of the matters or Claims discharged and released in this Agreement. This release shall not apply to any of the Company's obligations under this Agreement.

4. No Interference. Nothing in this Agreement or the Surviving Agreement is intended to interfere with Executive's right to report possible violations of federal, state or local law or regulation to any governmental or law enforcement agency or entity, or to make other disclosures that are protected under the whistleblower provisions of federal or state law or regulation. Executive further acknowledges that nothing in this Agreement is intended to interfere with Executive's right to file a claim or charge with, or testify, assist, or participate in an investigation, hearing, or proceeding conducted by, the Equal Employment Opportunity Commission (the "EEOC"), any state human rights commission, or any other government agency or entity. In making such disclosures, Executive need not seek prior authorization from the Company, and is not required to notify the Company of any such reports, disclosures or conduct. However, by executing this Agreement, Executive hereby waives the right to recover any damages or benefits in any proceeding Executive may bring before the EEOC, any state human rights commission, or any other government agency or entity or in any proceeding brought by the EEOC, any state human rights commission, or any other government agency or entity on Executive's behalf with respect to any Claim released in this Agreement; except that Executive may receive bounty money awarded by the U.S. Securities and Exchange Commission pursuant to Section 21F of the Securities Exchange Act of 1934 or any similar provision.

- **5.** Known Violations. Executive represents and warrants that Executive is not aware of any illegal acts committed by or on behalf of the Company and represents that if Executive is or had been aware of any such conduct, that Executive has properly reported the same to the Company's Chief Legal Officer in writing. Executive further represents and warrants that Executive is not aware of any (i) violations, allegations or claims that the Company has violated any federal, state, local or foreign law or regulation of any kind, or (ii) any facts, basis or circumstances relating to any alleged violations, allegations or claims that the Company has violated any federal, state, local or foreign law or regulation of any kind. If Executive learns of any such information, Executive shall immediately inform the Company's Chief Legal Officer.
- 6. Return of Company Property. Within three days of the Separation Date, Executive shall, to the extent not previously returned or delivered, without copying or retaining any copies: (a) return all equipment, records, files, documents, data, computer programs, programs or other materials and property in Executive's possession which belong to the Company or any one or more of its affiliates, including, without limitation, all computer access codes, messaging devices, credit cards, cell phones, laptops, computers and related equipment, keys and access cards; and (b) deliver all original and copies of The Company Confidential Information, notes, materials, records, reports, plans, data or other documents, files or programs (whether stored in paper form, computer form, digital form, electronically or otherwise or on Executive's personal computer or any other media) that relate or refer to (1) the Company or any one or more of its affiliates, or (2) the Company's or any one or more of its affiliates' financial information, financial data, financial statements, personnel information, business information, strategies, sales, customers, suppliers, The Company Confidential Information or similar information. Should Executive later discover additional items described or referenced in subsections (a) or (b) above, Executive will promptly notify the Company and return/deliver such items to the Company.
- 7. <u>Surviving Agreement</u>. Executive and the Company agree that the provisions in <u>Exhibit A</u> to the Employment Agreement, Supplementary Terms of Employment Chief Executive Officer, H.D. Vest, Inc. ("*Surviving Agreement*"), shall survive the termination of Executive's employment and this Agreement and shall remain in full force and effect as set forth therein except as hereby amended:

### "11. Non-Competition:

- (a) For a period of two (2) years following the Separation Date, Executive shall not, within the United States and Canada, accept employment with or provide services for any entity whose business is or which takes steps to establish a business that will be, or engage in any activities that are, competitive with the Company's business. For purposes of this paragraph 11, 'the Company's business' shall mean (a) preparation and tax preparation-related products and services provided to consumers and small businesses, and to or through tax professionals; (b) investment and insurance products or services, and related advice and brokerage services, provided to or through tax professionals or in conjunction with tax preparation services, and (c) providing tax advice, enabling the provision of tax advice or using tax relationships or data to inform or enable the provision of investing advice and guidance.
- (b) For a period of two (2) years following the Separation Date, Executive shall not work with or for, or provide any services for, whether paid or unpaid, Lightyear, Genstar, and Stone Point, and any entity that advises or sponsors First Global, Cetera or Kestra.

- (c) For a period of two (2) years following the Separation Date, Executive shall seek and obtain the Company's approval to provide any consulting services, whether paid or unpaid, to or for any person or entity."
- "12. Non-Solicitation: For a period of two (2) years following the Separation Date, on Executive's own behalf or on behalf of any other person or entity, Executive shall not solicit, induce or attempt to influence directly or indirectly any employee of the Company to work for Executive or any other person or entity for whom Executive works or intends to work, not shall Executive solicit, induce or attempt to influence directly or indirectly any customer, business partner, supplier or vendor of the Company to terminate his/her/its business relationship with the Company."
- **8** . <u>Non-Disparagement</u>. Executive agrees that the Company's goodwill and reputation are assets of great value to the Company which have been obtained and maintained through great costs, time and effort. Therefore, Executive shall not in any way disparage, libel or defame the Company, its business or business practices, its products or services, or its shareholders, managers, officers, directors, employees, investors or affiliates. The rights afforded the Company under this provision are in addition to any and all rights and remedies otherwise afforded by law. Nothing in this <u>Section 8</u> is intended to interfere with Executive's rights in <u>Section 4</u>.
- **9** . <u>Neutral Reference</u>. The Company agrees to provide a neutral reference regarding Executive's employment with the Company and HD Vest. The Company shall state only Executive's position, compensation, separation date and that Executive left the Company to pursue other opportunities.
- 10. Cooperation. As a further material inducement to the Company to provide Executive the consideration provided for in Section 2 and subject to Section 4, Executive agrees (A) to cooperate and provide reasonable assistance, at the request of the Company, in the transitioning of Executive's job duties and responsibilities, (B) to be reasonably available to the Company or its representatives (including attorneys) to provide general advice or assistance as requested by the Company, and (C) to cooperate and provide reasonable assistance, at the request of the Company, in any and all investigations or other legal, equitable or business matters or proceedings which involve any matters for which Executive worked on or had responsibility during Executive's employment with the Company. This includes but is not limited to testifying (and preparing to testify) as a witness in any proceeding or otherwise providing information or reasonable assistance to the Company in connection with any investigation, claim or suit, and cooperating with the Company regarding any investigation, litigation, claims or other disputed items involving the Company that relate to matters within the knowledge or responsibility of Executive. Specifically, Executive agrees (i) to meet with the Company's representatives, its counsel or other designees at reasonable times and places with respect to any items within the scope of this provision; (ii) to provide truthful testimony regarding the same to any court, agency or other adjudicatory body; (iii) to provide the Company with immediate notice of contact or subpoena by any non-governmental party; and (iv) to not voluntarily assist any non-governmental adverse party or such non-governmental adverse party's representatives. Executive acknowledges and understands that Executive's obligations of cooperation under this Section 10 are not limited in time and may include, but shall not be limited to, the need for or availability for testimony. Executive shall receive no additional compensation for time spent assisting the Company pursuant to this Section 10 other than the compensation and benefits provided for in this Agreement, provided that Executive shall be entitled to be reimbursed for any reasonable out-of-pocket expenses incurred in fulfilling Executive's obligations pursuant to subsections (i) and (ii) above. Nothing in this Section 10 is intended to interfere with Executive's rights in Section 4.

- 11. <u>No Assignment of Claims</u>. Executive represents that Executive has not transferred or assigned, to any person or entity, any claim involving the Company or the Released Parties, or any portion thereof or interest therein. The Parties acknowledge and agree that nothing in this Agreement shall prohibit payment of any amounts due to Executive under this Agreement to Executive's estate or legal guardian.
- 12. <u>Binding Effect of Agreement</u>. This Agreement shall be binding upon and shall inure to the benefit of the Parties and their respective successors, assigns, executors, administrators, heirs and estates. The Released Parties are third-party beneficiaries of this Agreement.
- 13. Controlling Law and Venue. This Agreement shall in all respects be interpreted, enforced, and governed under the laws of the State of Texas, without regard to any conflict of law principles. The Company and Executive agree that the language in this Agreement shall, in all cases, be construed as a whole, according to its fair meaning, and not strictly for, or against, either of the Parties. The Parties agree that the Arbitration provisions in Section V of the Surviving Agreement shall govern any dispute regarding this Agreement, the Surviving Agreement and any other claim or dispute between Executive and the Company or HD Vest. Venue of any claim or dispute that is not covered by the Arbitration provisions in Section V of the Surviving Agreement shall be in a state district court of competent jurisdiction in Dallas County, Texas, or the United States District Court for the Northern District of Texas.
- 1 4. Severability. Should any provision of this Agreement be declared or determined to be illegal or invalid by any government agency or court of competent jurisdiction, the validity of the remaining parts, terms or provisions of this Agreement shall not be affected and such provisions shall remain in full force and effect. Upon any finding by any government agency or court of competent jurisdiction that Section 3 above is illegal or invalid, Executive agrees to execute a valid and enforceable general release.
- 15. <u>Breach of Agreement</u>. In the event Executive breaches any portion, or challenges the enforceability, of this Agreement, Executive shall pay the Company (i) an amount equal to all Moving/Commuting Expenses the Company paid on Executive's behalf or reimbursed to Executive, (ii) all attorneys' fees, expenses and costs the Company incurs in such action, and (iii) any and all other damages to which the Company may be entitled at law or in equity as a result of a breach of this Agreement.
- 16. Knowing and Voluntary Waiver. Executive acknowledges that Executive has had an opportunity to review all aspects of this Agreement, the Company is advising and has advised Executive in writing (i.e., through this Agreement) to consult with an attorney of Executive's own choosing at Executive's cost, regarding the effect of this Agreement, and Executive has had a reasonable opportunity to do so, if so desired. Executive understands it is Executive's choice whether or not to enter into this Agreement and that Executive's decision to do so is voluntary and is made knowingly. Executive acknowledges and understands that this Agreement specifically releases and waives all rights and claims Executive may have under the Age Discrimination in Employment Act ("ADEA") prior to the date on which Executive signs this Agreement. Furthermore, Executive acknowledges that the promises and benefits provided for in Section 2 of this Agreement will be delayed until this Agreement becomes effective, enforceable and irrevocable.
- 17. Time for Consideration. Executive has knowingly and voluntarily entered into this Agreement, and acknowledges that Executive has been given a period of 21 days from the date Executive received this Agreement to review and consider this Agreement before executing it. Executive understands that Executive has the right to use as much or as little of the 21 day period as Executive wishes before executing this Agreement. Accordingly, Executive understands Executive may execute this Agreement as soon as Executive wishes to execute it within the 21 day period. The signed Agreement must be returned to the Company, ATTN: Ann Bruder, 6333 State Hwy 161, 6th floor, Irving, TX 75038, before the end of

such 21 day period. Executive further understands that Executive may revoke this his agreement to release claims under the ADEA, within seven days after signing this Agreement, in which case the consideration provided for in Section 2 of this Agreement shall be null and void, and Executive shall fully repay to the Company all Moving/Commuting Expenses as set forth in Exhibit C of the Employment Agreement within 30 days after the Separation Date. Revocation is only effective if Executive delivers a written notice of revocation to the Company, ATTN: Ann Bruder, 6333 State Hwy 161, 6 th floor, Irving, TX 75038, within seven days after executing the Agreement.

- 18. No Admission of Liability. This Agreement shall not in any way be construed as an admission by the Company or Executive of any acts of wrongdoing or violation of any statute, law, or legal right. Rather, the Parties specifically deny and disclaim that either has any liability to the other, but are willing to enter this Agreement at this time to definitely resolve once and forever this matter and to avoid the costs, expense, and delay of litigation.
- 19. Entire Agreement. This Agreement and the Surviving Agreement constitute the entire agreement and understanding between the Parties with respect to the subject matter hereof, and fully supersede all prior and contemporaneous negotiations, understandings, representations, writings, discussions and/or agreements between the Parties, whether oral or written, pertaining to or concerning the subject matter of this Agreement, including the Employment Agreement. No oral statements or other prior written material not specifically incorporated into this Agreement, except for the Surviving Agreements, shall be of any force and effect, and no changes in or additions to this Agreement shall be recognized, unless incorporated into this Agreement by written amendment, such amendment to become effective on the date stipulated in it. Any amendment to this Agreement must be signed by all Parties to this Agreement.
- **20.** Disclaimer of Reliance. Except for the specific representations expressly made by the Company in this Agreement, Executive specifically disclaims that Executive is relying upon or has relied upon on any communications, promises, statements, inducements, or representation(s) that may have been made, oral or written, regarding the subject matter of this Agreement. The Parties represent that they are relying solely and only on their own judgment in entering into this Agreement.
- 21. No Waiver. Failure of the Company to exercise and/or delay in exercising any right, power or privilege in this Agreement or the Surviving Agreements shall not operate as a waiver. No waiver of the Company's rights hereunder shall be effective unless it is in writing and signed by the Company. The Company's waiver of any provision of the Agreement or the Surviving Agreement shall not constitute (i) a continuing waiver of that provision, or (ii) a waiver of any other provision of this Agreement. Furthermore, no waiver of any breach of any provision shall be deemed to be a waiver of any preceding or succeeding breach of the same or any other provision.
- **22.** Counterparts. This Agreement may be executed by the Parties in multiple counterparts, whether or not all signatories appear on these counterparts (including via electronic signatures and exchange of PDF documents via email), each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

[Remainder of Page Intentionally Left Blank]

PLEASE READ CAREFULLY – THIS AGREEMENT INCLUDES A RELEASE OF CLAIMS, INCLUDING A RELEASE OF CLAIMS UNDER THE AGE DISCRIMINATION IN EMPLOYMENT ACT. BEFORE SIGNING THIS AGREEMENT, READ IT, AND CAREFULLY CONSIDER IT. IF YOU CHOOSE, DISCUSS THIS AGREEMENT WITH YOUR ATTORNEY (AT YOUR OWN EXPENSE).

MY SIGNATURE BELOW MEANS THAT I HAVE READ THIS AGREEMENT AND AGREE AND CONSENT TO ALL THE TERMS AND CONDITIONS CONTAINED IN THIS AGREEMENT.

# ACCEPTED AND AGREED TO BY: EXECUTIVE /s/ Robert D. Oros ROBERT D. OROS October 28, 2018 Date BLUCORA, INC. By: /s/Ann J. Bruder Title: Chief Legal Officer Date: October 28, 2018 H.D. VEST, INC. By: Scott Rawlins Title: President Date: October 28, 2018

| Exhibit 1 | 0.51 |
|-----------|------|
|-----------|------|

# Blucora Director Tax-Smart Deferral Plan

Effective January 1, 2019

h:\rick\forms\variable.doc

--

### TABLE OF CONTENTS

**Page** 

**ARTICLE 1DEFINITIONS** 1 **ARTICLE 2ELIGIBILITY, ENROLLMENT, AND PARTICIPATION** 7 2.1 Eligibility 7 2.2 Enrollment Requirements 7 2.3 Commencement of Participation 7 ARTICLE 3DEFERRAL COMMITMENTS/VESTING/CREDITING 8 3.1 Minimum Deferrals 8 3.2 Maximum Deferral 8 3.3 Deferral of Annual Equity Grant 9 3.4 Election to Defer; Effect of Election Form 9 3.5 Withholding and Crediting of Annual Deferral Amounts 9 3.6 **Vesting** 9 3.7 Crediting/Debiting of Account Balances 10 ARTICLE 4IN-SERVICE SCHEDULED DISTRIBUTION 11 4.1 In-Service Scheduled Distribution 11 4.2 Other Benefits Take Precedence Over In-Service Scheduled **Distributions** 12 4.3 Withdrawal Payout/Suspensions for Unforeseeable Financial Emergencies 12 **ARTICLE 5TERMINATION BENEFIT** 12 5.1 Termination Benefit 12 5.2 Payment of Termination Benefit 13 **ARTICLE 6DISABILITY WAIVER AND BENEFIT** 14 6.1 **Disability Benefit** 14 6.2 Deferral Following Disability 14 **ARTICLE 7SURVIVOR BENEFIT** 14 7.1 Survivor Benefit 14

| 7.2 Payment of Survivor Benefit 14                      |
|---|
| ARTICLE 8BENEFICIARY DESIGNATION 15                     |
| 8.1 Beneficiary 15                                      |
| 8.2 Beneficiary Designation; Change; Spousal Consent 15 |
| 8.3 Acknowledgment 15                                   |
| 8.4 No Beneficiary Designation 15                       |
| 8.5 Doubt as to Beneficiary 15                          |
| 8.6 Discharge of Obligations 15                         |
| ARTICLE 9TERMINATION, AMENDMENT OR MODIFICATION 16      |
| 9.1 Termination 16                                      |
| 9.2 <b>Amendment</b> 16                                 |
| 9.3 Effect of Payment 16                                |
| ARTICLE 10ADMINISTRATION 16                             |
| 10.1 Committee Duties 16                                |
| 10.2 Administration of the Plan 17                      |
| 10.3 Administration Upon Change In Control 17           |
| 10.4 Agents 18  |
| 10.5 Binding Effect of Decisions 18                     |
| 10.6 Indemnity of Committee 18                          |
| 10.7 Company Information 18                             |
| 10.8 Reliance Upon Information 18                       |
| ARTICLE 110THER BENEFITS AND AGREEMENTS 19              |
| 11.1 Coordination with Other Benefits 19                |
| ARTICLE 12CLAIMS PROCEDURES 19                          |
| 12.1 Presentation of Claim 19                           |
| 12.2 Notification of Decision 19                        |
| 12.3 Review of a Denied Claim 20                        |
| 12.4 Decision on Review 20                              |
| 12.5 Legal Action 20                                    |
| ARTICLE 13TRUST 21                                      |
| 13.1 Establishment of the Trust 21                      |

| 13.2 Interrelationship of the Plan and the Trust 21                       |                |
|---|----------------|
| 13.3 Distributions From the Trust 21                                      |                |
| 13.4 Participants' Rights under the Trust 21                              |                |
| ARTICLE 14MISCELLANEOUS 21  |                |
| <u>14.1 Status of Plan</u> 21   |                |
| 14.2 Limitation of Rights 21  |                |
| 14.3 Nonalienation of Benefits 22   |                |
| 14.4 Unsecured General Creditor 22  |                |
| <u>14.5 Waiver</u> 22   |                |
| 14.6 Company's Liability 22   |                |
| 14.7 Nonassignability 22  |                |
| 14.8 Not a Contract of Service 23   |                |
| 14.9 Furnishing Information 23  |                |
| <u>14.10</u> <b>Terms</b> 23  |                |
| <u>14.11</u> Captions 23  |                |
| <u>14.12</u> Governing Law 23   |                |
| <u>14.13</u> <u>Notice</u> 23   |                |
| <u>14.14</u> <u>Successors</u> 24   |                |
| 14.15 Spouse's Interest 24  |                |
| <u>14.16 <b>Validity</b></u> 24   |                |
| <u>14.17</u> <b>Incompetent</b> 24  |                |
| <u>14.18 Court Order</u> 24   |                |
| 14.19 <b>Distribution in the Event of Income Inclusion Under Code Sec</b> | etion etion    |
| 409A or Other Taxation 25   |                |
| <u>14.20</u> <b>Insurance</b> 25  |                |
| 14.21 Legal Fees To Enforce Rights After Change in Con                    | <u>trol</u> 25 |
|   |                |
|   |                |
|   |                |
|   |                |
|   |                |
|   |                |
| -   |                |

### BLUCORA DIRECTOR TAX-SMART DEFERRAL PLAN

Effective January 1, 2019

Blucora, Inc., a Delaware corporation (the "Company") desires to adopt this Blucora Director Tax-Smart Deferral Plan (the "Plan"), effective as of January 1, 2019 (the "Effective Date"), to provide benefits in the form of unfunded deferred compensation to Directors (as defined below).

### **Purpose**

The purpose of the Plan is to provide specified benefits to Directors who contribute materially to the continued growth, development and future business success of the Company and its subsidiaries, if any, that are participating employers in this Plan. The provisions of this Plan shall be interpreted in accordance with the requirements of Code section 409A and the Final Treasury Regulations thereunder, it being the intent of the parties that the Plan shall be in compliance with the requirements of said Code section and said Regulations.

# ARTICLE 1 Definitions

For the purposes of this Plan, unless otherwise clearly apparent from the context, the following phrases or terms shall have the following indicated meanings:

- 1.1 "Account Balance" shall mean, with respect to a Participant, a credit on the records of the Company equal to the balance of such Participant's Deferral Account. The Account Balance, and each other specified account balance, shall be a bookkeeping entry only and shall be utilized solely as a device for the measurement and determination of the amounts to be paid to a Participant, or his or her designated Beneficiary, pursuant to this Plan.
- 1.2 "Annual Deferral Amount" shall mean that portion of a Participant's Director Fees and/or Annual Equity Grant that a Participant defers in accordance with Article 3 for any one Plan Year. In the event of a Participant's Disability, death or a Separation from Service prior to the end of a Plan Year, such year's Annual Deferral Amount shall be the actual amount withheld prior to such event.
- 1.3 "Annual Equity Grant" means the annual grant of Restricted Stock Units made to the Directors during each Plan Year pursuant to the Equity Incentive Plan. For the avoidance of doubt, the term "Annual Equity Grant" does not include the initial grant of Restricted Stock Units made to new Directors.
- "Annual Installment Method" shall be an annual installment payment over the number of years selected by the Participant in accordance with this Plan, calculated as follows: (i) for the first annual installment, the Participant's Account Balance shall be calculated as of the close of business on or around the date on which the Participant's Benefit Distribution Date, and (ii) for remaining annual installments, the Participant's Account Balance shall be calculated on every applicable anniversary of such Benefit Distribution Date. Each annual installment shall be calculated by multiplying this balance by a fraction, the numerator of which is one and the denominator of which is the remaining number of annual payments due the Participant; provided that, all installment payments made with respect a deferred Annual Equity Grant shall be made in whole Shares, rounded down to the next whole Share (other than the last installment). By way of example, if the Participant elects a ten (10) year Annual Installment Method, the first payment shall be 1/10 of the Account Balance, calculated as described in this definition. The following year, the payment shall be 1/9 of the vested Account Balance, calculated as described in this definition. For purposes of this Plan, the right to receive a benefit payment in annual installments shall be treated as the entitlement to a single payment.
- 1.5 **"Beneficiary"** shall mean one or more persons, trusts, estates or other entities, designated in accordance with Article 8, that are entitled to receive benefits under this Plan upon the death of a Participant.
- 1.6 "Beneficiary Designation Form" shall mean the form established from time to time by the Committee that a Participant completes signs and returns to the Committee to designate one or more Beneficiaries.
- 1.7 **"Benefit Distribution Date"** shall mean the date upon which all or an objectively determinable portion of a Participant's vested benefits will become eligible for distribution.

- 1.8 **"Board"** shall mean the board of directors of the Company.
- 1.9 **"Change in Control"** shall mean the occurrence of any of the following events:
  - an acquisition by any Entity of beneficial ownership (within the meaning of Rule 13d-3 promulgated under the Exchange Act) of forty percent (40%) or more of either (1) the number of then outstanding shares of common stock of the Company (the "Outstanding Company Common Stock") or (2) the combined voting power of the then outstanding voting securities of the Company entitled to vote generally in the election of directors (the "Outstanding Company Voting Securities"), provided, however, that the following acquisitions shall not constitute a Change in Control: (i) any acquisition directly from the Company, other than an acquisition by virtue of the exercise of a conversion privilege where the security being so converted was not acquired directly from the Company by the party exercising the conversion privilege, (ii) any acquisition by the Company, (iii) any acquisition by any employee benefit plan (or related trust) sponsored or maintained by the Company or any Related Company, or (iv) an acquisition by any Entity pursuant to a transaction that meets the conditions of clauses (A), (B) and (C) set forth in Section 1.9(c) below;
  - (b) a change in the composition of the Board during any two-year period such that the individuals who, as of the beginning of such two-year period, constitute the Board (the "Incumbent Board") cease for any reason to constitute at least a majority of the Board; provided, however, that for purposes of this definition, any individual who becomes a member of the Board subsequent to the beginning of the two-year period, whose election, or nomination for election by the Company's stockholders, was approved by a vote of at least a majority of those individuals who are members of the Board and who were also members of the Incumbent Board (or deemed to be such pursuant to this proviso) shall be considered as though such individual were a member of the Incumbent Board; and provided further, however, that any such individual whose initial assumption of office occurs as a result of or in connection with an actual or threatened election contest with respect to the election or removal of directors or other actual or threatened solicitation of proxies or consents by or on behalf of an Entity other than the Board shall not be considered a member of the Incumbent Board; or
  - (c) the consummation of (i) a merger or consolidation of the Company with or into any other company; (ii) a sale in one transaction or a series of transactions undertaken with a common purpose of all of the Company's outstanding voting securities; or (iii) a sale, lease, exchange or other transfer in one transaction or a series of related transactions undertaken with a common purpose of all or substantially all of the Company's assets; excluding, however, in each case, a transaction pursuant to which:
    - (A) the Entities who are the beneficial owners of the Outstanding Company Common Stock and Outstanding Company Voting Securities immediately prior to such transaction will beneficially own, directly or indirectly, at least fifty percent (50%) of the outstanding shares of common stock, and the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors, of the Successor Company in substantially the same proportions as their ownership, immediately prior to such transaction, of the Outstanding Company Common Stock and Outstanding Company Voting Securities;
    - (B) no Entity (other than the Company, any employee benefit plan (or related trust) of the Company, a Related Company or a Successor Company) will beneficially own, directly or indirectly, forty percent (40%) or more of, respectively, the outstanding shares of common stock of the Successor Company or the combined voting power of the outstanding voting securities of the Successor Company entitled to vote generally in the election of directors unless such ownership resulted solely from ownership of securities of the Company prior to the transaction; and
    - (C) individuals who were members of the Incumbent Board will immediately after the consummation of the transaction constitute at least a majority of the members of the board of directors of the Successor Company.

Where a series of transactions undertaken with a common purpose is deemed to be a transaction described in this Section 1.9(c), the effective date of such transaction shall be the date on which the last of such transactions is consummated.

Notwithstanding the foregoing provisions of this "Change in Control" definition, to the extent necessary to comply with Section 409A of the Code, an event shall not constitute a "Change in Control" for purposes of the Plan, unless such event also constitutes a change in the Company's ownership, its effective control or the ownership of a substantial portion of its assets within the meaning of Section 409A of the Code.

1.10 "Claim" shall mean any claim, liability or obligation of any nature, arising out of or relating to this Plan or an alleged breach of this Plan or Election Form.

- 1.11 "Claimant" shall have the meaning set forth in Section 12.1 below.
- 1.12 "Code" shall mean the Internal Revenue Code of 1986, as it may be amended from time to time, and the regulations and other authority issued thereunder by the appropriate governmental authority. References herein to any Code section shall include references to any successor Code section or provision.
- 1.13 "Committee" shall mean the committee described in Article 10.
- 1.14 "Company" shall mean Blucora, Inc., a Delaware corporation, and any successor to all or substantially all of the Company's assets or business.
- 1.15 "Deferral Account" shall mean (i) the sum of all of a Participant's Annual Deferral Amounts, plus (ii) amounts credited or debited to the Participant's Deferral Account in accordance with this Plan, less (iii) all distributions made to the Participant or his or her Beneficiary pursuant to this Plan that relate to his or her Deferral Account.
- 1.16 "**Director**" shall mean any non-Employee member of the Board.
- 1.17 **"Director Fees"** shall mean the annual fees paid by Company, including retainer fees and meetings fees, as compensation for serving on its Board.
- 1.18 "**Disability**" or "**Disabled**" shall mean that a Participant has incurred a Total and Permanent Disability (as such term is defined in the Equity Incentive Plan as of the Effective Date), provided that, such term meets the definition of "disability" provided for under Section 409A of the Code and the Treasury Regulations or other guidance issued thereunder.
- 1.19 "**Disability Benefit**" shall mean the benefit set forth in Article 6.
- 1.20 "Effective Date" shall mean January 1, 2019.
- 1.21 "**Election Form**" shall mean the form established from time to time by the Committee that a Participant completes, signs and returns to the Committee to make an election under the Plan.
- 1.22 "Employee" shall mean a person who is an employee of any Employer.
- 1.23 "Employer(s)" shall be defined as follows:
  - (a) Except as otherwise provided in part (b) of this Section, the term "Employer" shall mean (i) the Company; (ii) any of its subsidiaries (now in existence or hereafter formed or acquired); and/or (iii) an entity that would be aggregated and treated as a single employer with the Company under Code section 414(b) (controlled group of corporations) or Code section 414(c) (a group of trades or businesses, whether or not incorporated, under common control).
  - (b) For the purpose of determining whether a Participant has experienced a Separation from Service, the term "Employer" shall mean:
    - (i) The entity for which the Participant performs services and with respect to which the legally binding right to compensation deferred or contributed under this Plan arises; and
    - (ii) All other entities with which the entity described above would be aggregated and treated as a single employer under Code section 414(b) (controlled group of corporations) and Code section 414(c) (a group of trades or businesses, whether or not incorporated, under common control), as applicable. In order to identify the group of entities described in the preceding sentence, the Committee shall use an ownership threshold of at least 50% as a substitute for the 80% minimum ownership threshold that appears in, and otherwise must be used when applying, the applicable provisions of (A) Code section 1563 for determining a controlled group of corporations under Code section 414(b), and (B) Treas. Reg. §1.414(c)-2 for determining the trades or businesses that are under common control under Code section 414(c).
- 1.24 "Entity" shall mean any individual, entity or group (within the meaning of Section 13(d)(3) or Section 14(d)(2) of the Exchange Act.
- 1.25 "Equity Incentive Plan" shall mean the Blucora, Inc. 2018 Long-Term Incentive Plan, as it may be amended from time to time.
- 1.26 "Exchange Act" shall means the United States Securities Exchange Act of 1934, as amended.
- 1.27 "In-Service Scheduled Distribution" shall mean the distribution set forth in Section 4.1 below.
- 1.28 "**Insolvent**" shall mean either (i) the Company is unable to pay its debts as they become due, or (ii) the Company is subject to a pending proceeding as a debtor in the United States Bankruptcy Code.

- 1.29 "Parent Company" means a company or other entity which as a result of a Change in Control owns the Company or all or substantially all of the Company's assets either directly or through one or more subsidiaries.
- 1.30 "Participant" shall mean any Director who submits an executed Election Form and Beneficiary Designation Form, which are accepted by the Committee.
- 1.31 "Plan" shall mean the Blucora Director Tax-Smart Deferral Plan, which shall be evidenced by this instrument, as it may be amended from time to time, and by any other documents that together with this instrument define a Participant's rights to amounts credited to his or her Account Balance.
- 1.32 "Plan Year" shall mean a period beginning on January 1 of each calendar year and continuing through December 31 of such calendar year.
- 1.33 "Related Company" means any subsidiary of the Company or any other entity that is directly or indirectly controlled by, in control of, or under common control with the Company, including, without limitation, any Employer.
- 1.34 "Restricted Stock Unit" shall mean a "Restricted Stock Unit" as defined in the Equity Incentive Plan.
- 1.35 "Separation from Service" shall mean a termination of services provided by a Participant to the Company, whether voluntarily or involuntarily, other than by reason of death or Disability, as determined by the Committee in accordance with Treas. Reg. §1.409A-1(h). In determining whether a Participant has experienced a Separation from Service, the following provisions shall apply:
  - (a) For a Participant who provides services as a Director, except as otherwise provided in part (b) of this Section, a Separation from Service shall occur upon the Director's cessation of services as a Director.
  - (b) If a Participant provides services for an Employer as both an Employee and as a Director, to the extent permitted by Treas. Reg. §1.409A-1(h)(5) the services provided by such Participant as a Director shall not be taken into account in determining whether the Participant has experienced a Separation from Service as an Employee, and the services provided by such Participant as an Employee shall not be taken into account in determining whether the Participant has experienced a Separation from Service as a Director.
- 1.36 "Shares" shall mean the shares of common stock of the Company, par value \$0.0001 per share, which the Company is authorized to issue, or any securities into which or for which the common stock of the Company may be converted or exchanged, as the case may be.
- 1.37 "Specified Employee" shall mean any Participant who is determined to be a "key employee" (as defined under Code section 416(i) without regard to paragraph (5) thereof) for the applicable period, as determined annually by the Committee in accordance with Treas. Reg. §1.409A-1(i). In determining whether a Participant is a Specified Employee, the following provisions shall apply:
  - (a) The Committee's identification of the individuals who fall within the definition of "key employee" under Code section 416(i) (without regard to paragraph (5) thereof) shall be based upon the 12-month period ending on each December 31st (referred to below as the "identification date"). In applying the applicable provisions of Code section 416(i) to identify such individuals, "compensation" shall be determined in accordance with Treas. Reg. §1.415(c)-2(a) without regard to (i) any safe harbor provided in Treas. Reg. §1.415(c)-2(d), (ii) any of the special timing rules provided in Treas. Reg. §1.415(c)-2(e), and (iii) any of the special rules provided in Treas. Reg. §1.415(c)-2(g); and
  - (b) Each Participant who is among the individuals identified as a "key employee" in accordance with part (a) of this Section 1.37 shall be treated as a Specified Employee for purposes of this Plan if such Participant experiences a Separation from Service during the 12-month period that begins on the April 1st following the applicable identification date.
- 1.38 "Successor Company" means the surviving company, the successor company or Parent Company, as applicable, in connection with a Change in Control.
- 1.39 "Survivor Benefit" shall mean the benefit set forth in Article 7.
- 1.40 "**Termination Benefit**" shall mean the benefit set forth in Article 5.
- 1.41 "Trust" shall mean one or more trusts established by the Company in accordance with Article 13.

- 1.42 "Trustee" shall mean the duly appointed and acting trustee of the Trust, and any successor thereto.
- 1.43 "Unforeseeable Financial Emergency" shall mean a severe financial hardship of the Participant resulting from (a) an illness or accident of the Participant, the Participant's spouse, the Participant's Beneficiary or the Participant's dependent (as defined in Code section 152 without regard to paragraphs (b)(1), (b)(2) and (d)(1)(b) thereof), (b) a loss of the Participant's property due to casualty, or (c) such other similar extraordinary and unforeseeable circumstances arising as a result of events beyond the control of the Participant, all as determined by the Committee based on the relevant facts and circumstances. The circumstances that will constitute an unforeseeable emergency will depend upon the facts of each case, but an Unforeseeable Financial Emergency shall not be deemed to exist to the extent that such hardship is or may be relieved:
  - (a) Through reimbursement or compensation by insurance or otherwise;
  - (b) By liquidation of the Participant's assets, to the extent the liquidation of such assets would not itself cause severe financial hardship;
  - (c) By cessation of deferrals under this Plan or elective contributions made by the Participant under any 401(k) plan or other deferred compensation plan;
  - (d) Distributions or nontaxable loans from any other employer's plans to the extent eligible;
  - (e) Loans from commercial sources at reasonable commercial terms;
  - (f) Bank or other savings accounts; or
  - (g) Reasonable, periodic payment arrangements with a creditor.

By way of example, the need to send a Participant's child to college or the desire to purchase or improve a home is not considered an Unforeseeable Financial Emergency.

1.44 "Year of Service" shall mean each year of continuous service by the Participant as a Director with the Company.

# ARTICLE 2

### Eligibility, Enrollment, and Participation

- 2.1 <u>Eligibility</u>. Participation in the Plan shall be limited to the Directors of the Company.
- 2.2 **Enrollment Requirements**. As a condition to participation, each Director shall complete, execute and return to the Committee, an Election Form and a Beneficiary Designation Form, all by the deadline(s) established by the Committee in accordance with the applicable provisions of the Plan. In addition, the Committee shall establish from time to time such other enrollment requirements as it determines in its sole discretion are necessary.
- 2.3 <u>Commencement of Participation</u>. Provided a Director has met all enrollment requirements set forth in this Plan and required by the Committee, including returning all required documents to the Committee within the specified time periods set forth in the Plan, that Director shall commence participation in the Plan on the first day of the month following the month in which the Director completes all enrollment requirements. If a Director fails to meet all such requirements within the periods required, in accordance with the provisions of the Plan, that Director shall not be eligible to participate in the Plan until the first day of the Plan Year following the delivery to and acceptance by the Committee of the required documents.

# ARTICLE 3

## **Deferral Commitments/Vesting/Crediting**

### 3.1 **Minimum Deferrals**.

(a) <u>Annual Deferral Amount</u>. For each Plan Year, a Participant may elect to defer, as his or her Annual Deferral Amount, Director Fees and the Annual Equity Grant in the following minimum amount or percentage, as applicable, for each deferral elected:

|               | Minimum Amount     |
|---------------|--------------------|
| Deferral      |                    |
| Director Fees | \$10,000 aggregate |

| Annual Equity Grant | 20% (rounded up to the nearest |
|---------------------|--------------------------------|
|                     | whole Share)                   |

If an election is made for less than the stated minimum amounts, or if no election is made, the amount deferred shall be zero.

(b) Short Plan Year. Notwithstanding the foregoing, if a Participant first becomes a Participant after the first day of a Plan Year the minimum Annual Deferral Amount for Director Fees shall be an amount equal to the minimum set forth above, multiplied by a fraction, the numerator of which is the number of complete months remaining in the Plan Year and the denominator of which is 12. No deferral of the Annual Equity Grant is permitted for a Plan Year on or after the first day of such Plan Year.

#### 3.2 **Maximum Deferral**.

(a) <u>Annual Deferral Amount</u>. For each Plan Year, a Participant may elect to defer, as his or her Annual Deferral Amount, Director Fees and the Annual Equity Grant in the following maximum percentages, as applicable, for each deferral elected:

| Deferral            | Maximum Percentage                           |
|---------------------|--|
| Director Fees       | 100%   |
| Annual Equity Grant | 100% (rounded up to the nearest whole Share) |

- (b) Short Plan Year. Notwithstanding the foregoing, if a Participant first becomes a Participant after the first day of a Plan Year, then to the extent required by Section 3.4 and Code section 409A and related Treasury Regulations, the maximum amount of the Participant's Director Fees that may be deferred by the Participant for the Plan Year shall be determined by applying the percentages set forth in Section 3.2(a) to the portion of such compensation attributable to services performed after the date that the Participant's deferral election is made. No deferral of the Annual Equity Grant is permitted for a Plan Year on or after the first day of such Plan Year.
- 3.3 **Deferral of Annual Equity Grant**. For each Plan Year, a Participant may elect to defer, as his or her Annual Deferral Amount, a portion of the Annual Equity Grant to be granted during such Plan Year, in the percentages as set forth above in Section 3.1 and Section 3.2. If any portion of an Annual Equity Grant is deferred pursuant to this Plan, then notwithstanding anything to contrary in the award agreement for such Annual Equity Grant, the Shares to be delivered upon the conversion of the Restricted Stock Units underlying the portion of the Annual Equity Grant that has been deferred, shall be withheld and delivered to the Participant on the Benefit Distribution Date designated by the Participant.

### 3.4 <u>Election to Defer; Effect of Election Form.</u>

- (a) First Plan Year. A Director who first becomes eligible to participate in the Plan on or after the beginning of a Plan Year, as determined in accordance with Treas. Reg. §1.409A-2(a)(7)(ii) and the "plan aggregation" rules provided in Treas. Reg. §1.409A-1(c)(2), may be permitted to make an election to defer the portion of the Director Fees (but not the Annual Equity Grant) attributable to services to be performed after such election, provided that the Participant submits an Election Form on or before the deadline established by the Committee, which in no event shall be later than 30 days after the Participant first becomes eligible to participate in the Plan. Any deferral election made in accordance with this Section 3.4(a) shall become irrevocable no later than the 30<sup>th</sup> day after the date the Director becomes eligible to participate in the Plan.
- (b) <u>Subsequent Plan Years</u>. For each succeeding Plan Year, the Participant's prior election shall remain in effect unless the Participant makes a subsequent irrevocable deferral election for such succeeding Plan Year, by timely delivering a new Election Form to the Committee, in accordance with its rules and procedures, before the December 31 preceding the Plan Year for which the election is made. If no such Election Form is timely delivered for a Plan Year, the Annual Deferral Amount shall be the same amount as elected for the immediately prior Plan Year.

- 3.5 Withholding and Crediting of Annual Deferral Amounts. For each Plan Year, (a) the Director Fees portion of the Annual Deferral Amount shall be withheld at the time the Director Fees are or otherwise would be paid to the Participant, whether or not this occurs during the Plan Year itself, and (b) the Annual Equity Grant portion of the Annual Deferral Amount shall be withheld at the time such Annual Equity Grant is made by the Company to the Participant, provided that, the Shares underlying the portion of such Annual Equity Grant shall be withheld at the time the Restricted Stock Units underlying the Annual Equity Grant become vested. Annual Deferral Amounts shall be credited to a Participant's Deferral Account at the time such amounts would otherwise have been paid to the Participant.
- 3.6 <u>Vesting</u>. A Participant shall at all times be 100% vested in his or her Deferral Account; provided that, the portion of the Deferral Account that relates to any Annual Equity Grant that has been deferred shall be subject to the vesting conditions otherwise applicable to such Annual Equity Grant.
- 3.7 <u>Crediting/Debiting of Account Balances</u>. In accordance with, and subject to, the rules and procedures that are established from time to time by the Committee, in its sole discretion, with respect to a Participant's Account Balance relating to the deferred Director Fees (but not the Annual Equity Grant), amounts shall be credited or debited to a Participant's Account Balance in accordance with the following rules:
  - (a) Measurement Funds. The Participant may elect one or more of the measurement funds selected by the Committee, in its sole discretion, which are based on certain mutual funds or such other financial measures as selected by the Committee in its sole discretion (the "Measurement Funds"), for the purpose of crediting or debiting additional amounts to his or her Account Balance. As necessary, the Committee may, in its sole discretion, discontinue, substitute or add a Measurement Fund. Each such action will take effect as of the first day of the first calendar quarter that begins at least 30 days after the day on which the Committee gives Participants advance written notice of such change.
  - (b) Election of Measurement Funds. A Participant, in connection with his or her initial deferral election in accordance with Section 3.4(a) above, shall elect, on the Election Form, one or more Measurement Fund(s) (as described in Section 3.7(a) above) to be used to determine the amounts to be credited or debited to his or her Account Balance. If a Participant does not elect any of the Measurement Funds as described in the previous sentence, the Participant's Account Balance shall automatically be allocated into the lowest-risk Measurement Fund, as determined by the Committee, in its sole discretion. The Participant may (but is not required to) elect, by submitting an Election Form to the Committee that is accepted by the Committee, to add or delete one or more Measurement Fund(s) to be used to determine the amounts to be credited or debited to his or her Account Balance, or to change the portion of his or her Account Balance allocated to each previously or newly elected Measurement Fund. If an election is made in accordance with the previous sentence, it shall apply as of the first business day deemed reasonably practicable by the Committee, in its sole discretion, and shall continue thereafter for each subsequent day in which the Participant participates in the Plan, unless changed in accordance with the previous sentence. Notwithstanding the foregoing, all investments hereunder shall be considered assets of the Company, and the Participant shall remain subject to all applicable provisions of the Plan, including, without limitation, Sections 13.4 and 14.2 below. Each Participant, as a condition to his or her participation in the Plan, agrees to indemnify and hold harmless the Committee, the Company, and each Employer, and their representatives, delegates and agents, from and against any investment losses or other damages of any kind relating to the deemed investment of the Participant's Account Balance under the Plan. No assurances are provided by any person or entity that any investment results of any Measurement Fund will be favorable and, as with most investments, there is a risk of loss.
  - (c) <u>Proportionate Allocation</u>. In making any election described in Section 3.7(b) above, the Participant shall specify on the Election Form, in increments of one percent (1%), the percentage of his or her Account Balance to be allocated to a Measurement Fund (as if the Participant was making an investment in that Measurement Fund with that portion of his or her Account Balance).
  - (d) <u>Crediting or Debiting Method</u>. The performance of each Measurement Fund (either positive or negative) will be determined by the Committee, in its sole discretion on a daily basis based on the manner in which such Participant's Account Balance has been hypothetically allocated among the Measurement Funds by the Participant.

(e) No Actual Investment. Notwithstanding any other provision of this Plan that may be interpreted to the contrary, the Measurement Funds are to be used for measurement purposes only, and a Participant's election of any such Measurement Fund, the allocation of his or her Account Balance thereto, the calculation of additional amounts and the crediting or debiting of such amounts to a Participant's Account Balance shall not be considered or construed in any manner as an actual investment of his or her Account Balance in any such Measurement Fund. In the event that the Company or the Trustee, in its own discretion, decides to invest funds in any or all of the investments on which the Measurement Funds are based, no Participant shall have any rights in or to such investments themselves. Without limiting the foregoing, a Participant's Account Balance shall at all times be a bookkeeping entry only and shall not represent any investment made on his or her behalf by the Company or the Trust; the Participant shall at all times remain an unsecured creditor of the Company.

# ARTICLE 4 In-Service Scheduled Distribution

- 4.1 **In-Service Scheduled Distribution.** In connection with each election to defer an Annual Deferral Amount, a Participant may irrevocably elect to receive an In-Service Scheduled Distribution from the Plan with respect to all or a portion of the Annual Deferral Amount. The In-Service Scheduled Distribution shall be a lump sum payment of cash and/or Shares (if an Annual Equity Grant has been deferred) in an amount (and/or number of Shares) that is equal to the portion of the Annual Deferral Amount that the Participant elected to have distributed as an In-Service Scheduled Distribution, plus amounts credited or debited in the manner provided in Section 3.7 above to that amount, calculated as of the close of business on or around the Benefit Distribution Date designated by the Participant in accordance with this Section 4.1 (a "Scheduled Distribution"). The Benefit Distribution Date for an amount subject to an In-Service Scheduled Distribution election shall be the first day of any Plan Year designated by the Participant, which may be no sooner than 3 Plan Years after the end of the Plan Year in which the Annual Deferral Amount is actually deferred. The Participant may elect different Benefit Distribution Dates for Deferred Fees and Annual Equity Grants deferred with respect to different Plan Years. Subject to the other terms and conditions of this Plan, each In-Service Scheduled Distribution elected shall be paid out during a 60-day period commencing immediately after the Benefit Distribution Date. By way of example, if an In-Service Scheduled Distribution is elected for Annual Deferral Amounts that are deferred in the Plan Year commencing January 1, 2019, the earliest Benefit Distribution Date that may be designated by a Participant would be January 1, 2023, and the In-Service Scheduled Distribution would be paid out during the 60-day period commencing immediately after such Benefit Distribution Date.
- 4.2 Other Benefits Take Precedence Over In-Service Scheduled Distributions. Should an event occur prior to any Benefit Distribution Date designated for a Scheduled Distribution that triggers a benefit under Articles 5, 6, or 7, any Annual Deferral Amount, plus amounts credited or debited thereon, that are subject to an In-Service Scheduled Distribution election under Section 4.1 above shall not be paid in accordance with Section 4.1 above, but shall be paid in accordance with the other applicable Article.
- 4.3 <u>Withdrawal Payout/Suspensions for Unforeseeable Financial Emergencies.</u>
  - (a) If the Participant experiences an Unforeseeable Financial Emergency prior to the occurrence of a distribution event under Articles 5, 6, or 7, the Participant may petition the Committee to suspend deferrals of Director Fees required to be made by such Participant, to the extent deemed necessary by the Committee to satisfy the Unforeseeable Financial Emergency. If suspension of deferrals is not sufficient to satisfy the Participant's Unforeseeable Financial Emergency, the Participant may further petition the Committee to receive a partial or full payout from the Plan. The Participant shall only receive a payout from the Plan to the extent such payout is deemed necessary by the Committee to satisfy the Participant's Unforeseeable Financial Emergency.
  - (b) The payout shall not exceed the lesser of (i) the Participant's vested Account Balance, calculated as of the close of business on or around the Benefit Distribution Date for such payout, as determined by the Committee in accordance with the provisions set forth below, or (ii) the amount reasonably needed to satisfy the Unforeseeable Financial Emergency, plus amounts necessary to pay Federal, state, or local income taxes or penalties reasonably anticipated as a result of the distribution.

(c) If the Committee, in its sole discretion, approves a Participant's petition for suspension, the Participant's deferrals under this Plan shall be suspended as of the date of such approval. If the Committee, in its sole discretion, approves a Participant's petition for suspension and payout, the Participant's deferrals under this Plan shall be suspended as of the date of such approval and the Participant's Benefit Distribution Date for such payout shall be the date on which such Committee approval occurs, and such payout shall be distributed to the Participant in a lump sum no later than 60 days after such Benefit Distribution Date.

# ARTICLE 5 Termination Benefit

5.1 Termination Benefit. A Participant who experiences a Separation from Service shall receive, as a Termination Benefit, his or her vested Account Balance in either a lump sum payment of cash and/or Shares (if an Annual Equity Grant has been deferred) or annual installment payments of cash and/or Shares (if applicable), as elected by the Participant in accordance with Section 5.2. A Participant's Termination Benefit shall be calculated as of the close of business on or around the applicable Benefit Distribution Date for such benefit. The Benefit Distribution Date shall be (i) the first day after the end of the six-month period immediately following the date on which the Participant experiences a Separation from Service, if the Participant is a Specified Employee, and (ii) for all other Participants, the date on which the Participant experiences a Separation from Service; provided, however, if a Participant changes the form of distribution for the Termination Benefit in accordance with Section 5.2(b), the Benefit Distribution Date for the Termination Benefit shall be determined in accordance with Section 5.2(b). Notwithstanding the foregoing and notwithstanding anything an Election Form to the contrary, if a Participant experiences a Separation from Service prior to the completion of 5 Years of Service with the Company, then such Participant's Termination Benefit shall be paid in a lump sum on the Benefit Distribution Date, without regard to any change in form of distribution made in accordance with Section 5.2(b).

### 5.2 Payment of Termination Benefit.

- (a) A Participant, in connection with his or her commencement of participation in the Plan, shall elect on an Election Form to receive the Termination Benefit upon Separation from Service in a lump sum payment of cash and/or Shares (if an Annual Equity Grant has been deferred) or pursuant to an Annual Installment Method of up to 15 years. If a Participant does not make any election with respect to the payment of the Termination Benefit, then such benefit shall be payable in a lump sum.
- (b) A Participant may change the form of payment for the Termination Benefit by submitting an Election Form to the Committee in accordance with the following criteria:
  - (i) The election shall not take effect until at least 12 months after the date on which the election is made;
  - (ii) The new Benefit Distribution Date for the Participant's Termination Benefit upon Separation from Service shall be 5 years after the Benefit Distribution Date that would otherwise have been applicable to such benefit; and
  - (iii) The election must be made at least 12 months prior to the Benefit Distribution Date that would otherwise have been applicable to the Participant's Termination Benefit upon Separation from Service.

For purposes of applying the provisions of this Section 5.2(b), a Participant's election to change the form of payment for the Termination Benefit upon Separation from Service shall not be considered to be made until the date on which the election becomes irrevocable. Such an election shall become irrevocable no later than the date that is 12 months prior to the Benefit Distribution Date that would otherwise have been applicable to the Participant's Termination Benefit upon Separation from Service. Subject to the requirements of this Section 5.2(b), the Election Form most recently accepted by the Committee that has become effective shall govern the form of payout of the Participant's Termination Benefit.

- (c) The lump sum payment shall be made, or installment payments shall commence, in cash and/or Shares (as applicable), no later than 60 days after the Participant's Benefit Distribution Date. Remaining installments, if any, shall be paid no later than 60 days after each anniversary of the Participant's Benefit Distribution Date.
- (d) Notwithstanding anything to the contrary contained herein, the Company may distribute a Participant's Account Balance if, at any time after such Participant's Separation from Service, the Account Balance does not exceed the limit in Code section 402(g)(1)(B) and such distribution would result in the termination of the Participant's entire interest in the Plan as provided under Code section 409A.

#### **Disability Waiver and Benefit**

- Balance, calculated as of the close of business on or around the Benefit Distribution Date for such benefit. The Benefit Distribution Date shall be the date that the Committee determines that such Participant is Disabled within the meaning of Section 1.18; provided, however, if the Participant experienced a Separation from Service prior to the date he or she was Disabled, then the Benefit Distribution Date shall be determined in accordance with Section 5.1 above, as applicable (and the Participant shall receive a Termination Benefit and not a Disability Benefit). A Participant, in connection with his or her commencement of participation in the Plan, shall elect on an Election Form to receive the Disability Benefit in a lump sum payment of cash and/or Shares (if an Annual Equity Grant has been deferred) or pursuant to an Annual Installment Method of up to 15 years. If a Participant does not make any election with respect to the payment of the Disability Benefit, then such benefit shall be payable in a lump sum. The lump sum payment shall be made, or installment payments shall commence, no later than 60 days after the Disabled Participant's Benefit Distribution Date, and any remaining installments, if any, shall be paid no later than 60 days after each anniversary of the Disabled Participant's Benefit Distribution Date.
- 6.2 <u>Deferral Following Disability</u>. If a Participant returns to service as a Director after a Disability ceases, the Participant may elect to defer an Annual Deferral Amount for the Plan Year following his or her return to service and for every Plan Year thereafter while a Participant in the Plan; provided such deferral elections are otherwise allowed and an Election Form is delivered to and accepted by the Committee for each such election in accordance with Section 3.4(b) above.

# ARTICLE 7 Survivor Benefit

- 7.1 Survivor Benefit. In the event of a Participant's death prior to the complete distribution of his or her vested Account Balance, the Participant's Beneficiary(ies) shall receive a Survivor Benefit which will be equal to the Participant's unpaid vested Account Balance, calculated as of the close of business on or around the Benefit Distribution Date for such benefit. The Benefit Distribution Date shall be the date on which the Committee is provided with proof that is satisfactory to the Committee of the Participant's death.
- 7.2 Payment of Survivor Benefit. The Survivor Benefit shall be paid to the Participant's Beneficiary(ies) in a lump sum payment of cash and/or Shares (if an Annual Equity Grant has been deferred) no later than 60 days after the Participant's Benefit Distribution Date.

# ARTICLE 8 Beneficiary Designation

- 8.1 **Beneficiary**. Each Participant shall have the right, at any time, to designate his or her Beneficiary(ies) (both primary as well as contingent) to receive any benefits payable under the Plan to a beneficiary upon the death of a Participant. The Beneficiary designated under this Plan may be the same as or different from the Beneficiary designation under any other plan of the Company or any Employer in which the Participant participates.
- 8.2 **Beneficiary Designation; Change; Spousal Consent**. A Participant shall designate his or her Beneficiary by completing and signing the Beneficiary Designation Form, and returning it to the Committee or its designated agent. A Participant shall have the right to change a Beneficiary by completing, signing and otherwise complying with the terms of the Beneficiary Designation Form and the Committee's rules and procedures, as in effect from time to time. If the Participant names someone other than his or her spouse as a Beneficiary, spousal consent in writing to the designation of a different Beneficiary is required to be provided in a form designated by the Committee, executed by such Participant's spouse and returned to the Committee. A Beneficiary designation that has received spousal consent cannot be changed without spousal consent. Upon the acceptance by the Committee of a new Beneficiary Designation Form, all Beneficiary designations previously filed shall be canceled. The Committee shall be entitled to rely on the last Beneficiary Designation Form filed by the Participant and accepted by the Committee prior to his or her death.
- 8.3 **Acknowledgment**. No designation or change in designation of a Beneficiary shall be effective until received and acknowledged in writing by the Committee or its designated agent.

- No Beneficiary Designation. If a Participant fails to designate a Beneficiary as provided in Sections 8.1, 8.2 and 8.3 above or, if all designated Beneficiaries predecease the Participant or die prior to complete distribution of the Participant's benefits, then the Participant's designated Beneficiary shall be deemed to be his or her surviving spouse. If the Participant has no surviving spouse, the benefits remaining under the Plan to be paid to a Beneficiary shall be payable to the executor or personal representative of the Participant's estate.
- 8.5 **Doubt as to Beneficiary.** If the Committee has any doubt as to the proper Beneficiary to receive payments pursuant to this Plan, the Committee shall have the right, exercisable in its discretion, to either (i) cause the Company to withhold such payments until this matter is resolved to the Committee's satisfaction or (ii) direct that the amount be paid into any court of competent jurisdiction in an interpleader action, and such payment shall be a full and complete discharge of any liability or obligation under the Plan or Trust Agreement to the full extent of such payment.
- 8.6 **Discharge of Obligations**. The payment of benefits under the Plan to a Beneficiary shall fully and completely discharge the Company (and all Employers) and the Committee from all further obligations under this Plan with respect to the Participant.

### **ARTICLE 9**

### **Termination, Amendment or Modification**

- 9.1 Termination. Although the Company anticipates that it will continue the Plan for an indefinite period of time, there is no guarantee that the Company will continue the Plan or will not terminate the Plan at any time in the future. Accordingly, the Company reserves the right to discontinue its sponsorship of the Plan and the Company reserves the right to terminate the Plan at any time with respect to any or all of its participating Directors, by action of its Board. However, after the Plan termination the Account Balances of such Participants shall continue to be credited with Annual Deferral Amounts attributable to a deferral election that was in effect prior to the Plan termination to the extent deemed necessary to comply with Code section 409A and related Treasury Regulations, and additional amounts shall continue to credited or debited to such Participants' Account Balances pursuant to Section 3.7. The Measurement Funds available to Participants following the termination of the Plan shall be comparable in number and type to those Measurement Funds available to Participants in the Plan Year preceding the Plan Year in which the Plan termination is effective. In addition, following a Plan termination, Participant Account Balances shall remain in the Plan and shall not be distributed until such amounts become eligible for distribution in accordance with the other applicable provisions of the Plan. Notwithstanding the preceding sentence, to the extent permitted by Treas. Reg. §1.409A-3(j) (4)(ix), the Company may provide that upon termination of the Plan, all Account Balances of the Participants shall be distributed, subject to and in accordance with any rules established by the Company deemed necessary to comply with the applicable requirements and limitations of Treas. Reg. §1.409A-3(j)(4)(ix).
- 9.2 Amendment. The Company may, at any time, amend or modify the Plan in whole or in part by the action of its Board; provided, however, that: (i) no amendment or modification shall be effective to decrease or restrict the value of a Participant's Account Balance in existence at the time the amendment or modification is made, and (ii) no amendment or modification of this Section 9.2 or Section 10.3 below shall be effective. Notwithstanding the foregoing provisions of this Section 9.2, by written instrument, the Plan may be amended by the Company at any time prior to a Change in Control to conform the Plan to the provisions and requirements of any applicable law. No such amendment described in the immediately preceding sentence shall be considered prejudicial to any interest of a Participant or a Beneficiary hereunder.
- 9.3 **Effect of Payment**. The full payment of the Participant's vested Account Balance under Articles 4, 5, 6, or 7 of the Plan shall completely discharge all obligations to a Participant and his or her designated Beneficiaries under this Plan shall terminate.

# ARTICLE 10 Administration

10.1 Committee Duties. Except as otherwise provided in this Article 10, this Plan shall be administered by a Committee, which shall consist of the Board of the Company, or such committee as the Board of the Company shall appoint. The Board of the Company may remove or replace any member of a committee appointed by the Board of the Company at any time in its sole discretion. Members of the Committee may be Participants under this Plan. Any individual serving on the Committee who is a Participant shall abstain from any discussion and vote, and shall not otherwise act on any matter relating directly to himself or herself. When making a determination or calculation, the Committee shall be entitled to rely on information furnished by a Participant or the Company. The members of the Committee shall not receive any special compensation for serving in their capacities as members of the Committee, but shall be reimbursed by the Company for any reasonable expenses incurred in connection therewith. No bond or other security need be required of the Committee or any member thereof.

- 10.2 Administration of the Plan. The Committee shall operate, administer, interpret, construe, and construct the Plan, including without limitation, the sole discretion and authority to (i) make, amend, interpret and enforce all appropriate rules and regulations for the administration of this Plan and (ii) decide or resolve any and all questions as may arise in connection with the Plan, including correcting any defect, supplying any omission, or reconciling any inconsistency. The Committee shall have all powers necessary or appropriate to implement and administer the terms and provisions of the Plan, including the power to make findings of fact.
- 10.3 Administration Upon Change In Control. For purposes of this Plan, the Committee shall be the "Administrator" at all times prior to the occurrence of a Change in Control Within 120 days following a Change in Control, an independent third party "Administrator" may be selected by the individual who, immediately prior to the Change in Control, was the Company's Chairman of the Board or, if not so identified, the Chairman of the Board's Nominating and Governance Committee (the "Ex-Director"), and approved by the Trustee. The Committee, as constituted prior to the Change in Control, shall continue to be the Administrator until the earlier of (i) the date on which such independent third party is selected and approved, or (ii) the expiration of the 120-day period following the Change in Control. If an independent third party is not selected within 120 days of such Change in Control, the Committee, as described in Section 10.1 above, shall be the Administrator. The Administrator shall have the discretionary power to determine all questions arising in connection with the administration of the Plan and the interpretation of the Plan and Trust including, but not limited to benefit entitlement determinations; provided, however, upon and after the occurrence of a Change in Control, the Administrator shall have no power to direct the investment of Plan or Trust assets or select any investment manager or custodial firm for the Plan or Trust. Upon and after the occurrence of a Change in Control, the Company must: (1) pay all reasonable administrative expenses and fees of the Administrator; (2) indemnify the Administrator against any costs, expenses and liabilities including, without limitation, attorney's fees and expenses arising in connection with the performance of the Administrator hereunder, except with respect to matters resulting from the gross negligence or willful misconduct of the Administrator or its employees or agents; and (3) supply full and timely information to the Administrator on all matters relating to the Plan, the Trust, the Participants and their Beneficiaries, the Account Balances of the Participants, the date and circumstances of the Disability, death or Separation from Service of the Participants, and such other pertinent information as the Administrator may reasonably require. Upon and after a Change in Control, the Administrator may be terminated (and a replacement appointed) by the Trustee only with the approval of the Ex-Director. Upon and after a Change in Control, the Administrator may not be terminated by the Company.
- 10.4 **Agents**. In the administration of this Plan, the Committee may, from time to time, employ agents and delegate to them such administrative duties as it sees fit (including acting through a duly appointed representative).
- 10.5 **Binding Effect of Decisions**. The determination of the Committee as to the proper interpretation, construction, or application of any term or provision in the Plan and the rules and regulations promulgated hereunder shall be final, binding, and conclusive with respect to all interested parties.
- 10.6 <u>Indemnity of Committee</u>. To the full extent permitted by applicable law, the Company and each Employer shall individually and collectively indemnify and hold harmless each past, present, and future member of the Committee, any Employee to whom the duties of the Committee may be delegated, and the Administrator against any and all claims, losses, damages, judgments, settlements, costs, expenses or liabilities (and all actions in respect thereof and any legal or other costs and expenses in giving testimony or furnishing documents in response to a subpoena or otherwise) arising from any action or failure to act with respect to this Plan, except in the case of gross negligence or willful misconduct by the Committee, any of its members, any such Employee or the Administrator, including the cost of investigating, preparing, or defending any pending, threatened or anticipated possible action, claim, suit or other proceeding, whether or not in connection with litigation in which the Committee, such Employee, or the Administrator is a party. The foregoing right of indemnification shall inure to the benefit of the successors and assigns, and the heirs, executors, administrators and personal representatives of the Committee, any such Employee, and the Administrator, and shall be in addition to all other rights to which the Committee, any such Employee, and the Administrator may be entitled as a matter of law, contract, or otherwise. The amount of indemnification during a calendar year shall not affect the fees and expenses eligible for reimbursement in any other calendar year. Indemnification shall be made on or before the last day of the calendar year following the calendar year in which the amounts or liabilities subject to such indemnification were incurred.
- 10.7 <u>Company Information</u>. To enable the Committee and/or Administrator to perform its functions, the Company shall supply full and timely information to the Committee and/or Administrator, as the case may be, on all matters relating to the compensation of its Participants, the date and circumstances of the Disability, death or Separation from Service of its Participants, and such other pertinent information as the Committee or Administrator may reasonably require.

10.8 **Reliance Upon Information**. No member of the Committee shall be liable for any decision, action, omission, or mistake in judgment in connection with the administration of the Plan, provided that he acted in good faith. Without limiting the generality of the foregoing sentence, any decision or action taken by the Committee in reasonable reliance upon information supplied to it by the Board, the Company, any Employer, legal counsel, or the Company's independent accountants shall be deemed to have been taken in good faith. The Committee may from time to time consult with counsel who may be counsel to the Company or other counsel, with respect to its obligations or duties hereunder, or with respect to any action, proceeding or question of law, and shall not be held liable with respect to any action taken or omitted, in good faith, pursuant to the advice of such counsel.

# ARTICLE 11 Other Benefits and Agreements

11.1 Coordination with Other Benefits. The benefits provided for a Participant and a Participant's Beneficiary under the Plan are in addition to any other benefits available to such Participant under any other plan or program for Directors of the Company. The Plan shall supplement and shall not supersede, modify or amend any other such plan or program except as may otherwise be expressly provided.

# **ARTICLE 12 Claims Procedures**

- 12.1 Presentation of Claim. Any Participant or Beneficiary of a deceased Participant (such Participant or Beneficiary being referred to below as a "Claimant") may deliver to the Committee a written claim for a determination with respect to the amounts distributable to such Claimant from the Plan. If such a claim relates to the contents of a notice received by the Claimant, the claim must be made within 60 days after such notice was received by the Claimant. All other claims must be made within 180 days of the date on which the event that caused the claim to arise occurred. The claim must state with particularity the determination desired by the Claimant.
- 12.2 Notification of Decision. The Committee shall consider a Claimant's claim within a reasonable time, but no later than 90 days after receiving the claim. If the Committee determines that special circumstances require an extension of time for processing the claim, written notice of the extension shall be furnished to the Claimant prior to the termination of the initial 90-day period. In no event shall such extension exceed a period of 90 days from the end of the initial period. The extension notice shall indicate the special circumstances requiring an extension of time and the date by which the Committee expects to render the benefit determination. The Committee shall notify the Claimant in writing:
  - (a) that the Claimant's requested determination has been made, and that the claim has been allowed in full; or
  - (b) that the Committee has reached a conclusion contrary, in whole or in part, to the Claimant's requested determination, and such notice must set forth in a manner calculated to be understood by the Claimant:
    - (i) the specific reason(s) for the denial of the claim, or any part of it;
    - (ii) specific reference(s) to pertinent provisions of the Plan upon which such denial was based;
    - (iii) a description of any additional material or information necessary for the Claimant to perfect the claim, and an explanation of why such material or information is necessary;
    - (iv) an explanation of the claim review procedure set forth in Section 12.3 below; and
    - (v) a statement of the Claimant's right to bring a civil action following an adverse benefit determination on review.
- 12.3 **Review of a Denied Claim.** On or before 60 days after receiving a notice from the Committee that a claim has been denied, in whole or in part, a Claimant (or the Claimant's duly authorized representative), if he or she disagrees with the claim denial, must file with the Committee a written request for a review of the denial of the claim. The Claimant (or the Claimant's duly authorized representative):
  - (a) may, upon request and free of charge, have reasonable access to, and copies of, all documents, records and other information relevant to the claim for benefits;
  - (b) may submit written comments or other documents; and/or
  - (c) may request a hearing, which the Committee, in its sole discretion, may grant.

- 12.4 <u>Decision on Review</u>. The Committee shall render its decision on review promptly, and no later than 60 days after the Committee receives the Claimant's written request for a review of the denial of the claim. If the Committee determines that special circumstances require an extension of time for processing the claim, written notice of the extension shall be furnished to the Claimant prior to the termination of the initial 60-day period. In no event shall such extension exceed a period of 60 days from the end of the initial period. The extension notice shall indicate the special circumstances requiring an extension of time and the date by which the Committee expects to render the benefit determination. In rendering its decision, the Committee shall take into account all comments, documents, records and other information submitted by the Claimant relating to the claim, without regard to whether such information was submitted or considered in the initial benefit determination. The decision must be written in a manner calculated to be understood by the Claimant, and it must contain:
  - (a) specific reasons for the decision;
  - (b) specific reference(s) to the pertinent Plan provisions upon which the decision was based;
  - (c) a statement that the Claimant is entitled to receive, upon request and free of charge, reasonable access to and copies of, all documents, records and other information relevant to the Claimant's claim for benefits; and
  - (d) a statement of the Claimant's right to bring a civil action.
- 12.5 <u>Legal Action</u>. A Claimant's compliance with the foregoing provisions of this Article 12 is a mandatory prerequisite to a Claimant's right to commence any legal action with respect to any claim for benefits under this Plan.

# ARTICLE 13 <u>Trust</u>

- 13.1 **Establishment of the Trust**. In order to provide assets from which to fulfill the obligations of the Participants and their beneficiaries under the Plan, the Company intends, pursuant to formal action by the Committee, to establish a trust by a trust agreement with a third party, the Trustee, to which the Company may, in its discretion, contribute cash or other property, including securities issued by the Company, to provide for the benefit payments under the Plan; provided, however, in the event of a Change in Control the Company must contribute sufficient assets to any such Trust equal to the aggregate value of all Account Balances determined as of the date of the Change in Control.
- 13.2 <u>Interrelationship of the Plan and the Trust</u>. The provisions of the Plan shall govern the rights of a Participant to receive distributions pursuant to the Plan. The provisions of the Trust shall govern the rights of the Company, Participants and the creditors of the Company to the assets transferred to the Trust. The Company shall at all times remain liable to carry out its obligations under the Plan.
- 13.3 **Distributions From the Trust**. The Company's obligations under the Plan may be satisfied with Trust assets distributed pursuant to the terms of the Trust, and any such distribution shall reduce the Company's obligations under this Plan.
- 13.4 Participants' Rights under the Trust. The assets of the Trust shall be held for the benefit of the Participants in accordance with the terms of the Plan and Trust Agreement. All the assets of the Trust shall remain subject to the claims of the general unsecured creditors of the Company, and the rights of the Participants to the assets of the Trust shall be limited as provided in the Plan and Trust Agreement in the event that the Company becomes Insolvent.

# ARTICLE 14 Miscellaneous

- 14.1 <u>Status of Plan</u>. The Plan is intended to be a plan that is not qualified within the meaning of Code section 401(a) and is exempt from the requirements of Title I of the Employee Retirement Income Security Act of 1974, as amended. The Plan shall be administered and interpreted (a) to the extent possible in a manner consistent with the intent described in the preceding sentence, and (b) in accordance with Code section 409A the Final Regulations thereunder and other related Treasury guidance.
- 14.2 **Limitation of Rights**. Nothing in this Plan shall be construed to:
  - (a) Give a Participant any rights, other than as an unsecured general creditor of the Company with respect to any amount credited to his or her Account, until such amount is actually distributed to him or her;
  - (b) Limit in any way the right of the Company to terminate a Participant's service;
  - (c) Give a Participant or any other person any interest in any fund, reserve or any specific asset of the Company or any entity under common ownership or control with the Company; or

- (d) Create a fiduciary relationship between the Participant and the Company.
- 14.3 Nonalienation of Benefits. No right or benefit under this Plan shall be subject to anticipation, alienation, attachment, garnishment, sale, transfer, assignment (either at law or in equity), levy, execution, pledge, encumbrance, charge, or any other legal or equitable process, and any attempt to do so will be void and without effect. No right or benefit hereunder shall in any manner be liable for or subject to any debts, contracts, liabilities, engagements, or torts of the Participant or Beneficiary entitled to such benefits. The recovery under the Plan of overpayments of benefits previously made to a Participant; the transfer of benefit rights from the Plan to another plan; or the direct deposit of benefit payments to an account in a banking institution (if not actually part of an arrangement constituting an assignment or alienation), shall not be construed as an assignment or alienation for purposes of the immediately preceding paragraph. The first sentence of this Section 14.3 shall not preclude (a) the Participant from designating a Beneficiary to receive any benefit payable hereunder upon his or her death, or (b) the executors, administrators, or other legal representatives of the Participant or his or her estate from assigning any rights hereunder to the person or persons entitled thereto. In the event that any Participant's or Beneficiary's benefits hereunder are garnished or attached by order of any court, the Company or the Trustee may bring an action or a declaratory judgment in a court of competent jurisdiction to determine the proper recipient of the benefits to be paid under the Plan. During the pendency of such action, any benefits that become payable shall be held as credits to the Participant's Account or, if the Company prefers, paid into the court as they become payable, to be distributed by the court to the recipient as the court deems proper at the close of such action.
- 14.4 <u>Unsecured General Creditor</u>. Participants and their Beneficiaries, heirs, successors and assigns shall have no legal or equitable rights, interests or claims in any property or assets of the Company. For purposes of the payment of benefits under this Plan, any and all of the Company's assets shall be, and remain, the general, unpledged unrestricted assets of the Company. The Company's obligation under the Plan shall be merely that of an unfunded and unsecured promise to pay money in the future.
- 14.5 <u>Waiver</u>. No term or condition of this Plan shall be deemed to have been waived, nor shall there be an estoppel against the enforcement of any provision of this Plan, except by written instrument of the party charged with such waiver or estoppel. No such written waiver shall be deemed a continuing waiver unless specifically stated therein, and each such waiver shall operate only as to the specific term or condition waived and shall not constitute a waiver of such term or condition for the future or as to any act other than that specifically waived. Any waiver by any party hereto of a breach of any provision of the Plan by any other party shall not operate or be construed as a waiver by such party of any subsequent breach thereof.
- 14.6 <u>Company's Liability</u>. The Company's liability for the payment of benefits shall be defined only by the Plan. The Company shall have no obligation to a Participant under the Plan except as expressly provided in the Plan.
- 14.7 Nonassignability. Neither a Participant nor any other person shall have any right to commute, sell, assign, transfer, pledge, anticipate, mortgage or otherwise encumber, transfer, hypothecate, alienate or convey in advance of actual receipt, the amounts, if any, payable hereunder, or any part thereof, which are, and all rights to which are expressly declared to be, unassignable and non-transferable. No part of the amounts payable shall, prior to actual payment, be subject to seizure, attachment, garnishment or sequestration for the payment of any debts, judgments, alimony or separate maintenance owed by a Participant or any other person, be transferable by operation of law in the event of a Participant's or any other person's bankruptcy or insolvency or be transferable to a spouse as a result of a property settlement or otherwise.
- 14.8 **Not a Contract of Service**. The terms and conditions of this Plan shall not be deemed to constitute a contract of service between the Company and the Participant. Such service relationship can be terminated at any time for any reason, or no reason, with or without cause, and with or without notice, unless expressly provided in a written agreement. Nothing in this Plan shall be deemed to give a Participant the right to be retained in the service of the Company, or to interfere with the right of the Company to discipline or discharge the Participant at any time.
- 14.9 **Furnishing Information**. A Participant or his or her Beneficiary will cooperate with the Committee by furnishing any and all information requested by the Committee and take such other actions as may be requested in order to facilitate the administration of the Plan and the payments of benefits hereunder, including but not limited to taking such physical examinations as the Committee may deem necessary.
- 14.10 <u>Terms</u>. Whenever any words are used herein in the masculine, they shall be construed as though they were in the feminine in all cases where they would so apply; and whenever any words are used herein in the singular or in the plural, they shall be construed as though they were used in the plural or the singular, as the case may be, in all cases where they would so apply.
- 14.11 <u>Captions</u>. Headings and paragraphs of this Plan are for convenience only and shall not control or affect the meaning or construction of any of its provisions.

- 14.12 Governing Law. The provisions of this Plan shall be governed by, construed, and enforced in accordance with the laws of the State of Texas (excluding any conflict of laws, rule or principle of Texas law that might refer the governance, construction, or interpretation of this Plan to the laws of another state). A Participant's sole remedy for any Claim shall be against the Company, and no Participant shall have any claim or right of any nature against any Related Company of the Company or any stockholder or existing or former director, officer or Employee of the Company or any Related Company of the Company. The individuals and entities described above in this Section 14.12 (other than the Company) shall be third-party beneficiaries of this Plan for purposes of enforcing the terms of this Section 14.12.
- 14.13 **Notice**. Any notice or filing required or permitted to be given to the Committee under this Plan shall be sufficient if in writing and hand-delivered, or sent by registered or certified mail, to the address below:

Blucora, Inc. Attn: General Counsel 6333 North State Highway 161, 4th Floor Irving, Texas 75038

Such notice shall be deemed given as of the date of delivery or, if delivery is made by mail, as of the date shown on the postmark on the receipt for registration or certification.

Any notice or filing required or permitted to be given to a Participant under this Plan shall be sufficient if in writing and hand-delivered, or sent by mail, to the last known address of the Participant.

- 14.14 **Successors**. The provisions of this Plan shall bind and inure to the benefit of the Company and its successors and assigns and the Participant and the Participant's designated Beneficiaries.
- 14.15 **Spouse's Interest**. The interest in the benefits hereunder of a spouse of a Participant who has predeceased the Participant shall automatically pass to the Participant and shall not be transferable by such spouse in any manner, including but not limited to such spouse's will, nor shall such interest pass under the laws of interstate succession.
- 14.16 <u>Validity</u>. In case any provision of this Plan shall be illegal or invalid for any reason, said illegality or invalidity shall not affect the remaining parts hereof, but this Plan shall be construed and enforced as if such illegal or invalid provision had never been inserted herein.
- 14.17 <u>Incompetent</u>. If the Committee determines in its discretion that a benefit under this Plan is to be paid to a minor, a person declared incompetent or to a person incapable of handling the disposition of that person's property, the Committee may direct payment of such benefit to the guardian, legal representative or person having the care and custody of such minor, incompetent or incapable person. The Committee may require proof of minority, incompetence, incapacity or guardianship, as it may deem appropriate prior to distribution of the benefit. Any payment of a benefit shall be a payment for the account of the Participant and the Participant's Beneficiary, as the case may be, and shall be a complete discharge of any liability under the Plan for such payment amount.
- 14.18 <u>Court Order</u>. The Committee is authorized to make any payments directed by court order in any action in which the Plan or the Committee has been named as a party. In addition, if a court determines that a spouse or former spouse of a Participant has an interest in the Participant's benefits under the Plan in connection with a property settlement or otherwise, the Committee, in its sole discretion, shall have the right, notwithstanding any election made by a Participant, to immediately distribute the spouse's or former spouse's interest in the Participant's benefits under the Plan to that spouse or former spouse.

### 14.19 Distribution in the Event of Income Inclusion Under Code Section 409A or Other Taxation.

- (a) <u>Code Section 409A</u>. If any portion of a Participant's Account Balance under this Plan is required to be included in income by the Participant prior to receipt due to a failure of this Plan to comply with the requirements of Code section 409A and related Treasury Regulations, the Committee may determine that such Participant shall receive a distribution from the Plan in an amount equal to the lesser of (i) the portion of his or her Account Balance required to be included in income as a result of the failure of the Plan to comply with the requirements of Code section 409A and related Treasury Regulations, or (ii) the unpaid Account Balance.
- (b) <u>Trust</u>. If the Trust terminates in accordance with its terms and benefits are distributed from the Trust to a Participant in accordance therewith, the Participant's benefits under this Plan shall be reduced to the extent of such distributions; provided, however, that such distributions may not occur, if they otherwise would violate Code section 409A and the related Treasury Regulations.

- 14.20 <u>Insurance</u>. The Company, on its own behalf or on behalf of the Trustee of the Trust, and, in their sole discretion, may apply for and procure insurance on the life of the Participant, in such amounts and in such forms as the Trust may choose. The Company or the Trustee of the Trust, as the case may be, shall be the sole owner and beneficiary of any such insurance. The Participant shall have no interest whatsoever in any such policy or policies, and at the request of the Company shall submit to medical examinations and supply such information and execute such documents as may be required by the insurance company or companies to whom the Company has applied for insurance.
- 14.21 Legal Fees To Enforce Rights After Change in Control. The Company is aware that upon the occurrence of a Change in Control, the Board of the Company (which might then be composed of new members) or a stockholder of the Company, or of any Successor Company might then cause or attempt to cause the Company or such successor to refuse to comply with its obligations under the Plan and might cause or attempt to cause the Company to institute, or may institute, litigation seeking to deny Participants the benefits intended under the Plan. In these circumstances, the purpose of the Plan could be frustrated. Accordingly, if, following a Change in Control, it should appear to any Participant that the Company or any Successor Company has failed to comply with any of its obligations under the Plan or any agreement thereunder or, if the Company or any other person takes any action to declare the Plan void or unenforceable or institutes any litigation or other legal action designed to deny, diminish or to recover from any Participant the benefits intended to be provided, then the Company irrevocably authorizes such Participant to retain counsel of his or her choice at the expense of the Company to represent such Participant in connection with the initiation or defense of any litigation or other legal action, whether by or against the Company, or any director, officer, stockholder or other person affiliated with the Company or any successor thereto in any jurisdiction. The amount of fees and expenses eligible for reimbursement during a calendar year shall not affect the fees and expenses eligible for reimbursement of eligible fees and expenses shall be made on or before the last day of the calendar year following the calendar year in which the fees or expenses were incurred.

\* \* \* \* \* \* \* \*

[Remainder of Page Intentionally Left Blank. Signature Page Follows.]

| IN WITNESS      | WHEREOF, the | Company l | has signed t | his Plan | document a | s of Decem | ber 13, | 2018, | to be | effective | as o | of the |
|-----------------|--------------|-----------|--------------|----------|------------|------------|---------|-------|-------|-----------|------|--------|
| Effective Date. |              |           |              |          |            |            |         |       |       |           |      |        |

# Company

Blucora, Inc.,

a Delaware corporation

By:/s/ John S. Clendening

Name: John S. Clendening

Title: President and CEO

4845-6968-8192 v.3

Signature Page to the Blucora Director Tax-Smart Deferral Plan

| Ex   | hi | hit | 10 | 1.52 |
|------|----|-----|----|------|
| 11/X |    | .,  |    | 172  |

## **Blucora Executive Tax-Smart Deferral Plan**

Effective January 1, 2019

--

### TABLE OF CONTENTS

**Page** 

**ARTICLE 1DEFINITIONS** 1 ARTICLE 2SELECTION, ENROLLMENT, AND ELIGIBILITY 9 2.1 Selection by Committee 9 2.2 Enrollment Requirements 9 2.3 Eligibility; Commencement of Participation 9 2.4 Termination of Participation and/or Deferrals 9 ARTICLE 3DEFERRAL COMMITMENTS/COMPANY CONTRIBUTION AMOUNTS/VESTING/CREDITING/TAXES 10 3.1 Minimum Deferrals 10 3.2 Maximum Deferral 10 3.3 Election to Defer; Effect of Election Form 10 3.4 Withholding and Crediting of Annual Deferral Amounts 11 3.5 Company Contribution Amount 11 3.6 **Vesting** 12 3.7 Crediting/Debiting of Account Balances 12 3.8 FICA and Other Taxes 13 ARTICLE 4IN-SERVICE SCHEDULED DISTRIBUTION; UNFORESEEABLE FINANCIAL EMERGENCIES; WITHDRAWAL ELECTION 14 4.1 In-Service Scheduled Distribution 14 4.2 Other Benefits Take Precedence Over In-Service Scheduled Distributions 15 4.3 Withdrawal Payout/Suspensions for Unforeseeable Financial Emergencies 15 **ARTICLE 5RETIREMENT BENEFIT** 15 5.1 Retirement Benefit 15 5.2 Payment of Retirement Benefit 16 **ARTICLE 6TERMINATION BENEFIT** 17 6.1 **Termination Benefit** 17

| 6.2 Payment of Termination Benefit 17                   |
|---|
| ARTICLE 7DISABILITY WAIVER AND BENEFIT 17               |
| 7.1 Disability Benefit 17                               |
| 7.2 Deferral Following Disability 17                    |
| ARTICLE 8SURVIVOR BENEFIT 18                            |
| 8.1 Survivor Benefit 18                                 |
| 8.2 Payment of Survivor Benefit 18                      |
| ARTICLE 9BENEFICIARY DESIGNATION 18                     |
| 9.1 Beneficiary 18                                      |
| 9.2 Beneficiary Designation; Change; Spousal Consent 18 |
| 9.3 Acknowledgment 18                                   |
| 9.4 No Beneficiary Designation 18                       |
| 9.5 Doubt as to Beneficiary 19                          |
| 9.6 Discharge of Obligations 19                         |
| ARTICLE 10LEAVE OF ABSENCE 19                           |
| 10.1 Paid Leave of Absence 19                           |
| 10.2 Unpaid Leave of Absence 19                         |
| ARTICLE 11TERMINATION, AMENDMENT OR MODIFICATION 19     |
| 11.1 Termination 19                                     |
| 11.2 <b>Amendment</b> 20                                |
| 11.3 Effect of Payment 20                               |
| ARTICLE 12ADMINISTRATION 20                             |
| 12.1 Committee Duties 20                                |
| 12.2 Administration of the Plan 21                      |
| 12.3 Administration Upon Change In Control 21           |
| <u>12.4 Agents</u> 22                                   |
| 12.5 Binding Effect of Decisions 22                     |
| 12.6 Indemnity of Committee 22                          |
| 12.7 Employer Information 22                            |
| 12.8 Reliance Upon Information 22                       |
| ARTICLE 130THER BENEFITS AND AGREEMENTS 23              |

| 13.1 Coordination with Other Benefits 23            |
|---|
| ARTICLE 14CLAIMS PROCEDURES 23                      |
| 14.1 Presentation of Claim 23                       |
| 14.2 Notification of Decision 23                    |
| 14.3 Review of a Denied Claim 24                    |
| 14.4 Decision on Review 24                          |
| 14.5 Legal Action 25                                |
| ARTICLE 15TRUST 25                                  |
| 15.1 Establishment of the Trust 25                  |
| 15.2 Interrelationship of the Plan and the Trust 25 |
| 15.3 Distributions From the Trust 25                |
| 15.4 Participants' Rights under the Trust 25        |
| ARTICLE 16MISCELLANEOUS 25                          |
| <u>16.1</u> Status of Plan 25                       |
| 16.2 Limitation of Rights 26                        |
| 16.3 Nonalienation of Benefits 26                   |
| 16.4 Unsecured General Creditor 26                  |
| <u>16.5</u> <b>Waiver</b> 27                        |
| 16.6 Employer's Liability 27                        |
| 16.7 Nonassignability 27                            |
| 16.8 Not a Contract of Employment 27                |
| 16.9 Furnishing Information 27                      |
| <u>16.10</u> <b>Terms</b> 27                        |
| <u>16.11</u> <u>Captions</u> 27                     |
| 16.12 Governing Law 28                              |
| <u>16.13</u> Notice 28                              |
| <u>16.14</u> <u>Successors</u> 28                   |
| 16.15 Spouse's Interest 28                          |
| <u>16.16 Validity</u> 28                            |
| <u>16.17</u> <b>Incompetent</b> 28                  |

16.18 **Court Order** 29

16.19 Distribution in the Event of Income Inclusion Under Code Section 409A or Other Taxation 29

16.20 **Insurance** 29

16.21 Legal Fees To Enforce Rights After Change in Control 29

16.22 Formal Action by an Employer 30

### BLUCORA TAX-SMART EXECUTIVE DEFERRAL PLAN

Effective January 1, 2019

Blucora, Inc., a Delaware corporation (the "Company") and each other Employer, desires to adopt this Blucora Tax-Smart Executive Deferral Plan (the "Plan"), effective as of January 1, 2019 (the "Effective Date"), to provide benefits in the form of unfunded deferred compensation to a select group of management and highly compensated Employees who contribute materially to the continued growth, development and business success of the Company and the Employers.

### **Purpose**

The purpose of the Plan is to provide specified benefits to a select group of management or highly compensated Employees who contribute materially to the continued growth, development and future business success of the Company and its subsidiaries, if any, that are participating employers in this Plan. The provisions of this Plan shall be interpreted in accordance with the requirements of Code section 409A and the Final Treasury Regulations thereunder, it being the intent of the parties that the Plan shall be in compliance with the requirements of said Code section and said Regulations.

# ARTICLE 1 Definitions

For the purposes of this Plan, unless otherwise clearly apparent from the context, the following phrases or terms shall have the following indicated meanings:

- 1.1 "Account Balance" shall mean, with respect to a Participant, a credit on the records of the Employer equal to the sum of (i) the Deferral Account balance, and (ii) the Company Contribution Account balance. The Account Balance, and each other specified account balance, shall be a bookkeeping entry only and shall be utilized solely as a device for the measurement and determination of the amounts to be paid to a Participant, or his or her designated Beneficiary, pursuant to this Plan.
- 1.2 "Annual Bonus" shall mean any compensation, in addition to Incentive Bonus and Base Salary, payable to a Participant during a Plan Year, under any Employer's annual bonus and cash incentive plans. The Administrator, in its sole discretion, shall determine whether any particular type or item of compensation shall be deemed an "Annual Bonus" for purposes of the Plan.
- 1.3 "Annual Deferral Amount" shall mean that portion of a Participant's Base Salary, Annual Bonus, and Incentive Bonus that a Participant defers in accordance with Article 3 for any one Plan Year. In the event of a Participant's Disability, death or a Separation from Service prior to the end of a Plan Year, such year's Annual Deferral Amount shall be the actual amount withheld prior to such event.

- "Annual Installment Method" shall be an annual installment payment over the number of years selected by the Participant in accordance with this Plan, calculated as follows: (i) for the first annual installment, the Participant's vested Account Balance shall be calculated as of the close of business on or around the date on which the Participant's Benefit Distribution Date, and (ii) for remaining annual installments, the Participant's vested Account Balance shall be calculated on every applicable anniversary of such Benefit Distribution Date. Each annual installment shall be calculated by multiplying this balance by a fraction, the numerator of which is one and the denominator of which is the remaining number of annual payments due the Participant. By way of example, if the Participant elects a ten (10) year Annual Installment Method, the first payment shall be 1/10 of the vested Account Balance, calculated as described in this definition. The following year, the payment shall be 1/9 of the vested Account Balance, calculated as described in this definition. For purposes of this Plan, the right to receive a benefit payment in annual installments shall be treated as the entitlement to a single payment.
- 1.5 "Base Salary" shall mean the annual cash compensation relating to services performed during any calendar year as reportable on the Employee's form W-2, excluding distributions from nonqualified deferred compensation plans, bonuses, commissions, overtime, fringe benefits, stock options, relocation expenses, incentive payments, non-monetary awards, director fees and other fees, and automobile and other allowances paid to a Participant for employment services rendered (whether or not such allowances are included in the Employee's gross income). Base Salary shall be calculated before reduction for payroll taxes and for compensation voluntarily deferred or contributed by the Participant pursuant to all qualified or nonqualified plans of any Employer and shall be calculated to include amounts not otherwise included in the Participant's gross income under Code sections 125, 132(f)(4), 402(e)(3), 402(h), or 403(b) pursuant to plans established by any Employer; provided, however, that all such amounts will be included in compensation only to the extent that had there been no such plan, the amount would have been payable in cash to the Employee.
- 1.6 "Beneficiary" shall mean one or more persons, trusts, estates or other entities, designated in accordance with Article 9, that are entitled to receive benefits under this Plan upon the death of a Participant.
- 1.7 **"Beneficiary Designation Form"** shall mean the form established from time to time by the Committee that a Participant completes signs and returns to the Committee to designate one or more Beneficiaries.
- 1.8 **"Benefit Distribution Date"** shall mean the date upon which all or an objectively determinable portion of a Participant's vested benefits will become eligible for distribution.
- 1.9 **"Board"** shall mean the board of directors of the Company or any Employer, or such other person or group of persons referred to in Section 16.23 hereof, in the case of the Company, or any other Employer, which is not a corporation.
- 1.10 "Change in Control" shall mean the occurrence of any of the following events:
  - an acquisition by any Entity of beneficial ownership (within the meaning of Rule 13d-3 promulgated under the Exchange Act) of forty percent (40%) or more of either (1) the number of then outstanding shares of common stock of the Company (the "Outstanding Company Common Stock") or (2) the combined voting power of the then outstanding voting securities of the Company entitled to vote generally in the election of directors (the "Outstanding Company Voting Securities"), provided, however, that the following acquisitions shall not constitute a Change in Control: (i) any acquisition directly from the Company, other than an acquisition by virtue of the exercise of a conversion privilege where the security being so converted was not acquired directly from the Company by the party exercising the conversion privilege, (ii) any acquisition by the Company, (iii) any acquisition by any employee benefit plan (or related trust) sponsored or maintained by the Company or any Related Company, or (iv) an acquisition by any Entity pursuant to a transaction that meets the conditions of clauses (A), (B) and (C) set forth in Section 1.10(c) below;
  - (b) a change in the composition of the Board during any two-year period such that the individuals who, as of the beginning of such two-year period, constitute the Board (the "Incumbent Board") cease for any reason to constitute at least a majority of the Board; provided, however, that for purposes of this definition, any individual who becomes a member of the Board subsequent to the beginning of the two-year period, whose election, or nomination for election by the Company's stockholders, was approved by a vote of at least a majority of those individuals who are members of the Board and who were also members of the Incumbent Board (or deemed to be such pursuant to this proviso) shall be considered as though such individual were a member of the Incumbent Board; and provided further, however, that any such individual whose initial assumption of office occurs as a result of or in connection with an actual or threatened election contest with respect to the election or removal of directors or other actual or threatened solicitation of proxies or consents by or on behalf of an Entity other than the Board shall not be considered a member of the Incumbent Board; or

- (c) the consummation of (i) a merger or consolidation of the Company with or into any other company; (ii) a sale in one transaction or a series of transactions undertaken with a common purpose of all of the Company's outstanding voting securities; or (iii) a sale, lease, exchange or other transfer in one transaction or a series of related transactions undertaken with a common purpose of all or substantially all of the Company's assets; excluding, however, in each case, a transaction pursuant to which:
  - (A) the Entities who are the beneficial owners of the Outstanding Company Common Stock and Outstanding Company Voting Securities immediately prior to such transaction will beneficially own, directly or indirectly, at least fifty percent (50%) of the outstanding shares of common stock, and the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors, of the Successor Company in substantially the same proportions as their ownership, immediately prior to such transaction, of the Outstanding Company Common Stock and Outstanding Company Voting Securities;
  - (B) no Entity (other than the Company, any employee benefit plan (or related trust) of the Company, a Related Company or a Successor Company) will beneficially own, directly or indirectly, forty percent (40%) or more of, respectively, the outstanding shares of common stock of the Successor Company or the combined voting power of the outstanding voting securities of the Successor Company entitled to vote generally in the election of directors unless such ownership resulted solely from ownership of securities of the Company prior to the transaction; and
  - (C) individuals who were members of the Incumbent Board will immediately after the consummation of the transaction constitute at least a majority of the members of the board of directors of the Successor Company.

Where a series of transactions undertaken with a common purpose is deemed to be a transaction described in this Section 1.10(c), the effective date of such transaction shall be the date on which the last of such transactions is consummated.

Notwithstanding the foregoing provisions of this "Change in Control" definition, to the extent necessary to comply with Section 409A of the Code, an event shall not constitute a "Change in Control" for purposes of the Plan, unless such event also constitutes a change in the Company's ownership, its effective control or the ownership of a substantial portion of its assets within the meaning of Section 409A of the Code.

- 1.11 "Claim" shall mean any claim, liability or obligation of any nature, arising out of or relating to this Plan or an alleged breach of this Plan or Election Form.
- 1.12 "Claimant" shall have the meaning set forth in Section 14.1 below.
- 1.13 "Code" shall mean the Internal Revenue Code of 1986, as it may be amended from time to time, and the regulations and other authority issued thereunder by the appropriate governmental authority. References herein to any Code section shall include references to any successor Code section or provision.
- 1.14 "Committee" shall mean the committee described in Article 12.
- 1.15 "Company" shall mean Blucora, Inc., a Delaware corporation, and any successor to all or substantially all of the Company's assets or business.
- 1.16 "Company Contribution Account" shall mean (i) the sum of the Participant's Company Contribution Amounts, plus (ii) amounts credited or debited to the Participant's Company Contribution Account in accordance with this Plan, less (iii) all distributions made to the Participant or his or her Beneficiary pursuant to this Plan that relate to the Participant's Company Contribution Account.
- 1.17 "Company Contribution Amount" shall mean, for any one Plan Year, the amount determined in accordance with Section 3.5 below.
- 1.18 "Deferral Account" shall mean (i) the sum of all of a Participant's Annual Deferral Amounts, plus (ii) amounts credited or debited to the Participant's Deferral Account in accordance with this Plan, less (iii) all distributions made to the Participant or his or her Beneficiary pursuant to this Plan that relate to his or her Deferral Account.
- 1.19 "Disability" or "Disabled" shall mean that a Participant is determined to be disabled in accordance with the applicable disability insurance program of such Participant's Employer, provided that the definition of "disability" applied under such disability insurance program complies with the requirements of Code section 409A and provided, further, that if such Participant is not eligible his or her Employer's disability insurance program, that the Participant is determined to be disabled by the Social Security Administration.

- 1.20 "**Disability Benefit**" shall mean the benefit set forth in Article 7.
- 1.21 "Effective Date" shall mean January 1, 2019.
- 1.22 "**Election Form**" shall mean the form established from time to time by the Committee that a Participant completes, signs and returns to the Committee to make an election under the Plan.
- 1.23 "Employee" shall mean a person who is an employee of any Employer.
- 1.24 "Employer(s)" shall be defined as follows:
  - (a) Except as otherwise provided in part (b) of this Section, the term "Employer" shall mean (i) the Company; (ii) any of its subsidiaries (now in existence or hereafter formed or acquired); and/or (iii) an entity that would be aggregated and treated as a single employer with the Company under Code section 414(b) (controlled group of corporations) or Code section 414(c) (a group of trades or businesses, whether or not incorporated, under common control).
  - (b) For the purpose of determining whether a Participant has experienced a Separation from Service, the term "Employer" shall mean:
    - (i) The entity for which the Participant performs services and with respect to which the legally binding right to compensation deferred or contributed under this Plan arises; and
    - (ii) All other entities with which the entity described above would be aggregated and treated as a single employer under Code section 414(b) (controlled group of corporations) and Code section 414(c) (a group of trades or businesses, whether or not incorporated, under common control), as applicable. In order to identify the group of entities described in the preceding sentence, the Committee shall use an ownership threshold of at least 50% as a substitute for the 80% minimum ownership threshold that appears in, and otherwise must be used when applying, the applicable provisions of (A) Code section 1563 for determining a controlled group of corporations under Code section 414(b), and (B) Treas. Reg. §1.414(c)-2 for determining the trades or businesses that are under common control under Code section 414(c).
- 1.25 "Entity" shall mean any individual, entity or group (within the meaning of Section 13(d)(3) or Section 14(d)(2) of the Exchange Act.
- 1.26 "ERISA" shall mean the Employee Retirement Income Security Act of 1974, as it may be amended from time to time, and the regulations and other authority issued thereunder by the appropriate governmental authority. References herein to any section of ERISA shall include references to any successor section or provision of ERISA.
- 1.27 "Exchange Act" shall means the United States Securities Exchange Act of 1934, as amended.
- 1.28 "Incentive Bonus" shall mean any cash compensation, in addition to Annual Bonus and Base Salary, earned by a Participant for services rendered during a Plan Year, under any Employer's monthly, quarterly, or other bonus and cash incentive plans. The Administrator, in its sole discretion, shall determine whether any particular type or item of compensation shall be deemed an "Incentive Bonus" for purposes of the Plan.
- 1.29 "In-Service Scheduled Distribution" shall mean the distribution set forth in Section 4.1 below.
- 1.30 "**Insolvent**" shall mean either (i) the Employer is unable to pay its debts as they become due, or (ii) the Employer is subject to a pending proceeding as a debtor in the United States Bankruptcy Code.
- 1.31 "Parent Company" means a company or other entity which as a result of a Change in Control owns the Company or all or substantially all of the Company's assets either directly or through one or more subsidiaries.
- 1.32 "**Participant**" shall mean any Employee (i) who is selected to participate in the Plan, and (ii) who submits an executed Election Form and Beneficiary Designation Form, which are accepted by the Committee.
- 1.33 "Plan" shall mean the Blucora Tax-Smart Executive Deferral Plan, which shall be evidenced by this instrument, as it may be amended from time to time, and by any other documents that together with this instrument define a Participant's rights to amounts credited to his or her Account Balance.
- 1.34 "Plan Year" shall mean a period beginning on January 1 of each calendar year and continuing through December 31 of such calendar year.
- 1.35 "Related Company" means any subsidiary of the Company or any other entity that is directly or indirectly controlled by, in control of, or under common control with the Company, including, without limitation, any Employer.

- 1.36 "**Retirement**" or "**Retires**" shall mean, with respect to a Participant, his or her Separation from Service for any reason other than death or Disability, on or after his or her attainment of (i) age sixty (60) and five (5) years of service with the Company or any Employer; (ii) age fifty-five (55) and ten (10) years of services with the Company or any Employer; or (iii) any age with twenty (20) years of service with the Company or any Employer.
- 1.37 "**Retirement Benefit**" shall mean the benefit set forth in Article 5.
- 1.38 "Separation from Service" shall mean a termination of services provided by a Participant to his or her Employer, whether voluntarily or involuntarily, other than by reason of death or Disability, as determined by the Committee in accordance with Treas. Reg. §1.409A-1(h). In determining whether a Participant has experienced a Separation from Service, the following provisions shall apply:
  - (a) For a Participant who provides services to an Employer as an Employee, except as otherwise provided in part (c) of this Section 1.38, a Separation from Service shall occur when such Participant has experienced a termination of employment with such Employer. A Participant shall be considered to have experienced a termination of employment when the facts and circumstances indicate that the Participant and his or her Employer reasonably anticipate that either (i) no further services will be performed for the Employer after a certain date, or (ii) that the level of bona fide services the Participant will perform for the Employer after such date (whether as an Employee or as an independent contractor) will permanently decrease to no more than 20% of the average level of bona fide services performed by such Participant (whether as an Employee or an independent contractor) over the immediately preceding 36-month period (or the full period of services to the Employer if the Participant has been providing services to the Employer less than 36 months).
  - If a Participant is on military leave, sick leave, or other bona fide leave of absence, the employment relationship between the Participant and the Employer shall be treated as continuing intact, provided that the period of such leave does not exceed 6 months, or if longer, so long as the Participant retains a right to reemployment with the Employer under an applicable statute or by contract. If the period of a military leave, sick leave, or other bona fide leave of absence exceeds 6 months and the Participant does not retain a right to reemployment under an applicable statute or by contract, the employment relationship shall be considered to be terminated for purposes of this Plan as of the first day immediately following the end of such 6-month period. In applying the provisions of this paragraph, a leave of absence shall be considered a bona fide leave of absence only if there is a reasonable expectation that the Participant will return to perform services for the Employer.
  - (b) If a Participant provides services for an Employer as both an Employee and as an independent contractor or a member of the Board, to the extent permitted by Treas. Reg. §1.409A-1(h)(5) the services provided by such Participant as an independent contractor or member of the Board shall not be taken into account in determining whether the Participant has experienced a Separation from Service as an Employee, and the services provided by such Participant as an Employee shall not be taken into account in determining whether the Participant has experienced a Separation from Service as an independent contractor or member of the Board.
- 1.39 "Specified Employee" shall mean any Participant who is determined to be a "key employee" (as defined under Code section 416(i) without regard to paragraph (5) thereof) for the applicable period, as determined annually by the Committee in accordance with Treas. Reg. §1.409A-1(i). In determining whether a Participant is a Specified Employee, the following provisions shall apply:
  - (a) The Committee's identification of the individuals who fall within the definition of "key employee" under Code section 416(i) (without regard to paragraph (5) thereof) shall be based upon the 12-month period ending on each December 31st (referred to below as the "identification date"). In applying the applicable provisions of Code section 416(i) to identify such individuals, "compensation" shall be determined in accordance with Treas. Reg. §1.415(c)-2(a) without regard to (i) any safe harbor provided in Treas. Reg. §1.415(c)-2(d), (ii) any of the special timing rules provided in Treas. Reg. §1.415(c)-2(e), and (iii) any of the special rules provided in Treas. Reg. §1.415(c)-2(g); and
  - (b) Each Participant who is among the individuals identified as a "key employee" in accordance with part (a) of this Section 1.39 shall be treated as a Specified Employee for purposes of this Plan if such Participant experiences a Separation from Service during the 12-month period that begins on the April 1st following the applicable identification date.
- 1.40 "Successor Company" means the surviving company, the successor company or Parent Company, as applicable, in connection with a Change in Control.
- 1.41 "Survivor Benefit" shall mean the benefit set forth in Article 8.

- 1.42 "**Termination Benefit**" shall mean the benefit set forth in Article 6.
- 1.43 "Trust" shall mean one or more trusts established by the Company in accordance with Article 15.
- 1.44 "Trustee" shall mean the duly appointed and acting trustee of the Trust, and any successor thereto.
- 1.45 "Unforeseeable Financial Emergency" shall mean a severe financial hardship of the Participant resulting from (a) an illness or accident of the Participant, the Participant's spouse, the Participant's Beneficiary or the Participant's dependent (as defined in Code section 152 without regard to paragraphs (b)(1), (b)(2) and (d)(1)(b) thereof), (b) a loss of the Participant's property due to casualty, or (c) such other similar extraordinary and unforeseeable circumstances arising as a result of events beyond the control of the Participant, all as determined by the Committee based on the relevant facts and circumstances. The circumstances that will constitute an unforeseeable emergency will depend upon the facts of each case, but an Unforeseeable Financial Emergency shall not be deemed to exist to the extent that such hardship is or may be relieved:
  - (a) Through reimbursement or compensation by insurance or otherwise;
  - (b) By liquidation of the Participant's assets, to the extent the liquidation of such assets would not itself cause severe financial hardship;
  - (c) By cessation of deferrals under this Plan or elective contributions made by the Participant under any 401(k) plan or other deferred compensation plan maintained by an Employer;
  - (d) Distributions or nontaxable loans from an Employer's other plans or any other employer's plans to the extent eligible;
  - (e) Loans from commercial sources at reasonable commercial terms;
  - (f) Bank or other savings accounts; or
  - (g) Reasonable, periodic payment arrangements with a creditor.

By way of example, the need to send a Participant's child to college or the desire to purchase or improve a home is not considered an Unforeseeable Financial Emergency.

1.46 "Year of Service" shall mean each year of continuous employment by the Participant with an Employer. For purposes of the Plan, the employment relationship is treated as continuing intact while the Participant is on military leave, sick leave, or other bona fide leave of absence if the period of such leave does not exceed six months, or if longer, so long as the Participant's right to reemployment is provided by either statute or contract.

# ARTICLE 2 Selection, Enrollment, and Eligibility

- 2.1 <u>Selection by Committee</u>. Participation in the Plan shall be limited to a select group of management and highly compensated Employees of the Employer, as determined by the Committee, in its sole discretion. From that group, the Committee shall select, in its sole discretion, Employees to participate in the Plan with respect to each Plan Year. The selection by the Committee of an Employee to participate in the Plan with respect to one Plan Year does not guarantee participation with respect to future Plan Years.
- 2.2 **Enrollment Requirements.** As a condition to participation, each selected Employee shall complete, execute and return to the Committee an Election Form and a Beneficiary Designation Form, all by the deadline(s) established by the Committee in accordance with the applicable provisions of the Plan. In addition, the Committee shall establish from time to time such other enrollment requirements as it determines in its sole discretion are necessary.
- 2.3 Eligibility; Commencement of Participation. Provided an Employee selected to participate in the Plan has met all enrollment requirements set forth in this Plan and required by the Committee, including returning all required documents to the Committee within the specified time periods set forth in the Plan, that Employee shall commence participation in the Plan on the first day of the month following the month in which the Employee completes all enrollment requirements. If an Employee fails to meet all such requirements within the periods required, in accordance with the provisions of the Plan, that Employee shall not be eligible to participate in the Plan until the first day of the Plan Year following the delivery to and acceptance by the Committee of the required documents.

2.4 Termination of Participation and/or Deferrals. If the Committee determines, in its sole discretion, that a Participant no longer qualifies as a member of a select group of management or highly compensated employees, as membership in such group is determined in accordance with ERISA Sections 201(2), 301(a)(3) and 401(a)(1), the Committee shall have the right, in its sole discretion, to (i) terminate any deferral election the Participant has made for the remainder of the Plan Year in which the Participant's membership status changes, (ii) prevent the Participant from making future deferral elections and/or (iii) to the extent permitted under Code section 409A, immediately distribute the Participant's then vested Account Balance as a Termination Benefit, and terminate the Participant's participation in the Plan.

### **ARTICLE 3**

## **Deferral Commitments/Company Contribution Amounts/Vesting/Crediting/Taxes**

### 3.1 **Minimum Deferrals**.

(a) **Annual Deferral Amount**. For each Plan Year, a Participant may elect to defer, as his or her Annual Deferral Amount, Base Salary, Annual Bonus, Incentive Bonus in the following minimum amounts for each deferral elected:

|   | Minimum Amount     |
|---|--------------------|
| Deferral                                      |                    |
| Base Salary, Annual<br>Bonus and/or Incentive | \$10,000 aggregate |
| Bonus   |                    |

If an election is made for less than the stated minimum amounts, or if no election is made, the amount deferred shall be zero.

(b) Short Plan Year. Notwithstanding the foregoing, if a Participant first becomes a Participant after the first day of a Plan Year the minimum Annual Deferral Amount shall be an amount equal to the minimum set forth above, multiplied by a fraction, the numerator of which is the number of complete months remaining in the Plan Year and the denominator of which is 12.

### 3.2 Maximum Deferral.

(a) Annual Deferral Amount. For each Plan Year, a Participant may elect to defer, as his or her Annual Deferral Amount, Base Salary, Annual Bonus, Incentive Bonus up to the following maximum percentages for each deferral elected:

| Deferral        | Maximum Percentage |
|-----------------|--------------------|
| Base Salary     | 90%                |
| Annual Bonus    | 100%               |
| Incentive Bonus | 100%               |

(b) Short Plan Year. Notwithstanding the foregoing, if a Participant first becomes a Participant after the first day of a Plan Year, then to the extent required by Section 3.3 and Code section 409A and related Treasury Regulations, the maximum amount of the Participant's Annual Deferral Amount, Base Salary, Annual Bonus, and Incentive Bonus that may be deferred by the Participant for the Plan Year shall be determined by applying the percentages set forth in Section 3.2(a) to the portion of such compensation attributable to services performed after the date that the Participant's deferral election is made.

### 3.3 Election to Defer; Effect of Election Form.

(a) First Plan Year. A selected Employee who first becomes eligible to participate in the Plan on or after the beginning of a Plan Year, as determined in accordance with Treas. Reg. §1.409A-2(a)(7)(ii) and the "plan aggregation" rules provided in Treas. Reg. §1.409A-1(c)(2), may be permitted to make an election to defer the portion of Annual Deferral Amount, Base Salary, Annual Bonus, and Incentive Bonus attributable to services to be performed after such election, provided that the Participant submits an Election Form on or before the deadline established by the Committee, which in no event shall be later than 30 days after the Participant first becomes eligible to participate in the Plan.

- Any deferral election made in accordance with this Section 3.3(a) shall become irrevocable no later than the 30<sup>th</sup> day after the date the selected Employee becomes eligible to participate in the Plan.
- (b) <u>Subsequent Plan Years</u>. For each succeeding Plan Year, the Participant's prior election shall remain in effect unless the Participant makes a subsequent irrevocable deferral election for such succeeding Plan Year, by timely delivering a new Election Form to the Committee, in accordance with its rules and procedures, before the December 31 preceding the Plan Year for which the election is made. If no such Election Form is timely delivered for a Plan Year, the Annual Deferral Amount shall be the same amount as elected for the immediately prior Plan Year.
- 3.4 Withholding and Crediting of Annual Deferral Amounts. For each Plan Year, the Base Salary portion of the Annual Deferral Amount shall be withheld from each regularly scheduled Base Salary payroll in equal amounts, as adjusted from time to time for increases and decreases in Base Salary. The Annual Bonus and Incentive Bonus portion of the Annual Deferral Amount shall be withheld at the time the Annual Bonus or Incentive Bonus are or otherwise would be paid to the Participant, whether or not this occurs during the Plan Year itself. Annual Deferral Amounts shall be credited to a Participant's Deferral Account at the time such amounts would otherwise have been paid to the Participant.

### 3.5 Company Contribution Amount.

- (a) For each Plan Year, an Employer may be required to credit amounts to a Participant's Company Contribution Account in accordance with employment or other agreements entered into between the Participant and the Employer. Such amounts shall be credited on the date or dates prescribed by such agreements.
- (b) For each Plan Year, an Employer, in its sole discretion, may, but is not required to, credit any amount it desires to any Participant's Company Contribution Account under this Plan, which amount shall be for that Participant the Company Contribution Amount for that Plan Year. The amount so credited to a Participant may be smaller or larger than the amount credited to any other Participant, and the amount credited to any Participant for a Plan Year may be zero, even though one or more other Participants receive a Company Contribution Amount for that Plan Year; provided, however, the Employer shall determine the amount credited to each Participant in a manner that does not violate any applicable nondiscrimination law.
- (c) The Company Contribution Amount described in this Section 3.5, if any, shall be credited on a date or dates to be determined by the Committee, in its sole discretion.

### 3.6 **Vesting**.

- (a) A Participant shall at all times be 100% vested in his or her Deferral Account.
- (b) A Participant shall be vested in his or her Company Contribution Account after one Year of Service, provided that, with respect to any Company Contributions made, the Committee may, in its sole discretion, declare that a different vesting schedule shall apply, which vesting schedule shall either solely apply to the Company Contributions made for such year, or to all future Company Contributions made thereafter, as determined by the Committee, in its sole discretion.
- (c) Notwithstanding anything to the contrary contained in this Section 3.6, upon a Change in Control or a Participant's death while employed by an Employer, Retirement, or Disability, a Participant's Company Contribution Account shall immediately become 100% vested (if it is not already vested in accordance with the above vesting schedules).
- 3.7 <u>Crediting/Debiting of Account Balances</u>. In accordance with, and subject to, the rules and procedures that are established from time to time by the Committee, in its sole discretion, amounts shall be credited or debited to a Participant's Account Balance in accordance with the following rules:
  - (a) Measurement Funds. The Participant may elect one or more of the measurement funds selected by the Committee, in its sole discretion, which are based on certain mutual funds or such other financial measures as selected by the Committee in its sole discretion (the "Measurement Funds"), for the purpose of crediting or debiting additional amounts to his or her Account Balance. As necessary, the Committee may, in its sole discretion, discontinue, substitute or add a Measurement Fund. Each such action will take effect as of the first day of the first calendar quarter that begins at least 30 days after the day on which the Committee gives Participants advance written notice of such change.

- Election of Measurement Funds. A Participant, in connection with his or her initial deferral election in accordance (b) with Section 3.3(a) above, shall elect, on the Election Form, one or more Measurement Fund(s) (as described in Section 3.7(a) above) to be used to determine the amounts to be credited or debited to his or her Account Balance. If a Participant does not elect any of the Measurement Funds as described in the previous sentence, the Participant's Account Balance shall automatically be allocated into the lowest-risk Measurement Fund, as determined by the Committee, in its sole discretion. The Participant may (but is not required to) elect, by submitting an Election Form to the Committee that is accepted by the Committee, to add or delete one or more Measurement Fund(s) to be used to determine the amounts to be credited or debited to his or her Account Balance, or to change the portion of his or her Account Balance allocated to each previously or newly elected Measurement Fund. If an election is made in accordance with the previous sentence, it shall apply as of the first business day deemed reasonably practicable by the Committee, in its sole discretion, and shall continue thereafter for each subsequent day in which the Participant participates in the Plan, unless changed in accordance with the previous sentence. Notwithstanding the foregoing, all investments hereunder shall be considered assets of the Company, and the Participant shall remain subject to all applicable provisions of the Plan, including, without limitation, Sections 15.4 and 16.2 below. Each Participant, as a condition to his or her participation in the Plan, agrees to indemnify and hold harmless the Committee, the Company, and each Employer, and their representatives, delegates and agents, from and against any investment losses or other damages of any kind relating to the deemed investment of the Participant's Account Balance under the Plan. No assurances are provided by any person or entity that any investment results of any Measurement Fund will be favorable and, as with most investments, there is a risk of loss.
- (c) <u>Proportionate Allocation</u>. In making any election described in Section 3.7(b) above, the Participant shall specify on the Election Form, in increments of one percent (1%), the percentage of his or her Account Balance to be allocated to a Measurement Fund (as if the Participant was making an investment in that Measurement Fund with that portion of his or her Account Balance).
- (d) <u>Crediting or Debiting Method</u>. The performance of each Measurement Fund (either positive or negative) will be determined by the Committee, in its sole discretion on a daily basis based on the manner in which such Participant's Account Balance has been hypothetically allocated among the Measurement Funds by the Participant.
- (e) No Actual Investment. Notwithstanding any other provision of this Plan that may be interpreted to the contrary, the Measurement Funds are to be used for measurement purposes only, and a Participant's election of any such Measurement Fund, the allocation of his or her Account Balance thereto, the calculation of additional amounts and the crediting or debiting of such amounts to a Participant's Account Balance shall not be considered or construed in any manner as an actual investment of his or her Account Balance in any such Measurement Fund. In the event that the Company or the Trustee, in its own discretion, decides to invest funds in any or all of the investments on which the Measurement Funds are based, no Participant shall have any rights in or to such investments themselves. Without limiting the foregoing, a Participant's Account Balance shall at all times be a bookkeeping entry only and shall not represent any investment made on his or her behalf by the Company or the Trust; the Participant shall at all times remain an unsecured creditor of the Company.

#### 3.8 FICA and Other Taxes.

- (a) Annual Deferral Amounts. For each Plan Year in which an Annual Deferral Amount is being withheld from a Participant, the Participant's Employer(s) shall withhold from that portion of the Participant's Base Salary, Annual Bonus and/or Incentive Bonus that is not being deferred, in a manner determined by the Employer(s), the Participant's share of FICA and other employment taxes on such Annual Deferral Amount. If necessary, the Committee may reduce the Annual Deferral Amount in order to comply with this Section 3.8.
- (b) <u>Company Contribution Account</u>. When a Participant becomes vested in a portion of his or her Company Contribution Account, the Participant's Employer(s) shall withhold from that portion of the Participant's Base Salary, Annual Bonus and/or Incentive Bonus that is not deferred, in a manner determined by the Employer(s), the Participant's share of FICA and other employment taxes on such Company Contribution Amount. If necessary, the Committee may reduce the vested portion of the Participant's Company Contribution Account in order to comply with this Section 3.8.
- (c) <u>Distributions</u>. The Participant's Employer(s), or the Trustee, shall withhold from any payments made to a Participant under this Plan all federal, state and local income, employment and other taxes required to be withheld by the Employer(s), or the Trustee, in connection with such payments, in amounts and in a manner to be determined in the sole discretion of the Employer(s) and the Trustee.

#### **ARTICLE 4**

# In-Service Scheduled Distribution; Unforeseeable Financial Emergencies; Withdrawal Election

- 4.1 **In-Service Scheduled Distribution.** In connection with each election to defer an Annual Deferral Amount, a Participant may irrevocably elect to receive an In-Service Scheduled Distribution from the Plan with respect to all or a portion of (i) the Annual Deferral Amount and (ii) the Company Contribution Amount. The In-Service Scheduled Distribution shall be a lump sum payment in an amount that is equal to the portion of the Annual Deferral Amount and the vested portion of the Company Contribution Amount that the Participant elected to have distributed as an In-Service Scheduled Distribution, plus amounts credited or debited in the manner provided in Section 3.7 above to that amount, calculated as of the close of business on or around the Benefit Distribution Date designated by the Participant in accordance with this Section 4.1 (a "Scheduled **Distribution**"). The Benefit Distribution Date for an amount subject to an In-Service Scheduled Distribution election shall be the first day of any Plan Year designated by the Participant, which may be no sooner than 3 Plan Years after the end of the Plan Year in which the Annual Deferral Amount is actually deferred or the vested portion of the Company Contribution Amount is actually contributed. The Participant may elect different Benefit Distribution Dates for amounts deferred with respect to different Plan Years. Subject to the other terms and conditions of this Plan, each In-Service Scheduled Distribution elected shall be paid out during a 60-day period commencing immediately after the Benefit Distribution Date. By way of example, if an In-Service Scheduled Distribution is elected for Annual Deferral Amounts that are deferred in the Plan Year commencing January 1, 2019, the earliest Benefit Distribution Date that may be designated by a Participant would be January 1, 2023, and the In-Service Scheduled Distribution would be paid out during the 60-day period commencing immediately after such Benefit Distribution Date.
- 4.2 Other Benefits Take Precedence Over In-Service Scheduled Distributions. Should an event occur prior to any Benefit Distribution Date designated for a Scheduled Distribution that triggers a benefit under Articles 5, 6, 7, or 8, any Annual Deferral Amount and/or Company Contribution Amount, plus amounts credited or debited thereon, that are subject to an In-Service Scheduled Distribution election under Section 4.1 above shall not be paid in accordance with Section 4.1 above, but shall be paid in accordance with the other applicable Article.

### 4.3 Withdrawal Payout/Suspensions for Unforeseeable Financial Emergencies.

- (a) If the Participant experiences an Unforeseeable Financial Emergency prior to the occurrence of a distribution event under Articles 5, 6, 7, or 8, the Participant may petition the Committee to suspend deferrals of Base Salary, Annual Bonus, and Incentive Bonus required to be made by such Participant, to the extent deemed necessary by the Committee to satisfy the Unforeseeable Financial Emergency. If suspension of deferrals is not sufficient to satisfy the Participant's Unforeseeable Financial Emergency, the Participant may further petition the Committee to receive a partial or full payout from the Plan. The Participant shall only receive a payout from the Plan to the extent such payout is deemed necessary by the Committee to satisfy the Participant's Unforeseeable Financial Emergency.
- (b) The payout shall not exceed the lesser of (i) the Participant's vested Account Balance, calculated as of the close of business on or around the Benefit Distribution Date for such payout, as determined by the Committee in accordance with the provisions set forth below, or (ii) the amount reasonably needed to satisfy the Unforeseeable Financial Emergency, plus amounts necessary to pay Federal, state, or local income taxes or penalties reasonably anticipated as a result of the distribution.
- (c) If the Committee, in its sole discretion, approves a Participant's petition for suspension, the Participant's deferrals under this Plan shall be suspended as of the date of such approval. If the Committee, in its sole discretion, approves a Participant's petition for suspension and payout, the Participant's deferrals under this Plan shall be suspended as of the date of such approval and the Participant's Benefit Distribution Date for such payout shall be the date on which such Committee approval occurs, and such payout shall be distributed to the Participant in a lump sum no later than 60 days after such Benefit Distribution Date.

ARTICLE 5
Retirement Benefit

Retirement Benefit. A Participant who experiences a Separation from Service that qualifies as a Retirement shall receive, as a Retirement Benefit, his or her vested Account Balance in either a lump sum or annual installment payments, as elected by the Participant in accordance with Section 5.2. A Participant's Retirement Benefit shall be calculated as of the close of business on or around the applicable Benefit Distribution Date for such benefit. The Benefit Distribution Date shall be (i) the first day after the end of the six-month period immediately following the date on which the Participant Retires, if the Participant is a Specified Employee, and (ii) for all other Participants, the date on which the Participant Retires; provided, however, if a Participant changes the form of distribution for the Retirement Benefit in accordance with Section 5.2(b), the Benefit Distribution Date for the Retirement Benefit shall be determined in accordance with Section 5.2(b).

### 5.2 Payment of Retirement Benefit.

- (a) A Participant, in connection with his or her commencement of participation in the Plan, shall elect on an Election Form to receive the Retirement Benefit upon Retirement in a lump sum or pursuant to an Annual Installment Method of up to 15 years. If a Participant does not make any election with respect to the payment of the Retirement Benefit upon Retirement, then such benefit shall be payable in a lump sum.
- (b) A Participant may change the form of payment for the Retirement Benefit upon Retirement by submitting an Election Form to the Committee in accordance with the following criteria:
  - (i) The election shall not take effect until at least 12 months after the date on which the election is made;
  - (ii) The new Benefit Distribution Date for the Participant's Retirement Benefit upon Retirement shall be 5 years after the Benefit Distribution Date that would otherwise have been applicable to such benefit; and
  - (iii) The election must be made at least 12 months prior to the Benefit Distribution Date that would otherwise have been applicable to the Participant's Retirement Benefit upon Retirement.

For purposes of applying the provisions of this Section 5.2(b), a Participant's election to change the form of payment for the Retirement Benefit upon Retirement shall not be considered to be made until the date on which the election becomes irrevocable. Such an election shall become irrevocable no later than the date that is 12 months prior to the Benefit Distribution Date that would otherwise have been applicable to the Participant's Retirement Benefit upon Retirement. Subject to the requirements of this Section 5.2(b), the Election Form most recently accepted by the Committee that has become effective shall govern the form of payout of the Participant's Retirement Benefit.

- (c) The lump sum payment shall be made, or installment payments shall commence, no later than 60 days after the Participant's Benefit Distribution Date. Remaining installments, if any, shall be paid no later than 60 days after each anniversary of the Participant's Benefit Distribution Date.
- (d) Notwithstanding anything to the contrary contained herein, the Company may distribute a Participant's Account Balance if, at any time after such Participant's Separation from Service, the Account Balance does not exceed the limit in Code section 402(g)(1)(B) and such distribution would result in the termination of the Participant's entire interest in the Plan as provided under Code section 409A.

# ARTICLE 6 Termination Benefit

- 6.1 <u>Termination Benefit</u>. A Participant who experiences a Separation from Service that does not qualify as a Retirement shall receive a Termination Benefit, which shall be equal to the Participant's vested Account Balance, calculated as of the close of business on or around the Benefit Distribution Date for such benefit. The Benefit Distribution Date shall be (i) the first day after the end of the six-month period immediately following the date on which the Participant experiences a Separation from Service, if the Participant is a Specified Employee, and (ii) for all other Participants, the date on which the Participant experiences a Separation from Service.
- 6.2 **Payment of Termination Benefit**. The Termination Benefit shall be paid to the Participant in a lump sum payment no later than 60 days after the Participant's Benefit Distribution Date.

ARTICLE 7

<u>Disability Waiver and Benefit</u>

- Disability Benefit. A Participant who experiences a Disability, shall receive, as a Disability Benefit his or her vested Account Balance, calculated as of the close of business on or around the Benefit Distribution Date for such benefit. The Benefit Distribution Date shall be the date that the Committee determines that such Participant is Disabled within the meaning of Section 1.19; provided, however, if the Participant experienced a Separation from Service prior to the date he or she was Disabled, then the Benefit Distribution Date shall be determined in accordance with Section 5.1 or 6.1 above, as applicable (and the Participant shall receive a Retirement Benefit or Termination Benefit, as applicable, and not a Disability Benefit). A Participant, in connection with his or her commencement of participation in the Plan, shall elect on an Election Form to receive the Disability Benefit in a lump sum or pursuant to an Annual Installment Method of up to 15 years. If a Participant does not make any election with respect to the payment of the Disability Benefit, then such benefit shall be payable in a lump sum. The lump sum payment shall be made, or installment payments shall commence, no later than 60 days after the Disabled Participant's Benefit Distribution Date, and any remaining installments, if any, shall be paid no later than 60 days after each anniversary of the Disabled Participant's Benefit Distribution Date.
- 7.2 **Deferral Following Disability**. If a Participant returns to employment with an Employer after a Disability ceases, the Participant may elect to defer an Annual Deferral Amount for the Plan Year following his or her return to employment or service and for every Plan Year thereafter while a Participant in the Plan; provided such deferral elections are otherwise allowed and an Election Form is delivered to and accepted by the Committee for each such election in accordance with Section 3.3(b) above.

# ARTICLE 8 Survivor Benefit

- 8.1 <u>Survivor Benefit</u>. In the event of a Participant's death prior to the complete distribution of his or her vested Account Balance, the Participant's Beneficiary(ies) shall receive a Survivor Benefit which will be equal to the Participant's unpaid vested Account Balance, calculated as of the close of business on or around the Benefit Distribution Date for such benefit. The Benefit Distribution Date shall be the date on which the Committee is provided with proof that is satisfactory to the Committee of the Participant's death.
- 8.2 **Payment of Survivor Benefit**. The Survivor Benefit shall be paid to the Participant's Beneficiary(ies) in a lump sum payment no later than 60 days after the Participant's Benefit Distribution Date.

# ARTICLE 9 Beneficiary Designation

- 9.1 **Beneficiary**. Each Participant shall have the right, at any time, to designate his or her Beneficiary(ies) (both primary as well as contingent) to receive any benefits payable under the Plan to a beneficiary upon the death of a Participant. The Beneficiary designated under this Plan may be the same as or different from the Beneficiary designation under any other plan of an Employer in which the Participant participates.
- 9.2 **Beneficiary Designation: Change: Spousal Consent**. A Participant shall designate his or her Beneficiary by completing and signing the Beneficiary Designation Form, and returning it to the Committee or its designated agent. A Participant shall have the right to change a Beneficiary by completing, signing and otherwise complying with the terms of the Beneficiary Designation Form and the Committee's rules and procedures, as in effect from time to time. If the Participant names someone other than his or her spouse as a Beneficiary, spousal consent in writing to the designation of a different Beneficiary is required to be provided in a form designated by the Committee, executed by such Participant's spouse and returned to the Committee. A Beneficiary designation that has received spousal consent cannot be changed without spousal consent. Upon the acceptance by the Committee of a new Beneficiary Designation Form, all Beneficiary designations previously filed shall be canceled. The Committee shall be entitled to rely on the last Beneficiary Designation Form filed by the Participant and accepted by the Committee prior to his or her death.
- 9.3 **Acknowledgment**. No designation or change in designation of a Beneficiary shall be effective until received and acknowledged in writing by the Committee or its designated agent.
- 9.4 **No Beneficiary Designation**. If a Participant fails to designate a Beneficiary as provided in Sections 9.1, 9.2 and 9.3 above or, if all designated Beneficiaries predecease the Participant or die prior to complete distribution of the Participant's benefits, then the Participant's designated Beneficiary shall be deemed to be his or her surviving spouse. If the Participant has no surviving spouse, the benefits remaining under the Plan to be paid to a Beneficiary shall be payable to the executor or personal representative of the Participant's estate.

- 9.5 **Doubt as to Beneficiary**. If the Committee has any doubt as to the proper Beneficiary to receive payments pursuant to this Plan, the Committee shall have the right, exercisable in its discretion, to either (i) cause the Participant's Employer to withhold such payments until this matter is resolved to the Committee's satisfaction or (ii) direct that the amount be paid into any court of competent jurisdiction in an interpleader action, and such payment shall be a full and complete discharge of any liability or obligation under the Plan or Trust Agreement to the full extent of such payment.
- 9.6 **Discharge of Obligations**. The payment of benefits under the Plan to a Beneficiary shall fully and completely discharge all Employers and the Committee from all further obligations under this Plan with respect to the Participant.

## ARTICLE 10 Leave of Absence

- 10.1 Paid Leave of Absence. If a Participant is authorized by the Participant's Employer to take a paid leave of absence from the employment of the Employer and such leave of absence does not constitute a Separation from Service, (i) the Participant shall continue to be considered eligible for the benefits provided in Articles 4, 5, 6, 7 or 8 in accordance with the provisions of those Articles, and (ii) the Annual Deferral Amount shall continue to be withheld during such paid leave of absence in accordance with Section 3.3 above.
- 10.2 <u>Unpaid Leave of Absence</u>. If a Participant is authorized by the Participant's Employer to take an unpaid leave of absence from the employment of the Employer for any reason and such leave of absence does not constitute a Separation from Service, such Participant shall continue to be eligible for the benefits provided in Articles 4, 5, 6, 7 or 8 in accordance with the provisions of those Articles. However, the Participant shall be excused from fulfilling his or her Annual Deferral Amount commitment that would otherwise have been withheld during the remainder of the Plan Year in which the unpaid leave of absence is taken. During the unpaid leave of absence, the Participant shall not be allowed to make any new or additional deferral elections. However, if the Participant returns to employment, the Participant may elect to defer an Annual Deferral Amount for the Plan Year following his or her return to employment and for every Plan Year thereafter while a Participant in the Plan; provided such deferral elections are otherwise allowed and an Election Form is delivered to and accepted by the Committee for each such election in accordance with Section 3.3 above.

# ARTICLE 11 Termination, Amendment or Modification

11.1 Termination. Although the Company anticipates that it will continue the Plan for an indefinite period of time, there is no guarantee that the Company will continue the Plan or will not terminate the Plan at any time in the future or that any Employer will continue to participate in the Plan or will not terminate its participation in the Plan at any time in the future. Accordingly, the Company reserves the right to discontinue its sponsorship of the Plan and each Employer reserves the right to withdraw its participation in the Plan and both the Company and each Employer reserves the right to terminate the Plan at any time with respect to any or all of its participating Employees, by action of its Board. However, after the Plan termination the Account Balances of such Participants shall continue to be credited with Annual Deferral Amounts attributable to a deferral election that was in effect prior to the Plan termination to the extent deemed necessary to comply with Code section 409A and related Treasury Regulations, and additional amounts shall continue to credited or debited to such Participants' Account Balances pursuant to Section 3.7. The Measurement Funds available to Participants following the termination of the Plan shall be comparable in number and type to those Measurement Funds available to Participants in the Plan Year preceding the Plan Year in which the Plan termination is effective. In addition, following a Plan termination, Participant Account Balances shall remain in the Plan and shall not be distributed until such amounts become eligible for distribution in accordance with the other applicable provisions of the Plan. Notwithstanding the preceding sentence, to the extent permitted by Treas, Reg. §1.409A-3(i) (4)(ix), the Employer may provide that upon termination of the Plan, all Account Balances of the Participants shall be distributed, subject to and in accordance with any rules established by such Employer deemed necessary to comply with the applicable requirements and limitations of Treas. Reg. §1.409A-3(j)(4)(ix).

- 11.2 Amendment. The Company may, at any time, amend or modify the Plan in whole or in part by the action of its Board; provided, however, that: (i) no amendment or modification shall be effective to decrease or restrict the value of a Participant's vested Account Balance in existence at the time the amendment or modification is made, and (ii) no amendment or modification of this Section 11.2 or Section 12.3 below shall be effective. Notwithstanding the foregoing provisions of this Section 11.2, by written instrument, the Plan may be amended by the Company at any time prior to a Change in Control if required to ensure that the Plan is characterized as a "top-hat plan" of deferred compensation maintained for a select group of management or highly compensated employees as described under ERISA Sections 201(2), 301(a)(3), and 401(a)(1), or to conform the Plan to the provisions and requirements of any applicable law (including ERISA and the Code). No such amendment described in the immediately preceding sentence shall be considered prejudicial to any interest of a Participant or a Beneficiary hereunder.
- 11.3 <u>Effect of Payment</u>. The full payment of the Participant's vested Account Balance under Articles 4, 5, 6, or 7 of the Plan shall completely discharge all obligations to a Participant and his or her designated Beneficiaries under this Plan shall terminate.

# ARTICLE 12 Administration

- 12.1 Committee Duties. Except as otherwise provided in this Article 12, this Plan shall be administered by a Committee, which shall consist of the Board of the Company, or such committee as the Board of the Company shall appoint. The Board of the Company may remove or replace any member of a committee appointed by the Board of the Company at any time in its sole discretion. Members of the Committee may be Participants under this Plan. Any individual serving on the Committee who is a Participant shall abstain from any discussion and vote, and shall not otherwise act on any matter relating directly to himself or herself. When making a determination or calculation, the Committee shall be entitled to rely on information furnished by a Participant or the Company. The members of the Committee shall not receive any special compensation for serving in their capacities as members of the Committee, but shall be reimbursed by the Company for any reasonable expenses incurred in connection therewith. No bond or other security need be required of the Committee or any member thereof.
- 12.2 Administration of the Plan. The Committee shall operate, administer, interpret, construe, and construct the Plan, including without limitation, the sole discretion and authority to (i) make, amend, interpret and enforce all appropriate rules and regulations for the administration of this Plan and (ii) decide or resolve any and all questions as may arise in connection with the Plan, including correcting any defect, supplying any omission, or reconciling any inconsistency. The Committee shall have all powers necessary or appropriate to implement and administer the terms and provisions of the Plan, including the power to make findings of fact.
- Administration Upon Change In Control. For purposes of this Plan, the Committee shall be the "Administrator" at all times 12.3 prior to the occurrence of a Change in Control Within 120 days following a Change in Control, an independent third party "Administrator" may be selected by the individual who, immediately prior to the Change in Control, was the Company's Chief Executive Officer, Chief Financial Officer, Chief Legal Officer or, if not so identified, the Company's highest ranking officer (the "Ex-Officer"), and approved by the Trustee. The Committee, as constituted prior to the Change in Control, shall continue to be the Administrator until the earlier of (i) the date on which such independent third party is selected and approved, or (ii) the expiration of the one hundred and twenty (120) day period following the Change in Control. If an independent third party is not selected within 120 days of such Change in Control, the Committee, as described in Section 12.1 above, shall be the Administrator. The Administrator shall have the discretionary power to determine all questions arising in connection with the administration of the Plan and the interpretation of the Plan and Trust including, but not limited to benefit entitlement determinations; provided, however, upon and after the occurrence of a Change in Control, the Administrator shall have no power to direct the investment of Plan or Trust assets or select any investment manager or custodial firm for the Plan or Trust. Upon and after the occurrence of a Change in Control, the Company must: (1) pay all reasonable administrative expenses and fees of the Administrator; (2) indemnify the Administrator against any costs, expenses and liabilities including, without limitation, attorney's fees and expenses arising in connection with the performance of the Administrator hereunder, except with respect to matters resulting from the gross negligence or willful misconduct of the Administrator or its employees or agents; and (3) supply full and timely information to the Administrator on all matters relating to the Plan, the Trust, the Participants and their Beneficiaries, the Account Balances of the Participants, the date and circumstances of the Disability, death or Separation from Service of the Participants, and such other pertinent information as the Administrator may reasonably require. Upon and after a Change in Control, the Administrator may be terminated (and a replacement appointed) by the Trustee only with the approval of the Ex-Officer. Upon and after a Change in Control, the Administrator may not be terminated by the Company.
- 12.4 **Agents**. In the administration of this Plan, the Committee may, from time to time, employ agents and delegate to them such administrative duties as it sees fit (including acting through a duly appointed representative).

- 12.5 <u>Binding Effect of Decisions</u>. The determination of the Committee as to the proper interpretation, construction, or application of any term or provision in the Plan and the rules and regulations promulgated hereunder shall be final, binding, and conclusive with respect to all interested parties.
- Indemnity of Committee. To the full extent permitted by applicable law, the Company and each Employer shall individually 12.6 and collectively indemnify and hold harmless each past, present, and future member of the Committee, any Employee to whom the duties of the Committee may be delegated, and the Administrator against any and all claims, losses, damages, judgments, settlements, costs, expenses or liabilities (and all actions in respect thereof and any legal or other costs and expenses in giving testimony or furnishing documents in response to a subpoena or otherwise) arising from any action or failure to act with respect to this Plan, except in the case of gross negligence or willful misconduct by the Committee, any of its members, any such Employee or the Administrator, including the cost of investigating, preparing, or defending any pending, threatened or anticipated possible action, claim, suit or other proceeding, whether or not in connection with litigation in which the Committee, such Employee, or the Administrator is a party. The foregoing right of indemnification shall inure to the benefit of the successors and assigns, and the heirs, executors, administrators and personal representatives of the Committee, any such Employee, and the Administrator, and shall be in addition to all other rights to which the Committee, any such Employee, and the Administrator may be entitled as a matter of law, contract, or otherwise. The amount of indemnification during a calendar year shall not affect the fees and expenses eligible for reimbursement in any other calendar year. Indemnification shall be made on or before the last day of the calendar year following the calendar year in which the amounts or liabilities subject to such indemnification were incurred.
- 12.7 **Employer Information**. To enable the Committee and/or Administrator to perform its functions, the Company and each Employer shall supply full and timely information to the Committee and/or Administrator, as the case may be, on all matters relating to the compensation of its Participants, the date and circumstances of the Disability, death or Separation from Service of its Participants, and such other pertinent information as the Committee or Administrator may reasonably require.
- 12.8 **Reliance Upon Information**. No member of the Committee shall be liable for any decision, action, omission, or mistake in judgment in connection with the administration of the Plan, provided that he acted in good faith. Without limiting the generality of the foregoing sentence, any decision or action taken by the Committee in reasonable reliance upon information supplied to it by the Board, the Company, any Employer, legal counsel, or the Company's independent accountants shall be deemed to have been taken in good faith. The Committee may from time to time consult with counsel who may be counsel to any Employer or other counsel, with respect to its obligations or duties hereunder, or with respect to any action, proceeding or question of law, and shall not be held liable with respect to any action taken or omitted, in good faith, pursuant to the advice of such counsel.

# ARTICLE 13 Other Benefits and Agreements

13.1 Coordination with Other Benefits. The benefits provided for a Participant and Participant's Beneficiary under the Plan are in addition to any other benefits available to such Participant under any other plan or program for employees of the Participant's Employer. The Plan shall supplement and shall not supersede, modify or amend any other such plan or program except as may otherwise be expressly provided.

# ARTICLE 14 Claims Procedures

- 14.1 Presentation of Claim. Any Participant or Beneficiary of a deceased Participant (such Participant or Beneficiary being referred to below as a "Claimant") may deliver to the Committee a written claim for a determination with respect to the amounts distributable to such Claimant from the Plan. If such a claim relates to the contents of a notice received by the Claimant, the claim must be made within 60 days after such notice was received by the Claimant. All other claims must be made within 180 days of the date on which the event that caused the claim to arise occurred. The claim must state with particularity the determination desired by the Claimant.
- 14.2 Notification of Decision. The Committee shall consider a Claimant's claim within a reasonable time, but no later than 90 days after receiving the claim. If the Committee determines that special circumstances require an extension of time for processing the claim, written notice of the extension shall be furnished to the Claimant prior to the termination of the initial 90-day period. In no event shall such extension exceed a period of 90 days from the end of the initial period. The extension notice shall indicate the special circumstances requiring an extension of time and the date by which the Committee expects to render the benefit determination. The Committee shall notify the Claimant in writing:

- (a) that the Claimant's requested determination has been made, and that the claim has been allowed in full; or
- (b) that the Committee has reached a conclusion contrary, in whole or in part, to the Claimant's requested determination, and such notice must set forth in a manner calculated to be understood by the Claimant:
  - (i) the specific reason(s) for the denial of the claim, or any part of it;
  - (ii) specific reference(s) to pertinent provisions of the Plan upon which such denial was based;
  - (iii) a description of any additional material or information necessary for the Claimant to perfect the claim, and an explanation of why such material or information is necessary;
  - (iv) an explanation of the claim review procedure set forth in Section 14.3 below; and
  - (v) a statement of the Claimant's right to bring a civil action under ERISA Section 502(a) following an adverse benefit determination on review.
- 14.3 **Review of a Denied Claim.** On or before 60 days after receiving a notice from the Committee that a claim has been denied, in whole or in part, a Claimant (or the Claimant's duly authorized representative), if he or she disagrees with the claim denial, must file with the Committee a written request for a review of the denial of the claim. The Claimant (or the Claimant's duly authorized representative):
  - (a) may, upon request and free of charge, have reasonable access to, and copies of, all documents, records and other information relevant to the claim for benefits;
  - (b) may submit written comments or other documents; and/or
  - (c) may request a hearing, which the Committee, in its sole discretion, may grant.
- 14.4 <u>Decision on Review</u>. The Committee shall render its decision on review promptly, and no later than 60 days after the Committee receives the Claimant's written request for a review of the denial of the claim. If the Committee determines that special circumstances require an extension of time for processing the claim, written notice of the extension shall be furnished to the Claimant prior to the termination of the initial 60-day period. In no event shall such extension exceed a period of 60 days from the end of the initial period. The extension notice shall indicate the special circumstances requiring an extension of time and the date by which the Committee expects to render the benefit determination. In rendering its decision, the Committee shall take into account all comments, documents, records and other information submitted by the Claimant relating to the claim, without regard to whether such information was submitted or considered in the initial benefit determination. The decision must be written in a manner calculated to be understood by the Claimant, and it must contain:
  - (a) specific reasons for the decision;
  - (b) specific reference(s) to the pertinent Plan provisions upon which the decision was based;
  - (c) a statement that the Claimant is entitled to receive, upon request and free of charge, reasonable access to and copies of, all documents, records and other information relevant (as defined in applicable ERISA regulations) to the Claimant's claim for benefits; and
  - (d) a statement of the Claimant's right to bring a civil action under ERISA Section 502(a).
- 14.5 <u>Legal Action</u>. A Claimant's compliance with the foregoing provisions of this Article 14 is a mandatory prerequisite to a Claimant's right to commence any legal action with respect to any claim for benefits under this Plan.

# ARTICLE 15 Trust

15.1 Establishment of the Trust. In order to provide assets from which to fulfill the obligations of the Participants and their beneficiaries under the Plan, the Company intends, pursuant to formal action by the Committee, to establish a trust by a trust agreement with a third party, the Trustee, to which each Employer may, in its discretion, contribute cash or other property, including securities issued by the Company, to provide for the benefit payments under the Plan; provided, however, in the event of a Change in Control the Company must contribute sufficient assets to any such Trust equal to the aggregate value of all Account Balances determined as of the date of the Change in Control.

- 15.2 <u>Interrelationship of the Plan and the Trust</u>. The provisions of the Plan shall govern the rights of a Participant to receive distributions pursuant to the Plan. The provisions of the Trust shall govern the rights of the Employers, Participants and the creditors of the Employers to the assets transferred to the Trust. Each Employer shall at all times remain liable to carry out its obligations under the Plan.
- 15.3 **Distributions From the Trust**. Each Employer's obligations under the Plan may be satisfied with Trust assets distributed pursuant to the terms of the Trust, and any such distribution shall reduce the Employer's obligations under this Plan.
- 15.4 Participants' Rights under the Trust. The assets of the Trust shall be held for the benefit of the Participants in accordance with the terms of the Plan and Trust Agreement. All the assets of the Trust shall remain subject to the claims of the general unsecured creditors of each Employer, and the rights of the Participants to the assets of the Trust shall be limited as provided in the Plan and Trust Agreement in the event that an Employer becomes Insolvent.

# ARTICLE 16 Miscellaneous

- 16.1 Status of Plan. The Plan is intended to be a plan that is not qualified within the meaning of Code section 401(a) and that "is unfunded and is maintained by an employer primarily for the purpose of providing deferred compensation for a select group of management or highly compensated employees" within the meaning of ERISA Sections 201(2), 301(a)(3) and 401(a)(1), and as such it is intended that the Plan be exempt from the participation, vesting, funding, and fiduciary responsibility requirements of Title I of ERISA. This Plan is also intended to qualify for simplified reporting under U.S. Department of Labor Regulation Section 2530.104-23, which provides for an alternative method of compliance for plans described in such regulation. The Plan shall be administered and interpreted (a) to the extent possible in a manner consistent with the intent described in the preceding sentence, and (b) in accordance with Code section 409A the Final Regulations thereunder and other related Treasury guidance.
- 16.2 <u>Limitation of Rights</u>. Nothing in this Plan shall be construed to:
  - (a) Give any individual who is employed by an Employer or any entity under common ownership or control with an Employer any right to be a Participant unless and until such person is designated pursuant to Section 2.1 above;
  - (b) Give a Participant any rights, other than as an unsecured general creditor of each Employer with respect to any amount credited to his or her Account, until such amount is actually distributed to him or her;
  - (c) Limit in any way the right of an Employer to terminate a Participant's employment;
  - (d) Give a Participant or any other person any interest in any fund, reserve or any specific asset of the Employer or any entity under common ownership or control with an Employer; or
  - (e) Create a fiduciary relationship between the Participant and any Employer.
- 16.3 Nonalienation of Benefits. No right or benefit under this Plan shall be subject to anticipation, alienation, attachment, garnishment, sale, transfer, assignment (either at law or in equity), levy, execution, pledge, encumbrance, charge, or any other legal or equitable process, and any attempt to do so will be void and without effect. No right or benefit hereunder shall in any manner be liable for or subject to any debts, contracts, liabilities, engagements, or torts of the Participant or Beneficiary entitled to such benefits. The withholding of taxes from benefit payments hereunder; the recovery under the Plan of overpayments of benefits previously made to a Participant; the transfer of benefit rights from the Plan to another plan; or the direct deposit of benefit payments to an account in a banking institution (if not actually part of an arrangement constituting an assignment or alienation), shall not be construed as an assignment or alienation for purposes of the immediately preceding paragraph. The first sentence of this Section 16.3 shall not preclude (a) the Participant from designating a Beneficiary to receive any benefit payable hereunder upon his or her death, or (b) the executors, administrators, or other legal representatives of the Participant or his or her estate from assigning any rights hereunder to the person or persons entitled thereto. In the event that any Participant's or Beneficiary's benefits hereunder are garnished or attached by order of any court, the Company or the Trustee may bring an action or a declaratory judgment in a court of competent jurisdiction to determine the proper recipient of the benefits to be paid under the Plan. During the pendency of such action, any benefits that become payable shall be held as credits to the Participant's Account or, if the Company prefers, paid into the court as they become payable, to be distributed by the court to the recipient as the court deems proper at the close of such action.

- 16.4 <u>Unsecured General Creditor</u>. Participants and their Beneficiaries, heirs, successors and assigns shall have no legal or equitable rights, interests or claims in any property or assets of an Employer. For purposes of the payment of benefits under this Plan, any and all of an Employer's assets shall be, and remain, the general, unpledged unrestricted assets of the Employer. An Employer's obligation under the Plan shall be merely that of an unfunded and unsecured promise to pay money in the future.
- 16.5 **Waiver**. No term or condition of this Plan shall be deemed to have been waived, nor shall there be an estoppel against the enforcement of any provision of this Plan, except by written instrument of the party charged with such waiver or estoppel. No such written waiver shall be deemed a continuing waiver unless specifically stated therein, and each such waiver shall operate only as to the specific term or condition waived and shall not constitute a waiver of such term or condition for the future or as to any act other than that specifically waived. Any waiver by any party hereto of a breach of any provision of the Plan by any other party shall not operate or be construed as a waiver by such party of any subsequent breach thereof.
- 16.6 <u>Employer's Liability</u>. An Employer's liability for the payment of benefits shall be defined only by the Plan, as entered into between the Employer and a Participant. An Employer shall have no obligation to a Participant under the Plan except as expressly provided in the Plan.
- 16.7 Nonassignability. Neither a Participant nor any other person shall have any right to commute, sell, assign, transfer, pledge, anticipate, mortgage or otherwise encumber, transfer, hypothecate, alienate or convey in advance of actual receipt, the amounts, if any, payable hereunder, or any part thereof, which are, and all rights to which are expressly declared to be, unassignable and non-transferable. No part of the amounts payable shall, prior to actual payment, be subject to seizure, attachment, garnishment or sequestration for the payment of any debts, judgments, alimony or separate maintenance owed by a Participant or any other person, be transferable by operation of law in the event of a Participant's or any other person's bankruptcy or insolvency or be transferable to a spouse as a result of a property settlement or otherwise.
- 16.8 Not a Contract of Employment. The terms and conditions of this Plan shall not be deemed to constitute a contract of employment between any Employer and the Participant. Such employment is hereby acknowledged to be an "at will" employment relationship that can be terminated at any time for any reason, or no reason, with or without cause, and with or without notice, unless expressly provided in a written employment agreement. Nothing in this Plan shall be deemed to give a Participant the right to be retained in the service of any Employer, or to interfere with the right of any Employer to discipline or discharge the Participant at any time.
- 16.9 **Furnishing Information**. A Participant or his or her Beneficiary will cooperate with the Committee by furnishing any and all information requested by the Committee and take such other actions as may be requested in order to facilitate the administration of the Plan and the payments of benefits hereunder, including but not limited to taking such physical examinations as the Committee may deem necessary.
- 16.10 **Terms**. Whenever any words are used herein in the masculine, they shall be construed as though they were in the feminine in all cases where they would so apply; and whenever any words are used herein in the singular or in the plural, they shall be construed as though they were used in the plural or the singular, as the case may be, in all cases where they would so apply.
- 16.11 <u>Captions</u>. Headings and paragraphs of this Plan are for convenience only and shall not control or affect the meaning or construction of any of its provisions.
- 16.12 Governing Law. Subject to ERISA, the provisions of this Plan shall be governed by, construed, and enforced in accordance with the laws of the State of Texas (excluding any conflict of laws, rule or principle of Texas law that might refer the governance, construction, or interpretation of this Plan to the laws of another state). A Participant's sole remedy for any Claim shall be against the Company, and no Participant shall have any claim or right of any nature against any Related Company of the Company or any stockholder or existing or former director, officer or Employee of the Company or any Related Company of the Company. The individuals and entities described above in this Section 16.12 (other than the Company) shall be third-party beneficiaries of this Plan for purposes of enforcing the terms of this Section 16.12.
- 16.13 <u>Notice</u>. Any notice or filing required or permitted to be given to the Committee under this Plan shall be sufficient if in writing and hand-delivered, or sent by registered or certified mail, to the address below:

Blucora, Inc. Attn: General Counsel 6333 North State Highway 161, 4th Floor Irving, Texas 75038 Such notice shall be deemed given as of the date of delivery or, if delivery is made by mail, as of the date shown on the postmark on the receipt for registration or certification.

Any notice or filing required or permitted to be given to a Participant under this Plan shall be sufficient if in writing and hand-delivered, or sent by mail, to the last known address of the Participant.

- 16.14 **Successors**. The provisions of this Plan shall bind and inure to the benefit of the Participant's Employer and its successors and assigns and the Participant and the Participant's designated Beneficiaries.
- 16.15 **Spouse's Interest**. The interest in the benefits hereunder of a spouse of a Participant who has predeceased the Participant shall automatically pass to the Participant and shall not be transferable by such spouse in any manner, including but not limited to such spouse's will, nor shall such interest pass under the laws of interestate succession.
- 16.16 <u>Validity</u>. In case any provision of this Plan shall be illegal or invalid for any reason, said illegality or invalidity shall not affect the remaining parts hereof, but this Plan shall be construed and enforced as if such illegal or invalid provision had never been inserted herein.
- 16.17 <u>Incompetent</u>. If the Committee determines in its discretion that a benefit under this Plan is to be paid to a minor, a person declared incompetent or to a person incapable of handling the disposition of that person's property, the Committee may direct payment of such benefit to the guardian, legal representative or person having the care and custody of such minor, incompetent or incapable person. The Committee may require proof of minority, incompetence, incapacity or guardianship, as it may deem appropriate prior to distribution of the benefit. Any payment of a benefit shall be a payment for the account of the Participant and the Participant's Beneficiary, as the case may be, and shall be a complete discharge of any liability under the Plan for such payment amount.
- 16.18 Court Order. The Committee is authorized to make any payments directed by court order in any action in which the Plan or the Committee has been named as a party. In addition, if a court determines that a spouse or former spouse of a Participant has an interest in the Participant's benefits under the Plan in connection with a property settlement or otherwise, the Committee, in its sole discretion, shall have the right, notwithstanding any election made by a Participant, to immediately distribute the spouse's or former spouse's interest in the Participant's benefits under the Plan to that spouse or former spouse.

### 16.19 Distribution in the Event of Income Inclusion Under Code Section 409A or Other Taxation.

- (a) Code Section 409A. If any portion of a Participant's Account Balance under this Plan is required to be included in income by the Participant prior to receipt due to a failure of this Plan to comply with the requirements of Code section 409A and related Treasury Regulations, the Committee may determine that such Participant shall receive a distribution from the Plan in an amount equal to the lesser of (i) the portion of his or her Account Balance required to be included in income as a result of the failure of the Plan to comply with the requirements of Code section 409A and related Treasury Regulations, or (ii) the unpaid vested Account Balance.
- (b) <u>Trust</u>. If the Trust terminates in accordance with its terms and benefits are distributed from the Trust to a Participant in accordance therewith, the Participant's benefits under this Plan shall be reduced to the extent of such distributions; provided, however, that such distributions may not occur, if they otherwise would violate Code section 409A and the related Treasury Regulations.
- 16.20 <u>Insurance</u>. The Employers, on their own behalf or on behalf of the Trustee of the Trust, and, in their sole discretion, may apply for and procure insurance on the life of the Participant, in such amounts and in such forms as the Trust may choose. The Employers or the Trustee of the Trust, as the case may be, shall be the sole owner and beneficiary of any such insurance. The Participant shall have no interest whatsoever in any such policy or policies, and at the request of the Employers shall submit to medical examinations and supply such information and execute such documents as may be required by the insurance company or companies to whom the Employers have applied for insurance.

- 16.21 Legal Fees To Enforce Rights After Change in Control. The Company and each Employer is aware that upon the occurrence of a Change in Control, the Board of the Company or the Board of a Participant's Employer (which might then be composed of new members) or a stockholder of the Company or the Participant's Employer, or of any successor corporation might then cause or attempt to cause the Company, the Participant's Employer or such successor to refuse to comply with its obligations under the Plan and might cause or attempt to cause the Company or the Participant's Employer to institute, or may institute, litigation seeking to deny Participants the benefits intended under the Plan. In these circumstances, the purpose of the Plan could be frustrated. Accordingly, if, following a Change in Control, it should appear to any Participant that the Company, the Participant's Employer or any successor corporation has failed to comply with any of its obligations under the Plan or any agreement thereunder or, if the Company, such Employer or any other person takes any action to declare the Plan void or unenforceable or institutes any litigation or other legal action designed to deny, diminish or to recover from any Participant the benefits intended to be provided, then the Company and the Participant's Employer irrevocably authorize such Participant to retain counsel of his or her choice at the expense of the Company and the Participant's Employer (who shall be jointly and severally liable) to represent such Participant in connection with the initiation or defense of any litigation or other legal action, whether by or against the Company, the Participant's Employer or any director, officer, stockholder or other person affiliated with the Company, the Participant's Employer or any successor thereto in any jurisdiction. The amount of fees and expenses eligible for reimbursement during a calendar year shall not affect the fees and expenses eligible for reimbursement in any other calendar year. Reimbursement of eligible fees and expenses shall be made on or before the last day of the calendar year following the calendar year in which the fees or expenses were incurred.
- 16.22 **Formal Action by an Employer.** Any formal action herein permitted or required to be taken by the Company or an Employer shall be:
  - (a) if and when a partnership, by written instrument executed by one or more of its general partners or by written instrument executed by a person or group of persons who has been authorized by written instrument executed by one or more general partners as having authority to take such action;
  - (b) if and when a partnership, by written instrument executed by one or more of its general partners or by written instrument executed by a person or group of persons who has been authorized by written instrument executed by one or more general partners as having authority to take such action;
  - (c) if and when a proprietorship, by written instrument executed by the proprietor or by written instrument executed by a person or group of persons who has been authorized by written instrument executed by the proprietor as having authority to take such action;
  - (d) if and when a corporation, by resolution of its board of directors or other governing board, or by written instrument executed by a person or group of persons who has been authorized by resolution of its board of directors or other governing board as having authority to take such action; or
  - (e) if and when a joint venture, by written instrument executed by one of the joint venturers or by written instrument executed by a person or group of persons who has been authorized by written instrument executed by one of the joint venturers as having authority to take such action.

\* \* \* \* \* \* \* \*

[Remainder of Page Intentionally Left Blank. Signature Page Follows.]

| <del></del> |  |
|-------------|--|
|             |  |
|             |  |
|             |  |
|             |  |
|             |  |
|             |  |
|             |  |

| IN WITNESS      | WHEREOF, the | Company l | has signed t | his Plan | document a | s of Decem | ber 13, | 2018, | to be | effective | as o | of the |
|-----------------|--------------|-----------|--------------|----------|------------|------------|---------|-------|-------|-----------|------|--------|
| Effective Date. |              |           |              |          |            |            |         |       |       |           |      |        |

# Company

**Blucora, Inc.**, a Delaware corporation

By: John S. Clendening

Name: John S. Clendening

Title: President and CEO

4842-1954-1370 v.4

Signature Page to the Blucora Executive Tax-Smart Deferral Plan

### Subsidiaries of the registrant

Go2Net, Inc., a Delaware corporation HDV Holdings, Inc., a Delaware corporation

H.D. Vest, Inc., a Texas corporation

H.D. Vest Investment Securities, Inc., a Texas corporation H.D. Vest Advisory Services, Inc., a Texas corporation

H.D. Vest Insurance Agency LLC, a Massachusetts Limited Liability Company

H.D. Vest Insurance Agency LLC, a Montana Limited Liability Company
H.D. Vest Insurance Agency LLC, a Texas Limited Liability Company
Project Baseball Sub, Inc., a Delaware corporation

TaxAct Holdings, Inc., a Delaware corporation

TaxAct, Inc., an Iowa corporation

TaxSmart Research, LLC, a Delaware corporation

Financial BluPrint, LLC, a Delaware corporation

SimpleTax Software, Inc., a British Columbia corporation

### CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We consent to the incorporation by reference in the following Registration Statements:

- Registration Statement (Form S-8 No. 333-169691) pertaining to the Blucora, Inc. Restated 1996 Flexible Stock Incentive Plan,
- Registration Statement (Form S-8 No. 333-198645) pertaining to the Blucora, Inc. Restated 1996 Flexible Stock Incentive Plan,
- Registration Statement (Form S-8 No. 333-204585) pertaining to the Blucora, Inc. 2015 Incentive Plan,
- Registration Statement (Form S-8 No. 333-209218) pertaining to the Blucora, Inc. 2016 Equity Inducement Plan,
- Registration Statement (Form S-8 No. 333-214117) pertaining to the Blucora, Inc. 2016 Equity Inducement Plan,
- Registration Statement (Form S-8 No. 333-211625) pertaining to the Blucora, Inc., 2015 Incentive Plan as Amended and Restated and 2016 Employee Stock Purchase Plan, and
- Registration Statement (Form S-3 No. 333-216984) pertaining to the resale of shares of common stock of Blucora, Inc. by certain selling stockholders;
- Registration Statement (Form S-8 No. 333-225495) pertaining to the Blucora, Inc 2018 Incentive Plan

of our reports dated March 1, 2019, with respect to the consolidated financial statements of Blucora, Inc. and the effectiveness of internal control over financial reporting of Blucora, Inc. included in this Annual Report (Form 10-K) of Blucora, Inc., for the year ended December 31, 2018.

/s/ ERNST & YOUNG LLP

Dallas, Texas March 1, 2019

# CERTIFICATION OF PRINCIPAL EXECUTIVE OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002 (EXCHANGE ACT RULES 13a-14(a) and 15d-14(a))

### I, John S. Clendening, certify that:

- 1. I have reviewed this Annual Report on Form 10-K of Blucora, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Dated: March 1, 2019

/s/ John S. Clendening

John S. Clendening Chief Executive Officer and President (Principal Executive Officer)

# CERTIFICATION OF PRINCIPAL FINANCIAL OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002 (EXCHANGE ACT RULES 13a-14(a) and 15d-14(a))

### I, Davinder Athwal, certify that:

- 1. I have reviewed this Annual Report on Form 10-K of Blucora, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Dated: March 1, 2019

/s/ Davinder Athwal

Davinder Athwal Chief Financial Officer (Principal Financial Officer)

# CERTIFICATION OF PRINCIPAL EXECUTIVE OFFICER PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002 (18 U.S.C. SECTION 1350)

I, John S. Clendening, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that the Annual Report on Form 10-K of Blucora, Inc. for the year ended December 31, 2018 fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and that information contained in such Annual Report on Form 10-K fairly presents, in all material respects, the financial condition and results of operations of Blucora, Inc.

Dated: March 1, 2019

By: /s/ John S. Clendening

Name: John S. Clendening

Title: Chief Executive Officer and President

(Principal Executive Officer)

### CERTIFICATION OF PRINCIPAL FINANCIAL OFFICER PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002 (18 U.S.C. SECTION 1350)

I, Davinder Athwal, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that the Annual Report on Form 10-K of Blucora, Inc. for the year ended December 31, 2018 fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and that information contained in such Annual Report on Form 10-K fairly presents, in all material respects, the financial condition and results of operations of Blucora, Inc.

Dated: March 1, 2019

By: /s/ Davinder Athwal

Name: Davinder Athwal

Title: Chief Financial Officer

(Principal Financial Officer)